

With support from



POLICY ANALYSIS FOR THE PUBLIC AND PRIVATE SECTORS RECYCLED WATER MASTER PLAN UPDATE EAST BAY MUNICIPAL UTILITY DISTRICT

FINAL INTERIM REPORT

2175 North California Blvd. | Suite 315 Walnut Creek, California 94596 925-627-4100

woodardcurran.com

COMMITMENT & INTEGRITY DRIVE RESULTS

0061010.00 EBMUD December 2018 This page is intentionally left blank.



TABLE OF CONTENTS

SEC	SECTION		PAGE NO.	
TAE	BLE OF CONTE	NTS	I	
1.	INTRODUCT	ION	1-1	
	1.1 M	aster Plan Update Goals	1-1	
2.	COST EVAL	UATION APPROACH	2-1	
	2.1 Se	cope and Estimate Classification	2-1	
	2.2 C	ost Estimating Approach	2-1	
	2.3 C	onstruction Component Cost	2-5	
3.	NON-POTAE	BLE REUSE OPPORTUNITIES	3-1	
	3.1 E	xisting Non-Potable Reuse Demands	3-1	
	3.2 D	evelopment of Non-Potable Reuse Alternatives		
	3.3 C	entralized Non-Potable Reuse Alternatives	3-8	
	3.4 Sa	atellite Recycled Water Projects	3-33	
	3.5 Fi	Il Stations within EBMUD Service Area	3-48	
	3.6 S	ummary of Non-Potable Project Alternatives	3-49	
4.	POTABLE R	EUSE OPPORTUNITIES	4-1	
	4.1 C	urrent State of Potable Reuse Regulations in California	4-1	
	4.2 D	evelopment of Potable Reuse Alternatives	4-7	
	4.3 Se	ources of Wastewater for Advanced Treatment	4-9	
	4.4 Ta	argets for Advanced Treated Water	4-13	
	4.5 A	dvanced Water Treatment Trains	4-23	
	4.6 C	onveyance Concepts	4-35	
	4.7 Po	otable Reuse Alternatives Summary	4-36	
	4.8 C	ost Evaluation for Potable Reuse Alternatives	4-57	
5.	EVALUATIO	N OF ALTERNATIVES	5-1	
	5.1 N	on-Cost Evaluation	5-1	
	5.2 E	conomic Value of Recycled Water Supply	5-11	
	5.3 C	omparison of Recycled Water Project Unit Costs to Other Supplies	5-17	
	5.4 Si	ummary of Alternatives Evaluation	5-22	
6.	RECOMMEN	IDED PROJECTS	6-1	
	6.1 D	ERWA/San Ramon Valley Recycled Water Project Phases 3 and 5	6-1	
	6.2 Ea	ast Bayshore Phase 2	6-3	
	6.3 C	hevron Refinery/Richmond WPCP Recycled Water Project	6-8	
	6.4 Pl	hillips 66 Refinery Recycled Water Project	6-11	
	6.5 C	ost of Recommended Projects	6-13	
	6.6 Fi	uture Potable Reuse Consideration	6-13	



7.	IMPLEMENTATION PLAN		7-1
	7.1	Project Phasing	7-1
	7.2	Institutional Needs and Customer Outreach	7-5
	7.3	Funding Opportunities	7-5
8.	REFER	ENCES	8-1

APPENDICES

- APPENDIX B: Cost Estimate Details for Potable Reuse Alternatives
- APPENDIX C: Economic Evaluation Willingness-To-Pay and Shortage Estimates
- APPENDIX D: Customer List for East Bayshore Recycled Water Project

TABLES

- Table 2-1: Cost Estimate Classification Matrix (AACE International)
- Table 2-2: Mobilization Allowances
- Table 2-3: Example of Cost Contingency and Implementation Factors
- Table 2-4: Land Use Designations
- Table 2-5: Costs for Open Cut Pipeline Construction
- Table 2-6: Costs for Trenchless Pipeline Construction
- Table 2-7: Annual Labor Requirements for Various Pump Station Sizes
- Table 2-8: Pre-Stressed Concrete Tank Cost Allowances
- Table 2-9: Costs for Groundwater Wells
- Table 2-10: Annual Labor Requirements for Groundwater Wells
- Table 2-11: Preliminary Advanced Treatment Processes Unit Costs
- Table 2-12: Preliminary Advanced Treatment Annual O&M Costs
- Table 3-1: Recycled Water Deliveries, Annual Averages, 2013-2017
- Table 3-2: Sources of Demand Estimates for Development of Non-Potable Project Alternatives
- Table 3-3: Typical Recycled Water Quality Objectives¹
- Table 3-4: Typical Raw WW Quality for Satellite Treatment Facility Sizing
- Table 3-5: Recycled Water Distribution Design Criteria
- Table 3-6: DERWA/San Ramon Project Status
- Table 3-7: DERWA/San Ramon Project Costs, District's Share
- Table 3-8: East Bayshore Recycled Water Project Costs (2017 dollars)¹
- Table 3-9: Chevron Richmond Refinery Recycled Water Project Demand Deficit
- Table 3-10: City of Richmond WPCP Recycled Water Project Capital Costs
- Table 3-11: Phillips 66 Refinery Recycled Water Project Supply and Demand
- Table 3-12: Phillips 66 Refinery Recycled Water Project Costs
- Table 3-13: Reliez Valley Recycled Water Project Annual Average Demand
- Table 3-14: CCCSD Regional Project Costs
- Table 3-15: Annual Average Irrigation Demands



Table 3-16: CCWD Pipeline in Canal ROW Project Costs Table 3-17: Diablo Country Club Satellite Project Costs Table 3-18: Moraga Country Club Satellite Project Costs Table 3-19: Moraga Area Distribution Customer Annual Average Demand Table 3-20: Moraga Area Recycled Water Expansion Costs Table 3-21: Mountain View/St. Mary's Cemeteries Satellite Annual Average Demand Table 3-22: Rossmoor Area Distribution Customer Annual Average Demand Table 3-23: UC Berkeley Main Campus Satellite Demand Table 3-24: UC Berkeley Main Campus Satellite Project Costs Table 3-25: Oakland Hills Satellite Project Non-Potable Water Demand Table 3-26: Oakland Hills Satellite Project Costs Table 3-27: Non-Potable Project Alternatives Cost Summary Table 4-1: Groundwater Augmentation Criteria for Potable Water Reuse Table 4-2: Summary of Reservoir Water Augmentation Criteria Table 4-3: Conceptual Requirements for Raw and Treated Drinking Water Augmentation Table 4-4: Surface Water Treatment Plant Capacity and Recent Production Table 4-5: Sources of Secondary Effluent for Potable Reuse Table 4-6: Briones Reservoir Augmentation Project Alternatives Table 4-7: San Pablo Reservoir Augmentation Project Alternatives Table 4-8: Upper San Leandro Reservoir Augmentation Project Alternatives Table 4-9: Treated Water Augmentation Connections Table 4-10: Pathogen Removal Credits for Advanced Treatment Trains Table 4-11: Potable Reuse Alternatives Table 4-12: Potable Reuse Alternatives Cost Summary Table 5-1: Summary of Evaluation Criteria Table 5-2: Non-Potable Reuse Scoring Rubric Table 5-3: Non-Potable Reuse Alternatives Non-Cost Evaluation Table 5-4: Ranked Scores for Non-Potable Reuse Alternatives Table 5-5: Potable Reuse Scoring Rubric Table 5-6: Potable Reuse Alternatives Non-Cost Evaluation Table 5-7: EBMUD Weighted-Average Alternative Water Supply Cost Estimate (2017 dollars) Table 5-8. Projected Growth in Recycled Water Capacity per 2015 UWMP (MGD) Table 5-9. Parameters Used to Calculate Willingness-to-Pay for Increments of New Recycled Water Capacity Table 5-10. Willingness-to-Pay for Increments of Potable Reuse Water (\$/AF) Table 5-11: Summary of Economic Evaluation Parameters Table 5-12: Highest Ranking, Lowest Cost Alternatives Table 5-13: Recommend Non-Potable Reuse Projects to Meet 20 MGD by 2040 Table 6-1: Capital and O&M Costs of Recommended Projects Table 6-2: Highest Ranking, Lowest Cost Potable Reuse Alternatives

Table 6-1: Phasing for Recommended Projects



FIGURES

Figure 2-1: Pump Station Construction Costs (with Bid Markups)

- Figure 2-2: Welded Steel Storage Tank Construction Costs (with Bid Markups)
- Figure 2-3: Pre-stressed Concrete Storage Tank Construction Costs
- Figure 3-1: Existing Centralized Recycled Water Projects (2013 to 2017)
- Figure 3-2: Conceptual Satellite Treatment Train Schematic
- Figure 3-3: DERWA/San Ramon Distribution System
- Figure 3-4: East Bayshore Distribution System
- Figure 3-5: Recycled Water Treatment System Reconfiguration for Satellite Treatment of Adeline Interceptor with MBR
- Figure 3-6: San Leandro Water Reclamation Facility Expansion Project
- Figure 3-7: Chevron Richmond Refinery Recycled Water Project
- Figure 3-8: Chevron Richmond Refinery Recycled Water Project Water Balance
- Figure 3-9: Phillips 66 Refinery Recycled Water Project
- Figure 3-10: Franklin Canyon Recycled Water Project
- Figure 3-11: Reliez Valley Recycled Water Project
- Figure 3-12: Diablo Country Club Satellite Project
- Figure 3-13: Moraga Satellite Project
- Figure 3-14: Mountain View/St. Mary's Cemeteries Satellite Project
- Figure 3-15: Rossmoor Country Club Satellite Project
- Figure 3-16: UC Berkeley Main Campus Satellite Project
- Figure 3-17: Oakland Hills Satellite Project
- Figure 4-1: Spectrum of Potable Reuse Applications
- Figure 4-2: Seasonality of Surface Water Treatment Plant Production, FY08-14
- Figure 4-3: Sources of Secondary Effluent for Potable Reuse
- Figure 4-4: Location of East Bay Plain Subbasin
- Figure 4-5: Briones Reservoir Operations for 30-MGD Recycled Water Project
- Figure 4-6: Treatment Train Type 1 Groundwater Augmentation
- Figure 4-7: Treatment Train Type 2 Reservoir Water Augmentation
- Figure 4-8: Treatment Train Type 3 Reservoir Augmentation with New MBR
- Figure 4-9: Treatment Train Type 4 Raw Water Augmentation
- Figure 4-10: Treatment Train Type 5 Raw Water Augmentation with New MBR
- Figure 4-11: Treatment Train Type 6 Treated Water Augmentation
- Figure 4-12: Treatment Train Type 7 Treated Water Augmentation with New MBR
- Figure 4-13: Potable Reuse Alternative Oro Loma Groundwater Augmentation
- Figure 4-14: Potable Reuse Alternative Richmond Groundwater Augmentation
- Figure 4-15: Potable Reuse Alternatives San Leandro WPCP as a Source
- Figure 4-16: Advanced Treatment Footprint at San Leandro WPCP
- Figure 4-17: Potable Reuse Alternatives Pinole WPCP as a Source
- Figure 4-18: Advanced Treatment Footprint at Pinole WPCP
- Figure 4-19: Potable Reuse Alternatives Richmond WPCP as a Source
- Figure 4-20: Advanced Treatment Footprint at Richmond WPCP



Figure 4-21: Potable Reuse Alternatives - West County WPCP as a Source Figure 4-22: Advanced Treatment Footprint at West County WWTP Figure 4-23: Potable Reuse Alternatives - Oro Loma WPCP as a Source Figure 4-24: Advanced Treatment Footprint at Oro Loma WPCP Figure 4-25: Potable Reuse Alternatives - CCCSD WWTP as a Source Figure 4-26: Advanced Treatment Footprint at Central San WWTP Figure 4-27: Potable Reuse Alternatives - SD-1 as a Source Figure 4-28: Advanced Treatment Footprint at SD-1 Figure 4-29: Potable Reuse Alternatives - LAVWMA as a Source Figure 4-30: Potable Reuse Alternative – Pt. Isabel Satellite WWTP as a Source Figure 4-31: Advanced Treatment Footprint at Pt. Isabel Figure 5-1: Non-Potable Reuse Alternative Comparison Figure 5-2: Potable Reuse Alternative Evaluation Summary Figure 5-3: Consumer Willingness to Pay to Avoid a Water Shortage Figure 5-4. Willingness-to-Pay for New Increment of Recycled Water under Rationing Figure 5-5: Economic Comparison of Non-Potable Reuse Alternatives Figure 5-6: Economic Comparison of Potable Reuse Alternatives Figure 5-7: Dry Year Economic Comparison of Non-Potable Reuse Alternatives Figure 5-8: Dry Year Economic Comparison of Potable Reuse Alternatives Figure 5-9: Recommend Recycled Water Projects Figure 6-1: DERWA/San Ramon Phases 3 and 5 Expansion Figure 6-2: East Bayshore Phase 1A, 1B and Phase 2 Project Figure 6-3: Proposed 4.5-MGD MBR Layout at EBRWF Figure 6-4: Potential Alignment to Expand EBRWF to San Pablo WTP Figure 6-5: Chevron Refinery/Richmond WPCP Pipeline to RARE Figure 6-6: Chevron Refinery/Richmond WPCP Site Layout Figure 6-7: P66 Refinery Alternative Site Location Figure 6-8: Phillips 66 Refinery Alternative Process Flow Diagram Figure 7-1: Planned Recycled Water Deliveries from Recommended Project List

Figure 7-2: Proposed Phasing of Recommended Projects



ACRONYMS

µS/cm	Micro Siemens per centimeter
AACE	Association for the Advancement of Cost Engineering
ADWF	Average dry weather flow
AF	Acre-feet
AF/MO	Acre-feet per month
AFY	Acre-feet per year
AOP	Advanced oxidation process
AWT	Advanced water treatment
BAC	Biologically activate carbon
BACWA	Bay Area Clean Water Agencies
BAF	Biologically active filtration
BOD	Biochemical oxygen demand
Canal	Contra Costa Canal
CCCSD	Central Contra Costa Sanitary District
CCI	Construction cost index
CCR	California Code of Regulations
CDPH	California Department of Public Health
CECs	Constituents of emerging concern
cfs	cubic feet per second
CIP	Capital improvement program
CNWS	Concord Naval Weapons Station
COC	Cycles of concentration
CSD	Community Services District
СТ	Concentration of chlorine x time of contact
DCC	Diablo Country Club
DDW	Division of Drinking Water
DERWA	DSRSD-EBMUD Recycled Water Authority
DERWA/San Ramon	San Ramon Valley Recycled Water Program
District	East Bay Municipal Utility District
DSRSD	Dublin San Ramon Services District
EBDA	East Bay Dischargers Authority
EBMUD	East Bay Municipal Utility District
EBRWF	East Bayshore Recycled Water Facility



EC	Electrical conductivity
EIR	Environmental Impact Report
ENR	Engineering News Record
ESB	Engineered storage buffer
ESP	Engineering Standard Practice
FAT	Full advanced treatment
fps	Feet per second
gpm	gallons per minute
H_2O_2	Hydrogen peroxide
hp	Horsepower
HVAC	Heating, ventilation, and air conditioning
LAVWMA	Livermore-Amador Valley Water Management Agency
LRVs	Log reduction values
MBR	Membrane bioreactor
MCC	Moraga Country Club
MCLs	Maximum contaminant levels
MF	Microfiltration
MG	Million gallons
mg/L	Milligrams per liter
MGD	Million gallons per day
MOU	Memorandum of Understanding
NaOCI	Sodium hypochlorite
NDMA	N-nitrosodimethylamine
NPDES	National Pollutant Discharge Elimination System
NRWRP	North Richmond Water Recycling Plant
O ₃	Ozonation
O&M	Operation and maintenance
O&P	Overhead and profit
OLSD	Oro Loma Sanitary District
RARE	Richmond Advanced Recycled Expansion
RCC	Richmond Country Club, Rossmoor Country Club
RO	Reverse osmosis
RRT	Response retention time
RWMP	Recycled Water Master Plan



SAR	Sodium adsorption ratio
SB	Senate Bill
SCC	Sequoyah Country Club
SD-1	EBMUD Main Wastewater Treatment Plant (Special District No. 1)
SLWRF	San Leandro Water Recycling Facility
SRT	Solids residence time
SRWTP	Satellite recycled water treatment plants
SWRCB	State Water Resources Control Board
SWA	Surface water source augmentation
SWTR	Surface Water Treatment Rule
TDS	Total dissolved solids
Title 22	California Recycled Water Regulations - Title 22, California Code of Regulations
TKN	Total Kjeldahl nitrogen
TOC	Total organic carbon
TSS	Total suspended solids
UCB	University of California Berkeley
URT	Underground retention time
UF	Ultrafiltration
USLWTP	Upper San Leandro Water Treatment Plant
UV	Ultraviolet light
VFDs	Variable frequency drives
West County	West County Water District
WCWD	West County Water District
WPCF	Water Pollution Control Facility
WPCP	Water Pollution Control Plant
WSMP	Water Supply Management Program
WTP	Water Treatment Plant
WWTP	Wastewater Treatment Plant

CONVERSION FACTORS

1 million gallons per day (MGD) = 1,120 acre-feet per year (AFY)

1 million gallons (MG) = 3.069 acre-feet (AF)



1. INTRODUCTION

The East Bay Municipal Utility District (District or EBMUD) is a publicly owned multipurpose agency that provides drinking water for 1.4 million customers in Alameda and Contra Costa counties. Approximately 90% of the District's source water comes from the Mokelumne River and is delivered to the District's service area by the Mokelumne Aqueducts. The balance of the District's supply comes from East Bay watersheds and the Freeport Regional Water Project, which delivers water from the Sacramento River during certain drought events. During the period 2009 through 2018, the District produced, on average, about 175 MGD of potable water.

The District's first comprehensive "Water Reclamation Master Plan" was developed in 1991. Its goals were to identify potential water reuse opportunities, develop and rank feasible projects, and provide recommendations for implementing high priority projects. The 1991 plan included irrigation projects as well as cooling tower and industrial use. In 1993 the District implemented the Water Supply Management Program (WSMP) and established a recycled water goal of 14 MGD by 2020. In 2012 the District's WSMP 2040 Plan updated the recycled water goal to a total of 20 MGD by 2040 and an updated "Recycled Water Master Plan" was developed. This new plan identified potential projects that could be implemented to meet the 20 MGD by 2040 recycled water goal.

The District currently has approximately 9 MGD of recycled water capability in place. Water use includes irrigation, office building toilet flushing, cooling towers, and industrial boilers.

1.1 Master Plan Update Goals

The goal of the Recycled Water Master Plan (RWMP) Update is to develop a comprehensive update to the District's 2010 Recycled Water Master Plan. The District's current recycled water goal of 20 million gallons per day (MGD) by 2040 is dependent entirely on non-potable reuse. Given numerous factors, including statewide population growth, climate change, ecosystem challenges, legislative and regulatory pressures, the District envisions that an additional expansion of water recycling efforts may be necessary in the future to ensure continued reliability of the water supply, which may include potable reuse as it becomes more prudent and feasible.

The first phase of the RWMP Update involved identifying and assessing opportunities for both non-potable reuse and potable reuse. It included development of a revised non-potable recycled water project list, prioritized based on feasibility and affordability, and revised non-potable reuse goals. The potable reuse assessment considers impacts on operations of existing conveyance, treatment, storage, and distribution systems. Potential sources of recycled water include the District's Main Wastewater Treatment Plant (Special District No. 1, or SD-1, which treats wastewater collected from a sub-area of the District's much larger potable water service area), as well as other nearby wastewater agencies. The assessment includes an economic evaluation to determine under what conditions potable reuse alternatives may become economically feasible for the District.

The purpose of this report is to define the District's portfolio of non-potable and potable water reuse options, perform a qualitative evaluation of each option, and define a shortlist of options for recycled water implementation. This report is organized as follows:

- Introduction
- Cost estimating approach
- Non-potable reuse alternatives
- Potable reuse alternatives
- Evaluation of alternatives
- Recommended Master Plan Projects

This page is intentionally left blank.



2. COST EVALUATION APPROACH

This section details the basis of the cost estimates for the potable and non-potable alternatives.

2.1 Scope and Estimate Classification

The Association for the Advancement of Cost Engineering International (AACE International) has developed a cost estimate classification system that provides guidelines for applying the general principles of estimate classification to project cost estimates. The five estimate classes are presented in AACE International Recommended Practice No. 18R-97 (Cost Estimate Classification System – As Applied in Engineering, Procurement, and Construction for the Process Industries). The guideline establishes a relationship between the project maturity (i.e., project definition as percent of complete definition) and the accuracy and methodology used to produce the cost estimate.

Table 2-1 provides a summary of the estimate classes and expected accuracy range. For Class 5 estimates, the expected accuracy range is -20% to -50% on the low end and +30% to +100% on the high end. The estimates developed for the RWMP Update will be Class 5 cost estimates.

Estimate Class	Level of Project Definition	Purpose of Estimate	Methodology	Expected Accuracy Range
Class 5	0% to 2%	Concept screening	Capacity factored, parametric models, judgement, or analogy	Low: -20% to -50% High: +30% to +100%
Class 4	1% to 15%	Study or feasibility	Equipment factored or parametric models	Low: -15% to -30% High: +20% to +50%
Class 3	10% to 40%	Budget authorization or control	Semi-detailed unit costs with assembly level line items	Low: -10% to -20% High: +10% to +30%
Class 2	30% to 75%	Control or bid/tender	Detailed unit cost with forced detailed take-off	Low: -5% to -15% High: +5% to +20%
Class 1	65% to 100%	Check estimate or bid/tender	Detailed unit cost with detailed take-cost	Low: -3% to -10% High: +3% to +15%

Table 2-1: Cost Estimate Classification Matrix (AACE International)

Source: AACE International Recommended Practice No. 18R-97

The expected accuracy range of the Class 5 cost estimates is similar to the "Preliminary" or "Conceptual" category as defined in the District's Engineering Standard Practice (ESP) 020.3, which has an expected accuracy range of -30% to +50% (EBMUD, 2008b).

2.2 Cost Estimating Approach

The RWMP Update includes the assessment of both potable reuse and non-potable reuse project alternatives. Most of the non-potable project descriptions are based on work previously developed by the District and other public water and wastewater agencies. Cost estimates for these projects have been reviewed for any major errors or omissions to the facilities or the unit costs. Once any necessary corrections were made, the raw construction costs were extracted and escalated to December 2017 dollars. The soft costs and allowances defined in this section were then applied to the raw construction costs, resulting in a capital cost estimate for use in the RWMP Update. Variations from this approach are highlighted for each project as needed in Section 3.2.3



For potable reuse project alternatives, cost estimates were based on preliminary facilities plans developed for Oro Loma Sanitary District and Central Contra Costa Sanitary District as well as new potable reuse alternatives developed under this RWMP Update. Construction costs were estimated using unit cost information provided by the District, which were developed from past construction projects and industry costs estimate resources (primarily RS Means Heavy Construction Cost Data). Operation and maintenance (O&M) costs are based on recent estimates at the District's facilities, estimated lab hours, equipment power needs, and chemical and other consumable demands.

2.2.1 Raw Construction Cost

Raw construction costs were estimated for each project component based on estimated unit costs multiplied by quantity take-offs. Unit costs were developed primarily using historical cost data from previous District projects, supplemented with the experience from projects of similar size or configuration. In some cases (i.e., for pumps, storage tanks and pipelines) these unit costs are based on construction bid data, which already include markups for contractor overhead and profit.

Based on the level of detail available for Class 5 estimates, allowances were used for some elements such as site work, as detailed further herein.

Engineering economic factors were utilized to develop and escalate unit costs when required to reflect the current construction cost, industry trends, and project location. These factors are incorporated into the unit costs and are represented in two categories:

- Engineering News Record's (ENR) Construction Cost Index (CCI) The ENR CCI is an index for construction cost inflation that is used to convert historic cost information to current value. The rate of construction cost inflation varies by geographical region and ENR publishes CCI values for major metropolitan cities. The ENR CCI for the 20-city average (10,870 in December 2017) was used for the RWMP Update.
- <u>RS Means Location Factor</u> The unit costs presented in RS Means represent the national average across the United States and Canada. A location factor, also referred to as City Cost Indexes in RS Means, is applied to account for variations in regional costs such as labor, equipment rental, raw materials, and freight. The Oakland, California location factor listed in RS Means is used for this Project, which has a corresponding location factor of 123.1 (RS Means 2017). This location factor represents a weighted average of both materials and labor cost across all divisions of construction. The location factor may be used to adjust cost estimates from other geographic areas (for example, to adjust capital cost estimates for potable reuse treatment trains in Southern California).

Unit costs were factored by the relative difference in CCI to escalate costs to the time of the estimate and the location factor to translate the cost given to an equivalent cost for the District's service area.

2.2.2 Allowances and Contingency

2.2.2.1 Construction Cost Allowances

Several allowances are applied to the raw construction cost subtotal to develop an estimated construction cost. The construction cost allowances used are listed below.

<u>Tax on Materials and Equipment Rental = 9%, applied to 50% of raw construction cost</u> – A Class 5 estimate uses installed unit cost metrics that include both raw materials and installation (i.e., labor and equipment) costs. As of December 2017, the sales tax rate was 8.25% in Contra Costa County and 9.25% in Alameda County. The regionally-averaged tax on materials was estimated as 9.0% (local tax) and applied to 50% of the raw construction cost.



- <u>Overhead and Profit = 15%</u> Overhead and profit (O&P) represents the general contractor's operating costs and estimated profit levels. The O&P factor typically varies between 10% and 25%, depending on the size of the project and market conditions, with larger projects typically having lower O&P factors. An O&P factor of 15% was applied to the raw construction cost.
- <u>Owner's Reserve for Change Orders = 0%</u> Change orders may be a result of the Owner's direction to implement additional work, differing field conditions that requires additional work, or an error in the project contract documents. District standard practice does not include a change order allowance for this level of cost estimate, so these were not included as a line item in RWMP Update cost estimates.

For components costs that were developed based on the historical cost data (bid amount of past projects), construction cost allowances and the estimating contingency were assumed to be included in the bid price and were not applied.

2.2.2.2 Estimating Contingency and Mobilization

In addition to the Construction Cost Allowances, a final estimating contingency was applied to generate the estimated total construction costs. The estimating contingency is defined as unknown costs due to lack of detailed engineering during the preliminary planning phase that are estimated as a percentage of defined project costs (i.e., the construction cost including construction cost implementation cost allowances). As the level of project definition and understanding increases and the level of unknown decreases, the estimating contingency typically decreases. For the RWMP Update, an estimating contingency of 25% was applied to the cost estimates after construction cost and implementation cost allowances were included. The contingency percentage is slightly higher than the maximum contingency of 20% designated for preliminary cost estimates (per ESP 020.3) and matches the cost estimating approach used for the Richmond Advanced Recycled Expansion project (RARE).

Mobilization involves the process of establishing resources at a project site that are to be used over the course of the project such as temporary office trailers, temporary utilities, and other equipment rental. For this project, mobilization was estimated as a fraction of the sum of construction cost with contingency based on the project size as summarized in **Table 2-2**.

Project Size (Construction Cost with Contingency)	Mobilization Allowance
< \$5M	10%
\$5M - \$10M	8%
> \$10M	5%

Table 2-2: Mobilization Allowances

For components costs that were developed based on the historical cost data (construction bids from past projects), construction cost allowances and the estimating contingency were assumed to be included in the bid price and were not applied.

2.2.2.3 Implementation Cost Allowances

To generate the estimated capital costs, implementation cost allowances such as environmental review, design, construction management, and other administrative costs associated with the project were included. Implementation costs are typically estimated as a percentage of total construction cost, after including all allowances described in Section 2.2.2.1. The implementation cost allowances used are summarized below and total 30%.



- Environmental Documentation and Permits = 5% Environmental documentation and permits involve producing environmental studies and acquiring any permits necessary to construct a project. A factor of 5% was applied to the total construction cost for environmental documentation and permits.
- <u>Design Cost = 15%</u> Engineering design services include field investigations (e.g., surveys, geotechnical reports, hazard materials investigations), preliminary and final design, contract document development (i.e., plans and specifications), preparation of detailed cost estimates, and project scheduling. An engineering services factor of 15% was applied to the total construction cost.
- Project Administration and Construction Management = 10% Costs for project administration includes planning, funding, design, and construction. Costs for construction management, including inspection, can vary greatly with project size and complexity and whether the Owner performs this work with in-house staff or through a consultant. A construction management factor of 10% was applied to the total construction cost.

2.2.3 Capital Cost Summary

A summary of the Allowances is shown in **Table 2-3**. The table also includes a set of example calculations based on a raw construction subtotal of \$1,000,000. All subtotals are rounded up to two significant figures.

Category	Factor	Example Cost
Raw Construction Cost Subtotal		\$1,000,000
Tax on Materials and Rental Equipment (Applied to 50% of raw		
construction cost)	9%	\$45,000
Overhead and Profit	15%	\$150,000
Estimated Construction Cost (Including Construction Cost Allowances)		\$1,200,000
Estimating Contingency	25%	\$300,000
Estimated Construction Cost (Including Contingency)		\$1,500,000
Mobilization	10%	\$150,000
Estimated Project Cost (Including Contingency and Mobilization)		\$1,700,000
Environmental Documentation and Permits	5%	\$85,000
Design Cost	15%	\$255,000
Administration and Construction Management Cost	10%	\$170,000
Estimated Capital Cost (Including Allowances and Contingency)	\$2,200,000	

Table 2-3: Example of Cost Contingency and Implementation Factors

Note: All line items in Class 5 planning-level estimates will be rounded to two significant figures.

2.2.4 Operations and Maintenance

Annual O&M requirements were derived from experience on similar projects and standard engineering methods. There is the potential for future increases in O&M unit costs, such as energy and labor costs, that are not accounted for in the O&M cost estimates but will be accounted for in the life-cycle cost development. The three components used to develop annual O&M costs were:

• <u>Labor</u> – Labor costs associated with the system O&M is calculated on an hourly basis. Where applicable, it was assumed that the maximum number of working hours per year is 2,080 hours. The average hourly cost of O&M personnel, which includes all wages and benefits to the operator, is estimated at \$135.



- <u>Electricity</u> The unit cost of electricity used was \$0.15/kWh and was based on the average electricity billing rate of new Pacific Gas and Electric customers. All power-intensive equipment in the Project (such as pumps, blowers, and ultraviolet [UV] disinfection lamps) were included in the electricity estimate. Equipment and systems that consume significantly less energy (such as lighting, chemical dosing systems, and valve actuators) are assumed to be negligible and were not included.
- <u>Consumables</u> Consumables are a major component of operation expenditures and include resources that are intended and expected to be used up relatively quickly. Example of consumables include chemicals, gaskets, and potable water. Appropriate consumable costs are discussed for each facility type in Section 2.3.

2.2.5 Life Cycle Costs

Cost estimates were converted to an annualized total cost following District guidance documents ESP 020.1 *Life Cycle Cost Analysis* (EBMUD, 2010b) and ESP 462.1 *Useful Life of Water Facilities* (EBMUD, 2011b) and using the following assumptions:

- Base year: December 2017
- Discount rate: 3% (net of inflation)
- Project service period: 30 years
- Useful life of advanced treatment plants: 30 years
- Useful life of other facilities will vary based on component type based on ESP 462.1 (e.g., 35 years for water treatment plant equipment; 75 years for polyvinyl chloride [PVC] pipelines). Useful life assumptions for equipment not listed on ESP 462.1 will be noted. For facilities with a useful life longer than 30 years, a salvage value with straight-line depreciation will be applied in year 30 where appropriate.

Annualized total costs were divided by projected annual recycled water deliveries to estimate the unit cost per acrefoot of water delivered (\$/AF). Financing costs for loan or bond repayment are not included in the annualized total costs, but financing options are discussed as part of the master plan implementation.

2.3 Construction Component Cost

2.3.1 Site Work

Site work includes all work related to the civil construction of the Project such as excavation, off haul and disposal, grading, paving, shoring dewatering and backfill. Assumptions regarding site work are described within each project type (pipeline, pump station, etc.) in the following sections.

2.3.2 Pipelines

Pipeline capital costs were based on construction cost data provided by the District. Pipeline capital costs were determined based on pipe diameter and the development density of the project location. To determine the development density, a spatial analysis was conducted in ArcGIS using available land use data and the pipe alignments. Parcels in the District's service area were reassigned to one of three land use categories – High Density Urban, Low Density Urban, and Non-Urban – based on their 2015 Assessors Land use code. Assessor land use codes were consolidated into these three categories as summarized in **Table 2-4**.



Table 2-4: Land Use Designations

Assessor Land Use Category (EBMUD WSMP 2040)	Land Use Description (Contra Costa County)	Recategorized Land use
EV, EOS	Agricultural, Open Space, Parks and Recreation, Public and Semi-Public, and Watershed	Non-Urban
EHW, EMUR3, EP, EPI, ER, ER1, ER2, ER3, ER4, ERAW, ERW, ES	Commercial, Residential Mixed Use, Business Park, Commercial, Commercial Recreation, Delta Recreation, Downtown/Waterfront, Mobile Homes, non-high density Multiple-Family Residential, Office, Single Family Residential, and all other specific area designations	Low-Density Urban
ER5, ER6, EO, EC, EIL, EOH	Industry, Multiple-Family Residential – Very High Density and Very High Special, and Pleasant Hill BART – Mixed Use	High Density Urban

For the purposes of this master plan, it was assumed that open cut pipeline installation could be used for pipe aligned within urban areas and easily accessible non-urban (non-hilly) areas. Pipeline cost estimates for open cut construction were based on the costs presented in **Table 2-5**.

Table 2-5: Costs for Open Cut Pipeline Construction

Project Location	Unit	Unit Cost
Non-Urban	\$ / inch-diameter / linear foot	\$30
Urban – Low Density	\$ / inch-diameter / linear foot	\$40
Urban – High Density	\$ / inch-diameter / linear foot	\$50

For trenchless installation, the cost estimates were based on the costs presented in **Table 2-6**. For the purposes of this RWMP, it was assumed that trenchless crossings would be microtunneled for all pipelines with a diameter over 24-inches and horizontal directional drilling (HDD) would be used for trenchless crossings for pipelines 24-inches and smaller. Determination about pipeline construction approach (i.e., when to use trenchless construction or tunneling) is discussed in Section 4.6.3. Additionally, District staff identified that the existing San Pablo Tunnel could be rehabilitated and used for recycled water transmission.

Element	Unit	Unit Cost
San Pablo Tunnel Rehabilitation and new pipe, or new tunnel ¹	Linear foot	\$3,500
Microtunnel Launch and Receiving Pit	lump sum	\$620,000
Microtunnel Casing and Pipe (> 24-inch diameter)	Linear foot	\$2,800
Horizontal Directional Drilling (HDD) (≤ 24-inch diameter)	Linear foot	\$2,200

Note:

1. EBMUD, 2018, "San Pablo Tunnel - Full Seismic Retrofit", Cost estimate by District staff based on March 9, 2006 Northern Pipeline cost estimates. An additional 25% contingency was added to the District estimate.



2.3.2.1 Annual Pipeline O&M

Pipelines require a minimal amount of operational labor resources, as most of the operations occur at the pump station or at the discharge point (i.e., reservoir or storage tank). Therefore, it is assumed that there are no operational labor requirements for pipelines. Pipelines would require regularly scheduled maintenance that may include the exercising of valves, appurtenance inspections (including customer turnouts), and flushing procedures at dead ends. It is estimated that it would require two percent of the construction cost for annual maintenance. No consumables or electrical needs are identified specific to pipelines.

2.3.3 Pump Stations

Pump stations include a variety of elements depending on the type of pumps, pump station arrangement, surge control systems, and project characteristics (i.e., wet wells or canned pumps, available layout, pump station turndown, electrical equipment location, etc.). For this RWMP Update, pump station capital costs were based on the construction cost curve provided by the District (**Figure 2-1**). The cost estimates shown below will be marked up using the implementation cost allowances listed in Section 2.2.2.3.





Note: Pumping plan horsepower includes spare (standby) installed pumps

2.3.3.1 Annual Pump Station O&M

Pump station operations and maintenance includes labor, electricity, and consumables.

• <u>Labor</u> – The annual labor requirements of a pump station mainly depend on the amount of equipment at the pump station, as well as the level of automation that is implemented at the pump station. Other minor factors, such as pump station location, contingency measures, and age of pump station, would also affect the labor



demands. Operators are expected to regularly tend to the pump stations to operate valves, start and stop pumps, and examine flow data. Routine maintenance may include the inspection of equipment, exercising of valves, and servicing instrumentation. Estimates for operation and maintenance labor requirements are tabulated below in **Table 2-7**.

Pump Station Capacity (gpm)	Annual Operator Hours	Annual Maintenance Hours	Total Annual O&M Hours	
0 to 2,500	400	100	500	
2,500 and up	800	200	1,000	

- <u>Electricity</u> Pump station electricity consumption is estimated by multiplying the pump design point (in delivered head, feet), average flow (cubic feet per second [cfs]) and the annual hours in operation and dividing by the pump efficiency. The calculation also includes conversion factors to produce a result in kilowatt hour (kWh). An example calculation is shown below:
 - Example Pump Station
 - Flow: 1 MGD (1.55 cfs)
 - Operating Hours: 12 months, constant (8760 hours)
 - \circ Head delivered: 100 ft
 - Pump Efficiency: 80%

$$\frac{1.55 \ [cfs] * 100 \ [ft] * \ 62.4 \left[\frac{lb}{ft^3}\right] * 0.7457 \left[\frac{kWh}{hp}\right] * 8760 \ [hr]}{550 \ \left[\frac{ft * lb}{hp * sec}\right] * 0.8 \ [-]} = 143,600 \ [kWh]$$

• **<u>Consumables</u>** – Pump station consumables are to be estimated as 5% of the estimated construction cost.

2.3.4 Storage Tanks

Storage tanks require significant site work and piping. For this project, the District has provided two construction cost curves for tanks (welded steel and pre-stressed concrete). The welded steel storage tank curve (**Figure 2-2**) includes costs of site work and piping. The pre-stressed concrete tank curve (**Figure 2-3**) does not include site work, piping, contractor overhead, sales tax or estimating contingency. Instead, allowances for those items are defined in **Table 2-8** In addition to the previously described allowances for the contractor overhead, sales tax and estimating contingency. The cost of the storage tank will be based on the lowest generated cost from the two curves for the desired tank size (concrete tanks for a volume of 2 MG or less and steel for 2 MG or more). Tank material will not be indicated at this time as that decision will depend on a more detailed site assessment. The cost estimates will be marked up using the implementation cost allowances listed in Section 2.2.2.3.



es
(

Element	Allowance
Site Work	3% of Curve Cost
Site Piping/Fittings	\$150,000

2.3.4.1 Annual Storage Tank O&M

The annual O&M requirements of a storage tank are estimated at 1% of capital costs. No consumables or electrical needs are identified specific to storage tanks.



Figure 2-2: Welded Steel Storage Tank Construction Costs (with Bid Markups)







2.3.5 Groundwater Wells

Cost estimates for groundwater wells were adapted from the Oro Loma Sanitary District *Recycled Water Feasibility Study* (2016). Preliminary unit costs (escalated from the 2016 study) are presented in **Table 2-9**.

Table 2-9: Costs for Groundwater Wells

Element	Description	Unit Cost
Groundwater Injection Well	Assumes 1 MGD capacity, approx. 400-500 ft deep	\$1,679,000
Groundwater Extraction Well	Assumes 2 MGD capacity, approx. 400-500 ft deep and includes wellbead treatment for Mangapese	\$4 068 000

2.3.5.1 Annual Groundwater Well O&M

Groundwater well operations and maintenance includes labor, electricity, and consumables.

<u>Labor</u> – The annual labor requirements of a groundwater well mainly depend on the type of well: injection or extraction. Other minor factors, such as the well location, contingency measures, and the age of the well, would also affect the labor demands. Operators are expected to regularly tend to the groundwater wells to operate valves, start and stop pumps, and examine flow data. Routine maintenance may include the inspection of equipment, exercising of valves, and servicing instrumentation as well as backwashing for extraction wells. Estimates for operation and maintenance labor requirements are tabulated below in Table 2-10.



		•	
Well Type	Annual Operator Hours (Backwash)	Annual Maintenance Hours	Total Annual O&M Hours
Injection	0	104	104
Extraction	52	104	156

<u>Electricity</u> – Groundwater well electricity consumption is estimated only for extraction wells, as it is assumed that the injection wells are fed by head provided at the water source pump station. Electricity consumption for extraction wells is estimated by multiplying the pumping design point (in delivered head, feet), average flow (cubic feet per second [cfs]) and the annual hours in operation and dividing by the well efficiency. The calculation also includes conversion factors to produce a result in kilowatt hour (kWh). For this project, it has been assumed that the electricity consumption of the wellhead treatment is small in comparison to the extraction usage and has been accounted for in a conservative estimate of the depth of extraction. An example calculation is shown below:

- Example Groundwater Extraction Well
- Flow: 1 MGD (1.55 cfs)
- Operating Hours: 12 months, constant (8760 hours)
- Depth of extraction: 100 ft
- Pump Efficiency: 80%

$$\frac{1.55 \ [cfs] * 100 \ [ft] * \ 62.4 \left[\frac{lb}{ft^3}\right] * 0.7457 \left[\frac{kWh}{hp}\right] * 8760 [hr]}{550 \ \left[\frac{ft * lb}{hp * sec}\right] * 0.8 \ [-]} = 143,600 \ [kWh]$$

• <u>Consumables</u> – Groundwater well consumables are to be estimated as 0.5% of the estimated construction cost.

2.3.6 Advanced Treatment

The RWMP Update includes an assessment of potable reuse alternatives, each of which will require advanced treatment of municipal wastewater. Unit costs presented in this section for standard treatment processes are based on those developed by the consultant team for other advanced treatment projects, including the Santa Clara Valley Water District, Sacramento Regional County Sanitation District, and others. Preliminary unit costs, presented in **Table 2-11**, are based on a surface water augmentation treatment train, which is one of the alternatives under development for the RWMP Update. Costs for membrane bioreactors (MBRs) were developed differently because MBR technology is not specific to advanced treatment, and there are many facilities already constructed and operating. MBR costs were developed in bins based on facility size due to significant impact of economies of scale on the cost.



Treatment Process	Capital (\$M/MGD)
MBR (0.5 - 1 MGD)	16.0
MBR (1 - 5 MGD)	15.0
MBR (5 - 10 MGD)	11.0
MBR (>10 MGD)	10.0
Ozone	0.34
Biologically Activated Carbon (BAC)	0.30
Microfiltration (MF)	1.2
Reverse Osmosis (RO)	1.5
Advanced Oxidation and Disinfection	0.44
Free Chlorine Disinfection	0.25
Chemicals (storage and use)	0.13
Sitework/Piping/Structures	3.2

Table 2-11: Preliminary Advanced Treatment Processes Unit Costs

2.3.6.1 Annual Advanced Treatment O&M

In addition to O&M costs for all the treatment processes (including electricity and consumables), it was assumed that labor requirements for O&M at the facility will be approximately one full-time employee per MGD (or 2,080 hours/MGD). Lastly, for those alternatives which utilize existing water treatment plants, the additional O&M costs for those plants was included based on the District's fiscal year 2016 Annual Energy Report. The O&M costs are presented in below in **Table 2-12**.



Treatment Process	Annual O&M
AWT Facility	
MBR (0.5 - 1 MGD)	0.56 \$M/MGD
MBR (1 - 5 MGD)	0.54 \$M/MGD
MBR (5 - 10 MGD)	0.46 \$M/MGD
MBR (>10 MGD)	0.44 \$M/MGD
Ozone	0.09 \$M/MGD
Biologically Activated Carbon (BAC)	0.13 \$M/MGD
Microfiltration (MF)	0.34 \$M/MGD
Reverse Osmosis (RO)	0.57 \$M/MGD
Advanced Oxidation and Disinfection 0.07 \$M/MGD	
Free Chlorine Disinfection	0.03 \$M/MGD
Chemicals (storage and use)	0.12 \$M/MGD
Labor (assumes 1040 hours/MGD)	0.14 \$M/MGD
Major Equipment Maintenance and Repair	2% of applicable capital cost
Surface Water Treatment Plant O&M	
Walnut Creek WTP	0.03 \$M/MGD
Orinda WTP	0.03 \$M/MGD
Upper San Leandro WTP	0.09 \$M/MGD
Sobrante WTP	0.11 \$M/MGD

Table 2-12: Preliminary Advanced Treatment Annual O&M Costs

2.3.7 Non-Potable Treatment

Costs for non-potable treatment trains are based on previous study information, where available (adjusted to December 2017 dollars). The unit costs for MBR treatment presented in **Table 2-11** were applied for projects without previously estimated secondary treatment costs.

2.3.7.1 Annual Non-Potable Treatment O&M

O&M costs for non-potable treatment was based on previous study information, where available (adjusted to December 2017 dollars). Unit costs for O&M for MBR treatment were based on those listed in **Table 2-12**. Additionally, it was assumed that labor requirements for O&M at non-potable treatment facilities would be approximately one half of one full-time employee per MGD (or 1,040 hours/MGD).

This page is intentionally left blank.



3. NON-POTABLE REUSE OPPORTUNITIES

The District has been recycling water for non-potable uses within its water service area since 1971. The District would like to continue to plan, develop and implement recycled water projects throughout the water service area to offset potable water demands. This section evaluates the District's existing non-potable reuse projects, identifies new non-potable reuse projects, and provides an initialize screening and prioritization of projects to evaluate.

3.1 Existing Non-Potable Reuse Demands

The District's current recycled water goal is based on implementing a variety of non-potable reuse projects through 2040. The District has built infrastructure with the capability to provide over 9 MGD of recycled water for irrigation, commercial and industrial uses as summarized below. Full descriptions of each project are provided in Section 3.3.

- San Ramon Valley Recycled Water Program (DERWA/San Ramon) DSRSD and the District created the San Ramon Valley Recycled Water Program in 1995 through a joint powers authority referred to as the DSRSD-EBMUD Recycled Water Authority (DERWA). Phase 1 of the DERWA/San Ramon project currently provides between 0.5 MGD and 0.7 MGD of recycled water for irrigation. The recent completion of Phase 2B brings the total capacity to 0.8 MGD, and Phase 2A (currently in progress) will bring the total capacity to 1.3 MGD.
- <u>East Bayshore Recycled Water Facility (EBRWF)</u> The EBRWF began delivery of recycled water in 2008 and currently supplies recycled water primarily for landscape irrigation in Oakland and Emeryville, with annual average demands in 2017 at 0.14 MGD. The current project capacity is 0.2 MGD.
- <u>San Leandro Water Reclamation Facility (SLWRF)</u> The SLWRF was constructed in 1998 and provides secondary-treated and disinfected recycled water produced for irrigation purposes. The SLWRF has the capacity to supply up to 0.4 MGD of recycled water to customers (0.2 MGD annual average capacity). However, the recycled water demand has decreased in recent years. The SLWRF supply decreased from 0.1 MGD in 2015 to no use by 2017. The Chuck Corica Golf Complex currently uses a combination of surface water and groundwater to satisfy irrigation demands; and the Metropolitan Golf Links currently uses groundwater for irrigation. Therefore, the recycled water facility is not likely to be operated in the next year or two (see Section 3.3.3 for additional details).
- North Richmond Water Recycling Plant (NRWRP) The NRWRP as built in 1996 and currently supplies tertiary recycled water for cooling towers at the Chevron Richmond Refinery. NRWRP has a design capacity of 5.4 MGD, but typically produces about 4 MGD. In 2016 and 2017, the District's NRWRP experienced interruption of influent supply from West County due to construction shutdown. Therefore, the District had to supplement Chevron Richmond Refinery with potable water. The NRWRP is expected to be back in service by late 2018.
- <u>Richmond Advanced Recycled Expansion (RARE) Project</u> The RARE Project, constructed in 2010, supplies high-purity recycled water for boilers at the Chevron Richmond Refinery. RARE can produce up to 3.5 MGD but could easily expand to 4 MGD. A small portion (i.e., <10 percent) of RARE's recycled water demand was also supplemented with potable water due to water supply and water quality issues.

Figure 3-1 shows the annual average consumption for the existing recycled water projects, with corresponding values listed in **Table 3-1**. Capacities shown in Figure 3-1 are average annual production capacities. As shown, the recycled water demand for some projects has decreased in recent years. Other projects have experienced water supply and quality issues, requiring potable water make-up to meet customer demands.





Figure 3-1: Existing Centralized Recycled Water Projects (2013 to 2017)

Table 3-1: Recycled Water Deliveries, Annual Averages, 2013-2017

		Capacity	Recy	cled Wa	ater Deli	vered (I	MGD)
Project	Type of Use	(MGD)	2013	2014	2015	2016	2017
DERWA / San Ramon Valley Phase 1 & 2B	Landscape Irrigation	0.8 ¹	0.69	0.64	0.5	0.5	0.67
East Bayshore Phase 1A	Irrigation, Toilet Flushing, Industrial	0.2	0.15	0.16	0.14	0.13	0.14
San Leandro WRP	Golf Course and Landscape Irrigation	0.2	0.31	0.29	0.1	0.01	0
North Richmond WRP	Chevron Refinery Cooling Towers	4.0	3.64	3.64	4.0	0.5	0.0
RARE	Chevron Refinery Boiler Makeup	3.5	3.25	3.26	1.8	3.1	3.4
Total		8.7	8.0	8.0	6.5	4.2	4.2

Notes: 1. Capacity will increase to 1.3 MGD upon completion of Phase 2A (in progress). Total capacity will increase to 9.2 MGD.

3.2 Development of Non-Potable Reuse Alternatives

A full range of projects were considered including those that were identified as part of the WSMP 2040 as well as new opportunities that have been identified. The non-potable reuse projects have been categorized into centralized and satellite treatment facilities. Centralized projects consist of locating recycled water treatment facilities (e.g., filtration and disinfection) and distribution (e.g., recycled water pumping station) at a wastewater treatment plant (WWTP). A centralized project assumes that secondary treatment occurs at the WWTP and recycled water treatment facilities would only address processes needed to meet Title 22 regulations and/or customer water quality objectives. A centralized facility also includes the purple pipe distribution system to reach the recycled water customers.



Satellite treatment facilities consist of locating treatment and distribution facilities adjacent or in close proximity to the targeted recycled water customers, typically outside of the WWTP. The development of project alternatives includes consideration and evaluation of satellite treatment facilities in lieu of extending the purple pipe distribution system from a centralized facility.

This section summarizes the key assumptions and planning basis used to develop and evaluate proposed non-potable reuse projects. The assumed wastewater characteristics, water quality objectives and design basis used to size new potential satellite treatment facilities is discussed.

3.2.1 Demand Projections

Recycled water project demands were primarily updated from WSMP 2040 and/or other more recent District studies. In cases where a previous study was not performed, recycled water demands were estimated using historical water meter data, information from District Staff, and/or estimating demands based on irrigated area. When feasible, historical District potable water meter data was reviewed and used to update and/or develop customer irrigation demands. Three (2014 to 2016) to five (2012 to 2016) years of water meter data was used to estimate annual average and peak month demands for each customer. Irrigation water demand varies significantly between winter and summer months. The typical irrigation season is from April through October and it is the highest in July. The peak month demand was also calculated in some cases using a peaking factor of 1.9, which is based on the 95th percentile of daily demand to average annual demand ratios from January 1 to December 31, 2014. This peaking factor was developed for the *East Bayshore Recycled Water Expansion Study Hydraulic Analysis of Future Pipelines and Demands* (EBMUD, 2017a).

Recycled water demand updates were also coordinated with District to reflect new or updated conditions. For example, some customers in recent years have implemented water conservation measures or are using stormwater runoff or groundwater for irrigation in lieu of potable water. In cases where potential customers are currently using alternatives to potable water, the recycled water demands were eliminated and/or adjusted accordingly.

Table 3-2 summarizes project annual average demands in acre-feet per year (AFY) and relevant information sources.

 Refer to each project description for additional details.

	Annual Demand,	
Project	AFY	Source
DERWA Phase 1. Existing	600 to 800	District Recycled Water Program Data (2013 to 2017)
DERWA Phase 2. Bishop Ranch	800	WSMP 2040
DERWA Phase 3. Danville East	800	DERWA/San Ramon Valley Recycled Water Program Treatment and Distribution Costs (May 2018)
DERWA Phase 4. Blackhawk East	300	DERWA/San Ramon Valley Recycled Water Program
DERWA Phase 5. Blackhawk West	300	DERWA/San Ramon Valley Recycled Water Program
East Bayshore Phase 1A. Existing.	150 to 200	District Recycled Water Program Data (2013 to 2017)
East Bayshore Phase 1A	300 (1)	East Bayshore Recycled Water Quality Improvements Study (Brown and Caldwell, 2018a)
East Bayshore Phase 1B	1,064 (1)	East Bayshore Recycled Water Quality Improvements Study (Brown and Caldwell, 2018a)
East Bayshore Phase 2	2,867(1)	East Bayshore Recycled Water Quality Improvements Study (Brown and Caldwell, 2018a)
San Leandro WRF Expansion	0	District Recycled Water Program Data (2017)

Table 3-2: Sources of Demand Estimates for Development of Non-Potable Project Alternatives



Project	Annual Demand, AFY	Source
		Supply based on 2016 City of Richmond Facility Plan
Chevron/RARE Expansion	4,284	Demand adjusted based as described in Table 3-9.
Richmond Country Club	100	District Water Meter Data (2015 to 2016)
Point Richmond	120	WSMP 2040
Phillips 66 Refinery	Up to 4,144	ConocoPhillips San Francisco Refinery High-Purity Recycled Water Project Technical Study (Brown and Caldwell, 2007)
Franklyn Canyon	300	WSMP 2040
Lamorinda / Reliez Valley	100	District Water Meter Data (2012 to 2016)
Central San Regional	22,400	CCCSD's Final Comprehensive Wastewater Master Plan (Carollo and CH2M, 2017).
Contra Costa Pipeline in Canal ROW	900	CCWD Final Untreated Water Facilities Improvement Program Plan Update. (Carollo, 2013)
UCB Global Campus, Richmond	1,040	District water supply assessment (EBMUD, 2013).
Rolling Hills Cemetery	200	WSMP 2040
Diablo Country Club	250	Diablo Country Club Satellite Recycled Water Treatment Plant Feasibility Study (Brezack & Associates Planning, LLC, 2013)
Morana Area ²	250	District water meter data (2012 to 2016) for MCC and Moraga Commons. Miramonte Highschool and St. Mary's College do not have irrigation- specific accounts. Therefore, irrigation demands were calculated based on grass area
Orinda Country Club	0	District information (irrigation supply is creek water)
Mountain View/St. Man/'s	0	
Cemeteries	40	District water meter data (2014 to 2016)
Rossmoor Country Club	90	District water meter data (2012 to 2016)
UCB Main Campus, Berkelev	900	East Bayshore Recycled Water Expansion Study (EBMUD, 2017)
Oakland Hills	350	Oakland Hills Alternative Water Supply Feasibility Study (West Yost, 2017).

Notes:

1. East Bayshore Phase 1B includes Phase 1A demands. East Bayshore Phase 2 includes Phase 1A and Phase 1B demands.

2. Moraga Area customers include the Moraga Country Club (MCC), Moraga Commons, Miramonte High School and St. Mary's College.

3.2.2 Water Quality Objectives

The primary non-potable end use identified for the non-potable reuse alternatives is landscape irrigation; however, there are several projects that consider recycled water for cooling tower makeup water and/or boiler feed water. In the absence of specific customer recycled water quality objectives, it is assumed that recycled water would meet the water quality objectives shown in **Table 3-3**. The values in **Table 3-3** reflect the objectives established as part of the East Bayshore Recycled Water Facility (EBRWF), which assume tertiary treatment and disinfection to meet the Title 22 regulations for non-potable unrestricted reuse as well as total dissolved solids (TDS) and chloride concentrations to meet customer end use requirements (BC, 2018a). Unless otherwise noted, project costs in this study assume that the objectives can be met without additional treatment for salt removal. As shown in **Table 3-3**, the principal recycled water quality constituents of concern for landscape irrigation are total dissolved solids (TDS), chloride, sodium and boron. The sodium adsorption ratio (SAR) is also a concern for irrigation. Recycled water quality criteria vary and are



dependent on the type landscape vegetation and the method of irrigation (drip irrigation or sprinkler irrigation). The projects described in Section 3.3 assume, unless noted otherwise, that the focus for landscape irrigation is non-salt sensitive species.

The required recycled water quality characteristics for cooling tower makeup water may vary for different customers and is dependent on the age, materials of construction of the cooling towers, operating characteristics of the cooling towers, and the level of pretreatment currently performed on the circulation water. In general, the primary issues of concern in industrial cooling tower applications are scaling, fouling and corrosion. The cycles of concentration (COC) setpoint is an operating characteristic of the cooling tower and is established based on the makeup water quality and water pretreatment. The COCs are equal to the ratio of circulating cooling water concentrations to fresh makeup water concentration. Since recycled water generally has higher mineral and nutrient level compared to potable water, fewer COCs are recommended. Alternatively, chemical pre-treatment and/or production of higher quality recycled water are alternatives to reducing the COC. The District had previously established water quality objectives based on 3.5 COC (EBMUD, 2016) for light industrial/commercial applications. With heavy industrial customers, such as refineries, water quality objectives were established based on discussions with the industrial customer. Because existing industrial/cooling customers use (high quality) potable water, additional treatment to reduce ammonia, metals and salt concentrations would likely be needed, similar to the District's existing facilities delivering recycled water to the Chevron Richmond Refinery.

	Irrigation		Industrial Cooling	
Parameter	Grasses ²	Sensitive Species ³	Towers and HVAC ⁴	
Ammonia (mgN/L)	NA	NA	0.6	
Chloride (mg/L)	<350	100	<71	
Total Dissolved Solids (mg/L)	<1,670⁵	1,000 to 2,000	<430	
Sodium Adsorption Ratio (SAR)	<9	3	NA	
Boron (mg/L)	2.0 to 4.0	0.5 to 1.0	NA	

Table 3-3: Typical Recycled Water Quality Objectives¹

Notes:

1. Objectives in this table were used for projects that do not have established customer specific recycled water quality objectives. Based on Brown and Caldwell, 2018a.

2. This includes general turf grasses and native species such as California hairgrass, California melic, and pine bluegrass.

3. This includes butterfly bush, trumpet vine, liquidamber, ginkgo, roses, and Chinese pistache.

4. Industrial uses include industrial cooling (based on 3.5 COCs [EBMUD, 2016a]) and building heating, ventilation, and air conditioning (HVAC).

5. This value assumes a leaching fraction (i.e., the amount of additional irrigation water that must be applied) of 15 percent to flush salt below the root zone and minimize detrimental impacts to the vegetation.

3.2.3 Project Costs

This RWMP Update includes the evaluation of a wide range of potential recycled water projects from a variety of resources, references and authors. Section 2 includes the methods for reviewing and updating costs for projects developed as part of previous work as well as methods for developing costs for new projects for both capital and operations and maintenance (O&M) costs. Specific information developed for each project is included in the cost spreadsheets in Appendix A.

In general, non-potable reuse alternative project descriptions and costs are based on work previously developed by the District and other agencies. Cost estimates for these projects were reviewed for any major omissions and revised



if needed. Whenever possible, the raw construction costs were extracted and updated to December 2017 dollars using 20-City Average Engineering News Record's (ENR) Construction Cost Index (CCI) ratios. The soft costs and allowances defined in the Section 2 were then applied to the raw construction costs to estimate capital costs. Adjustments due to location were not necessary for non-potable reuse alternatives.

3.2.4 Raw Wastewater Characteristics

Similar to the demands and project costs, when facility sizing was not available for the non-potable reuse projects, additional evaluation was performed to size the treatment facilities. This primarily was limited to select satellite treatment facilities. **Table 3-4** presents the raw wastewater characteristics assumed for sizing satellite treatment facilities. The data presented in **Table 3-4**, was based on recent data collected from the District's Adeline Interceptor and SD-1 raw influent and used for the EBRWF Water Quality Improvements Project. As part of the EBRWF Water Quality Improvements Project. As part of the EBRWF Water 2017. Samples were collected from the Adeline Interceptor and the SD-1 raw influent using portable composite samplers. Analyses included TDS, total suspended solids (TSS), biochemical oxygen demand (BOD), total Kjeldahl nitrogen (TKN) and ammonia. Recommended sampling parameters were summarized in the *East Bayshore Recycled Water Facility Water Quality Evaluation Draft Technical Memorandum* dated October 2017.

It should be noted that unless sewer system flow data was available at the diversion point for the satellite facility, it was assumed that there was adequate flow in the sewer system for the satellite facility. As the non-potable projects are evaluated and further developed, this assumption should be confirmed.

Parameter	Value	Source ¹
Maximum Month Flow, MGD	varies	Max Month condition
Biochemical Oxygen Demand, mg/L	410	90th percentile SD-1 Raw Influent
Ammonia, mgN/L	37.6	90th percentile SD-1 Raw Influent
Total Suspended Solids, mg/L	360	Max. value at Adeline Interceptor
Total Phosphorus, mgP/L	7.0	Assumed
Alkalinity, mgCaCO₃/L	330	90th percentile SD-1 Raw Influent
Total Dissolved Solids, mg/L	370	Max. value at Adeline Interceptor
Total Kjeldahl Nitrogen, mgN/L	59.5	90th percentile SD-1 Raw Influent

Table 3-4: Typical Raw WW Quality for Satellite Treatment Facility Sizing

Notes:

1. District data for SD-1 raw influent and Adeline Interceptor, except where noted.

3.2.5 Treatment System

As noted earlier, several non-potable reuse project alternatives have been developed by the District in earlier studies. Facility sizing was developed for those project alternatives that did not have additional information. Recycled water treatment facilities were assumed to operate 24-hours per day and were sized to meet the peak (or maximum) month recycled water demand. It was assumed that potable water could be used as an emergency backup supply such that a fully redundant treatment train was not included for the facilities. Equalization for influent flows was not included but would need to be verified as projects are further developed. For the estimating size and cost of satellite treatment facilities, it was assumed that an MBR followed by UV disinfection would be used. **Figure 3-2** shows a conceptual diagram of the treatment train.





Figure 3-2: Conceptual Satellite Treatment Train Schematic

Note: Raw wastewater would be diverted from sanitary sewer system. Excess sludge would be discharged to the sewer downstream of influent diversion point.

MBR systems are common for satellite treatment facilities because an MBR combines biological treatment with solidsliquid separation and filtration (eliminating the need for separate secondary clarifier and filtration structures), and it produces high quality effluent with a reduced footprint. Fine screening (1 to 3 mm) is placed upstream of the MBR system to prevent clogging of the filters. Waste activated sludge is assumed to be discharged to the local sewer, downstream of the influent diversion point. UV disinfection is assumed because of its small footprint and safety benefits from avoiding the storage of chemicals. The UV system was sized based on a 65 percent UV transmittance and 80 mJ/cm² UV dose. To maintain a disinfection residual in the distribution system, an onsite hypochlorite generation system was included. The hypochlorite system was sized to deliver a 5 mg/L chlorine residual at max month conditions.

3.2.6 Distribution System

Recycled water produced would be diverted to a recycled water storage tank, as the treatment system was assumed to operate over a 24-hour period. The distribution system was sized to meet peak hour demands, assuming that irrigation occurs over an 11-hour period. Industrial demands were assumed to occur over a 24-hour period and dual plumbing demands were typically assumed to occur over an 8-hour period. The proposed storage tank was sized to provide one day of storage at max month demands. The addition of booster pumps, on-site customer retrofits, and storage along the distribution system are specific to each project alternative and are noted (if needed) under the project descriptions. **Table 3-5** summarizes recycled water distribution system design criteria, including pipe material and discharge pressure.



Item	Planning Basis	
Storage Tank Sizing	1 day of max month demands	
Storage Tank Material ¹	Welded Steel or Pre-Stressed Concrete	
Distribution System Sizing	Peak Hour	
Pump Redundancy Criteria	2 duty, 1 standby unit	
Pump Efficiency	50 percent	
Demand Period ²		
Irrigation	8 to 11-hours per day	
Dual Plumbing	8-hours per day	
Industrial	24-hours per day	
Discharge pressure (at customer site) ³	30 to 50 psi	
Pipeline Material	High Density Polyethylene (HDPE)	
Assumed Pipeline Headloss	1 foot per 1,00 feet of pipe	
Maximum Pipeline Velocity at Peak Hour	5 ft/sec	

Table 3-5: Recycled Water Distribution Design Criteria

Notes:

- 1. The most cost-effective material was selected based on the size of tank (concrete for <2 MG and steel for >2 MG).
- Typical demand period by customer type assumed unless noted otherwise. An irrigation demand period between 8 to 11 hours per day is assumed. The upper range is based on the peak hour factor of 2.2 (24 hrs/2.2 = 11 hrs per day, average peak hour demand to daily demand ratio) used in the East Bayshore Recycled Water Expansion Study Hydraulic Analysis for Future Pipelines and Demands (EBMUD, 2017a).
- 3. Assumed recycled water discharge pressure at the customer site unless noted otherwise.

3.3 Centralized Non-Potable Reuse Alternatives

This section summarizes the centralized non-potable reuse alternatives evaluated. Project descriptions include annual average demands, available cost information and non-economic considerations.

3.3.1 San Ramon Valley Recycled Water Program

DSRSD and the District created the multi-phase San Ramon Valley Recycled Water Program in 1995 through a joint powers authority referred to as DERWA. DERWA was established to supply recycled water through the construction and operation of a water recycling facility with a planned capacity of up to 5.7 million gallons per day. The water recycling facility started operation on February 1, 2006. DSRSD currently supplies water to parts of Dublin and the Dougherty Valley, while EBMUD serves recycled water to portions of San Ramon (DERWA/San Ramon).

The project has historically been planned, designed and constructed in a series of numbered phases; Phase 1 is complete while Phase 2 is nearing completion. **Figure 3-3** shows the pipeline alignments and **Table 3-6** summarizes the status for each phase. In 2016, the District completed installation of a pipeline in the Bishop Ranch Business Park area of San Ramon (DERWA/San Ramon Phase 2). Per the current construction schedule, subsequent phases will be constructed in the following order: Phase 3, Phase 5, and Phase 4. In sum, Phases 1 through 5 will serve an annual average of 2.5 MGD (2800 AFY) of recycled water to the District irrigation customers in parts of Blackhawk, Danville and San Ramon.



Phase	Status ¹	Annual Demand (AFY)
1. Existing	In operation since 2006	600 to 800 ²
2. Bishop Ranch	Near completion	800 ³
3. Danville East	Pipeline construction FY 24-25	8004
4. Blackhawk East	st Pipeline construction FY 33-34	3004
5. Blackhawk We	st Pipeline construction FY 28-29	3004
	Total	2,800

Table 3-6: DERWA/San Ramon Project Status

Notes:

1. Based on information provided by District staff. Timing of phases 3 to 5 will depend on securing supplemental supplies.

2. Source: District Recycled Water Consumption Data (2013 to 2017).

3. Source: WSMP 2040.

4. Source: DERWA/San Ramon Valley Recycled Water Program Treatment and Distribution Costs (May 2018).

DERWA is currently expanding its recycled water treatment plant to increase treatment capacity from 9.7 MGD to 16.2 MGD and meet future recycled water demands. The Phase 2 Recycled Water Treatment Plant Expansion Project construction began in 2017 and the new system is expected to be on-line for the 2018 irrigation season. Per the DERWA FY18-19 budget, the partners are allocating capital cost share and any funding secured as follows: DSRSD – 46%, the District – 27%, City of Pleasanton – 27%. The expanded treatment capacity is expected to be fully utilized by 2020 during the summer irrigation season.

DERWA has experienced peak month supply shortfalls during the summer season, requiring supplementation with potable water. Therefore, DERWA is considering requesting customers to reduce use and switching a few customers to potable supply to meet summer demands if needed. DERWA is also exploring other additional supply opportunities, including groundwater, recycled water from CCCSD, and diversion of raw wastewater from CCCSD's adjacent sewerage to supplement DERWA's recycled water supply.





Figure 3-3: DERWA/San Ramon Distribution System

Table 3-7 summarizes EBMUD's share of capital and O&M costs associated with this multiphase project (treatment and distribution). All costs are future costs based on FY 2018 capital improvement projects (CIP) budget (September 2016 dollars updated to December 2017 dollars). For budget purposes, Phases 1 and 2 are complete. Approximately 2.8 miles of distribution pipeline have been completed for Phase 3 with a corresponding investment of \$2.6 million. An additional \$22 million of future capital will be invested to build 5.4 miles of distribution of pipeline for Phase 3. This assumes that the pipeline in Crow Canyon Rd. west of Dougherty Road (shown as Optional in **Figure 3-3**, above) will not be constructed. The District will also contribute \$5.6 million for DERWA Phase 3 Treatment. The District's share of Phase 3 O&M for distribution and treatment are \$0.15 million and \$0.35 million, respectively. These are the costs to operate the defined project phase and do not include the costs to operate the existing system.


	2018 Dollars ¹			201	7 Dollars ²	
Phase	Capital Cost (\$M)³	O&M (\$M/yr)	Capital Cost (\$M)	O&M (\$M/yr)	Annual Demand (AFY)	Annualized Total Cost (\$/AF)4
Phase 3						
Treatment/Supplemental Supply	2.2	0.35	2.0	0.35		600
Distribution (5.8 miles pipe & Pump Station R3000)	23.9	0.15	23.0	0.15		1,300
Total	26.1	0.50	25.0	0.50	800	1,900
Phase 4			•			
Treatment Expansion	2.5	0.13	2.0	0.13		400
Distribution (1.4 miles pipe & Pump Station R4000)	15.4	0.05	15.0	0.05		2,500
Total	17.9	0.18	17.0	0.18	300	2,900
Phase 5						
Treatment Expansion	4.1	0.13	4.0	0.13		1,000
Distribution (2.8 miles pipe, no pump station)	4.2	0.02	4.1	0.02		600
Total	8.3	0.15	8.1	0.15	300	1,600

Table 3-7: DERWA/San Ramon Project Costs, District's Share

Notes:

1. Source: DERWA/San Ramon Valley Recycled Water Program Treatment and Distribution Costs (September 2018). District's share for capital and O&M are shown. All costs are future costs based on FY2020 CIP budget (September 2018 dollars).

2. Capital and O&M costs (from source) were updated to 2017 dollars using ENR CCI ratios for 20 Cities Average.

3. All capital costs include planning, design, construction of infrastructure and retrofits. Assumes the pipeline in Crow Canyon Rd. west of Dougherty Rd is not constructed.

4. Annualized total cost is calculated using methodology outlined in Section 2.2.

These projects are expansions of committed projects currently in progress or construction. The programs are EIRcertified, and the recycled water is distributed to a wide service area. Overall, this is a well-established program and these projects should continue to be high priority for the District as long as funding and source supply are available. It is worth noting that Phase 4 has the highest unit costs between the three remaining phases as it includes installation of a pump station (R4000, per District nomenclature). Compared with Phase 5, Phase 4 capital and O&M are higher for the same annual demand, resulting in a higher unit cost.

3.3.2 East Bayshore Recycled Water Project

The EBRWF currently supplies recycled water primarily for landscape irrigation in Oakland and Emeryville. The recycled water quality meets DDW Title 22 requirements for unrestricted non-potable reuse. The EBRWF provides microfiltration (MF) and chlorine disinfection of the District's secondary effluent at the Main WWTP (SD-1). A portion of recycled water has been used for cooling tower makeup and for toilet/urinal flushing, but these uses were largely discontinued due to water quality issues. Recycled water is currently used at SD-1 for in-plant uses such as pump seal water and irrigation; SD-1 demands range from 0.5 MGD to 1.0 MGD.



In 2017, annual average recycled water demands outside of SD-1 were 0.14 MGD, which is lower than the initial goal of 0.5 MGD for the first phase of the project (Phase 1A). The District wants to maximize recycled water use in the EBRWF service area by expanding the distribution system to Berkeley, Albany and Alameda and by expanding the customer base to include commercial cooling towers, dual plumbing and other industrial uses.

The *East Bayshore Recycled Water Quality Improvements Study* is evaluating EBRWF expansions and treatment upgrade requirements based on these target end uses. The EBRWF Study has not yet been finalized (as of December 2018), but the draft report identified short-, intermediate- and long-term scenarios to improve recycled water quality and recommended next steps to ultimately identify a path forward for the EBRWF recycled water program (Brown and Caldwell, 2018a).

Two alternatives were developed: Alternative 1 consists of delivering recycled water for landscape irrigation of nonsensitive species only, while Alternative 2 consists of delivering recycled water for both industrial purposes and irrigation. Alternative 1 and Alternative 2 would both require treatment upgrades to improve recycled water quality and meet the established recycled water quality objectives. Alternative 1 would require partial RO to reduce chloride concentrations in the recycled water. Alternative 2 would require ammonia removal and chloride reduction. Two concepts were reviewed for ammonia and chloride reduction; the first concept considers treating tertiary effluent with RO followed by ion exchange (IX). The second concept considers constructing an MBR at SD-1 to treat wastewater from the Adeline Interceptor; an initial review of wastewater quality from the Adeline Interceptor indicates that chloride concentrations are low enough that RO would not be needed downstream of the MBR. Additional characterization of the Adeline Interceptor wastewater is recommended to confirm chloride concentrations and confirm that RO is not needed.

A decision to move forward with Alternative 1 or Alternative 2 has not been made yet. Additional studies and pilot testing are recommended to confirm that recycled water (after the proposed water quality improvements) is suitable for use in the commercial cooling systems. For this reason, a decision on whether to move forward with Alternative 1 or Alternative 2 has not been made and for planning purposes, Alternative 2 is referenced in this report as it is more conservative (i.e., Alternative 2 has higher capital cost estimates than Alternative 1). Alternative 2 also offers the benefit of a more diverse customer base, year-round demands, and requires less distribution system expansion than Alternative 1.

The recommended treatment upgrade for Alternative 2 is the MBR treatment facility at SD-1 capable of producing 4.5 MGD of maximum month flow. This assumes that Adeline Interceptor wastewater quality is such that RO treatment is not needed. The MBR offers the advantage of eliminating the need for RO and it achieves nutrient removal at SD-1 (i.e., nutrients are not returned to the plant in RO waste streams). The MBR can also accommodate variability in influent ammonia loadings while still meeting recycled water quality objectives. The long-term project (Alternative 2, Phase 2) would achieve annual average deliveries of 2.6 MGD to irrigation and industrial customers.

Alternative 2 consists of delivering recycled water to industrial and landscape irrigation customers. The phasing plan targets the following deliveries (not including SD-1 in-plant use):

- Phase 1A (Short-term project). 500 AFY (0.44 MGD annual average) to existing distribution system customers (not including SD-1 in-plant use) plus delivery to new customers within the existing distribution network and Frontage Road (I-80) pipeline alignment up to University Village.
- **Phase 1B** (Intermediate-term project). 1,100 AFY (0.95 MGD annual average). This phase includes all facilities in Phase 1A plus minimal expansion of the distribution system in Oakland and Berkeley to reach new irrigation and industrial users.
- **Phase 2** (Long-term project). 2,900 AFY (2.6 MGD annual average). This phase includes all facilities in Phase 1A and Phase 1B plus expansion to UC Berkeley, Albany and Alameda.



Figure 3-4 shows the existing and proposed recycled water distribution system for Phases 1A, 1B and 2, which are discussed in further detail below. **Appendix D** summarizes the list of customers for East Bayshore Alternative 2. The long-term project (Phase 2) includes all customer demands and the associated cost with the ultimate distribution system expansion.

EBRWF Alternative 2 Short-term Project (Phase 1A)

Short-term demands were identified that reach customers along the Frontage Road (I-80) pipeline alignment and adjacent to the existing distribution network to limit recycled water distribution system expansions. In addition to the EBMUD Administration Building, new industrial customers will be connected in Emeryville and an allowance for recycled water use was included for the Sherwin Williams redevelopment site. Treatment upgrades would be required to meet new end user water quality objectives for chloride and ammonia. While the MBR is the recommended alternative in the long-term, cost estimates were not prepared for an MBR in the short-term because the MBR would be implemented as part of the larger project when demands are higher. The cost associated with this phase include treatment upgrades assuming RO and ion exchange as a placeholder. Because full-scale implementation of Alternative 2 will take time, EBMUD could consider early extension of the distribution system to Brooklyn Basin with potable water blending to meet chloride objectives until treatment upgrades are constructed.

EBRWF Alternative 2 Intermediate-Term Project (Phase 1B)

As shown in **Figure 3-4**, Phase 1B would include minimal expansion of the existing recycled water distribution system in Oakland and Berkeley and a new separate recycled water treatment facility sized to meet the projected max month demand of 1.5 MGD (not including SD-1 reuse demands). Note that under the MBR scenario, recycled water for inplant use at SD-1 is assumed to come from secondary effluent. Recycled water used at SD-1 would not meet Title 22 requirements for unrestricted reuse; SD-1 would use recycled water under the same protocol that is currently used when EBRWF is offline.

A new recycled water treatment facility would be sited at the existing East Bayshore site at SD-1 and would use an MBR to treat raw wastewater from the Adeline Interceptor. A diversion pump station would convey raw water from the Adeline Interceptor to the MBR treatment process at SD-1. Land to the east of SD-1 would have to be acquired from the California Department of Transportation (Caltrans) in order to construct the diversion pump station. The MBR alternative is comprised of screening, grit removal, activated sludge basins, MBR tanks, blowers, and chlorine disinfection. The existing disinfection basins would be repurposed. The MBR effluent would be disinfected, stored and pumped using the existing EBRWF facilities. MBR treatment would address water quality improvements needed to serve irrigation and industrial customers. With this alternative, the existing MF facilities would no longer be needed. **Figure 3-5** shows a schematic for the treatment upgrades assumed in this master plan update.

EBRWF Alternative 2 Long-term Project (Phase 2)

The long-term project includes extension of the distribution system to Powell Street, Channing Way and Alameda. Phase 2 includes a satellite MBR treatment facility at SD-1 with a capacity of 4.5 MGD. Because of the larger footprint required for the long-term MBR facility, the existing EBRWF disinfection facilities would need to be relocated. The MBR facility would be located at the EBRWF location. It is recommended that alternate sites be considered, as there is limited land available and constructability is a potential issue.

The Phase 2 distribution system expansion would require a significant amount of pipeline construction through congested urban areas, so construction challenges are anticipated. Phase 2 will expand service to include University of California Berkeley (UCB). If the UCB Satellite Treatment Project were to be selected for implementation, its demands would have to be subtracted from the EBRWF Phase 2 demands.



The East Bayshore Recycled Water Project Phases 1A, 1B and 2 are expansions of a committed project, and therefore score well with respect to institutional complexity. Because the system serves a large number of customers, it also meets the District's environmental and social objectives.

The proposed satellite treatment at SD-1 would consist of a smaller, separate MBR treatment facility which would require construction and implementation of new processes. The benefit of this option is that recycled water produced at SD-1 would have lower salt and nutrient concentrations which would also facilitate long-term plans for potable reuse. Implementation of new processes like MBR will impact the District from a long-term operational complexity, but could set up the District to gain experience on new processes which may be implemented to address future nutrient regulations.

As noted in the EBRWF study, it is recommended that the District continue to perform outreach to future customers to confirm interest, demands, and recycled water quality requirements. In addition to continued outreach, it is also recommended that the District work with the targeted cities to identify opportunities to provide recycled water connections with new construction and/or upgrades/improvements at public facilities.





Figure 3-4: East Bayshore Distribution System





Figure 3-5: Recycled Water Treatment System Reconfiguration for Satellite Treatment of Adeline Interceptor with MBR

Table 3-8 (next page) shows the costs associated with each of the phases for the EBRWF Water Quality Improvements Project. The cost estimates shown are standalone and do not consider a phased approach for each alternative (i.e., Phase 2 includes Phase 1A and 1B; the costs are not additive).



Phase	Capital Cost (\$M)	O&M Cost (\$M/yr)	Annual Demand (AFY)²	Annualized Total Cost (\$/AF)⁴
EBRWP – Alternative 2 Phase 1A				
Treatment Upgrades	6	0.35		1,300
Recycled Water Conveyance	10	0.15		1,100
Phase 1A Total	16	0.50	500	2,400
EBRWP – Alternative 2 Phase 1B ³				
Treatment Upgrades	13	0.38		900
Recycled Water Conveyance	27	0.45		1,500
Phase 1B Total	40	0.83	1,100	2,400
EBRWP – Alternative 2 Phase 2 ³				
Treatment Upgrades	22	1.2		800
Recycled Water Conveyance	108	1.7		2,200
Phase 2 Total	130	2.9	2,900	3,000

Table 3-8: East Bayshore Recycled Water Project Costs (2017 dollars)¹

Notes:

1. Source: (Brown and Caldwell, 2018b)

2. Annual demands include existing uses, but do not include in-plant uses at SD-1.

3. Costs for Phase 1B include all costs required to provide recycled water for Phase 1A and Phase 1B customers.

Costs for Phase 2 include all costs required to provide recycled water for Phase 1A, Phase 1B and Phase 2 customers.

4. Annualized total cost is calculated using methodology outlined in Section 2.2.

3.3.3 San Leandro Water Reclamation Facility Expansion Project

Since 1988, the San Leandro Water Reclamation Facility (SLWRF) has been providing secondary-treated and disinfected recycled water produced by the City of San Leandro's Water Pollution Control Plant (WPCP) to customers for irrigation purposes. The District constructed facilities to convey recycled water to the Chuck Corica Golf Complex and roadway medians along Harbor Bay Parkway in Alameda and to the Metropolitan Golf Links in Oakland. The Monarch Bay Golf Club in San Leandro is also a recycled water customer, but it is supplied directly by the City of San Leandro.

The SLWRF has the capacity to supply up to 0.4 MGD of recycled water to customers. However, the recycled water demand has decreased in recent years from 0.1 MGD in 2015 to 0.01 MGD in 2016. The Chuck Corica Golf Complex has invested in alternative landscaping and stormwater capture, and currently uses a combination of surface water and groundwater to satisfy their irrigation demands. The Metropolitan Golf Links currently uses groundwater to satisfy their irrigation demands. Both golf courses intend to close their recycled water accounts, but the meters will stay in place just in case future use is needed. Harbor Way alone does not create enough demand to warrant turning on the recycled water pump station, therefore, the recycle water facility is likely not to be operated unless a more consistent demand can be identified along the existing recycled water distribution system.

In 2016, the City of San Leandro Recycled Water Market Assessment Study was completed (Carollo, 2016b). Based on 2002 to 2005 water demand data, 15 potential recycled water customers were identified for various uses (irrigation, commercial, food and light industrial). The projected annual average and max month demands were 0.15 MGD (171 AFY) and 0.56 MGD, respectively. However, many potential customers were not interested in non-potable reuse. **Figure 3-6** shows the existing and potential customers for the SLWRF. While the Oakland International Airport is



approximately one mile from the SLWRF, demands are low and currently not plumbed with purple pipe. A future expansion of the airport in the form of a third terminal could potentially provide a demand for non-potable reuse (i.e., toilet flushing), but such an expansion is not planned at this time. The Waste Management Facility (Davis Street Resource Recovery Complex and Transfer Station), sited across the street from SLWRF, is another potential customer however, additional facilities would be required to comply with the disinfection requirements (i.e., a contact basin would be required) because of the limited contact time in the distribution system.

There is insufficient customer demand to make this recycled water project viable at this point in time.







3.3.4 Chevron Richmond Refinery Recycled Water Project

Since 1996, the District has delivered recycled water to the Chevron Richmond Refinery. The District currently operates two treatment plants, the Richmond Advanced Recycled Expansion (RARE) Water Project and the NRWRP, both supplied primarily by secondary-treated effluent produced by the West County WPCP. The RARE Water Project supplies high-purity recycled water for the high-pressure boilers and the NRWRP supplies tertiary recycled water for the cooling towers at the refinery. **Figure 3-7** shows the existing recycled water distribution system.



Figure 3-7: Chevron Richmond Refinery Recycled Water Project

The RARE Water Project has a capacity of 3.5 MGD, but the facility can be easily expanded to 4.0 MGD with the installation of additional microfiltration modules. Further expansion would require the construction of additional facilities. The NRWRP is more than 20 years old, requires significant maintenance and is challenged by the variable quality of the secondary effluent it receives making it difficult to meet Chevron's water quality requirements at times. In 2017, the



West County WPCP completed several plant expansion and upgrade projects, including converting the aeration basin configuration to Modified Ludzack-Ettinger (MLE) mode with enhanced nitrification reliability and facilitated denitrification. The expansion and reduced nutrient loads and ultimately improved the water quality of the secondary effluent treated at NRWRP and RARE.

The District is interested in exploring an expansion of recycled water use at the refinery with a goal of 10 MGD, however additional water supply is needed. **Table 3-9** summarizes the current and future recycled water demands and supply. Based on historical data review of the WCWD effluent flows (2011 to 2014), the annual average and minimum month flows are 8 MGD and 6.4 MGD, respectively. During the dry months, there is a projected deficit of 4.5 MGD in meeting future refinery demands. Given the deficit in WCWD effluent supply, the District has identified the refinery WWTP effluent and the City of Richmond's WPCP effluent as potential sources.

Demand and Supply	Current (MGD)	Future (MGD)
Chevron Refinery Cooling Towers Demand	3.5	5.0
Chevron Refinery Boilers Demand	3.0	5.0
Total Chevron Refinery Demand	6.5	10.0
Total Supply to NRWRP/RARE ¹	7.0 ¹	10.9 ¹
WCWD Effluent ²	6.4 ²	6.4 ²
Supply Deficit	0.6	4.5

Table 3-9: Chevron Richmond Refinery Recycled Water Project Demand Deficit

Notes:

 The District's RARE system has an overall recovery of 85 percent. To meet current and future boiler demands of 3.0 and 5.0 MGD, the feed flow to RARE would be 3.5 MGD and 5.9 MGD, respectively. Feed flow to NRWRP is 3.5 MGD (current) and 5.0 MGD (future).

2. Value represents the minimum monthly flow between 2011 to 2014. The minimum monthly flow in 2015 was 5.9 MGD.

3.3.4.1 Chevron Refinery WWTP Effluent

In 2016, the District completed a study to evaluate the feasibility of increasing recycled water production at the RARE Water Project to 10 MGD (BC, 2016). The study evaluated the feasibility of using the refinery's process flows or WWTP effluent as an additional influent water source. RO concentrate from the RARE Water Project is currently treated at the refinery's WWTP. Therefore, pilot testing would be required to determine the feasibility of WWTP effluent as a source for the RARE Water Project's MF/RO system, as well as the impacts back at the WWTP from the expanded need for RO concentrate management. In addition, further investigation is needed to determine the potential NPDES permit modifications that would be required. The additional studies needed would require significant input and financial support from Chevron to move forward and is therefore not considered viable at this time.

3.3.4.2 Richmond WPCP Effluent

Another potential source of recycled water for the RARE Water Project is effluent from the City of Richmond WPCP. In 2015, the annual average flow and ADWF were 6 MGD and 4.4 MGD, respectively. Per the 2016 Facility Plan, the projected 2040 ADWF is 7.4 MGD (Carollo, 2016a). In order to use available Richmond WPCP effluent, treatment upgrades would be required to meet the District's RARE influent water quality limits for salinity and ammonia. The 2016 Facility Plan analyzed alternatives for reducing both constituents and identified the need for a 5-MGD MBR followed by RO and UV. A portion of the UV-disinfected effluent would be chlorinated to meet the District's chlorine residual requirement. This project would allow the RARE facility to expand to 5.0 MGD recycled water delivery to Chevron, therefore the costs associated with the RARE Water Project expansion were also included. The Richmond WPCP would be the primary supply for the RARE Water Project and a portion of the flow originally supplied by WCWD could



be utilized at the NRWRP. **Figure 3-8** depicts the water balance for the project. As shown, the District would be able to produce up to 10 MGD recycled during months of minimum supply from WCWD with additional supply from Richmond WPCP effluent. **Table 3-10** summarizes the capital costs of the facility improvements at the Richmond WPCP and the conveyance system to the RARE facility. The cost estimates presented assume that Richmond WPCP effluent meets the RARE feed requirements. Impacts to the District's long-term operation are anticipated to be minimal since the new treatment facilities will be operated by the City of Richmond.

It is worth noting that the recycled water project was not a recommended project in the 2016 Facility Plan and therefore was not carried forward into the City of Richmond's capital improvement plan (CIP). The primary reasons for not recommending the project included the need for interagency coordination and financial support for the treatment and distribution facilities. The 2016 Facility Plan stated that these challenges might be resolved when the need for recycled water becomes sufficiently critical.

Other projects competing for the same source of water include the Richmond Country Club and the Rolling Hills Cemetery as discussed below.

	2016 Dollars ¹		2017 D	2017 Dollars ²		
Element	Capital Cost (\$M)	O&M (\$M/yr)	Capital Cost (\$M)	O&M (\$M/yr)	Annual Demand (AFY) ⁶	Annualized Total Cost (\$/AF) ⁷
Treatment Upgrades ³	91.2	NA	96	4.8		2,300
RARE Expansion ⁴	2.0	0.45	2.1	0.47		
Recycled Water Conveyance⁵	11.4	0.090	11.9	0.44		300
Total	105	0.54	110	5.7	4,300	2,600

Table 3-10: City of Richmond WPCP Recycled Water Project Capital Costs

Notes:

1. Source: City of Richmond and Veolia Water WWTP Facility Plan. Final Draft. September 2016.

2. 2016 capital costs (from source) were updated to December 2017 dollars using ENR CCI ratios for 20 Cities Average.

3. Includes ammonia removal, salinity reduction and disinfection facilities. Two options were presented to meet nutrient removal: expansion of existing conventional activated sludge (CAS) process (aeration basins and secondary clarifiers) and split flow CAS/MBR which includes expansion of existing CAS process and a separate MBR train. The cost for the split CAS/MBR with RO/UV facilities, but without the aeration basin improvements to achieve Level 3 nutrient requirements is shown here. Annual O&M cost were not provided in the original estimate. 2017 O&M cost estimates for this study are presented in Appendix A.

4. Includes costs to expand RARE facilities from 3.5 MGD to 5.0 MGD production rate (Source: BC, 2016).

- 5. Includes cost for constructing a new pipeline to RARE and a new recycled water pump station to pump secondary effluent to RARE. The annual O&M costs in the original estimate were based on the power required to operate the pump station. The 2017 conveyance O&M for the highest capital alternative was estimated for power and labor to operate and maintain the pump station and the pipeline (see Appendix A).
- Annual demand is based on delivering Richmond WPCP effluent to RARE to address the 4.5 MGD supply deficit noted in Table 3-9. 4.5 MGD of effluent and 85% recovery rate = 3.8 MGD (4,300 AFY) average annual recycled water delivered to Chevron.

7. Annualized total cost is calculated using methodology outlined in Section 2.2.





Figure 3-8: Chevron Richmond Refinery Recycled Water Project Water Balance

(1) Recycled water production during month with minimum supply of WCWD effluent with additional supply from Richmond WPCP effluent.

(2) WCWD and Richmond WPCP effluent flow in excess of demand for recycling is discharged to San Francisco Bay.

(3) Richmond WPCP effluent will be treated via MBR, RO and UV disinfection to meet the District's RARE influent water quality limits for TDS, ammonia and disinfection.



3.3.5 Richmond Country Club Water Recycling Project

In 1984, the District began operating its first golf course irrigation project at the Richmond Country Club using recycled water supplied from the West County WPCP (see **Figure 3-7**). The Richmond Country Club was using an average of 0.18 MGD of recycled water to irrigate approximately 150 acres. The Richmond Country Club owned the pumped station, transmission pipeline and a 3-acre storage pond on-site and the District was contracting the maintenance and operation of the pump station to WCWD.

Once RARE came online, the Richmond Country Club was returned to potable water service due to the limited WCWD effluent supply during the irrigation season. The Richmond Country Club has expressed interest in returning to recycled water service, if available. The Richmond Country Club irrigation demand has varied from an annual average of 0.19 MGD (2012 to 2014) to 0.10 MGD (2015 to 2016). Due to the small recycled water demands, the limited recycled water supply and the investment the District has made to serve the Chevron Richmond Country Club Water Recycling Project. Therefore, this project would not be included in the non-potable reuse portfolio at this time. However, this site may be served with recycled water should supply become available or the refinery demand be reduced in the future.

3.3.6 Point Richmond Recycled Water Project

Recycled water service to Point Richmond customers was previously investigated as part of the NRWRP Expansion Study and screened out because of limited supply and distance to recycled water source. As shown in **Figure 3-7**, the closest source of water is the City of Richmond WPCP. Per the WSMP 2040, the annual potential average demand for recycled water was in the range of 0.07 to 0.1 MGD or 80 to 120 AFY.

Potential recycled water customers in the Point Richmond area include the Terminal One Project and Bottoms Property Redevelopment. The Terminal One Project proposes the development of approximately 14 acres of property, including 316 residential units, commercial space, a waterfront park, as well as road, trail, and other improvements. Compared to a conventional development project, the Terminal One Project would use a reduced water demand for irrigation in compliance with the District's Regulations Section 31 (ESA, 2016). On December 13, 2017, a public hearing was held to consider a design review permit for the project. The Bottoms Property Residential Project proposes to build a residential development of 60 dwelling units. The project will be developed within approximately 6 acres of the 25 acresite. In 2014, the City of Richmond Planning Department prepared a Recirculated Draft EIR for the project.

Like the Richmond Country Club, the Chevron Richmond Refinery recycled water demands are prioritized over the Point Richmond demands. Therefore, this project would not be included in the recycled water portfolio. However, this site may be served with recycled water should supply become available or the refinery demand be reduced in the future

3.3.7 Phillips 66 Refinery Recycled Water Project

The Phillips 66 Recycled Water Project could utilize up to 3.7 MGD of recycled water at the Phillips 66 Refinery (Phillips 66) in Rodeo for use in the refinery high-pressure boilers and cooling towers, if sufficient supply were available. A new recycled water facility would treat disinfected secondary effluent from the Pinole-Hercules and Rodeo treatment plants (see **Figure 3-8**). In 2005, the District and Phillips 66 executed a Memorandum of Understanding (MOU) to evaluate the feasibility of developing this project. A 2007 feasibility study identified alternatives and costs for the treatment and use of recycled water at the refinery (BC, 2007). The project is technically feasible; however, the available water supply has decreased in recent years.

As shown in **Table 3-11**, the combined supply of final effluent from the Pinole-Hercules WPCP and Rodeo Sanitary District treatment plants is large enough (dry weather flow of 2.7 MGD) to produce sufficient supply of 1,340 gpm (1.9 MGD) for the boiler feed water treatment system, assuming a recovery rate of 90% for MF and 85% for RO. Remaining



flow, if available, could be used to satisfy a portion of the cooling tower makeup water demand (i.e., 600 gpm or 0.86 MGD) for a total of 2.6 MGD or 2,912 AFY (Project Phase 1). In the future, if sufficient flows were available, the remaining cooling tower demand could be met (Project Phase 2). Cost estimates presented in this study include annual average recycled water delivery of up to 3.7 MGD (4,144 AFY). Under Phase 1 and Phase 2 project scenarios, the final effluent from both facilities would be pumped by the Rodeo Pump Station to Phillips 66.

	ADWF,	MGD	Minimum Mont	thly Flow, MGD
Treatment Facility	2004-2005	2016/17	2004-2005	2016/17
Pinole-Hercules WPCP	3.2	2.3	3.0	2.2
Rodeo WWTP	0.6	0.5	0.6	0.5
Total	3.8	2.8	3.6	2.7

Table 3-11: Phillips 66 Refinery Recycled Water Project Supply and Demand

Source: BACWA, 2017. Group Annual Report. Nutrient Watershed Permit Annual Report. October 2017.







Table 3-12 summarizes the capital and annual O&M costs for the project. The high-purity recycled water project would consist of the following process units: MF, biological active filtration (BAF), reverse osmosis (RO) and ultraviolet (UV) disinfection. Secondary effluent will be pumped from the Rodeo Pump Station through a new pipe that will deliver water to the refinery fence line. An existing tank will be used for effluent equalization prior to treatment (MF, RO, BAF). A portion of MF filtrate would be treated via RO and another portion would go to a BAF unit. The BAF unit is necessary to remove ammonia to meet water quality requirements for the cooling towers. Effluent from the BAF unit will be disinfected with an in-line UV system to meet Title 22 requirements for disinfected tertiary recycled water. For cost estimating purposes, it was assumed that existing carbon steel pipe would be reused as much as possible for the distribution piping to deliver product water. Capital cost also include site preparation and electrical system. The District is currently exploring options for funding the project.



	2007 D	ollars ¹	2017 Dollars ²						
Phase	Capital Cost (\$M)	O&M (\$M/yr)	Capital Cost (\$M)	O&M (\$M/yr)	Annual Demand (AFY)	Annualized Total Cost (\$/AF)⁵			
Phase 1. Boilers and Half	of Cooling Tov	wers							
Treatment ³	27	1.2	48	2.0		1,000			
Conveyance to Refinery ⁴	3.0	0.03	5.4	0.1		100			
Phase 2. Remaining Half of Cooling Towers. Provisions included above for future expansion									
Total	30	1.2	53	2.1	4,100	1,100			
Notes:									

Table 3-12: Phillips 66 Refinery Recycled Water Project Costs

1. Source: BC, 2007. ConocoPhillips Recycled Water Project Technical Study. August 2007.

2. Raw construction costs (from source) were updated to December 2017 dollars using ENR CCI ratios for 20 Cities Average. O&M costs were updated based on current electricity and labor costs.

3. Treatment includes BAF, MF, RO, UV, secondary effluent tank, secondary effluent transfer pumps, secondary effluent piping, utilities and product water piping. O&M includes electrical costs, labor, maintenance, chemicals, UV energy and replacement costs. Brine disposal not included.

4. Conveyance to Refinery includes pipeline and modifications to Rodeo Sanitary District (RSD) pump station. O&M includes RSD pump station electrical cost. Pipe maintenance not included.

5. Annualized total cost is calculated using methodology outlined in Section 2.2.

This project is retained for further consideration. The project is technically feasible; however, the available water supply has decreased in recent years. From an implementation complexity and environmental justice, this project is similar to the Chevron/Richmond WPCP alternative. The Refinery has expressed interest in pursuing this project, but additional partnerships with Rodeo and Pinole would be required. The 2005 MOU addressed the technical feasibility of the project, hence new agreements to define responsibility, operation and cost would need to be developed. The District could use the Chevron Richmond Refinery agreement as a reference for this project. The District should continue conversations with the Refinery to further refine this project.

At this time, no other non-potable water projects are competing for the same source of water. However, the treated wastewater from Rodeo WWTP and/or Pinole-Hercules could be used for a potable reuse project as described in Section 4.

3.3.8 Franklin Canyon Recycled Water Project

The proposed project would supply 0.2 to 0.3 MGD (200 to 300 AFY) of recycled water from the Rodeo WPCF through a 4.5-mile pipeline to the Franklyn Canyon Golf Course for irrigation (WSMP, 2040). As shown in **Figure 3-9**, this project would use the same source of water as the Phillips 66 Refinery Recycled Water Project. The project cannot be served directly from the Pinole effluent pipeline, which carries secondary effluent. Due to insufficient flows for both projects and the long pipeline length, it was determined that the Phillips 66 would be prioritized over the Franklin Canyon Recycled Water Project. Other potential sources of wastewater include the Mountain View Sanitary District and Central Contra Costa Sanitary District (CCCSD) however, approximately 9 miles of pipeline would be required to convey the recycled water to the golf course. Therefore, no economically feasible options are available at this time and the project will not be included in the recycled water portfolio.







3.3.9 Lamorinda Recycled Water Project

The District has considered several variations of a recycled water project to serve customers in Walnut Creek, Pleasant Hill, and the Lafayette, Moraga, and Orinda (Lamorinda) portions of its service area. This would require a partnership with CCCSD for distribution of their recycled water to District customers. The original market assessment and project alternatives for the Lamorinda area was developed in 1996 and proposed an extensive distribution system through the District's service area using the abandoned Shell high-pressure fuel line (purchased by CCCSD) as the major transmission line for the project. At that time, the proposed projects did not move forward as they were cost prohibitive at \$7,500 to \$13,300 per AF.

In 2004, after a new market assessment was prepared, it was determined that a project to serve wide reaching areas of Lafayette, Moraga and Orinda was likely not cost-effective. In 2015, CCCSD re-evaluated the cost to rehabilitate the



Shell pipeline to serve the Moraga Country Club and the Rossmoor Golf Course. The costs remained prohibitively high and there are no current plans to further evaluate this project. Therefore, a recycled water project using the existing Shell Pipeline is considered not feasible at this time and will not be included in the recycled water portfolio.

Several new smaller alternatives were considered in 2004 around CCCSD's existing recycled water distribution system, including the Reliez Valley Recycled Water Project discussed below. New projects would then be limited to the areas near the northern boundary of the District's water service area and closest to the CCCSD's recycled water system to take advantage of existing infrastructure (see **Figure 3-10**).

3.3.10 Reliez Valley Recycled Water Project

For this project, the District would partner with CCCSD to obtain recycled water from their existing system (Zone 1 pipeline) and distribute it to a limited set of customers in the District's service area in the northern portion of Lafayette. As described in the WSMP, potential landscape irrigation customers included two cemeteries, a golf course, and the City of Pleasant Hill with an estimated recycled water demand of up to 0.2 MGD (250 AFY). Figure 3-10 shows the proposed distribution system from CCCSD to three potential "anchor" customers. As shown in Table 3-13, the combined annual average irrigation demand for the two cemeteries is 0.10 MGD. The Grayson Wood Golf Course is currently not in operation and there is no District irrigation account for this facility. The property is currently on the market for development, but it is unlikely that a new golf course or other project with a substantial irrigation demand will be operated in the future in this property. Other potential irrigation customers within the District's service area, such as Brookwood Park or the Sports Field Complex in Lafayette, do not have significant demands. Without an anchor customer, this project is not viable.

	Irrigation Demand (MGD) ¹						
Customer	2012	2013	2014	2015	2016	Average	
Oakmont Memorial Park	0.07	0.09	0.06	0.04	0.05	0.06	
Queen of Heaven Cemetery	0.05	0.04	0.03	0.03	0.03	0.04	
Grayson Woods Golf Course	Not in operation						
Total						0.10	

Table 3-13: Reliez Valley Recycled Water Project Annual Average Demand

Notes:

1. Source: District water meter data (2012 to 2016).







3.3.11 CCCSD Regional Project

The CCCSD is planning to expand its recycled water facilities to augment regional water supplies and is exploring wholesale recycled water opportunities such as supplying recycled water to nearby refineries. For this alternative, the District would pay CCCSD to deliver recycled water to nearby refineries which would free up potable water for transfer to the District. This alternative would require agreements between the District, CCCSD and CCWD. It is assumed that CCCSD would enter into agreements with the refineries. A detailed analysis of water resource benefits of this option is outside the scope of this study.

The District can accept water from CCWD at the existing EBMUD-CCWD intertie. The Los Vaqueros Reservoir Expansion Project Draft Supplement to the Final EIS/EIR dated June 2017 describes the project components for Los Vaqueros Reservoir Expansion Project. A new high-lift pump station would be required to lift water from CCWD's Los



Vaqueros Pipeline to the District's Mokelumne Aqueduct #2. The District would need to purchase land for the pump station. Variable Frequency Drives (VFDs) would be installed at the District's Walnut Creek Pumping Plant to assist with managing flow rates on the Mokelumne Aqueduct. This would require construction of two new buildings on the District property. Treatment upgrades would also be required to allow the District to treat water from Delta sources which have a different water quality. It is assumed that the costs associated with these facilities are included under a separate project.

CCCSD currently produces an average of 1.6 MGD of recycled water. Most of it is sent to Zone 1 customers for landscape irrigation. The 2035 projected average dry weather flow at CCCSD are 41 MGD and the projected seasonal average Title 22 recycled water demand is 5.5 MGD, hence an average of 35.5 MGD is available for recycled water production in the future. The currently available dry weather flow is approximately 24 MGD (see **Table 4-5**). There are opportunities to offset raw water use at neighboring refineries by supplying high-quality recycled water. The refineries currently use a combined total of 20 MGD of raw canal water supplied by CCWD for their cooling tower, boiler feedwater and other processes. Supplying recycled water to the refineries would require treatment upgrades to remove ammonia, and possibly total nitrogen and dissolved salts. The level of treatment depends on the water quality the refinery will accept. If water quality equivalent to canal raw water is required, chloride would need to be removed. However, the refinery may be able to accept a lower water quality. This would need to be negotiated. The two costs discussed below represent the range of costs based on water quality requirements.

The capital costs to produce 20 MGD of recycled water for the refineries was previously presented in CCCSD's *Recycled Water Wholesale Opportunities Report* dated March 2016, adapted from the *November 2013 Refinery Recycled Water Update*. The report finalized in March 2016 summarizes the capital and O&M costs for tertiary treatment with nitrification, disinfection, cloth filtration, pipeline rehabilitation and modifications to the refineries' facilities, but do not include RO because it assumes a lower water quality would be acceptable. The 2016 capital and O&M costs were \$135 million and \$11 million per year respectively.

More recently, the CCCSD's Comprehensive Wastewater Master Plan dated June 2017, presented the capital costs to produce 20 MGD of recycled water assuming a water quality equivalent to canal water. Planned treatment upgrades include addition of an MBR, UV and RO for refinery recycled water. The MBR/RO system could be expanded in 5 MGD increments according to water demands from the refineries. Recycled water would be disinfected using UV. The cost for the 20-MGD alternative are summarized on the next page in **Table 3-14**. These costs are higher compared to the previous 2016 costs because a higher recycled water quality requirement is assumed. Both costs included costs to rehabilitate the existing distribution pipeline to the refineries.

This is an institutionally complicated project that will require multiple agreements and treatment upgrades to implement. Upgrades at CCCSD are needed to meet refinery water quality objectives. Infrastructure requirements from the District perspective are anticipated to be minimal and are assumed to be included under a separate project. Additionally, financial agreements between CCWD and CCCSD and the District will be needed to address operation of facilities and the cost of water. It is recommended that this project continue to be carried forward so that these details can be further developed to determine the feasibility of the project.



	June 2017 Dollars		Dec. 2017	Dec. 2017 Dollars ⁴		
	Capital Cost (\$M)	O&M (\$Mil/yr)	Capital Cost (\$M)	O&M (\$M/yr)	Annual Demand (AFY)	Annualized Total Cost (\$/AF)⁵
Treatment ^{1, 2}	286	8.9	291	7.5		1,000
Recycled Water Conveyance ³	29	0.1	29	1.6		100
Total	315	9.0	320	9.1	22,400	1,100

Table 3-14: CCCSD Regional Project Costs

Notes:

1. Source: Carollo and CH2M, 2017. Central Contra Costa Sanitary District's Comprehensive Wastewater Master Plan. Final. June 2017.

2. Capital costs include treatment and conveyance to supply 20 MGD of recycled water to refineries. Costs include a 23-MGD MBR/UV, a 14-MGD RO, electrical upgrades, and distribution pipeline rehabilitation.

3. Assume existing distribution system (i.e., CCWD Recycled Water Pipelines to Shell and Tesoro) would be used to convey recycled water to the refineries, but pipeline would require rehabilitation.

4. Capital and O&M costs (from source) were updated to December 2017 dollars using ENR CCI ratios for 20 Cities Average.

5. Annualized total cost is calculated using methodology outlined in Section 2.2.

3.3.12 Contra Costa Pipeline in Canal Right of Way

The Contra Costa Canal (Canal) is an aqueduct in Contra Costa County that is used for agricultural, industrial and nonpotable municipal water purposes. Prior to the construction of the Shortcut Pipeline and the Multi-Purpose Pipeline, the Canal was the central conveyance facility. Since 2012, its primary purpose is to provide redundancy to the Shortcut Pipeline and to convey raw water to approximately 40 customers with a total demand of approximately 0.83 MGD. Due to high maintenance costs and relatively low water sales for a 25-mile canal, CCWD is considering alternatives to reduce its operational costs.

The Canal provides water for a portion of the CCWD's service area, but it also borders the District's service area as it crosses through Walnut Creek and Pleasant Hill. According to District Water Conservation staff, Diablo Hills Golf Club is a District customer that has been using water from the canal for irrigation. Since CCWD is looking at decommissioning the Canal, the Diablo Hills Golf Course would need to connect to the District's potable water system.

Instead of repurposing the canal to distribute recycled water from CCCSD, CCWD's preference is to install a new pipeline in the canal right of way. This would prevent the recycled water from ending in Mallard Reservoir. Furthermore, the use of a pipeline would allow the water to be delivered at pressure – eliminating the need for a booster pump at each location along the canal. The new recycled water pipeline could be used to serve the Diablo Hill Golf Course and other nearby District customers such as Heather Farms. **Table 3-15** shows the annual average irrigation demand for the golf course and the portion of Heather Farm Park that is in the District's service area. As shown, the total District customer demand is small (i.e., 51,000 gpd) compared to the total customer demand of 0.83 MGD. **Table 3-16** summarizes the costs for this alternative.

The recycled water pipeline originating in the canal near the CCCSD wastewater treatment plant and ending at the Concord Naval Weapons Station (CNWS), is an attractive alternative if interagency agreements can be reached between the CCWD, CCCSD, and the CNWS developer. This alternative could be implemented independently of the Naval base conversion and be used to help CCWD meet its 2020 water conservation goals. It could also be implemented in a phased approach – an untreated water pipeline can be constructed and then converted to recycled water later. The District would be able to offset approximately 50,000 gpd of potable water use in the area. Without the



connection of the recycled water pipeline to the other CCWD customers, this alternative would not be economically feasible due to the low demands.

The project will deliver recycled water to approximately 40 customers. However, District's customers estimated annual average demand is only 51,000 gpd. Therefore, the benefits to the District are limited. Without recycled water delivery to CCWD customers this project would not be economically feasible; the low unit cost indicated in **Table 3-16** is based on participation of numerous CCWD customers. It is recommended that the District continue discussions with CCWD to better define the project.

Table 3-15: Annual Average Irrigation Demands

District Customer	Irrigation Demand (gpd)
Diablo Hills Golf Course	47,000 ¹
Heather Farm Park	4,000 ²
Total District Customer Demand	51,000

Notes:

1. 2011 Data. Source: Carollo, 2013. Contra Costa Water District Untreated Water Facilities Improvement Program Plan Update. Final. July 2013.

2. 2010 Data. Source: EBMUD, 2016b. Using Water from Contra Costa Canal Preliminary Evaluation Summary. June 2016.

Table 3-16: CCWD Pipeline in Canal ROW Project Costs

	2013 Dollars ¹		2017 Dollars⁵			
	Capital Cost (\$M)	O&M (\$M/yr)	Capital Cost (\$M	O&M (\$M/yr)	Annual Demand (AFY)	Annualized Total Cost (\$/AF) ⁷
Treatment ²	NA	NA	NA	NA		NA
Recycled Water Conveyance ^{3, 4}	27	NA	30	0.78		2,000
Total	27	NA	30	0.78	900 ⁶	2,000

Notes:

1. Source: Carollo, 2013. Contra Costa Water District Untreated Water Facilities Improvement Program Plan Update. Final. July 2013.

2. Cost to augment CCCSD tertiary treatment capacity, if needed, is not included.

Recycled water conveyance capital costs include 18.5 miles of 28-inch, 18-inch, and 6-inch diameter HDPE pipeline, a
pump station and a hydropneumatic tank. Costs do not include shortcut pipeline redundancy alternative. Conversion of
Castle Rock Water Company customers to treated water is also not included.

4. Recycled water conveyance O&M costs were not available. Operations and maintenance costs are reduced significantly because the pump station and pipeline maintenance are relatively minimal. Recycled water would be purchase from CCCSD at approximately \$200 per acre-foot (compared to current cost of up to \$80 per acre-foot). Therefore, an annual cost increase of up to \$140,000 would have to be absorbed by CCWD or passed on to customers.

- Capital costs (from source) were updated to December 2017 dollars using ENR CCI ratios for 20 Cities Average. 2017 O&M cost estimates for the conveyance system were estimated using methodology outline in this study and include labor, electricity and consumables (see Appendix A for details).
- 6. EBMUD customers make up less than 60 AFY of demand. Unit costs would be considerably higher without CCWD participation.
- 7. Annualized total cost is calculated using methodology outlined in Section 2.2.



3.3.13 UC Berkeley's Global Campus, Richmond

The UC Berkeley Global Campus at Richmond Bay was originally developed as a project concept with funding from Lawrence Berkeley National Lab and known as the Richmond Bay Campus. The project proposed to serve as a new research and action hub in Richmond with a focus on global issues, culture and collaboration. The proposed project site is approximately 133 acres, it is located in the City of Richmond and within the District's ultimate service area. In 2013, the District prepared a water supply assessment for the Richmond Bay Campus 2013 Long Range Development Plan. At the time, the average water use was approximately 52,000 gallons per day, and future average water demands for the proposed project were approximately 932,000 gpd (EBMUD, 2013). This demand was accounted for in District's Urban Water Management Plan (EBMUD, 2015b).

The District's current recycled water distribution pipeline ends approximately 3 miles from the project site. Due to the proximity to the District's existing recycled water distribution system, projects in this area present opportunities for recycled water uses that can be served by expanding recycled water pipelines in the future.

In 2016, UC Berkeley indefinitely suspended plans to physically develop the Berkeley Global Campus at Richmond Bay due budgetary challenges confronting the University. Therefore, this project will not be included in the recycled water portfolio at this time.

3.4 Satellite Recycled Water Projects

Satellite recycled water treatment plants (SRWTP) take raw sewage from a sewer pipeline and treat it to meet the Title 22 standard required for a specific project. These systems can serve large water users that are located far from a centralized treatment facility. The District has identified several potential satellite recycled water treatment plant projects that could provide recycled water to customers.

3.4.1 Rolling Hills Cemetery

Rolling Hills Memorial Park is a cemetery in Richmond, Contra Costa County. The WSMP 2040 demand estimates were 0.05 to 0.18 MGD (50 to 200 AFY). The cemetery is in WCWD service area and WCWD wastewater flows are dedicated to the Chevron Richmond Refinery Project. Therefore, the main constraint is the wastewater availability in the area. Because of limited supply, the Chevron Richmond Refinery Project would be prioritized over the Rolling Hills Cemetery satellite project. Therefore, this project was not included in the water reuse portfolio.

3.4.2 Diablo Country Club

The Diablo Country Club (DCC) is a 120-acre golf course located in Contra Costa County, California, at the base of Mount Diablo. The District provides potable water service to the DCC and CCCSD provides wastewater collection and treatment services. The DCC's major water use is golf course irrigation. As early as 2004, the DCC has been investigating the use of recycled water for golf course irrigation. The DCC has been proactive in pursing measures to improve irrigation efficiencies, reducing potable water demand and decreasing its operating costs.

In 2012, the District signed a MOU with the DCC and CCCSD in which the parties agreed to cooperate on a feasibility study to evaluate the use of a satellite recycled water treatment plant to provide a portion of the irrigation water for the DCC golf course. Approximately 25 percent of the irrigation demand would continue to be purchased from the District. The feasibility study recommended a satellite project that could recycle up to approximately 0.51 MGD of sewage from the CCCSD collection system to provide water for irrigation. This would be equivalent to an annualized average flow of 0.22 MGD. A second MOU to better define the project, responsibilities, and fees was executed in 2015. In June 2017, the DCC in coordination with CCCSD requested proposals for design-build services to design and construct the Satellite Water Recycling Facility, including the diversion structure, pump station and force main at the club. The RFP also requested proposals for financing options. At the time the RFP was released, the environmental review process



was still ongoing. The RFP will be re-issued in 2018. It is worth noting that while the project will be located on DCC property, the project will eventually be owned and operated by CCCSD.

Table 3-17 summarizes the project costs. Capital costs are based on an MBR system sized to provide 100 percent of the irrigation demand. **Figure 3-12** shows the potential location of the satellite facility and points of sewage diversion. Untreated wastewater would be diverted from the sewage diversion station to the satellite facility and would be split into two parallel treatment trains to provide system redundancy. The treatment train also includes fine screen headworks, and UV disinfection. The disinfected irrigation water would be pumped to the existing golf course ponds, where it would be blended with potable water, and potentially surface water flows or groundwater. To reduce regrowth in the recycled water storage and distribution system, minimal chlorine dosing with on-site generated sodium hypochlorite is recommended. Debris from the fine screens would be hauled off-site for disposal. Waste activated sludge would be returned to the sewer collection system.

DCC is pursuing a self-financing model. Their studies have shown that the satellite project would pay for itself, while eliminating the risks associated with drought restrictions.

	2013 Dollars ¹		2017 Dolla	ars ⁴		
	Capital Cost (\$M)	O&M Cost (\$M/yr)	Capital Cost (\$M)	O&M Cost (\$M/yr)	Annual Demand (AFY)	Annualized Total Cost (\$/AF) ⁶
Treatment ²	7.6	0.35	10.4	0.43		3,600
Recycled Water Conveyance ³	0.4	0.01	0.6	0.02		300
Total	8.0	0.36	11	0.45	250	3,900

Table 3-17: Diablo Country Club Satellite Project Costs

Notes:

1. Source: Brezack & Associates Planning, LLC, 2013. Diablo Country Club Satellite Recycled Water Treatment Plant Feasibility Study. September 2013.

2. Treatment capital costs include sewer diversion and treatment process facilities. 2013 O&M costs were based on 5 percent of construction cost.

3. Recycled water conveyance capital costs include recycled water storage, pump station and pipeline.

4. Construction costs (with sales tax and overhead profit) were updated to December 2017 dollars using ENR CCI ratios for 20 Cities Average. O&M costs were updated from original estimate based on ENR ratios and updated energy costs.

5. Annualized total cost is calculated using methodology outlined in Section 2.2. Overall methodology, including the use of a higher discount rate (i.e., 3 percent vs 2 percent) compared to original estimate, results in a higher annualized total cost than presented in the 2013 Feasibility Study.





Figure 3-12: Diablo Country Club Satellite Project

3.4.3 Moraga Country Club and Nearby Potential Customers

The District, the Moraga Country Club (MCC) and CCCSD entered into a MOU that set forth principles for the MCC satellite treatment project. The original project was defined to serve only the MCC. Future phases may be implemented to serve other nearby potential recycled water users such as Miramonte High School, Moraga Commons, and St. Mary's College. **Figure 3-13** shows the location of the proposed satellite facility for the MCC and the potential distribution system to other customers. This section descries two alternatives: one for the country club only, and one to serve MCC and nearby users in addition to the country club.





Figure 3-13: Moraga Satellite Project

3.4.3.1 Moraga Country Club

The MCC project would provide up to 0.5 MGD of recycled water for golf course irrigation, to meet the annual average demand of 0.16 MGD (180 AFY). In 2009, MCC completed a feasibility study (RMC, 2009) to determine the preliminary sizing of the satellite facility to serve the golf course. Based on monthly potable water use for golf course irrigation between 2006 and 2008, the peak water demand occurs in the month of July and a 0.5-MGD satellite facility would produce the average day water demand in the peak month. Recycled water storage would be used to satisfy peak day demands. The existing storage ponds, adjacent to Hole 9, are used to store potable water for irrigation. In the future, these ponds could be used for recycled water storage. Based on 2006 to 2008 data provided by CCCSD, the Moraga



Pump Station, has adequate wastewater flow to satisfy the demand. However, hourly flow data would need to be reviewed to confirm these assumptions.

The flow diverted and pumped from the Moraga Pump Station to the satellite facility would be split into two separate trains within the satellite treatment plant. The process train includes fine screening (1 to 3 mm screens), biological process (anoxic and aeration tanks), membrane filtration, and UV disinfection system. Following disinfection, the recycled water would be routed to the Hole 9 storage ponds for distribution using existing system. Screenings would be compacted and hauled offsite. Waste sludge, fine screen washwater and membrane cleaning solutions would be pumped to the sanitary sewer.

Table 3-18 summarizes costs for the satellite project. Capital cost includes two new feed pumps to be installed at the existing Moraga Pump Station, new pipeline alignments (raw wastewater, recycled water and waste), and treatment facilities for a traditional design-bid-build project delivery approach. However, MCC is a private entity and can elect to do a design-build contact, sole-source the project to expedite the design and construction processes and ultimately reduce capital costs. The District is looking for the MCC to self-finance the project.

Construction of a satellite recycled water facility at MCC would require an agreement between the District and MCC defining ownership, operations, and cost responsibilities. There are several examples in the Bay Area of public-private partnerships that could be used as a reference. One example within the District's service area is the DCC. The District provides potable water service to DCC and CCCSD provides wastewater collection and treatment services to the club. If the DCC project moves forward, it is anticipated that CCCSD would eventually own and operate the satellite recycled water treatment plant at DCC. It is worth noting that the annual average demands used in this study are based on District water meter data (2012 to 2016) which are lower than previously evaluated. Current demands are slightly above 150 AFY (0.13 MGD), which is the threshold set in this study to retain projects for further consideration based on the project recycled water demand. Because the recycled water demands have decreased but the capital costs remain the same, the project would have a higher unit cost than previous studies.



	2009 Dollars ¹		2017	Dollars ⁵		
	Capital Cost (\$M)	O&M Cost (\$M/yr)	Capital Cost (\$M)	O&M Cost (\$M/yr)	Annual Demand (AFY) ⁶	Annualized Total Cost (\$/AF) ⁷
Treatment ^{2, 3}	9.9	0.12	14	0.15		4,600
Recycled Water Conveyance ⁴	0	0	0	0		0
Total	10	0.12	14	0.15	180	4,600

Table 3-18: Moraga Country Club Satellite Project Costs

Notes:

1. Source: RMC, 2009. Moraga Country Club. Satellite Recycled Water Treatment Plant. Detailed Feasibility Study. September 2009.

 Treatment capital costs includes two submersible pumps, pipeline alignments (raw wastewater, recycled water and waste) and treatment facilities for a traditional design-bid-build project (above ground). Capital costs were \$9.9 and \$10.2 million for above-ground and below-ground construction, respectively. The cost for above ground are used in this analysis.

- 3. Treatment O&M costs include energy, chemical use, membrane replacement, UV replacement, and labor.
- 4. Recycled water to be used on-site. Therefore, conveyance costs are assumed to be minimal compared to treatment.

5. Construction costs (with contingency) and O&M costs were updated to December 2017 dollars using ENR CCI ratios for 20 Cities Average.

6. Annual demands were updated based on (2012 to 2016) District water meter data (see Table 3-17).

7. Annualized total cost is calculated using methodology outlined in Section 2.2. Overall methodology and the use of lower demands results in a higher annualized total cost than presented in the 2009 Feasibility Study.

3.4.3.2 Moraga Area Expansion

In the future, the MCC Satellite Recycled Water Treatment Plant could serve other nearby potential recycled water customers such as Miramonte High School, Moraga Commons and St. Mary's College (see Figure 3-12). Table 3-19 summarizes the annual average irrigation demands for each of the potential customers, including MCC, for a total of 0.22 MGD (or 250 AFY). The irrigation water demands for MCC have decreased in recent years. Based on District water meter data, the club uses about 0.16 MGD (annual average) with a peak month consumption of about 0.4 MGD in the month of July. As a result of these decreased demands, a 0.5-MGD MCC satellite facility would produce enough water to meet the estimated peak month demands for both the club and other nearby customers. This would require extension of the recycled water pipeline to serve other customers.

Table 3-20 summarizes the cost for this alternative, including the additional cost to distribute recycled water (i.e., distribution pumps, distribution pipeline and electrical infrastructure). A distribution booster pump at the satellite facility would pump recycled water for delivery to other customers. Electrical infrastructure needed for implementation includes installation of electrical equipment and power distribution. In addition to the recycled water storage provided at the satellite facility site, each customer would need to provide a storage tank for recycled water storage prior to distribution. This alternative assumes adequate wastewater flows are available to satisfy the demand. However, as noted in Section 3.4.3.1, the available wastewater flows (including hourly data) would need to be verified to confirm the assumption that local flows could support a 0.5 MGD facility.

Compared to the MCC satellite facility only serving MCC, service to other customers is a very different model with institutional challenges. MCC is a private entity and the satellite treatment facility would be located on their property, but the recycled water would be distributed to other customers in the area. Similar to other private-public projects,



agreement between the District, MCC and other customers would need to define ownership, operations and cost responsibilities. Because of these complexities, this project is rated lower than MCC.

	Irrigation Demand (gpd) ¹					
Customer	2012	2013	2014	2015	2016	Average
MCC	157,900	184,500	165,700	125,900	141,600	155,100
Moraga Commons	13,400	16,300	11,200	9,700	11,100	12,300
Miramonte	No irrigation specific account					16,500 ²
St. Mary's College	No irrigation specific account				34,700 ²	
Total						218,600

Table 3-19: Moraga Area Distribution Customer Annual Average Demand

Notes:

1. District water meter data.

2. Based on grass area calculation. Assumes 0.1 in/acre/day irrigation and 95 recycled water percentage.

Table 3-20: Moraga Area Recycled Water Expansion Costs

	2017 Dollars			Annualized
	Capital Cost (\$M)	O&M Cost (\$M/yr)	Annual Demand (AFY)	Total Cost (\$/AF)
Treatment ¹	19	19		4,500
Recycled Water Conveyance/Storage ²	4	4		1,200
Total	23	23	250	5,700

Notes:

1. Treatment costs were calculated for this alternative using unit costs for MBR/UV/Chlorine train, which resulted in similar unit costs but higher total capital and O&M costs compared with the MCC alternative.

2. Recycled water distribution costs include pump station, piping and storage tanks for each customer location. Capital cost estimates do not include land acquisition. O&M costs include pumping energy and pipeline maintenance.

3.4.4 Orinda Country Club

The Orinda Country Club is a private club located in the foothills east of the San Francisco Bay. The club is sited in a 250-acre property and includes a golf course. The club currently uses creek water for irrigation and recycled water use is not anticipated in the future. Therefore, this project was not retained for further consideration.

3.4.5 Mountain View and St. Mary's Cemeteries

The Mountain View Cemetery and St. Mary's Catholic Cemetery are located adjacent to one another in the foothills of Oakland (see **Figure 3-14**). The Mountain View Cemetery is owned by the Mountain View Cemetery Associated and is the larger of the two cemeteries. The cemeteries were previously evaluated as part of the 2005 Satellite Treatment Feasibility Study. A demonstration satellite project was proposed at the 223 acre-site to develop between 100 and 200 AFY of recycled water for irrigation. No fatal flaws were found. However, at the time, the Mountain View Cemetery was pursuing other potential alternatives to conserve water and was looking to implement lower-cost measures.

Table 3-21 summarizes the annual average irrigation demands for both cemeteries. The 2040 projections were based on demand data from 1999 to 2003. More recent data (2014 to 2016) shows the irrigation demands are lower than previously projected. The Mountain View Cemetery currently uses onsite lakes to collect stormwater runoff. This water



is pumped back through the on-site irrigation system to supplement well water irrigation during dry summer months. Because the cemetery is now using surface water runoff and groundwater for irrigation, the current demands have decreased significantly compared to values used in the WSMP 2040. The demands for St. Mary's Cemetery alone would not make this project feasible. Therefore, this project was not included in the recycled water portfolio.

Гable 3-21: Mountain View/St. Mary's Се	meteries Satellite Annual Average Demand
---	--

	Irrigation Demand (gpd)		
Customer	1999 to 2003 ¹	2040 Projections ²	2014 to 2016 ³
Mt. View Cemetery	160,000	191,000	9,200
St. Mary's Cemetery	40,000	63,774	26,000
Total	200,000	254,774	35,200

Notes:

1. EBMUD, 2005. Satellite Recycled Water Treatment Plant Feasibility Study. Draft.

2. WSMP 2040

3. District water meter data (2014 to 2016).







3.4.6 Rossmoor Country Club

Rossmoor Country Club (RCC) is located in the retirement community of Rossmoor in Walnut Creek. For this master plan update, a brief evaluation was conducted regarding the feasibility of using a satellite treatment facility for irrigation of the RCC golf course, the adjacent Community Gardens (1800 Rossmoor Pkwy) and the adjacent Tice Valley Park. **Figure 3-15** shows the potential location of the satellite facility within the golf course site, the point of sewer diversion and the distribution pipeline to the nearby customers.

District water meter data was reviewed for each of the potential customers. **Table 3-22** summarizes the customers annual average demands (2012 to 2016). On average, all three customers use less than 100,000 gallons per day (or less than 0.1 MGD) of potable water. The maximum month demand during dry weather months (June to August) is approximately 0.3 MGD (compared to 0.35 MGD potable water use previously reported). A small satellite facility could be sized to meet the max month demand of the club and the nearby customers (about 0.3 MGD), with peak flow demands supplemented with supply from the on-site reservoir.



CCCSD's average dry weather sewer flow from the entire Rossmoor Retirement Community was reported to be approximately 0.5 MGD (based on flows measured at the Rossmoor sewer shed outlet as indicated in **Figure 3-15**), which would be sufficient to serve the estimated peak day demands.

	Irrigation Demand (gpd) ¹					
Customer	2012	2013	2014	2015	2016	Average
RCC ²	76,100	53,000	55,900	62,300	89,400	67,300
Community Gardens	6,100	6,600	4,700	2,100	7,300	5,400
Tice Valley Park	12,000	14,100	5,700	7,400	10,200	9,900
Total						82,600

Table 3-22: Rossmoor Area Distribution Customer Annual Average Demand

Notes:

1. District water meter data.

2. Includes all Golden Rain Foundation accounts.

Due to limited cost effectiveness, only users with average annual demand greater than 150 AFY (0.13 MGD) were retained for further consideration in this study. Since the annual demand was below this threshold, this project was not retained for further consideration. However, the District would continue to support this project under a custom self-financing model similar to Diablo Country Club.







3.4.7 UC Berkeley Main Campus

In 2005, the District completed a study to determine the feasibility of constructing a satellite demonstration project at two alternative service area locations within the University of California Berkeley (UCB) (EBMUD, 2005). Based on the study results, a small-scale demonstration recycled water facility was recommended to be installed at the Berkeley campus. The intend was to gain experience from the operation of a small-scale demonstration project to evaluate if a larger-scale project would be feasible in the future. Due to issues related to siting and unexpected construction costs, the District and UCB jointly decided to stop pursuing the small-scale demonstration project in 2006.



As part of the EBRWF Water Quality Improvements Project, the District is currently evaluating extension of the recycled water pipeline to serve UCB main campus irrigation demands. The proposed project will extend the EBRWF distribution system to the UCB main campus. If the UCB Satellite project were to proceed, the EBRWF Phase 2 demands would need to be adjusted to remove deliveries to UCB main campus. **Table 3-23** summarizes the 2040-projected irrigation demands for the entire UCB campus. The annual average and peak month demands are about 0.8 MGD and 1.5 MGD, respectively. The proposed project would divert raw wastewater from the existing local sewers to the new satellite recycled water treatment plant located at UCB. **Figure 3-16** shows the potential location of the satellite facility and sewer diversion point. Sewer trunk lines and property are managed by UCB which makes the potential development of a satellite treatment facility on the UCB campus more straightforward.

Table 3-24 summarizes the costs for the 1.5-MGD satellite facility project. Raw water will be pumped to the satellite facility for treatment. The proposed recycled water facility would be equipped with a fine screen for solids removal, an MBR, UV disinfection and a recycled water storage tank. An onsite hypochlorite generation system will provide sodium hypochlorite for disinfection residual in the recycled water storage tank and for potable water backup addition. To be consistent with the previous (2005) evaluation, it is assumed that land for a pump station, a satellite treatment facility and a storage facility is provided free of charge by UCB.

	Projected Irrigation Demand (gpd)			
UC Berkeley Customer	2040 Average Day	2040 Max Month		
UC Berkeley – Hearst Ave.	40,176	76,334		
UC Berkeley – Oxford St.	9,829	18,675		
UC Berkeley – Warring St.	39,317	74,702		
UC Berkeley – Bancroft Way	545,972	1,037,347		
UC Berkeley – Clark Kerr	143,157	271,998		
Total Irrigation	778,451	1,479,056		

Table 3-23: UC Berkeley Main Campus Satellite Demand

Source: EBMUD, 2017. East Bayshore Recycled Water Expansion Study. Hydraulic Analysis of Future Pipelines and Demands.

	2017 Dollars			
	Capital Cost (\$M)	O&M Cost (\$M/yr)	Annual Demand (AFY)	Annualized Total Cost (\$/AF)
Treatment ¹	25	0.63		1,900
Recycled Water Conveyance ²	11	0.21		1,00
Total	36	0.84	900	2,900

Table 3-24: UC Berkeley Main Campus Satellite Project Costs

Notes:

1. Treatment capital cost estimate includes raw wastewater diversion and treatment facilities, and pipelines. Cost of land not included.

2. Recycled water conveyance cost estimate includes recycled water storage tank, pump station and pipeline within UCB campus.

Agreements between the District and UC Berkeley would need to be developed. Similar to the DCC satellite project, the District would be interested in having UC Berkeley self-finance construction of the satellite plant in lieu of paying the standard water rate. The responsibility of operation and maintenance of the treatment and sewer line diversion facilities would need to be discussed.





Figure 3-16: UC Berkeley Main Campus Satellite Project

3.4.8 Oakland Hills

In 2017, the District completed a feasibility study of the Oakland Hills alternative water supply (West Yost, 2017). The project would serve three large irrigation customers: Oakland Zoo, the Sequoyah Country Club (SCC) and the future Oak Knoll Development. A new satellite treatment plant would be located at SCC (highest elevation of the three) and distribute to serve all three customers (see **Figure 3-17**). It is assumed that Lake Chabot Golf Course will continue to use raw water from Lake Chabot for golf course irrigation. This option is one of several potential alternatives discussed in the District's 2017 Oakland Hills Alternative Water Supply Feasibility Study.

SCC, a privately-owned golf course and country club, has been interested in obtaining a non-potable water supply for golf course irrigation. In 2005, the District prepared the Satellite Recycled Water Treatment Plant Feasibility Study that



evaluated a satellite facility to provide recycled water to SCC. The satellite project was technically feasible; however, the project was not implemented due to high cost and lack of financial support. In parallel to the District's latest efforts, SCC conducted its own study to identify recycled water alternatives for SCC. The draft report is on hold as SCC's consultant further investigates site constraints.

The Oakland Zoo is a non-profit organization and has been in its current location since 1936. Oakland Zoo is expanding to nearly double its footprint to encompass up to 100 acres by 2018. With this expansion, it is assumed that the existing irrigation water demand will double. The planned development of Oak Knoll Community would convert approximately 190 acres of land into a mixed-use community with 935 residential units, trails, parks and more. It is assumed that irrigation demand would be equivalent to 50 percent of the total projected water use per the 2016 Project Water Supply Assessment. A master plan for the Oak Knoll development was just approved, but it has not been decided if recycled water will be required and/or that dual plumbing of new buildings will be required.

Table 3-25 summarizes the satellite project non-potable water demand. The annual average and peak month demands are about 0.31 MGD and 0.57 MGD, respectively. The proposed project would divert raw wastewater from a low point in the existing local sewers (i.e., Mountain Boulevard) to the new satellite recycled water treatment plant located at SCC. The estimated average and peak wastewater flows are about 0.58 MGD and 0.93 MGD, respectively. These sewer flows are sufficient to supply SCC and Oakland Zoo, even if the Oak Knoll Development were not to move forward. Nevertheless, sewer flows require confirmation.

_	Projected Irrigation Demand (gpd)		
Customer	Annual Average	Peak Month	
Sequoyah Country Club ¹	120,400	238,000	
Oakland Zoo ²	87,800	135,000	
Future Oak Knoll Development ³	103,500	198,600	
Total	311,700	571,600	

Table 3-25: Oakland Hills Satellite Project Non-Potable Water Demand

Notes:

1. Based on District meter data (2013 to 2015). Oakland Hills Alternative Water Supply Feasibility Study. April 2017.

2. Based on District meter data (2013 to 2015) and projected demand associated with zoo expansion.

3. Assumed to be fifty percent of the site's projected potable water demand.

Table 3-26 summarizes the costs for the proposed 0.5 MGD satellite project which provides flexibility to treat higher flows in the future if needed. The treatment facilities would be designed to meet the average annual demand, because the peak demand occurs only a few times a year. Capital costs include sewer diversion structure at Mountain Boulevard, diversion pumps, diversion pipeline from sewer to satellite plant at SCC, treatment facilities and recycle storage at SCC. The satellite plant will consist of a screening (bar screen and fine screen to remove debris), packaged MBR system, UV disinfection system and distribution pumps all within an enclosed building. Waste sludge would be returned to SD-1. Tertiary effluent would be stored in a 0.5 MG tank to provide one-day storage at SCC. A booster pump would be required to delivered recycled water to customers. It is assumed that each customer will construct their own recycled water storage tank, hence only the cost for the SCC storage tank is included in the estimates.

Agreements between the District and customers would need to be developed. Similar to the DCC satellite project, the District would be interested in having the Oakland Hills customers self-finance construction of the satellite plant in lieu of paying the standard water rate. The responsibility of operation and maintenance of the treatment and sewer line diversion facilities would need to be discussed. In addition, there may be an opportunity for the District to explore a raw water expansion project from Lake Chabot to the Oakland Zoo.


	April 201	7 Dollars ¹	December 2017 Dollars⁵			
	Capital (\$M)	O&M⁴ (\$M/yr)	Capital⁵ (\$M)	O&M (\$M/yr)	Annual Demand (AFY)	Annualized Total Cost (\$/AF) ⁶
Treatment ²	14.8	0.14	15.8	0.15		2,400
Recycled Water Conveyance ³	5.8	0.02	6.2	0.02		800
Total	20.6	0.16	22.0	0.17	350	3,200

Table 3-26: Oakland Hills Satellite Project Costs

Notes:

1. Source: West Yost, 2017. Oakland Hills Alternative Water Supply Feasibility Study. April 2017.

2. Treatment cost estimate includes raw wastewater diversion, treatment facilities and electrical infrastructure. Costs do not include land acquisition.

3. Recycled water conveyance cost estimate includes recycled water distribution pumps and pipelines.

4. O&M costs include energy, operations and maintenance for treatment facilities, and pumping energy for conveyance.

5. Construction costs (with sales tax and contractor overhead profit) and O&M costs were updated to December 2017 dollars using ENR CCI ratios for 20 Cities Average.

 Annualized total costs were calculated using methodology described in Section 2.2. Cost methodology differs from 2017 Feasibility Study. Methodology for this study includes the use of higher markups, salvage value and a 3 percent discount rate (compared to 3.5 discount rate used in the 2017 Feasibility Study), resulting in a lower annualized unit cost.





Figure 3-17: Oakland Hills Satellite Project

3.5 Fill Stations within EBMUD Service Area

There are several commercial and residential recycled water fill stations within the EBMUD service area. Recycled water for trucks is available at EBMUD's SD-1, at CCCSD's WWTP, and at the DSRSD WWTP and recycled water hydrants within their service area. Each purveyor implements a permitting program with specific regulations and applicable fees. The volume of recycled water used by commercial and construction customers is small in comparison to recycled water pipeline projects and fluctuates based on the timing and location of construction.

EBMUD, CCCSD, and DSRSD have all opened residential fill stations since 2014. While the fill stations offered a tremendous opportunity to educate the public on the benefits of recycled water to preserve landscaping during the recent drought, the volume of recycled water distributed from residential fill stations is small, and is not available in all



years. DSRSD's Residential Recycled Water Fill Station is currently closed, as the Tri-Valley's water supply has returned to pre-drought status.

3.6 Summary of Non-Potable Project Alternatives

The non-potable reuse alternatives were screened and prioritized based on the size of the project, wastewater supply limitations, cost, and other non-cost factors, such as institutional complexity. Due to limited cost-effectiveness, only projects with average annual demand greater than 150 AFY (0.13 MGD) were considered. **Table 3-27** summarizes the capital, O&M and unit cost for viable alternatives discussed above. These alternatives are retained for further consideration. The results of the non-cost evaluation are included in Section 5.1.1.



			Annual		Unit Cost (\$/	/AF)	
Projects	Capital (\$M)	O&M (\$M/yr)	Demand (AFY)	Treatment	Distribution	Total	Dry Year
Centralized Treatment							
DERWA/San Ramon Valley Phase 3	25	0.49	800	550	1,350	1,900	6,300
DERWA/San Ramon Valley Phase 4	17	0.18	300	400	2,500	2,900	9,700
DERWA/San Ramon Valley Phase 5	8.1	0.15	300	1,000	600	1,600	5,400
East Bayshore Phase 1A	16	0.50	500	1,300	1,100	2,400	7,400
East Bayshore Phase 1B	40	0.83	1,100	940	1,460	2,400	7,800
East Bayshore Phase 2	130	2.9	2,900	800	2,200	3,000	9,400
Chevron Refinery/Richmond WPCP	110	5.7	4,300	2,300	300	2,600	8,600
P66 Rodeo Refinery	53	2.1	4,100	1,000	100	1,100	3,700
Central San Regional Project	320	9.1	22,400	1,000	100	1,100	3,400
CCWD Pipeline in Canal ROW	30	0.78	900	0	2,000	2,000	6,800
Satellite Treatment			·				
Diablo Country Club	11	0.42	250	3,600	300	3,900	12,000
Moraga Country Club	14	0.15	180	4,600	0	4,600	15,000
Moraga Area Expansion	22	0.26	250	4,500	1,200	5,700	18,000
UC Berkeley Main Campus	36	0.84	880	1,900	1,000	2,900	8,800
Oakland Hills	21	0.17	350	2,400	800	3,200	11,000

Based on limited recycled water demands, competing uses for the same wastewater source, and technical feasibility issues as presented in the project descriptions, the following alternatives were screened out as not viable for further consideration:

- San Leandro Water Reclamation Facility Expansion Project
- Chevron Refinery Process Water or WWTP Effluent
- Richmond Country Club Water Recycling Project
- Point Richmond Recycled Water Project
- Franklyn Canyon Recycled Water Project
- Lamorinda Recycled Water Project
- Reliez Valley Recycled Water Project
- UC Berkeley's Global Campus Richmond Project



- Rolling Hills Cemetery Satellite Recycled Water Project
- Orinda Country Club Satellite Recycled Water Project
- Mountain View and St. Mary's Cemetery Satellite Recycled Water Project
- Rossmoor Country Club Satellite Recycled Water Project

This page is intentionally left blank.



4. POTABLE REUSE OPPORTUNITIES

4.1 Current State of Potable Reuse Regulations in California

There are four primary types of potable reuse:

- **Groundwater augmentation** through surface spreading or subsurface injection;
- **Reservoir water augmentation** to a surface water reservoir that is used as a potable water supply source;
- Raw water augmentation upstream of a surface water treatment plant; and
- **Treated drinking water augmentation** directly to a potable water distribution system.

Both groundwater and reservoir water augmentation include the use of an environmental buffer (an aquifer in the case of groundwater augmentation and a surface water reservoir in the case of reservoir water augmentation) and are therefore generally known as indirect potable reuse or IPR. Raw water and treated water augmentation do not use an environmental buffer, relying instead upon engineered storage buffers (ESBs), and therefore considered direct potable reuse of DPR.

The terms "indirect potable reuse" and "direct potable reuse" are used throughout the industry, but the spectrum of possible reuse projects can make it difficult to distinguish between the two. Assembly Bill 574 (Quirk-Hayward), which was signed in October 2017 and went into effect on January 1, 2018, further clarified this distinction by formally establishing the four categories listed above (Note, while AB 574 uses the term "reservoir water augmentation," the recently adopted updates to Title 22 use "surface water augmentation," a term used in this report only when referring to the regulations.) The terms "indirect potable reuse" and "direct potable reuse" will not be used extensively in this report, but the terms are important for historical context.

The overriding regulatory criteria governing wastewater reuse are found in the California Code of Regulations (CCR), Title 22, Division 4, Section 60301, et seq., commonly referred to as Title 22 (Title 22). Title 22 has long established water quality requirements for non-potable reuse. In 2014, the State Water Resources Control Board (SWRCB) revised and adopted uniform recycling criteria for groundwater replenishment.

In 2010, the California Legislature enacted Senate Bill (SB) 918 directing California Department of Public Health (CDPH) (now Division of Drinking Water, or DDW) to establish Surface Water Augmentation Regulations (for Reservoir Water Augmentation) and to investigate the feasibility of developing uniform water recycling criteria for direct potable reuse by December 2016. SB 918 also included the requirements to convene an expert panel. In 2013, the Legislature enacted SB 322 which required an advisory group be convened to advise the expert panel and DDW in the development of the feasibility report for direct potable reuse criteria.

In December 2016, the SWRCB released a primary Report to the Legislature (SWRCB, 2016a) as well as summary reports from the Expert Panel and Advisory Group as follows:

- Investigation on the Feasibility of Developing Uniform Water Recycling Criteria for Direct Potable Reuse
- Appendix A: Evaluation of the Feasibility of Developing Uniform Water Recycling Criteria for Direct Potable Reuse. Expert Panel Final Report.
- Appendix B: Recommendations of the Advisory Group on the Feasibility of Developing Uniform Water Recycling Criteria for Direct Potable Reuse. Advisory Group Final Report.



The Expert Panel determined that "it is feasible to develop uniform water recycling criteria for [Direct Potable Reuse] that would incorporate a level of public health protection as good as or better than what is currently provided in California by conventional drinking water supplies..." The panel noted that the functionality provided by the environmental buffer (i.e., storage, attenuation, and response time) in an indirect potable reuse project must be addressed by other means. The panel also noted that any project that cannot obtain two months of retention in the environmental buffer should be classified as direct potable reuse.

Given the lack of an environmental buffer, the Expert Panel stressed that reliability would be the overarching goal for a direct potable reuse option to consistently achieve the desired water quality in the product water. The panel suggested that direct potable reuse regulations provide "reliability" by:

- Providing multiple, independent treatment barriers;
- Incorporating the frequent monitoring of surrogate parameters at each step to ensure treatment processes are performing properly; and
- Developing and implementing rigorous response protocols (such as a formal Hazard Analysis Critical Control Point system).

After reviewing the recommendations of the Expert Panel Report and Advisory Group Report, DDW concluded that it is feasible to begin the process of developing direct potable reuse regulations. However, DDW believes the additional research and knowledge gaps identified in the Expert Panel and Advisory Group reports must be addressed prior to the adoption of criteria.

Concurrently, in July 2017, the SWRCB released proposed regulations for public comment that would establish uniform recycling criteria for reservoir water augmentation; the regulations use the term "Surface Water Source Augmentation" rather than reservoir water augmentation. (Note: This report uses the term "reservoir water augmentation," which allows a clearer distinction between projects that go into surface water aqueducts from those that go to surface water reservoirs). After public comment and revisions, the final regulations were adopted and became effective October 1, 2018.

AB 574 requires the SWRCB to establish uniform water recycling criteria for raw water augmentation by the end of 2023. There is currently no timeline for developing recycling criteria for treated drinking water augmentation. Without regulations, the information provided in the Expert Panel and DDW reports were used to develop conceptual requirements for raw water and treated water augmentation alternatives, as further described in Section 4.1.3.

The following sections provide an overview and applicable regulations for each of the four types of potable reuse, shown below in **Figure 4-1**.







4.1.1 Groundwater Augmentation

Regulations for groundwater augmentation (also called groundwater recharge or replenishment) became effective in June 2014. Groundwater augmentation has been implemented for many years, most notably by the Groundwater Replenishment System in Orange County and the Montebello Forebay Project in Los Angeles County. The groundwater augmentation regulations are organized by type of project:

- Surface application (surface spreading), and
- Subsurface application (injection or vadose zone wells).

Subsurface application (injection) of recycled water directly into the groundwater basin requires full advanced treatment (FAT) that includes RO and an advanced oxidation process (AOP) equivalent to at least 0.5-log reduction of 1,4dioxane. The regulations specify requirements for the RO process, including bench-scale testing. For surface application via surface spreading, additional treatment is provided through soil aquifer treatment (percolation) and dilution of the recycled water with groundwater in the groundwater basin. Due to space constraints and local hydrogeology, this RWMP Update only considers alternatives with subsurface application via injection wells.

Regarding pathogen control, Title 22 requires that recycled municipal wastewater used for groundwater augmentation goes through a treatment train with at least three separate treatment processes (multiple barriers) to achieve at least:



- 12-log enteric virus reduction;
- 10-log *Giardia* cyst reduction; and
- 10-log *Cryptosporidium* oocyst reduction.

This is commonly referred to as 12/10/10 log reduction. Each barrier must achieve a minimum of 1-log reduction and will not be credited for more than a 6-log reduction for each of the pathogens listed above. Underground retention time (URT) may be credited with 1 log/month for virus reduction. URT must be verified using an added tracer study to receive credit for the full log reduction (1 log/month). Depending on the method used for project planning purposes, the regulations give partial log-reduction credit for intrinsic tracer studies (0.67 log/month), numerical modeling (0.5 log/month), or analytical modeling (0.25 log/month). The regulations require that a tracer study be initiated within three months of project start-up.

For both surface and subsurface applications of advanced treated water to drinking water aquifers, the California Division of Drinking Water requires low levels of total organic carbon (TOC) at 0.5 milligrams per liter (mg/L) and total nitrogen at 10 mg/L.

A minimum of two months of URT is required before extraction for potable use, which provides "response retention time" to monitor water quality and respond to water quality concerns.

Parameter	Surface Application	Subsurface Application		
Minimum Treatment	Tertiary Filtration + Disinfection	Full Advanced Treatment		
Minimum Retention Time	2 months			
Virus	≥ 12-log redu	ction		
Giardia	≥ 10-log reduction			
Cryptosporidium	≥ 10-log redu	ction		
Safe Drinking Water Act Contaminants	Meets all MC	CLs		
Total Nitrogen	≤ 10 mg/L			
Total Organic Carbon	\leq 0.5 mg/L (after SAT and Dilution)	≤ 0.5 mg/L		

Groundwater augmentation criteria are summarized below in Table 4-1.

Table 4-1: Groundwater Augmentation Criteria for Potable Water Reuse

4.1.2 Reservoir Water Augmentation

Regulations for reservoir augmentation became effective October 1, 2018. The City of San Diego's North City Pure Water Project is anticipated to be the first project permitted under the adopted regulations in the near future.

The regulations include the following requirements:

• **Full Advanced Treatment** must include RO and an AOP that achieves at least 0.5-log reduction of 1,4-dioxane, similar to the requirements for groundwater augmentation via injection;



- **Retention time**: a minimum theoretical reservoir retention time of 180 days, which may be reduced to 60 days with additional treatment and SWRCB approval;
- Dilution requirements:
 - 1% (1:100) dilution of any 24-hour inflow of advanced treated water, measured at the outlet; or
 - 10% (1:10) dilution any 24-hour inflow of advanced treated water, measured at the outlet, plus one additional independent log-reduction of all three organisms.
- Pathogen removal requirements:
 - The regulations state that recycled municipal wastewater delivered to an augmented reservoir with 1% dilution shall go through a treatment train to achieve:
 - 8-log enteric virus reduction;
 - 7-log *Giardia* reduction; and
 - 8-log *Cryptosporidium* reduction.
 - At least two barriers must achieve a minimum of 1-log reduction and no barrier will be credited for more than a 6-log reduction for each of the pathogens listed above.
 - The Surface Water Treatment Rule then requires the surface water treatment plant to provide treatment to remove 4-log virus, 3-log *Giardia*, and minimum 2-log *Cryptosporidium*. The pathogen requirements for recycled water listed for reservoir augmentation projects are less than those required for groundwater augmentation because surface water is further treated through a surface water treatment plant prior to potable use.

The regulations also contained additional criteria on reservoir ownership, operational history, hydrodynamic modeling, and tracer studies. Reservoir water augmentation criteria are summarized on the next page in **Table 4-2**.

Parameter	1% Dilution in Reservoir	10% Dilution in Reservoir			
Minimum Retention Time	6 months (may be reduced to 2 months with DDW approval)				
Virus	\geq 8-log reduction	\geq 9-log reduction			
Giardia	\geq 7-log reduction	\geq 8-log reduction			
Cryptosporidium	\geq 8-log reduction	\geq 9-log reduction			
Safe Drinking Water Act Contaminants	Meets all MCLs				
Total Nitrogen	No requirement in regulations				
	NPDES Permit may limit nitrogen as a biostimulatory substance				
Total Organic Carbon	≤ 0.25 mg/L to be verified during startup				

Table 4-2: Summary of Reservoir Water Augmentation Criteria

In addition to meeting Title 22 recycling criteria, reservoir augmentation projects will require a National Pollutant Discharge Elimination System (NPDES) permit for discharge to surface water – the District's reservoirs are named water bodies with beneficial uses listed in the Water Quality Control Plan for the San Francisco Bay Basin ("Basin Plan," San Francisco Bay Regional Water Quality Control Board, 2017). This may result in additional requirements related to protection of aquatic life, including strict limits on residual chlorine in advanced treated water added to a reservoir and limits intended to prevent reservoir eutrophication (i.e., low nitrogen and phosphorus).



4.1.3 Raw Water and Treated Drinking Water Augmentation

Within California, uniform recycling criteria for raw water augmentation are expected by the end of 2023, as required by AB 574. While not in California, raw water augmentation is currently being implemented by the City of Big Springs in Texas.

There is currently no timeline for developing recycling criteria for treated drinking water augmentation in California. Without codified regulations, the information provided in the Expert Panel and DDW reports was used to develop conceptual requirements for raw water and treated drinking water augmentation facilities, as listed in **Table 4-3** (SWRCB, 2016 and 2016a). Log reduction values for pathogens are the sum of the proposed requirements for reservoir water augmentation (10% dilution) and Surface Water Treatment Rule requirements (4-log virus and 3-log *Giardia* and minimum 2-log *Cryptosporidium*).

Table 4-3: Conceptual Requirements for Raw and Treated Drinking Water Augmentation

Parameter	Criterion
Minimum Retention Time using ESB	> Failure Response Time of Advanced Treatment System
Virus	≥ 13-log reduction
Giardia	≥ 11-log reduction
Cryptosporidium	≥ 11-log reduction

The SWRCB feasibility report (SWRCB, 2016) identifies several key areas requiring additional research to support development of the regulations, including:

- Consideration of a probabilistic method (Quantitative Microbial Risk Assessment) to better quantify treatment train performance and confirm necessary virus, giardia, and cryptosporidium log removal values.
- Improving the understanding of pathogens in raw wastewater to develop better empirical data on concentrations and variability, including during community outbreaks of disease.
- Continued review of the risks of emerging constituents to public health, based on convening the "blue ribbon" panel every 5 years. The focus is intended to be on new compounds that may pose health risks from short term exposures, low molecular-weight compounds that are not rejected by reverse osmosis, and screening for unknown compounds using non-targeted analysis.

More specific research topics include:

- Continuing to evaluate the assignment of log removal credits to specific treatment technologies, such as RO, MBR, and microfiltration, based on periodic integrity testing and real-time validation.
- Investigating the use of surrogate parameters to ensure removal of trace organic compounds and pathogens.

District staff should strongly consider participating in the research supporting development of these regulations, given the numerous opportunities for raw water augmentation and treated drinking water augmentation identified in this RWMP Update.



4.2 Development of Potable Reuse Alternatives

4.2.1 Approach to Development of Alternatives

Potable reuse alternatives were developed using the following three-step process:

- **Identify Sources**. The first step was to identify all possible sources of treated municipal wastewater within or immediately adjacent to the District's water service area (Section 4.3).
- Identify Targets. Possible targets based on potable reuse type were:
 - Groundwater augmentation: Targets are groundwater basins in the District's water service area (see Section 4.4.1).
 - Reservoir water augmentation: Targets are District's surface water reservoirs (see Section 4.4.2).
 - Raw water augmentation: Targets are District's surface water treatment plants or aqueducts (see Section 4.4.3).
 - Treated drinking water augmentation: Targets are large pipelines or tanks within the District treated water distribution system (see Section 4.4.4).

Conveyance needs (i.e., pipeline sizes) were also assessed to connect the sources and targets, as discussed in Section 4.6.

• Determine Treatment. Seven treatment trains were developed based on the combination of source and target for each alternative (see Section 4.5). All treatment trains provide Full Advanced Treatment and include reverse osmosis. The production capacity of each treatment train was based on the lower of the source and target capacities. In most cases, the source (i.e., the available wastewater supply) is the limiting factor, except for a few reservoir augmentation alternatives where reservoir operations limit the ability to accept recycled water.

4.2.2 Surface Water Treatment Plant Constraints

Potable reuse alternatives were developed with the goal of minimizing changes needed to the District's existing surface water treatment plants and their operating schedules. Specifically:

- San Pablo Water Treatment Plant (WTP) was not assumed to be operational.
- Lafayette WTP was not assumed to be a direct target for any potable reuse alternatives since it is not located near any WWTPs to serve as a source or receive treatment byproducts such as RO concentrate. However, it could receive advanced treated water placed in the Mokelumne Aqueduct.
- Water treatment plants supplied by reservoirs with local runoff (Upper San Leandro and Sobrante WTPs) would continue to use these supplies. To the extent practicable, seasonal operation was retained for Upper San Leandro and Sobrante WTPs, because these treatment plants are more expensive to operate, and they are typically not needed to meet winter demands (it is operationally simpler to run fewer WTPs). Alternatives that retained this seasonal use were scored higher than alternatives that require a shift to year-round production.
- The volume of recycled water sent directly to surface water treatment plants via raw water augmentation, or indirectly sent to water treatment plants via reservoir water augmentation, is theoretically limited by water treatment plant production capacity, although in practice this constraint is rarely applied since total treated water demand remains constant.



The capacity and recent production rates for the District's surface water treatment plants is listed for reference in **Table 4-4**. Monthly production at the District's surface water treatment plants is shown for reference in **Figure 4-2**. The ability of surface water treatment plants to accept advanced treated water is discussed further in Section 4.4.3.

Capacity	Walnut Creek	Lafayette	Orinda	Sobrante	Upper San Leandro
Plant Capacity (MGD)	90	25	190	50	45
Average Annual Production, FY08-14 (MGD)	43	9	111	15	6
Plant Capacity (AF/MO)	8,400	2,333	17,733	4,667	4,200
Average Annual Production, FY08-14 (AF/MO)	4,030	860	10,330	1,370	580

Table 4-4: Surface Water Treatment Plant Capacity and Recent Production



Figure 4-2: Seasonality of Surface Water Treatment Plant Production, FY08-14

The cost of treating water at surface water treatment plants was included in the O&M cost estimates for reservoir water augmentation and raw water augmentation alternatives. The conventional surface water treatment plants (USL and Sobrante) are more expensive to operate than the in-line plants (Walnut Creek and Orinda) due to the additional treatment steps included at those plants (see Section 4.4.3.1 and 4.4.3.2) (EBMUD, 2017b), so sending more water to the conventional surface water treatment plants would increase the District's operating costs. Regardless of treatment steps, all the WTPs meet applicable minimum Surface Water Treatment Rule requirements. The additional treatment steps provided at the conventional surface water treatment plants could be beneficial in a potable reuse context -- for example by providing additional pathogen removal and removing taste & odor-causing compounds and other trace organics.



4.3 Sources of Wastewater for Advanced Treatment

4.3.1 Availability of Treated Effluent

Eleven WWTPs in or near the District's water service area were considered as potential water sources for potable reuse, as shown in **Figure 4-3** and listed below in **Table 4-5**. For each WWTP, the estimated firm supply available for potable reuse is based on the dry weather flow discharged to surface water in the summer of 2015, which was an exceptionally dry year. In most cases, these wastewater flows are considerably lower than those listed in the District's WSMP 2040 (Appendix D, TM-4, "Future Recycled Water Potential Analysis") due to increased water conservation and water use efficiency and/or reductions in groundwater infiltration over the last decade. Current commitments for non-potable reuse were also subtracted out to estimate the firm daily supply available on a year-round basis. The following are peak month non-potable demands (approximately double the average annual demand), unless specified, for each potential potable reuse water source.

- **SD-1:** 5 MGD reserved for the EBRWF.
- City of San Leandro Water Pollution Control Plant (WPCP): 0.6 MGD reserved for the Monarch Bay Golf Club, and up to 2 MGD reserved for future users of the San Leandro Recycled Water Project. This is a conservative assumption, given that the San Leandro Recycled Water Project is not currently operational (see Section 3.3.3)
- CCCSD WWTP: Existing commitments of 3 MGD based on 1.1 MGD for in-plant use, 1.2 MGD for irrigation in Zone 1, and 0.5 MGD for serving the Shell Refinery (CCCSD, 2016). Concord Naval Weapons Station demand was not included, as the project will also produce additional wastewater supplies.
- DSRSD and Livermore: These two plants send treated secondary effluent to the Livermore-Amador Valley Water Management Agency (LAVWMA) pipeline, which connects to the East Bay Dischargers Authority (EBDA) deep water outfall (located in the District's water service area). During the summer irrigation season, very little flow is directed to the LAVWMA pipeline. Up to 10 MGD is available in the winter (November through April) from the two plants combined (West Yost 2017).
- West County WPCP: Most of the flow from this plant currently goes to the NRWRP and Richmond Advanced Recycled Expansion (RARE) to supply the Chevron Richmond refinery. For the potable reuse evaluation, however, no flow was assumed to go to RARE. The potable reuse alternatives are assumed to be a substitute for RARE in the event that the Refinery is no longer operational.

No recycled water commitments were subtracted from the available supply for Pinole/Hercules WPCP, Richmond WPCP, Oro Loma WPCP, and Crockett Community Services District (CSD) Water Treatment Facility.

Satellite treatment alternatives were also considered using raw wastewater from the District's collection system (including the interceptors). Two locations were identified:

- Raw wastewater delivered to SD-1: In lieu of upgrading the entire SD-1 plant to achieve nutrient removal (see next section), a portion of the influent stream could be routed to a new wastewater treatment facility. A 4-MGD and 10-MGD MBR plant were assumed for two alternatives at SD-1. Such a facility could also serve a dual use providing non-potable recycled water for the EBRWF provided the established non-potable recycled water quality objectives are met. A satellite facility would also allow dewatering return streams to be separated out from the plant influent, potentially reducing the load of constituents of concern from the District's biosolids program.
- Raw wastewater near Albany: Collection system modeling performed by Woodard & Curran in support of the District's Infiltration/Inflow Control Project Program - Data Assessment and Modeling Project



indicates the following estimated dry weather flow available in or near the North Interceptor in the vicinity of Pt. Isabel:

- **1.5 MGD near Pt. Isabel/Central Ave**, reflecting flows from Stege Sanitary District only.
- 4 MGD near Buchanan Street, where there are contributions from both Stege Sanitary District and City of Albany. On this basis, a satellite treatment facility was sized at 4 MGD. Since actual flow data was not used, this estimate would have to be confirmed using monitoring data prior to developing a more detailed project description for this alternative.

The estimated firm daily supply for each wastewater source was reduced by 20% to account for advanced treatment train waste streams (approximately 5% for microfiltration or MBR, and 15% for reverse osmosis).

In addition to the 11 WWTP's, the East Bay Discharges Authority (EBDA) disposal system also runs through a portion of EBMUD's service area. As discussed below in Section 4.3.2, it is assumed that secondary treatment upgrades would be required to produce effluent appropriate for advanced treatment. Because EBDA conveys comingled wastewater from numerous agencies and would therefore require upgrades at multiple WWTPs and also conveys the brine concentrate stream from Zone 7's demineralization plant, the EBDA outfall is not considered further.

4.3.2 Water Quality Considerations of Treated Effluent

A high-quality feed water is essential for advanced treatment. Each of the potential wastewater sources for advanced treatment considered in this study currently provides secondary treatment without significant nitrification or denitrification. Additional treatment was assumed to be required to make this feed water suitable for advanced treatment, based on recommendations from the Advisory Group on the Feasibility of Developing Uniform Water Recycling Criteria for Direct Potable Reuse in California (SWRCB, 2016a). The required additional treatment consists of longer solids residence time (SRT) and partial removal of nitrogen (nitrification/denitrification).

Separately from this study, the Bay Area Clean Water Agencies (BACWA) recently completed a Nutrient Reduction Study that quantifies opportunities for removing nutrients from wastewater discharged to San Francisco Bay, either through optimizing existing operations or upgrading treatment plants (BACWA, 2018). The study includes cost estimates for upgrading each of the 37 municipal wastewater treatment plants to in the Bay area to achieve a total nitrogen concentration below 15 mg/L. This corresponds to "Level 2" nitrogen removal, and for most of the plants in this potable reuse study it will be achieved by adding aeration tankage to allow for Modified Ludzack-Ettinger process (anoxic zone upstream of aeration), while also achieving a longer SRT overall.

"Level 2" nitrogen removal upgrades are used for the potable reuse alternatives because it corresponds to a sufficient long SRT (greater than 5 days, preferably 7-10 days in dry weather), which is known to result in improved effluent quality suitable for advanced treatment and containing fewer trace pollutants (Chemicals of Emerging Concern, or CECs). Lower nitrogen is also a benefit when there is an environmental buffer (groundwater or surface water reservoir).

Upgrades for nutrient removal were assumed to be driven by NPDES permit requirements, so the costs are not included in the capital cost estimates for each alternative. However, they are an important consideration for the timing of implementation – potable reuse treatment trains are more sensibly added on if and when nutrient upgrades have already been completed.

Dissolved solids (expressed as hardness, TDS, salinity, or conductivity) are also a concern for advanced treatment, as they control sizing and design of the RO and chemical feed systems. In the study area, most of the wastewater sources have a specific conductance in the range of about 900-1,300 microSiemens/centimeter (μ S/cm), although SD-1 is significantly higher (effluent exceeds 2,000 μ S/cm). For cost estimating purposes, all RO systems were sized using the same recovery rate regardless of the wastewater source.







Note: City of Livermore is not shown; flows are combined with DSRSD in the LAVWMA pipeline.



Wastewater Supply Source	Permitted ADWF Treatment Capacity (MGD)	2015 ADWF Discharged to Bay (MGD)	Estimated Supply Available for Reuse (MGD)	Estimated Supply Available for Potable Reuse Supply (MGD)	Effluent Disposal Method	Notes
Oro Loma WPCP	20	10.3	10	8.0	SF Bay via EBDA outfall	1
City of San Leandro WPCP	7.6	4.3	1.7	1.4	SF Bay via EBDA outfall	1, 3, 6
DSRSD WWTP	17	2	Included with Livermore, below	Included with Livermore, below	SF Bay via LAVWMA/EBDA	2, 8
Livermore Water Reclamation Plant	8.5	3.0	0 (summer) 10 (winter)	0 (summer) 8 (winter)	SF Bay via LAVWMA/EBDA	4, 8
		Included with				
Richmond WPCP	16	West County, below	4.5	3.6	SF Bay	4
West County WCPC	12.5	5.6	1.1	0.9	SF Bay via Richmond	4, 6
Crockett CSD Water Treatment Facility (C&H)	1.8	0.8	0.8	0.6	Carquinez Strait	4
EBMUD Main WWTP (SD-1)	120	43	38	30	SF Bay	7
Pinole/Hercules WPCP	4.1	2.1	2.1	1.7	San Pablo Bay via Joint Outfall	7
Rodeo WPCF	1.1	0.5	0.5	0.4	San Pablo Bay via Joint Outfall	7
CCCSD WWTP	54	27	24	19	Suisun Bay	7, 5

Table 4-5: Sources of Secondary Effluent for Potable Reuse

Notes:

1. EBDA - NPDES Permit for Common Outfall. Order No. R2-2017-0016, NPDES No. CA0037869. San Francisco Bay Regional Water Quality Control Board, 2017.

2. Dublin San Ramon Services District - NPDES Permit. Order No. R2-2017-0017, NPDES No. CA0037613. San Francisco Bay Regional Water Quality Control Board, 2017.

3. City of San Leandro Recycled Water Market Assessment Study. Prepared by Carollo Engineers for City of San Leandro. Draft, January 2016.

4. CIWQS. Available at http://www.waterboards.ca.gov/water_issues/programs/ciwqs/

5. (CCCSD, 2016)

6. EBMUD Water Supply Management Program 2040 Plan, Appendix D. April 2012.

7. 2016 Nutrient Watershed Permit Group Annual Report.

8. (West Yost, 2017)



4.4 Targets for Advanced Treated Water

This section presents additional detail on the targets for potable reuse projects within the District's water service area.

4.4.1 Groundwater Augmentation Targets

Both the southern and northern most portions of the East Bay Plain Subbasin were identified as candidates for potable reuse as discussed below (see **Figure 4-4**).







4.4.1.1 Southern portion of the East Bay Plain Subbasin

The East Bay Plain Subbasin extends from the East Bay foothills west to San Francisco Bay, approximately from Richmond south to Hayward. The East Bay Plain Subbasin has three main aquifer units, including the Deep Aquifer which is believed to produce the highest yield and be the most continuous aquifer unit. The Deep Aquifer is located about 500 feet below grade.

As part of the Bayside Groundwater Project, the District constructed a 1-MGD aquifer storage and recovery well facility at the Oro Loma Sanitary District (OLSD) WWTP site in 1997, with the intent of supplementing the groundwater basin with potable water during wet years and extracting the water during dry years (EBMUD, 2005). The facility became operational in 2010, and test injections with potable water most recently occurred in February 2017. When operational, the facility injects and extracts from the deep aquifer unit. Extracted water is treated at a wellhead treatment facility before being conveyed to the potable water distribution system. Wellhead treatment includes manganese removal and disinfection.

For the OLSD Recycled Water Facility Study (OLSD, 2016), the District's Bayside Groundwater Project concept was extended to a series of injection and extraction wells using advanced treated OLSD effluent as the source of water for injection. All wells were assumed to operate year-round at a constant flow rate. This differs from the Bayside Groundwater Project, in which injection and extraction occur at different times. The extracted water would undergo wellhead treatment (manganese removal and disinfection, as in the Bayside Groundwater Project) and then be added directly into the potable water distribution system near the point of extraction.

The OLSD study assumed a project size of 10 MGD for injection and 9 MGD for extraction (90% recovery rate). This study assumes a slightly smaller injection rate (8 MGD) based on new wastewater flow data (the average dry weather flow rate is now about 10 MGD, not 12 MGD as stated in the OLSD study). Pipeline and well facilities and cost estimates from the OLSD study were used after adjusting for flow rates (i.e., 80% of the OLSD costs were used, corresponding to four 2-MGD injection wells and eight 1-MGD extraction wells). Preliminary pipeline and well locations are shown in **Figure 4-13** (see page 4-39).

4.4.1.2 North Richmond Area of the East Bay Plain Subbasin

The East Bay Plain Subbasin in the vicinity of North Richmond is not currently used for potable water production, although there are non-potable wells in the area. This portion of the groundwater basin was historically used for potable water production – for example, the San Pablo Well Field historically produced about 2 MGD before eventually pulling in seawater (EBMUD, 2001). While this is a small amount, it is not known whether the production rate was limited by the technology in use at the time or the inherent properties of the groundwater basin.

The District has identified an area in North Richmond as potentially favorable for the development of new wells, based on local geology and historic groundwater use (EBMUD, 2001). West County Wastewater District is located within this area, and the Richmond WPCP is also located nearby. Either or both of these facilities could theoretically serve as a source of advanced treated water for injection into the groundwater basin. However, there are key unknowns affecting the maximum size of the project, including:

- Accurate characteristics of local groundwater basin, including thickness of aquifer units and properties;
- Available storage capacity;
- Hydrogeology; and
- Water quality data.

The condition of existing groundwater quality including potential contamination sources that could interact with the project is also unknown; as described in the *Regional Hydrogeologic Investigation* completed for the District (EBMUD,



2001), the area is "virtually blanketed with potential shallow water contaminant sites" in addition to a large plume from Chevron that is adjacent to the west side of the basin. Due to the numerous unknowns that directly relate to facility costs, such as well location and depth, it is not possible to develop a cost estimate for this alternative. However, it may be considered for further study in the future.

4.4.2 Reservoir Augmentation Targets

Four of the District's five East Bay terminal reservoirs were considered as targets for reservoir augmentation, due to their larger size and configuration: Briones Reservoir, San Pablo Reservoir, Upper San Leandro Reservoir, and Lake Chabot (see **Figure 4-3** on page 4-11). Lafayette Reservoir was not considered because it is significantly smaller and is not used as a part of routine water supply operations. The amount of recycled water that each reservoir can accept was determined based on the proposed surface water source augmentation regulation (SWRCB, 2017), reservoir storage capacity, and downstream treatment plant capacity. Each reservoir was considered independently due to the unique interaction of these limitations at each proposed location; multi-reservoir options were not included in the initial development of Surface Water Source Augmentation (SWA) project alternatives.

For Briones, San Pablo, and Upper San Leandro Reservoirs, the District's Water Supply Engineering Statistical Reports for fiscal years 2008 through 2014 were used to represent the interaction of advanced treated water with the proposed reservoirs and their associated treatment plants (EBMUD 2009, 2010a, 2011a, 2012b, 2013a, and 2014a). These operations reports were averaged to provide a typical operation pattern for each reservoir, which was then re-evaluated with recycled water as an additional supply.

Simulated reservoir operations were used to estimate dilution and residence time for each alternative considered. Residence times are shortest (i.e., most likely to hit the 6-month minimum) when reservoirs are drawn down at a high rate to meet peak summer demands, typically from June to August. Dilution is lowest (i.e., the least dilution, approaching the 1% dilution criteria) when the reservoir volume is at the annual minimum, typically in the late fall.

Advanced treated water was assumed to be added year-round at a constant flow rate in order to better understand the other limiting factors. For alternative development, this approach minimizes the unit cost of providing advanced treated water, since the treatment trains can be sized to meet just one annual average demand.

The assessment for Lake Chabot was more simplistic, as this reservoir is not currently used for potable water supply.

4.4.2.1 Briones Reservoir

Briones Reservoir was considered as a target for advanced treated water from SD-1, CCCSD, Richmond WPCP, West County WPCP, or Pinole WPCP.

Briones Reservoir was created in 1964 with the construction of Briones Dam on Bear Creek. Briones Reservoir has a storage capacity of 60,500 AF, the largest of the five terminal reservoirs currently operated by the District. Briones has the highest filling priority of the terminal reservoirs, as it holds nearly half of the total standby storage and has sufficient elevation to feed all the District's water treatment plants. Additionally, Briones has the best water quality because it receives so little local inflow from its small watershed (EBMUD, 2014b).

From 2007 to 2014, an average volume of 11,000 AFY of Mokelumne Aqueduct water cycled through Briones Reservoir, and an average volume of 3,600 AFY of watershed runoff reached the reservoir. Advanced treated water sent to Briones Reservoir was assumed to replace the average volume of Mokelumne Aqueduct water.

Advanced treated water sent to Briones Reservoir may be drafted back to Briones and Orinda Centers through the Briones Aqueduct, and then continue to the Orinda WTP. This is the most direct method to connect Briones Reservoir to a WTP for potable reuse. Additionally, water in the Briones and Orinda Centers spills into San Pablo Reservoir,



which continues into the Sobrante WTP. Historically, Briones has not spilled water directly into San Pablo Reservoir; although facilities exist for this purpose, they are primarily designed for emergencies. Routine spilling water of water from Briones into San Pablo Reservoir via the Briones Dam drain valve was not considered as part of this study.

From July 2007 to June 2014, Briones Reservoir was maintained with an average monthly volume exceeding 40,300 AF. During this same period, Briones reached a maximum storage capacity of 60,130 AF. The largest possible project that could be developed in Briones Reservoir under the proposed SWA regulations is approximately 58 MGD.

Proposed reservoir augmentation project sizes included in the project alternatives are as follows:

- 30 MGD the largest project using SD-1, with wastewater supply as the limiting factor. A 30-MGD project
 would entirely replace all the Mokelumne Aqueduct Water that is currently cycled through the reservoir
 on an average annual basis (and more the annual volume cycled through the reservoir would
 substantially increase).
- 10 MGD this project would entirely replace the approximate volume of Mokelumne Aqueduct water currently cycling through the reservoir on an average annual basis.

The proposed SWA regulations require a minimum theoretical retention time and a minimum dilution of the recycled water. For Briones Reservoir, the residence time requirement is the more restrictive of these two regulations. Briones Reservoir could accept a maximum of 58 MGD of recycled water under these regulations, compared to the status quo supply from the Mokelumne Aqueduct of about 10 MGD. This larger flow rate was determined by finding the month with the most restrictive residence time parameters and limiting recycled water production to meet this most conservative limit. This larger inflow volume would require drafting from Briones Reservoir at approximately four times the current rate.

Although the largest project alternative is limited by the residence time requirement, the dilution requirement is still a relevant requirement. The actual dilution could be considerably smaller than the predicted dilution achieved due to short circuiting and stratification of the reservoir. The predicted dilution under these operating conditions is 0.45% and the regulatory maximum is 1%, so if actual mixing conditions are far from ideal, this amount of inflow may need to be reduced or additional pathogen LRV credits should be added to increase the allowable dilution requirement up to 10%.

Summary information about each alternative (and the larger 58-MGD project, for reference) is presented below in **Table 4-6.**

	Project Size Recycled Water		Minimum	Estimated Dilution	Increased Flow Through
Project Description	AFY	MGD	(months)	(Max % RW by Volume)	Reservoir (AFY)
Largest Possible Project – Limited by SWA Regulations Only	64,900	58	6.4	0.45%	54,000
Limited by Wastewater Supply	33,600	30	10.5	0.23%	21,500
Smaller Size – Approximately equal to current use of Aqueduct Water	11,200	10	23	0.06%	~0

Table 4-6: Briones Reservoir Augmentation Project Alternatives



Example simulated inflow and outflow patterns for the 30-MGD project size are presented below in **Figure 4-5**. Simulations were prepared for each of the alternatives for Briones, San Pablo, and USL Reservoirs, but the results are omitted from this report for brevity.



Figure 4-5: Briones Reservoir Operations for 30-MGD Recycled Water Project

To summarize, Briones Reservoir can accept a large volume of recycled water – more than is available from either SD-1 or Central Contra Costa Sanitary District alone – while complying with the SWA regulations. Because of the limited size of the watershed, the reservoir inflows are already highly controlled, and there is very little risk that advanced treated water would be spilled downstream.

The reservoir's ability to be used as an emergency standby storage would not be impaired by the addition of advanced treated water. In an emergency, Briones Reservoir can be used as a supply for all of the District's WTPs, with a maximum drawdown rate of 80 MGD for the auxiliary blow off, 100 MGD for the multi-jet sleeve valve, and 130 MGD for the 36-inch cone valve. These outflow rates result in a residence time less than 6 months. Therefore, in an emergency where Briones Reservoir is rapidly drawn down, it may be necessary to reduce inflows of advanced treated water to continue complying with the proposed SWA project residence time regulations.

4.4.2.2 San Pablo Reservoir

San Pablo Reservoir was considered as a target for advanced treated water from SD-1, Richmond WPCP, West County WPCP, Pinole WPCP, or a new satellite treatment plant near Pt. Isabel.

San Pablo Reservoir was created in 1920 with the construction of the earthen San Pablo Dam on San Pablo Creek. The reservoir has a storage capacity of 38,600 AF (EBMUD, 2014b). San Pablo Dam was reinforced once in 1979, and again in 2008, without increasing the storage of the reservoir during either construction.

San Pablo Reservoir supplies water directly to the Sobrante WTP and receives wash water from the Orinda WTP. Briones Reservoir can supply San Pablo Reservoir with spill water, but historically has not done so. Additionally, San Pablo Reservoir is connected to the San Pablo WTP, which is currently mothballed. From 2007 to 2014, an average volume of 14,000 AFY of Mokelumne Aqueduct water and wash water from Orinda WTP water cycled through San Pablo Reservoir, while an average volume of 12,000 AFY of watershed runoff reached the reservoir. Hydraulically, San



Pablo Reservoir sits at the end of the District's aqueducts, so water spills to the reservoir whenever aqueduct flows exceed the deliveries to the District's other facilities (Orinda and Walnut Creek WTPs, and Briones and Upper San Leandro Reservoirs). Some spillage is operationally necessary for system hydraulics, although an exact volume is not known.

San Pablo Reservoir was maintained with an average monthly volume of at least 18,000 AF between July 2007 and June 2014. During this same period, San Pablo reached a maximum storage of 38,100 AF.

San Pablo Reservoir is potentially limited in the amount of additional water it can accept by several factors including facilities limitations, regulatory guidelines, and operational limitations. The project alternatives developed here include 4 MGD (the maximum amount under Surface Water Augmentation regulations without replacing any of the Aqueduct Inflow) and 12.5 MGD (approximately equal to replacing the amount of aqueduct water currently cycled through the reservoir). Summary information about each alternative is presented in **Table 4-7**.

_	Project Size Recycled Water		Minimum Residence	Estimated Dilution	Increased Flow Through	
Project Description	AFY	MGD	lime (months)	(Max % RW by Volume)	Reservoir (AFY)	
Limited by SWA Regulation, Downstream WTP, and Current			•	• • • • •		
Aqueduct Use	5,000	4	6 months	0.08%	7,000	
Approximately equal to current use of Aqueduct Water	14,000	12.5	9 months	0.16%	~0	

Table 4-7: San Pablo Reservoir Augmentation Project Alternatives

Under the 4-MGD alternative, recycled water would be supplied to the reservoir in addition to the average amount of water currently supplied by the Mokelumne Aqueduct due to spillage and/or intentional filling. To accommodate the extra water added to the reservoir, Sobrante WTP would have to be operated at higher production rates in the fall and spring. The alternative was developed with the goal of limiting winter (December-March) reservoir outflows to the status quo outflow volume, concentrating the increased production in the spring and fall months (November, April, and May). Running the Sobrante WTP year-round would also be feasible from a Surface Water Augmentationregulatory perspective, though not in line with current District operations. Due to the increased supply from recycled water, annual production at Sobrante WTP would have to increase by approximately 40%.

Entirely replacing the inflow from the Mokelumne Aqueduct with a constant supply of recycled water would result in a 12.5-MGD project. This size requires no dramatic change in the annual production at Sobrante WTP, but reservoir and aqueduct operations would significantly differ from the status quo.

Since there is considerable uncertainty about the District's ability to decrease deliveries from the Mokelumne Aqueduct to San Pablo Reservoir, the alternatives developed for cost and non-cost evaluation were limited to the more easily operable 4-MGD alternative.

One potential benefit of targeting San Pablo Reservoir is the potential to reuse the San Pablo Tunnel, which had been used to convey water from the reservoir to the San Pablo WTP until the plant was decommissioned. For those reservoir augmentation alternatives with Richmond WPCP, SD-1 or the Point Isabel satellite facility as the source, it was assumed that the San Pablo tunnel would be rehabilitated and reused to serve as part of the pipeline alignment.



4.4.2.3 Upper San Leandro

Upper San Leandro Reservoir was considered as a target for advanced treated water from Oro Loma WPCP, San Leandro WPCP, or SD-1.

Upper San Leandro Reservoir was created in 1929 with the construction of the earthen San Leandro Dam on San Leandro Creek. A second dam was constructed in 1978 directly downstream of the first San Leandro Dam, after it was determined to be seismically inadequate. The reservoir has a storage capacity of 38,000 AF (EBMUD, 2014b).

Upper San Leandro Reservoir supplies water directly to the Upper San Leandro Water Treatment Plant (USLWTP). Upper San Leandro Reservoir can also spill to Chabot Reservoir (originally named Lower San Leandro Reservoir), but this has only historically happened during wet months as a form of regulating the storage in the reservoir and is not an annual practice. Upper San Leandro Reservoir stores mainly local runoff, with relatively small amounts of Mokelumne Aqueduct water supplementing the supply. From 2007 to 2014, an average volume of 1,800 AFY of Mokelumne Aqueduct water cycled through the reservoir, compared to more than 12,000 AFY of watershed runoff.

Upper San Leandro was maintained with an average monthly volume of at least 21,120 AF between July 2007 and June 2014. During this same period, Upper San Leandro reached a maximum storage of 37,600 AF, just 6 inches below the top of San Leandro Dam.

The amount of recycled water than can be feasibly added to Upper San Leandro Reservoir is limited by USLWTP capacity and SWA regulatory limits. Upper San Leandro Reservoir is located directly upstream from the USLWTP, which could reliably take on a larger volume of water than it currently does. The maximum amount of water that USLWTP can accept is 4,200 AF/month. The maximum amount of recycled water that Upper San Leandro Reservoir can accept in every month before violating SWA regulations is 3,520 AF/month. This regulatory limit is higher in other months, depending on the amount of storage in the reservoir at the time of the measurement and the monthly flow to USLWTP.

The project alternatives developed here include 34 MGD (the maximum amount under SWA regulations), 20 MGD (a medium-sized project), and 1.6 MGD (the current use of aqueduct water). Summary information about each alternative is presented below in **Table 4-8**.

	Project Size Recycled Water		Minimum Residence	Estimated Dilution (Max % BW by	Increased Flow Through Beservoir
Project Description	AFY	MGD	(months)	Volume)	(AFY)
Limited by SWA Regulation Only	39,000	34	6	0.51%	37,200
Medium Size	23,300	20	6	0.30%	21,500
Approximately equal to current use of Aqueduct Water	1,800	1.6	15	0.02%	~0

Table 4-8: Upper San Leandro Reservoir Augmentation Project Alternatives

The maximum volume of recycled water that could be put into the reservoir and comply with proposed SWA project residence time and dilution requirements is 34 MGD, the largest project size considered. This is about twenty times more water than the amount of Mokelumne Aqueduct water that currently cycles through the reservoir. This alternative exceeds the capacity of USLWTP and was developed primarily to demonstrate the effect of residence time limits. This larger inflow volume requires drafts from the reservoir to increase by about a factor of three to four.



The volume of recycled water that could be added to the reservoir while complying with proposed SWA regulations, staying within the capacity of USLWTP, and staying with the seasonal operation strategy for USLWTP is 20 MGD. For this alternative, USLWTP was assumed to be offline from November through February, and the outflow during this period was limited to the status quo reservoir spill rate. The 20-MGD input of recycled water requires drafts from Upper San Leandro Reservoir to increase by a factor of more than three.

Simply replacing the inflow from the Mokelumne Aqueduct with a constant supply of recycled water would result in a 1.6 MGD project. This size requires no re-working of the draft amounts to the USLWTP, but the storage pattern in the reservoir would differ from the status quo.

Upper San Leandro Reservoir is home to native rainbow trout (*Oncorhynchus mykiss*), whose migration was blocked by the construction of the dam. The rainbow trout in the reservoir are a rare un-hybridized population. Due to the presence of these rainbow trout, Upper San Leandro Reservoir is considered more environmentally sensitive than other reservoir augmentation alternatives.

4.4.2.4 Lake Chabot

Lake Chabot was created in 1874 with the construction of Chabot Dam. The dam was reconstructed and raised in 1980, and a seismic retrofit of the dam was recently completed in late 2017. The lake serves as a standby terminal reservoir to be used in an emergency, during which chlorinated raw lake water would be routed directly into the major distribution systems. However, the reservoir is currently not connected via pipelines to any distribution facilities or Upper San Leandro Reservoir (EBMUD, 2014b). The reservoir has a storage capacity of 10,400 AF.

Lake Chabot receives inflows from rainfall, runoff from a small watershed area of Upper San Leandro Creek, and, in some years, releases from Upper San Leandro Reservoir. It does not receive any water directly from the Mokelumne Aqueduct. The reservoir is currently used to meet three non-potable water demands:

- Irrigation at Lake Chabot Golf Course (123 AFY; West Yost, 2017);
- Irrigation at Redwood Canyon Golf Course (133 AFY; West Yost, 2017);
- Support of in-stream flows for fish habitat in Upper San Leandro Creek downstream of Chabot Dam (Approx. 130 AFY, pers. Comm. M. Tognolini, Nov. 8, 2017).

The April 2017 *Oakland Hills Alternative Water Supply Feasibility Study* (West Yost, 2017) identified Lake Chabot as a possible source of water for additional non-potable irrigation demands at Sequoyah Country Club, the Oakland Zoo, and the planned Oak Knoll Development. However, the report states that Lake Chabot's supply is insufficient to meet these additional demands.

Recycled water could be added to Lake Chabot to support additional in-stream flows in Upper San Leandro Creek, as well as additional non-potable irrigation demands such as those considered in the 2017 Oakland Hills study. Lake Chabot reservoir augmentation was included as a <u>potable</u> (rather than non-potable) reuse project alternative for two main reasons:

- Although Lake Chabot is not currently used for potable water supply, the reservoir is an emergency potable water supply for the region.
- The project could be used as the first phase of a reservoir augmentation project for Upper San Leandro Reservoir, allowing the District to gain experience with operating an advanced water treatment plant prior to adding it to the potable water supply.

Several sources were considered to support the Lake Chabot alternative, including the LAVWMA pipeline, which has about 10 MGD of available secondary-treated wastewater from DSRSD and the City of Livermore. The supply is limited



to the winter months, November through April (Figure 5, West Yost Associates, 2017), and there is very little supply available during irrigation season. A satellite treatment facility could be constructed adjacent to the LAVWMA pipeline, with MF and RO concentrate being returned to the pipeline for disposal through the EBDA outfall.

At a flow rate of 10 MGD, the residence time in Lake Chabot, which has a minimum storage volume of about 7,000 AF, would be 7.5 months, comfortably complying with the reservoir augmentation criteria of 6 months. However, Lake Chabot does not currently meet the proposed SWA requirement of operating as an approved surface water source for at least two years. As stated previously, while it is designated as an emergency water supply, it is not currently connected to the District's potable water distribution system and is not operated for water supply at this time.

Advanced treated water sent to Lake Chabot could be used to support:

- Additional non-potable demands;
- In-stream flows; and
- Supply to USLWTP, if and only if Lake Chabot is classified as an approved drinking water source.

No pipelines to connect to non-potable customers or USLWTP were included in the Lake Chabot alternative.

4.4.3 Raw Water Augmentation Targets

For raw water augmentation, advanced treated water could be delivered to the District's raw water aqueducts, or directly to a surface water treatment plant. San Pablo WTP was not considered as a raw water augmentation target since it is currently not in use and Lafayette WTP was not considered because of its location far from any wastewater source.

Advanced treated water can be added to the District's conventional surface water treatments plants or the "in-line" filtration plants. The additional treatment steps (flocculation and sedimentation) provided at the conventional surface water treatment plants are to remove fine particulates from the existing sources (which is not a concern in advanced treated water). These processes must be retained for compliance with the Surface Water Treatment Rule. These processes are not expected to significantly alter the quality of the advanced treated water under normal operating conditions, when the advanced treated water will be essentially particle-free after MF/RO/AOP. However, in the event of a failure of the advanced treatment system, the additional sediment removal and ozonation available at the conventional WTPs would provide more treatment system redundancy than is available at the in-line water treatment plants.

4.4.3.1 Conventional Surface Water Treatment Plants

USLWTP and Sobrante WTP are fed primarily from local runoff and local reservoirs. USLWTP is fed by Upper San Leandro Reservoir, while Sobrante WTP is fed from San Pablo Reservoir. These two WTPs provide full conventional treatment consisting of five basic steps—coagulation, flocculation, sedimentation, filtration, and disinfection, as well as ozonation for taste and odor.

USLWTP was considered as a target for advanced treated water from Oro Loma WPCP, San Leandro WPCP, or SD-1. Sobrante WTP was considered as a target for advanced treated water from Pinole WPCP, Richmond WPCP, or West County WPCP.

4.4.3.2 In-Line Water Treatment Plants

Orinda, Walnut Creek, and Lafayette WTPs, the "in-line filtration" water treatment plants, are fed directly from the Mokelumne Aqueduct, rather than being fed primarily from local runoff and local reservoirs. The process train at the



in-line WTPs includes only coagulation, filtration, and disinfection. Flocculation and sedimentation are not needed at the inline filtration plants because they were permitted as alternative technology facilities. Orinda WTP was considered as a target for advanced treated water from SD-1. Walnut Creek WTP was considered as a target for advanced treated water from CCCSD via the Mokelumne Aqueduct, as described below.

4.4.3.3 Mokelumne Aqueduct

Advanced treated water can be added directly to the Mokelumne Aqueduct and used at any of the District's surface water treatment plants. This alternative involves adding water near Mallard Reservoir, consistent with a previous potable reuse study conducted by CCCSD (CCCSD, 2016). The hydraulic grade line of the aqueduct is approximately 400 feet at this location. The water could be added to Mokelumne Aqueduct No. 2, similar to the transmission concept developed for the District to accept deliveries from the Los Vaqueros Reservoir Expansion project (CCWD, 2017). The water could be isolated for delivery to Walnut Creek WTP or distributed throughout the raw water system including Orinda WTP, Lafayette WTP, and the terminal reservoirs.

4.4.4 Treated Water Augmentation Targets

The targets for treated water augmentation are connection points within the treated water distribution system. Potential connection points were identified using a process similar to that described in a 2016 District assessment of potable reuse from SD-1 to the West-of-Hills distribution system (Maggiore, 2016) as well as consultation with District staff. The preliminary assessment presented in the 2016 memo was used as the basis for the siting and sizing of potable water system connections. Potential connections to the potable system were selected based on recommendations provided by District staff to maximize blending and distribution of the treated water within the distribution system while also minimizing negative impacts on the system hydraulics. Potential connection pipelines were sized per the standard maximum velocity for a connection (7 fps). Connections to the potable system would require a new pump station to provide similar head at the connection point as under existing conditions. Alternatively, the connection to the potable system could be routed through an existing treated water tank, requiring a longer connection pipeline but potentially less operational complexity.

A potable water system connection was identified for each of the wastewater sources aside from the CCCSD WWTP, which is not within the District's treated water service area. Connection pipe sizes were based on the velocity criteria discussed in Section 4.6.2. Pump stations were sized based on the approximate head at the connection point, per District pressure zone and hydraulic model figures. Treated water augmentation targets are summarized below in **Table 4-9**.

Wastewater Treatment Facility	Supply (MGD)	Proposed Connection Point	Head Required at Connection Point (ft.)
San Leandro WPCP	1.4	Dunsmuir Reservoir	222
Pinole/Hercules WPCP	1.7	Maloney Reservoir	324
Richmond WPCP	3.6	Wildcat Aqueduct (36-in)	205
West County WPCP	4.7	Wildcat Aqueduct (36-in)	205
Oro Loma WPCP	8.0	South Reservoir	222
SD-1	10 or 30	Claremont Center Distribution Pipeline (36-in)	345

Table 4-9: Treated Water Augmentation Connections



4.5 Advanced Water Treatment Trains

Advanced treatment trains were developed to bridge the gap between the wastewater source and the potable reuse target. Potable reuse treatment technologies have been documented in both demonstration and full-scale applications through years of research and performance monitoring. The treatment trains documented here were developed primarily for cost-estimating purposes and to demonstrate the viability of the potable reuse alternatives. Advanced treatment for potable reuse is still a relatively new application, so additional technologies are expected to be available in coming years – particularly those related to continuous online monitoring that can demonstrate increase levels of pathogen removal and reduced response time.

The unit processes used within the proposed treatment trains are described in the sections below. Pathogen log reduction values (LRVs) are identified for each unit process based on other potable reuse projects currently in planning stages and submitted to DDW for review. A rationale for the sizing of each treatment train component is also provided.

4.5.1 Unit Processes

4.5.1.1 Upgraded Secondary Treatment

Each wastewater source considered for advanced treatment currently has secondary treatment, as discussed in Section 4.3. Secondary improvements to achieve partial denitrification down to 15 mg/L Total Nitrogen and longer SRT were assumed to align with year-round "Level 2" nutrient removal as determined in the BACWA Nutrient Reduction Study. Cost estimates for these improvements are available in the BACWA Nutrient Reduction Study Report (BACWA, 2018). Capital cost estimates range from \$19M for the Pinole WPCP to \$2.2B for SD-1.

Some facilities (i.e., SD-1, CCCSD WWTP, and others) have existing tertiary filtration systems for production of nonpotable recycled water; however, these tertiary filtration systems were <u>not</u> considered to be a component of the advanced treatment trains. This assumption is important from a cost-estimating perspective; the existing tertiary filtration systems are typically smaller than the full treatment capacity, and it is more conservative from a cost estimating perspective to exclude them. The decision to include / exclude existing tertiary treatment an advanced treatment may be re-assessed for a select group of alternatives.

Overall Performance for Pathogen Reduction: A maximum of 2.0 log reduction for virus, 2.0 log for giardia, and 1.0 log reduction for *Cryptosporidium* (2/2/1 LRV) is expected to be available for a secondary treatment system with long solid retention time, based on Table 2-2 in the Expert Panel Report (SWRCB, 2016b). Given uncertainty about DDW granting credits for pathogen removal in secondary treatment, however, the total LRV for each treatment train exceeded the goal by at least the amount credited here (i.e., these are "bonus" removal credits).

Sizing Criteria: In virtually all cases, the full flow of the WWTP was assumed to be upgraded for Level 2 nutrient removal, given the operational difficulty of re-configuring a wastewater plant for split treatment. The only exception was for alternatives at SD-1 with flow rates of 10 MGD or less, which were assumed to use an MBR to produce nitrified and denitrified secondary effluent.

4.5.1.2 Membrane Bioreactor

An MBR is a technology that is typically used to treat BOD, COD, solids (TSS and turbidity), and nitrogen (ammonia and nitrates) in raw wastewater or primary effluent. The process essentially consists of aerobic microbial treatment followed by MF. In the aerobic treatment component of the process, microbes in a suspended growth environment metabolize organic compounds and, depending on the system design, provide nitrification and denitrification. Effluent from the aerobic treatment is then filtered through the MF system. Filtrate from the MF system is conveyed as treated effluent. Solids filtered (both suspended microbes and solids) screened by the MF system are regularly flushed from



the membrane surfaces and returned to the front end of the aerobic treatment system. Excess biomass and solids are periodically discharged as a waste stream to maintain a balance of solids inventory of the MBR.

Several of the alternatives start with raw sewage using a satellite treatment plant concept. Those alternatives were assumed to include an MBR to provide biological treatment with nitrification and denitrification at least equivalent to the upgraded secondary treatment, and membrane filtration.

Treatment trains with an MBR do <u>not</u> include separate standalone microfiltration. Note that for treated water augmentation treatment train with MBR providing secondary treatment upstream of ozone and BAC, colloidal material may slough off the BAC, requiring additional maintenance. For this case, additional measures may be required to protect the RO process.

Overall Performance for Pathogen Reduction: A maximum of 1.5/2/2 LRVs for virus/*GiardialCryptosporidium* is expected to be available for MBR. The LRVs are based on the Australian WaterVal program's recently published Tier 1 validation protocol, which is currently under consideration by DDW, and has recently been approved on an individual project basis. LRVs are more difficult to verify in MBR than in other MF systems, where pressure decay testing is an option, and research on verifiable LRVs in MBR is an area of active research.

Sizing Criteria: Satellite treatment using an MBR was sized by alternative-specific criteria, as follows:

- Satellite treatment at Pt. Isabel was sized according to the available dry weather flow;
- Satellite treatment at SD-1 for reservoir augmentation to San Pablo Reservoir was sized according to the constraints at San Pablo Reservoir (4 MGD);
- Other satellite treatment at SD-1 was sized at 10 MGD a somewhat arbitrary number that was selected to represent a medium-sized alternative, less than the full 30 MGD available.

The minimum space requirements for MBR treatment were assumed to be approximately 3,700 square feet per MGD.

4.5.1.3 Ozonation with Biologically Active Carbon

Ozonation (O_3) followed by biologically active carbon (BAC) is a treatment process intended to facilitate the breakdown of large, more recalcitrant organic molecules into simpler organic compounds that can be removed more easily through microbial consumption. In this process, O_3 is injected into the feed stream to break down organic compounds and trace pollutants through oxidation. The smaller molecules are then metabolized as a food source by the biofilm developed in the BAC media, which consists of granular activated carbon (GAC). The primary benefits of O_3 /BAC treatment are as follows:

- Reduction of TOC and trace pollutants through the mechanism described above.
- Reduction of pathogens, as ozone is a potent disinfectant.
- Improvements to the performance of downstream membrane filtration (MF/RO)

For this RWMP Update, Ozone/BAC was included only for the treated water augmentation alternatives. The treatment trains for groundwater augmentation, reservoir augmentation, and raw water augmentation derive sufficient pathogen removal from groundwater travel or a surface water treatment plant, such that O₃-BAC is not expected to be needed. Although not included in the raw water augmentation alternatives developed for this study, it is possible that the raw water augmentation regulations expected in 2023 will necessitate the addition of O₃-BAC to the treatment train.

Overall Performance for Pathogen Reduction: The LRVs assumed for the O₃-BAC process are as follows:

• O₃ – 6/6/1 LRVs for virus/*Giardia*/*Cryptosporidium* (Trussell, 2015). Higher LRVs for cryptosporidium are achievable with higher dosing.



BAC – 0/0/0 LRVs for virus/Giardia/Cryptosporidium. Although 1/2/2 LRVs can be achieved through filtration
with an effluent turbidity of 0.1 NTU per the USEPA Long Term 2 Enhanced Surface Water Treatment Rule,
this mode of operation was not assumed for this analysis. As the GAC media would be intended for use as a
microbial substrate rather than filtration as its primary function, biomass could potentially slough from the
media and into the BAC effluent. Therefore, the 0.1 NTU BAC effluent turbidity requirement might not be met
during portions of the BAC operation.

Sizing Criteria:

- Preliminary design and planning level information were used as the basis for estimating the equipment
 requirements for this effort. These sources include the Expedited Purified Water Program, Ford and Coyote
 Facilities Plan Final Report (Woodard & Curran, 2017a) and Oro Loma Sanitary District Potable Reuse
 Evaluation (HDR, 2016). Note that the information from these sources was also used to estimate the
 equipment requirements for the microfiltration, reverse osmosis, and advanced oxidation processes discussed
 in the sections that follow.
- O₃:TOC ratio of 1:1 and a contact time of 2 minutes for the O₃ contactor.
- Empty bed contact time of 20 minutes for the BAC.

4.5.1.4 Microfiltration

In a conventional FAT train configuration, low pressure membrane filtration (either ultrafiltration (UF) or microfiltration (MF)) systems are typically installed as the first treatment operation of the FAT process. These are physical filters which remove suspended solids and colloidal particulates from the process water upstream of the RO system. If left in the process water, these solids could impair the operation of the RO process by organic fouling or plugging of the RO membrane surfaces.

Membranes used for MF applications have a nominal pore size rating of 0.1 μ m, whereas UF membrane have smaller nominal pore rating of 0.01 μ m. Both MF and UF membranes are robust technologies that have been proven to be effective to remove *Giardia* and *Cryptosporidium*, algae, and some bacterial species. The notable distinction between MF and UF is that MF is not an effective barrier to viruses, whereas UF has been shown to have some effectiveness with virus removal, although Integrity testing to demonstrate virus removal is an additional technical challenge. MF/UF processes have not been shown to remove a significant amount of chemical pollutants.

Overall Performance for Pathogen Reduction: 0/4/4 LRVs for virus/*Giardia*/*Cryptosporidium*. Both MF and UF have been demonstrated to achieve 4 LRVs for *Giardia* and *Cryptosporidium* (Reardon, 2005). No LRV for virus was assumed for the MF treatment step.

Sizing Criteria:

• Assumed use of MF membranes; hollow fiber configuration; flux of 4-25pprox.. 32 gallons per square foot membrane area per day (gfd) average.

4.5.1.5 Reverse Osmosis

The reverse osmosis (RO) process in a potable reuse treatment train provides for removal of salt (measured as TDS and electrical conductivity (EC)), organics (measured as TOC), and pathogens. RO removes at least 95 percent of incoming salt. This treatment operation in the FAT process is sequenced between the MF/UF system and the AOP. In operation, the TDS and other dissolved components in the RO feed stream are pressurized above the osmotic pressure of the water. Once this threshold is overcome, water begins passing through the RO membrane to become permeate. The bulk of the TDS and dissolved components cannot pass through the RO membrane, and concentrate in the



reduced volume of remaining feed water. Upon exiting the RO process, the permeate is conveyed to the AOP process for further treatment. The remaining water exiting the RO system containing the concentrated TDS and dissolved components is discharged to waste as brine. Depending on the feed water quality, RO permeate can have a TDS concentration of less than 50 mg/L.

Overall Performance for Pathogen Reduction: 2/2/2 LRVs for virus/*GiardialCryptosporidium*. These LRV values were selected based on the minimum values reported in the literature (Trussell, 2015) to ensure a conservative basis when evaluating the overall performance of each alternative treatment process.

Note that the LRV allowable for RO membranes is not governed by the ability of an intact membrane to reject pathogens; it is governed by the ability to monitor the membrane integrity. Worn or damaged seals or membrane damage to the due to oxidization or abrasive from solids can allow pathogens to leak past the RO membrane. The monitoring tools currently used, EC meters and TOC meters, can measure up to 99 percent of both parameters through the RO process. Findings from demonstration tests and existing permitted installations (e.g. the Orange County Water District Groundwater Replenishment System) have demonstrated that these monitoring methods can provide virus/*Giardia*/*Cryptosporidium* LRVs of 2 for each (Trussell, 2015). Furthermore, emerging alternative technologies currently in various stages of testing and development have the potential to provide higher LRVs due to greater levels of accuracy in assessing membrane integrity. For example, Nalco has stated that test findings for their proprietary Trasar® fluorescent dye system can provide sufficient resolution to monitor more than 3 LRV for virus, *Giardia*, and *Cryptosporidium*. The assumption of 2/2/2 LRVs for RO assumes advancements in monitoring methodology will be made prior to implementation of potable reuse by the District, as currently LRVs based on TOC metering are limited to 1.5/1.5/1.5.

Sizing Criteria:

- 8-inch diameter spiral wound RO membranes of thin film composite construction. Assumed low pressure type RO membranes.
- Estimated RO recovery of 85%.

Disposal Considerations:

In addition to TDS and TOC removal, RO removes trace level pollutants such as metals and organic pollutants, including hormones, pharmaceuticals, and personal care products. RO concentrate disposal is a potential concern for NPDES permit compliance due to possible violations of numeric effluent limits, increased monitoring requirements, and/or the presence of chronic toxicity. Therefore, concentrate disposal was included as a factor in the non-cost evaluation in Section 5.1.2.

All of the alternatives involve disposal to a deep-water Bay outfall with dilution of at least 10:1. Alternatives sized to use most of a wastewater treatment plant's available dry weather flow were scored lower due to concerns over concentrate disposal, since less flow would be available for dilution prior to discharge. Alternatives with wastewater available for dilution prior to discharge were scored higher, including those at Oro Loma WPCP (dilution from other EBDA dischargers) and alternatives at SD-1 and CCCSD using less than one half of total wastewater available.

4.5.1.6 Advanced Oxidation Process

AOP is the third treatment operation of the FAT process, sequenced downstream of the RO system. It is intended to provide a high level of disinfection and to break down any trace organic compounds that pass through the RO membrane. The AOP assumed for this analysis consists of ultraviolet light system combined with hydrogen peroxide (UV-H₂O₂), which is the industry standard for FAT processes. For this type of AOP, the H₂O₂ is dosed to the RO permeate entering the UV reactor. Exposure to the intense UV light in the reactor causes the H₂O₂ to form hydroxyl



radicals, which are extremely potent oxidizers. They react with any organic compounds and pathogens present in the water, breaking them down into smaller organic molecules and, ultimately, into water and carbon dioxide. Any remaining hydroxyl radicals rapidly recombine to H_2O_2 after the water exits the UV reactor.

Other types of AOP such as UV-Chlorine are also being demonstrated elsewhere for lower cost than $UV-H_2O_2$ and could be evaluated during the design phase.

Overall Performance for Pathogen Reduction: 6/6/6 LRVs for virus/*Giardia*/*Cryptosporidium* as allowed by DDW (Trussell, 2015).

UV/AOP reliably provides at least 6-log disinfection of both protozoa and virus. The same system is assumed to reduce NDMA to <10 ng/L and destroy at least 0.5-log of 1,4-dioxane, thus also reducing other trace level pollutants. Online dose monitoring systems, using real time inputs of UV, UV intensity, flow, and oxidant dosing, is recommended for continuous confidence in UV AOP performance.

Sizing Criteria: UV dose of 900 mJ/cm². Although a UV dose of 235 mJ/cm² can provide 6 LRVs for virus, *Giardia*, and *Cryptosporidium*, a higher UV dose is needed to address 1,4-dioxane and NDMA. Although neither currently has a drinking water MCL, DDW requires that at least a 0.5 log reduction of 1,4-dioxane is provided. Additionally, industry practice is to reduce NDMA to below its DDW notification level of 10 ng/L. 900 mJ/cm² is an average value needed to photolytically degrade NDMA from approximately 150 ng/L to the DDW notification level of 10 ng/L. The 150 ng/L value was taken from data for the secondary effluent from another treatment plant in the Bay Area and may not be accurate for the wastewater sources considered in this study.

The minimum space requirements for FAT (MF/RO/AOP taken together) were assumed to be approximately 5,000 ft² per MGD.

4.5.1.7 Free Chlorine

For the reservoir water augmentation and raw water augmentation alternatives, the use of free chlorine was added to the overall treatment train to ensure that a 13 LRV for virus is achieved across the entire process. The use of free chlorine is an industry standard practice for disinfection, with well-documented methodologies to calculate the contact time (CT) requirements to achieve a desired virus LRV. The LRV for virus could range from 2 to 6, the maximum allowable value. Dechlorination is required prior to discharging for reservoir augmentation, but no dechlorination step would be required for raw water augmentation.

Free chlorine is not expected to be needed to meet pathogen removal requirements for the treated water augmentation alternatives due to the inclusion of the O_3/BAC process, which provides an LRV of 6 for virus. However, it could be included for redundancy without a major impact on the overall treatment cost. The free chlorine contact time of 80 (mg-min)/L used for sizing is conservative for cost estimating purposes. A smaller contact time of 5-10 min may be feasible, and design values would be established during a later phase.

Overall Performance for Pathogen Reduction: 2/0/0 for virus/*Giardia*/*Cryptosporidium*. An LRV of 2 was deemed to be sufficient to ensure that the 13 virus LRV requirement for the overall treatment process can be met.

Sizing Criteria:

- Preliminary design and planning level information from the Expedited Purified Water Program, Direct Potable Reuse Facilities Plan Final Report (Woodard & Curran, 2017b)
- CT of 80 (mg-min)/L (Woodard and Curran, 2017b).
- Contact Tank 30-minute modal contact time, 70% baffling factor for tank.



4.5.1.8 Engineered Storage Buffers

The raw water and treated water augmentation alternatives include ESBs. For both types of potable reuse, an ESB with a total volume equivalent to 6 hours at peak production was included. Cost estimates for ESBs assume the use of 3 storage tanks, each sized for a storage volume equivalent to 2 hours at peak production. This configuration allows one storage tank to be filled, one storage tank to be tested, and one storage tank to be drained at any given time, consistent with guidance (WRF, 2016). The minimum failure response time required for this configuration is 2 hours. Microfiltration is typically the most difficult process for which to verify pathogen removal, and pressure decay testing is not likely to be feasible every 2 hours. The 2-hour failure response time assumes that an advanced monitoring technique such as particle-counting will become available to allow verification of the MF process within 2 hours. A DDW-approved technology is not currently available but is an area of active research and development.

Raw water augmentation alternatives would have an additional 1 to 3 hours of response time during travel in the transmission pipelines to the closest surface water treatment plant.

The minimum space required for ESB storage tanks were estimated based on a tank height of 30 feet (1400 ft² per million gallons of storage).

4.5.1.9 Wellhead Treatment

The treatment train for groundwater augmentation using the Oro Loma WPCP was assumed to include wellhead treatment for manganese and disinfection following extraction and prior to addition to the treated water distribution system (OLSD, 2016).

4.5.2 Treatment Trains

Seven unique treatment trains were developed to meet all possible combinations of sources and targets, as shown in **Figure 4-6** through **Figure 4-12**. These treatment trains are as follows:

- Treatment Train Type 1 Groundwater Augmentation: Partially denitrified secondary effluent treated by the FAT process (MF/RO/AOP) and free chlorine disinfection, followed by well injection to the groundwater basin. Groundwater would be extracted from downgradient potable wells for delivery to the potable water system. Potable groundwater wells would be located sufficiently far enough downgradient of the injection location to meet underground retention time requirements.
- Treatment Train Type 2 Reservoir Water Augmentation: Partially denitrified secondary effluent treated by the FAT process and free chlorine disinfection, followed by dechlorination and discharge to a surface water reservoir. Chlorine disinfection would be for advanced water treatment and to maintain a residual in the conveyance pipeline. Dechlorination would take place at the reservoir site prior to point of discharge. After mixing and a minimum theoretical retention time of 180 days, surface water from the reservoir would be treated at a surface water treatment plant followed by delivery to the potable water system.
- Treatment Train Type 3 Reservoir Water Augmentation with New MBR: Raw wastewater treated by an MBR system followed by RO-AOP, chlorine disinfection, dechlorination, and discharge to a surface water reservoir. Chlorine disinfection would be for advanced water treatment and to maintain a residual in the conveyance pipeline. After mixing and a minimum theoretical retention time of 180 days, surface water from the reservoir would be treated at a surface water treatment plant followed by delivery to the potable water system.



- Treatment Train Type 4 Raw Water Augmentation: Partially denitrified secondary effluent treated by the FAT process and chlorine disinfection, followed by an ESB. Water from the ESB (6-hour volume, 2-hour response time) would be blended with raw water prior to a surface water treatment plant.
- Treatment Train Type 5 Raw Water Augmentation with New MBR: Raw wastewater treated by an MBR system followed by RO-AOP, chlorine disinfection, and an ESB (6-hour volume, 2-hour response time). Water from the ESB would be blended with raw water prior to a surface water treatment plant.
- Treatment Train Type 6 Treated Water Augmentation: Partially denitrified secondary effluent treated by O₃-BAC and FAT processes, followed by addition of a chloramine disinfection residual, and an ESB (6-hour volume, 2-hour response time). Water from the ESB would be delivered directly to the (6-hour volume, 2-hour response time potable water distribution system.
- Treatment Train Type 7 Treated Water Augmentation with New MBR: Raw wastewater treated by an MBR system followed by O₃-BAC, RO-AOP, addition of a chloramine disinfection residual, and an ESB (6-hour volume, 2-hour response time). Water from the ESB would be delivered directly to the potable water distribution system.

The LRVs for virus/*Giardia*/*Cryptosporidium* for each treatment train type is presented in **Table 4-10**. Also presented are the LRVs for virus/*Giardia*/*Cryptosporidium* for each treatment operation, along with the LRV treatment goal for the overall train.

For reservoir augmentation, the LRV requirements are implemented separately for the Advanced Water Treatment and Surface Water Treatment Plant components, as noted in Section 4.1.2. No comparable approach has been established for raw water augmentation or treated water augmentation, so it is not known how those advanced treatment LRV requirements would be implemented.



	2º Treat-		Adv	Advanced Water Treatment (AWT)									I RV	
Pathogen	ment ¹	MBR	O 3	BAC	MF	RO	AOP	Cl ₂	URT	SAT	SWTP	LRV	Goal	
1. Groun	1. Groundwater Augmentation by Well Injection													
Virus	2				0	2	6	6	2			18	12	
Giardia	2				4	2	6	0	0			14	10	
Crypto	1				4	2	6	0	0			13	10	
2. Reserv	oir Augm	entatior	า											
Virus	2				0	2	6	2 ²			4	16	13	
Giardia	2				4	2	6	0			3	17	11	
Crypto	1				4	2	6	0			2 ³	15	11	
3. Reserv	voir Augmo	entatior	n with l	MBR										
Virus		1.5				2	6	2 ²			4	15.5	13	
Giardia		2				2	6	0			3	13	11	
Crypto		2				2	6	0			2 ³	12	11	
4. Raw W	ater Augn	nentatio	n											
Virus	2				0	2	6	2 ²			4	16	13	
Giardia	2				4	2	6	0			3	17	11	
Crypto	1				4	2	6	0			2 ³	15	11	
5. Raw W	later Augn	nentatio	on with	MBR										
Virus		1.5				2	6	2 ²			4	15.5	13	
Giardia		2				2	6	0			3	13	11	
Crypto		2				2	6	0			2 ²	12	11	
6. Treate	d Water Au	ugment	ation											
Virus	2		6	0	0	2	6					16	13	
Giardia	2		6	0	4	2	6					20	11	
Crypto	1		1	0	4	2	6					14	11	
7. Treate	d Water Au	ugment	ation v	vith MBF	2									
Virus		1.5	6	0		2	6					15.5	13	
Giardia		2	6	0		2	6					16	11	
Crypto		2	1	0		2	6					11	11	

Table 4-10: Pathogen Removal Credits for Advanced Treatment Trains

Notes:

1. Secondary treatment is assumed to include nutrient removal.

2. Free chlorine for may be designed for 2-6 LRVs.

3. Conventional surface water treatment plants (USLWTP and Sobrante WTP) provide 2.5-LRV for Cryptosporidium.




Figure 4-6: Treatment Train Type 1 - Groundwater Augmentation

Figure 4-7: Treatment Train Type 2 - Reservoir Water Augmentation







Figure 4-8: Treatment Train Type 3 - Reservoir Augmentation with New MBR









Figure 4-10: Treatment Train Type 5 - Raw Water Augmentation with New MBR

Figure 4-11: Treatment Train Type 6 - Treated Water Augmentation







Figure 4-12: Treatment Train Type 7 - Treated Water Augmentation with New MBR



4.5.3 Basis of Cost Estimates

Opinions of probable capital and O&M cost for each of the treatment alternatives were developed following the cost estimating approach as detailed in **Section 2**.

4.6 Conveyance Concepts

This section presents the methods used to develop the transmission pipeline alignments and sizing for all potable reuse alternatives as summarized in **Table 4-11**.

4.6.1 Conveyance Routing

Pipeline routes for potable reuse projects were evaluated in ArcGIS by connecting recycled water sources and corresponding reuse targets with the shortest path between them, as follows:

- **Groundwater Augmentation:** Pipeline lengths and sizing for connecting the Oro Loma WPCP with injection wells were taken from the Oro Loma *Recycled Water Feasibility Study Final Report* (OLSD, 2016). The Oro Loma study did not include facilities required to connect the extraction wells to the District's distribution system, so a new tank, pump station and connection pipeline have been added to the cost estimate for this master plan.
- Reservoir, Raw Water, and Treated Water Augmentation: These direct alignment lengths were increased by 25 percent to serve as a preliminary, planning-level, pipeline length. For this RWMP Update, these preliminary alignment lengths could be used for high-level cost comparisons between project alternatives without developing in-depth alignments for each project alternative. Any route that crossed San Francisco Bay or a District reservoir was adjusted to avoid those intersections while maintaining the shortest possible route.
- Alternatives using the CCCSD WWTP as a source were an exception as pipeline alignments for these alternatives were taken from CCCSD's *Recycled Water Wholesale Opportunities* report (CCCSD, 2016) and were not increased by 25 percent.

4.6.2 Basis of Pipeline Sizing

Pipelines for each alternative were sized based on the alternative flowrate and a maximum allowable velocity criterion of 5 fps.

4.6.3 Basis of Trenchless Crossing Analysis and Trenchless Pipeline Construction

The number of trenchless crossings required for each project alternative was estimated by using ArcGIS to assess the number of streams, California Highways, railroads, and BART right of ways that were intersected by the preliminary pipeline alignments. In those cases where a pipeline required multiple trenchless crossings within a short distance (approximately 200 feet), it was assumed that these individual crossings could be combined into one larger trenchless crossing. Other redundant crossings, such as multiple creek crossings due to an alignment running roughly parallel to the meandering creek, were removed following individual analysis.

An estimate of the total length of trenchless crossings required by each project alternative was developed by measuring the approximate alignment length within restricted boundaries of streams, California Highways, railroads, and BART right of ways. A minimum trenchless crossing distance of 200 feet was imposed to consider the limitations of current trenchless technology. Measurements were made based on a review of the ArcGIS alignments and available aerial images.



In addition to including trenchless crossings for streams, California Highways, railroads, and BART right of ways, tunneled lengths were estimated for those projects which would either take advantage of the existing San Pablo Tunnel or could not be easily routed through the coastal hills and would require a new tunnel (i.e., to Upper San Leandro Reservoir). For those alternatives which would utilize the San Pablo Tunnel, it was assumed that the full tunnel length (17,600 feet) would be rehabbed. For those alternatives requiring a new tunnel, a length of 7,000 feet was assumed based on the existing tunnel length at San Pablo Reservoir.

4.6.4 Pump Station Sizing

Pump stations for each alternative were sized based on the project flow and estimated head required. Required head was based on the approximate pipeline starting and ending elevations with the addition of an assumed one foot of friction head per 1,000 feet of pipe and a pump efficiency of 75%. Pump configurations included one standby pump and a minimum of two duty pumps. The maximum allowable individual pump size was set at 600 horsepower (hp).

4.7 Potable Reuse Alternatives Summary

Based on the sources, targets, and treatment trains described above, the potable reuse alternatives carried forward for cost and non-cost evaluation are described in **Table 4-11**. The following summary information is provided for each alternative:

- A short name (for use in other tables);
- The source of wastewater;
- The target for advanced treated wastewater;
- The type of potable reuse;
- The maximum production rate, assumed to be constant year-round exempt for the alternative from LAVWMA as a source. This was used to size treatment processes.
- The total production volume per year (yield)
- The pipeline diameter and length required for conveyance from the source to the target;
- The number and length of trenchless crossings and tunnels required for conveyance;
- The required pump station size for conveyance from the source to the target;
- The treatment train configuration (1 through 7, see Figure 4-6 through Figure 4-12);
- The name of the figure(s) in which the alternative is depicted (see Figure 4-13 through Figure 4-30);
- Notes about sizing of facilities required for the alternative
- Sites with limited space available are identified. San Leandro WPCP, Oro Loma WPCP, West County WPCP, and CCCSD WWTP appear to have space available for advanced treatment. Pinole WPCP, Richmond WPCP, and SD-1 are noticeably space-constrained. This information was used to qualitatively evaluate the alternatives (see Table 5-1 for evaluation criteria).

Table 4-11: Potable Reuse Alternatives														
Name	Target	Туре	Produc- tion Rate (MGD)	Yield (AFY)	Pipeline Dia. (in)	Total Pipe Length ¹ (ft)	Tunneled Length	No. of Trenchless Crossings	Length of Trenchless Crossings	Installed Pump hp	Treatment Train Type	Overview Figure	AWT Facilities Size Figure	Notes
San Leandro WP	CP-Based Alternative	es												
SL-Raw-1	USL WTP	Raw	1.4	1,570	10	28,500	0	5	3,350	225	4 (Figure 4-9)	Figure 4-15	Figure 4-16	More water (up to 3.4 MGD) is available if current non- potable system is not utilized.
SL-ResU-1	USL Res.	Reservoir	1.4	1,570	10	27,700	7,000	9	3,200	300	2 (Figure 4-7)	Figure 4-15	Figure 4-16	More water (up to 3.4 MGD) is available if current non- potable system is not utilized.
SL-Chabot-1	Lake Chabot	Reservoir	1.4	780	10	26,400	0	7	2,600	120	2 (Figure 4-7)	Figure 4-15	Figure 4-16	Less over-sized than other Lake Chabot alternatives (based on non-potable demand).
SL-Treat-1	Dunsmuir Res.	Treated	1.4	1,570	10	17,300	0	5	2,420	120	6 (Figure 4-11)	Figure 4-15	Figure 4-16	More water (up to 3.4 MGD) is available if current non- potable system is not utilized.
Pinole WPCP-Bas	sed Alternatives				· · ·							· · ·	· · ·	Pinole WPCP site has noticeable space constraints
Pin-Raw-2	Sobrante WTP	Raw	1.7	1,900	10	29,300	0	3	1,100	180	4 (Figure 4-9)	Figure 4-17	Figure 4-18	Mutually exclusive with Non-Potable Reuse options.
Pin-ResB-2	Briones Res.	Reservoir	1.7	1,900	10	77,800	0	6	1,700	375	2 (Figure 4-7)	Figure 4-17	Figure 4-18	Mutually exclusive with Non-Potable Reuse options.
Pin-ResSP-2	San Pablo Res.	Reservoir	1.7	1,900	10	42,600	0	6	1,700	225	2 (Figure 4-7)	Figure 4-17	Figure 4-18	Mutually exclusive with Non-Potable Reuse options.
Pin-Treat-2	Maloney Res.	Treated	1.7	1,900	10	2,700	0	2	550	225	6 (Figure 4-11)	Figure 4-17	Figure 4-18	Mutually exclusive with Non-Potable Reuse options.
Richmond WPCP-Based Alternatives								· · · · · · · · · · · · · · · · · · ·			Richmond WPCP site has noticeable space constraints			
Rich-Raw-4	Sobrante WTP	Raw	3.6	4,030	16	49,200	0	8	2,050	375	4 (Figure 4-9)	Figure 4-19	Figure 4-20	
Rich-ResB-4	Briones Res.	Reservoir	3.6	4,030	16	65,900	0	8	2,450	900	2 (Figure 4-7)	Figure 4-19	Figure 4-20	
Rich-ResSP-4	San Pablo Res.	Reservoir	3.6	4,030	16	36,000	0	6	2,050	450	2 (Figure 4-7)	Figure 4-19	Figure 4-20	
Rich-Treat-4	Wildcat Aq.	Treated	3.6	4,030	16	1,600	0	0	0	300	6 (Figure 4-11)	Figure 4-19	Figure 4-20	
West County WP	CP-Based Alternative	es		•	· · ·	• •			· · · · ·	•				
WC-GW	Injection Wells	Ground Water	4.7	5,260	20	n/a	n/a	n/a	n/a	n/a	1 (Figure 4-6)	Figure 4-14	Figure 4-22	Injection and extraction wells from Richmond area of East Bay Plan Groundwater Basin. Technical information for sizing wells is not available, so this alternative was not evaluated further.
WC-Raw-5	Sobrante WTP	Raw	4.7	5,260	20	56,500	0	3	3,400	600	4 (Figure 4-9)	Figure 4-21	Figure 4-22	Mutually exclusive with RARE.
WC-ResB-5	Briones Res.	Reservoir	4.7	5,260	20	99,900	17,600	5	2,200	1200	2 (Figure 4-7)	Figure 4-21	Figure 4-22	Mutually exclusive with RARE.
WC-ResSP-5	San Pablo Res.	Reservoir	4.7	5,260	20	50,500	17,600	3	2,000	600	2 (Figure 4-7)	Figure 4-21	Figure 4-22	Mutually exclusive with RARE.
WC-Treat-5	Wildcat Aq.	Treated	4.7	5,260	20	11,300	0	2	1,200	375	6 (Figure 4-11)	Figure 4-21	Figure 4-22	Mutually exclusive with RARE.
Oro Loma WPCP	-Based Alternatives													
Oro-GW	Injection Wells	Ground Water	8	8,060	24	41,200	0	5	1,500		1 (Figure 4-6)	Figure 4-13	Figure 4-24	Injection and extraction wells from South East Bay Plan Groundwater Basin. Similar to 10-MGD "Recommended Project" from Oro Loma Recycled Water Feasibility Study.
Oro-Raw-8	USL WTP	Raw	8	8,960	24	40,400	0	7	3,750	1200	4 (Figure 4-9)	Figure 4-23	Figure 4-24	
Oro-ResU-8	USL Res.	Reservoir	8	8,960	24	57,800	7,000	9	3,450	1500	2 (Figure 4-7)	Figure 4-23	Figure 4-24	
Oro-Chabot-8	Lake Chabot	Reservoir	8	8,960	24	27,500	0	8	2,300	750	2 (Figure 4-7)	Figure 4-23	Figure 4-24	
Oro-Treat-8	South Reservoir	Treated	8	8,960	24	10,200	0	1	300	750	6 (Figure 4-11)	Figure 4-23	Figure 4-24	



		•	Dreduc		Dineline	Total Dina		No. of	I enote of	Installed			•	
			tion Rate	Yield	Dia.	Length ¹	Tunneled	Trenchless	Trenchless	Pump	Treatment Train	Overview	AWT Facilities	
Name	Target	Туре	(MGD)	(AFY)	(in)	(ft)	Length	Crossings	Crossings	hp	Туре	Figure	Size Figure	Notes
CCCSD WWTP-Ba	sed Alternatives													
CC-Raw-19	Mokelumne Aq.	Raw	19	21,280	36	18,500	0	3	2,500	2400	4 (Figure 4-9)	Figure 4-25	Figure 4-26	HGL for Mokelumne Aqueduct was estimated as 400 feet.
CC-Raw-10	Mokelumne Aq.	Raw	10	11,200	24	18,500	0	3	2,500	1500	4 (Figure 4-9)	Figure 4-25	Figure 4-26	HGL for Mokelumne Aqueduct was estimated as 400 feet.
CC-ResB-19	Briones Res.	Reservoir	19	21,280	36	54,800	0	8	3,300	3600	2 (Figure 4-7)	Figure 4-25	Figure 4-26	
CC-ResB-10	Briones Res.	Reservoir	10	11,200	24	54,800	0	8	3,300	2000	2 (Figure 4-7)	Figure 4-25	Figure 4-26	
EBMUD Main WW	TP (SD-1)-Based Al	ternatives												SD-1 site has noticeable space constraints
SD1-Raw-30	Orinda WTP	Raw	30	33,600	42	58,900	17,600	4	2,200	3600	4 (Figure 4-9)	Figure 4-27	Figure 4-28	At 30 MGD, assume the entire plant is upgraded for nutrient removal.
SD1-Raw-10	Orinda WTP	Raw	10	11,200	24	58,900	17,600	2	2,200	1500	5 (Figure 4-10	Figure 4-27	Figure 4-28	Assumes 10 MGD satellite treatment
SD1-ResU-30	USL Res.	Reservoir	30	33,600	42	48,400	7,000	7	3,550	4200	2 (Figure 4-7)	Figure 4-27	Figure 4-28	At 30 MGD, assumes the entire plant is upgraded for nutrient removal.
SD1-ResB-30	Briones Res.	Reservoir	30	33,600	42	67,800	17,600	4	2,400	5400	2 (Figure 4-7)	Figure 4-27	Figure 4-28	At 30 MGD, assumes the entire plant is upgraded for nutrient removal.
SD1-ResSP-4	San Pablo Res.	Reservoir	4	4,480	16	35,400	17,600	6	2,400	600	3 (Figure 4-8)	Figure 4-27	Figure 4-28	At 4 MGD, assumes satellite treatment, possibly pulling preferentially from the low-salinity Adeline Interceptor (4- 5 MGD available).
SD1-ResU-10	USL Res.	Reservoir	10	11,200	24	59,400	0	7	3,550	1800	3 (Figure 4-8)	Figure 4-27	Figure 4-28	Assumes 10 MGD satellite treatment
SD1-ResB-10	Briones Res.	Reservoir	10	11,200	24	67,800	17,600	1	2,400	2000	3 (Figure 4-8)	Figure 4-27	Figure 4-28	Assumes 10 MGD satellite treatment
SD1-Treat-30	Claremont Center	Treated	30	33,600	42	6,300	0	3	1,500	3000	6 (Figure 4-11)	Figure 4-27	Figure 4-28	At 30 MGD, assumes the entire plant is upgraded for nutrient removal.
SD1-Treat-10	Claremont Center	Treated	10	11,200	24	6,300	0	1	1,500	1200	7 (Figure 4-12)	Figure 4-27	Figure 4-28	Assumes 10 MGD satellite treatment
Satellite Treatmen	t Alternatives													
LAVWMA Castro Valley (LA- Chabot-1)	Lake Chabot	Reservoir	10	4,480	24	13,100	0	0	0	225	2 (Figure 4-7)	Figure 4-29	-	Requires both the Livermore and DSRSD plants to be upgraded for nitrogen removal.
Satellite – Pt. Isabel (Sat- ResSP-4)	San Pablo Res.	Reservoir	4	4,480	16	13,700	17,600	5	1,700	600	2 (Figure 4-7)	Figure 4-30	Figure 4-31	This alternative assumes the feed pipeline from San Pablo Reservoir to San Pablo WTP can be re-purposed for advanced treated water.

1. Pipeline length is further categorized by land use type for cost estimating purposes as discussed in Section 2.3.2







Figure 4-13: Potable Reuse Alternative - Oro Loma Groundwater Augmentation





Figure 4-14: Potable Reuse Alternative – Richmond Groundwater Augmentation





Figure 4-15: Potable Reuse Alternatives - San Leandro WPCP as a Source





Figure 4-16: Advanced Treatment Footprint at San Leandro WPCP











Figure 4-18: Advanced Treatment Footprint at Pinole WPCP





Figure 4-19: Potable Reuse Alternatives - Richmond WPCP as a Source





Figure 4-20: Advanced Treatment Footprint at Richmond WPCP





Figure 4-21: Potable Reuse Alternatives - West County WPCP as a Source





Figure 4-22: Advanced Treatment Footprint at West County WWTP





Figure 4-23: Potable Reuse Alternatives - Oro Loma WPCP as a Source





Figure 4-24: Advanced Treatment Footprint at Oro Loma WPCP





Figure 4-25: Potable Reuse Alternatives - CCCSD WWTP as a Source





Figure 4-26: Advanced Treatment Footprint at Central San WWTP





Figure 4-27: Potable Reuse Alternatives - SD-1 as a Source





Figure 4-28: Advanced Treatment Footprint at SD-1











Figure 4-30: Potable Reuse Alternative – Pt. Isabel Satellite WWTP as a Source





Figure 4-31: Advanced Treatment Footprint at Pt. Isabel

4.8 Cost Evaluation for Potable Reuse Alternatives

Capital and operating costs for each potable alternative were developed based on the methodology presented in Section 2. A summary of project capital, operating and unit costs are presented in **Table 4-12**. More detailed cost estimates for each project can be found in Appendix B.



	Viold	Capital	Annual	Treatment	Conveyance	Total Unit	Dry Year
Name	(AFY)	(\$M)	(\$M/yr.)	(\$/AF)	(\$/AF)	(\$/AF)	(\$/AF)
Oro-GW	8,060	250	15	2,100	1,200	3,300	9,900
SL-Raw-1	1,570	59	3	2,200	1,400	3,600	11,000
SL-ResU-1	1,570	82	3	1,900	2,400	4,300	13,000
SL-Chabot-1	780	44	3	3,800	1,800	5,600	18,000
SL-Treat-1	1,570	51	4	2,700	1,200	3,900	11,000
Pin-Raw-2	1,900	53	4	2,100	1,000	3,100	8,900
Pin-ResB-2	1,900	64	3	1,900	1,400	3,300	10,000
Pin-ResSP-2	1,900	49	3	1,900	900	2,800	8,400
Pin-Treat-2	1,900	40	4	2,600	600	3,200	8,400
Rich-Raw-4	4,030	110	7	2,000	1,000	3,000	8,300
Rich-ResB-4	4,030	110	7	1,900	1,100	3,000	9,100
Rich-ResSP-4	4,030	88	6	2,000	500	2,500	7,700
Rich-Treat-4	4,030	65	7	2,400	100	2,500	7,300
WC-Raw-5	5,260	160	9	1,900	1,100	3,000	8,900
WC-ResB-5	5,260	260	10	1,900	2,100	4,000	13,000
WC-ResSP-5	5,260	220	10	1,900	1,700	3,600	11,000
WC-Treat-5	5,260	98	9	2,300	400	2,700	7,600
Oro-Raw-8	8,960	200	14	1,900	700	2,600	7,400
Oro-ResU-8	8,960	230	14	1,900	900	2,800	8,600
Oro-Chabot-8	8,960	160	13	1,900	400	2,300	6,700
Oro-Treat-8	8,960	160	15	2,200	400	2,600	7,400
CC-Raw-19	21,280	310	31	1,900	300	2,200	6,300
CC-Raw-10	11,200	180	17	2,000	300	2,300	6,500
CC-ResB-19	21,280	380	33	1,900	500	2,400	6,900
CC-ResB-10	11,200	220	17	1,900	500	2,400	7,100
SD1-Raw-30	33,600	650	50	1,900	500	2,400	7,000
SD1-Raw-10	11,200	510	20	2,900	1,000	3,900	12,000
SD1-ResU-30	33,600	570	49	1,800	500	2,300	6,700
SD1-ResB-30	33,600	690	53	1,900	600	2,500	7,300
SD1-ResSP-4	4,480	300	9	3,600	1,500	5,100	16,000
SD1-ResU-10	11,200	430	18	2,800	700	3,500	10,000
SD1-ResB-10	11,200	510	20	2,900	1,000	3,900	12,000
SD1-Treat-30	33,600	480	54	2,100	200	2,300	6,700
SD1-Treat-10	11,200	360	20	3,200	200	3,400	9,800
LA-Chabot-10	4,480	130	12	3,800	200	4,000	12,000
Sat-ResSP-4	4,480	170	8	1,900	1,400	3,300	10,000

Table 4-12: Potable Reuse Alternatives Cost Summary



5. EVALUATION OF ALTERNATIVES

5.1 Non-Cost Evaluation

In addition to developing cost estimates for each alternative, a non-cost evaluation was conducted to capture the environmental and social objectives and the complexity and risk of the alternatives. The evaluation criteria are detailed in **Table 5-1** as well as the criteria weights developed by District staff.

Criteria	Description	Weight									
	Environmental and Social Objectives										
Environmental justice	Assessment of what regions/populations are served/impacted by this new supply and how the water quality of different regions/populations impacted by this new supply.	20%									
Environmental impacts from construction	Assessment of the potential environmental challenges during construction of the alternative and the mitigation that may be necessary for any impacts.	5%									
Energy use	Assessment of the energy usage during operations (GHG impacts).	10%									
Wastewater discharge	Assessment of reduced nutrient discharges	5%									
Complexity and Risk											
Institutional	Assessment of the time, challenges and requirements to implement the project either internally or in coordination with external partners.	15%									
Regulatory	Assessment of the time, challenges and requirements to implement the project from a planning and permitting perspective prior to construction and ongoing as part of operations.	15%									
Construction	Assessment of the time, challenges and requirements to design and construct the project.	5%									
Operational	Assessment of the impacts of the alternative on existing operation of the District's water and wastewater systems. Assessment of the complexity of the alternative and how challenging it will be for District staff to manage any new processes or operations.	25%									
	Total	100%									

5.1.1 Evaluation of Non-Potable Reuse Alternatives

This section summarizes the results of the non-cost evaluation for non-potable reuse alternatives using the criteria defined above. Scoring rubrics for the non-cost criteria are presented in **Table 5-2**. Proposed recycled water projects were scored 1 to 5 based on how effectively each project met the evaluation criteria. A high score indicated high response to the criteria and a low score indicated a low response to the criteria (5 = Most Favorable, 1 = Least Favorable). The rubric includes a brief description of the metrics used to score each alternative, differentiated for non-potable and potable projects. A final alternative score was then calculated using the scores for each criterion, combined with the criteria weights. **Table 5-3** summarizes the results of the ranking process for non-potable reuse alternatives.

This page is intentionally left blank.

Table 5-2: Non-Potable Reuse Scoring Rubric

Cuitouia	(most favorable)		Scoring Rubric		(least favorable)
Criteria	5	4	3	2	1
Environmental and Social Objectives					
Environmental justice	Project benefits many different types of customers and/or the potable water use offset provides increased reliability for a significant portion of the District service area.		Project benefits some of customers and/or the potable water use offset provides increased reliability for some of the District service area.		Project does not significantly impact the supply reliability for the District aside from 1-2 (likely private) customers.
Environmental impacts from construction	Project construction will have limited environmental impacts (few, if any, stretches of alignment near streams, wetlands or other habitat).		Project construction will have some environmental impacts (may include some stretches of alignment near streams, wetlands or other habitat).		Project construction will have significant environmental impacts (may include significant stretches of alignment near streams, wetlands or other habitat).
Energy use	Project operation will require low energy for treatment and conveyance.		Project operation will require "average" energy for treatment and conveyance.		Project operation will require significant energy for treatment and conveyance.
Wastewater discharge	Project provides denitrification or serves a large irrigation customer (>2 MGD)		Project provides partial denitrification or serves an irrigation customer		Project provides no denitrification and does not serve an irrigation customer.
Complexity and Risk					
Institutional Complexity	Project serves only District facilities or 1- 2 customers.		Project may serve DISTRICT facilities and 3-5 customers.		Project may serve DISTRICT facilities and 5+ customers.
Regulatory Complexity	Project requires limited number of permits, easements, documentation, etc., resulting in less effort to coordinate with state agencies and local stakeholders and minimal required annual monitoring/permitting.		Project requires some permits, easements, documentation, etc., resulting in some effort to coordinate with state agencies and local stakeholders and minimal required annual monitoring/permitting.		Project requires many permits, easements, documentation, etc., resulting in significant effort to coordinate with state agencies and local stakeholders and significant required annual monitoring/permitting.
Construction Risk	Project includes limited number of unique facilities, facility siting concerns, or other special circumstances such as trenchless crossings.		Project includes some unique facilities, facility siting concerns, or other special circumstances such as trenchless crossings.		Project includes many unique facilities, facility siting concerns, or other special circumstances such as trenchless crossings.
Operational Complexity	Project will require limited change to existing District operations (changes are limited to the expansion of an existing treatment facility).		Project will require limited change to existing District operations (changes are limited to some changes to a treatment facility and/or distribution).		Project will require the operation of a new, independent facility and conveyance system.



	_			Crit	eria					
	Enviro	nmental and	Social Ob	ojectives		Complexit	y and Risk			
Project	Environmental Justice	Environmental Impacts from Construction	Energy Use	Wastewater Discharge (Nutrients)	Institutional	Regulatory	Construction	Operational Complexity	Normalized Score (out of 100)	Notes
Criteria weights	20%	5%	10%	5%	15%	15%	5%	25%		
DERWA/San Ramon Phase 3	4	5	5	3	4	5	4	5	90	Expansion of committed project in progress/construction, with completion planned FY24-25. Established District partnership with DSRSD. Program is EIR certified
DERWA/San Ramon Phase 5	4	5	5	3	4	5	5	5	91	Expansion of committed project with implementation planned FY 28-29. Established District partnership with DSRSD. Potential supply limitations.
DERWA/San Ramon Phase 4	4	5	5	3	4	5	4	5	90	Expansion of committed project with implementation planned FY 34-35. Established District partnership with DSRSD. Phase 4. EIR certified. Potential supply limitations.
East Bayshore. Phase 1A	5	4	5	4	5	4	4	4	89	Expansion of committed project within existing distribution system and Frontage Road pipeline alignment for irrigation and industrial demands. Adequate supply availability. Requires treatment upgrades to meet water quality objectives and minor expansion of distribution system.
East Bayshore. Phase 1B	5	4	4	4	5	4	3	3	81	Expansion of committed project for irrigation and industrial demands. Adequate supply availability. Requires treatment upgrades and expansion of distribution system to Oakland and Berkeley.
East Bayshore. Phase 2	5	3	3	4	5	4	2	2	72	Expansion of committed project for irrigation and industrial demands. Adequate supply availability, but limited room for expansion within existing site. Requires treatment upgrades and expansion of the distribution system, including service to UCB and Alameda.
Chevron/Richmond	3	4	2	1	3	3	3	4	62	Expansion of committed project. Customer is established District project partner. Requires partnership with City of Richmond. Requires treatment upgrades at City of Richmond and distribution system.
P66 Rodeo Refinery	3	4	4	1	3	3	3	4	66	MOU between P66 and EBMUD. Project in planning phase, technically feasible but with supply limitations. District is exploring funding options.

Table 5-3: Non-Potable Reuse Alternatives Non-Cost Evaluation



				Crit	eria					
	Enviro	nmental and	l Social Ob	iectives		Complexit	v and Risk			
				<u>д</u> роц. тос				-		
Project	Environmental Justice	Environmental Impact from Construction	Energy Use	Wastewater Discharge (Nutrients)	Institutional	Regulatory	Construction	Operational Complexity	Normalized Score (out of 100)	Notes
Criteria weights	20%	5%	10%	5%	15%	15%	5%	25%		
Central San Regional	4	4	1	1	1	3	3	3	51	Requires agreements w/CCCSD and CCWD. May require CCWD approval from USBR for wheeling. May be able to complete under Cat Ex for CEQA. Assume District would accept water from CCWD at the existing District-CCWD intertie. Requires new high-lift pump station/land purchase and installation of VFDs at WC pumping plant. Assume WTP expansion would address WQ delta (i.e., higher salt content)
CC Pipeline in Canal ROW	2	4	3	3	2	3	4	4	60	Total water sales = 0.83 MGD, but only two District customers. Limited benefits to District. Agreements between CCWD, CCCSD and the CNWS developer. Impacts on existing operations on District minimal, but improvement on CCWD's operations. Project phases other potential impediments.
DCC Satellite Project	2	4	5	3	3	5	5	4	74	RFP has been issued/re-issued. Most likely to move forward among satellite projects. Wastewater supply availability. Self-financing model.
MCC Satellite Project	2	4	5	3	3	4	4	4	68	Project in planning phase. MOU between EBMUD, MCC and CCCSD. Wastewater supply availability. Funding issues. MCC to self-finance/construct/operate satellite facility.
Moraga Area Satellite Project	3	3	5	3	2	2	3	3	56	Conceptual phase. Institutional challenging model (various customers). Wastewater supply availability. Moraga customers to self-finance/operate satellite facility.
Oakland Hills Satellite Project	2	3	5	3	2	2	3	3	54	Conceptual phase. Institutional challenging model (various customers). Wastewater supply availability. Oakland Hill customers to self-finance/operate satellite facility.
UCB Satellite Project	3	2	4	4	3	3	2	4	66	Project in conceptual phase. Previous studies (2005) performed. Site constraints. Difficulty in finding a suitable site for satellite facility. UCB to self-finance/construct/operate satellite facility.



This page is intentionally left blank.



 Table 5-4 provides a summary of the non-cost scoring in ranked order. The normalized non-cost evaluation scores were plotted against the project unit costs to determine the highest ranking, lowest cost alternatives as presented in Figure 5-1.

Project	Normalized Score	Project Unit Cost (\$/AF)	Dry Year Unit Cost (\$/AF)
DERWA/San Ramon Phase 5	91	1,600	5,400
DERWA/San Ramon Phase 3	90	1,900	6,300
DERWA/San Ramon Phase 4	90	2,900	9,700
East Bayshore. Phase 1A	89	2,400	7,400
East Bayshore. Phase 1B	81	2,400	7,800
DCC Satellite Project	74	3,900	12,000
East Bayshore. Phase 2	72	3,000	9,400
MCC Satellite Project	68	4,600	15,000
P66 Rodeo Refinery	66	1,100	3,700
Chevron/Richmond	62	2,600	8,600
UCB Satellite Project	66	2,900	8,800
CC Pipeline in Canal ROW	60	2,000	6,800
Moraga Area Satellite Project	56	5,700	18,000
Oakland Hills Satellite Project	54	3,200	11,000
Central San Regional	51	1,100	3,400

Table 5-4: Ranked Scores for Non-Potable Reuse Alternatives

Note: For DERWA/San Ramon Valley Recycled Water Project Phases, unit costs shown are based on District's share for capital and O&M costs.

Committed centralized projects, such as DERWA/San Ramon Valley Recycled Water Projects, ranked the highest. Committed centralized projects are expansion of committed projects in progress/construction, have established partnerships, and the programs are EIR certified. It is important to note that wastewater supply from DSRSD is limited and the DERWA supply agreement currently defines recycled water service to the two-member agencies as first-come, first-served. This means that the availability of water is dependent upon how quickly the District expands its distribution system versus how quickly DSRSD develops its program. An important focus for the DERWA program in the next few years is to secure supplemental supplies.

Satellite recycled water treatment plant projects can cost-effectively serve large water users that are located far from a centralized treatment facility. Satellite treatment projects with agreements in place, technically feasible, and with funding support are rated the highest. Among the satellite treatment recycled water projects, DCC Satellite Project is the most likely to move forward. DCC is pursuing a self-financing model. Their studies have shown that the satellite project would pay for itself, while eliminating the risks associated with drought restrictions.





Figure 5-1: Non-Potable Reuse Alternative Comparison

Note: For DERWA/San Ramon Valley Recycled Water Project Phases, unit costs shown are based on District's share for capital, average O&M and average energy costs.

 \Box = Satellite RWTP, Δ = East Bayshore RWF, \circ = DERWA/San Ramon, \diamond = Refinery, x = CCCSD and CC Pipeline in Canal ROW.


5.1.2 Evaluation of Potable Reuse Alternatives

Using the scoring rubric presented in **Table 5-5**, the potable reuse alternatives were also scored for each non-cost criteria on a scale of 1 to 5. The criteria scores were combined with the criteria weights to calculate the overall project score (out of 100). A summary of the scores for each alternative and supporting notes on the scoring and any deviation from the rubric is presented in **Table 5-6**.

The non-cost evaluation for potable reuse alternatives does not explicitly include uncertainty associated with the timeline for adoption of state regulations governing raw water augmentation (expected in 2023) and treated drinking water augmentation (no timeline exists). There are currently no regulations for these types of potable reuse, but if the best alternatives are raw water and/or treated water augmentation, then it may be in the District's best interest to wait for regulations to come into effect before implementing potable reuse. Also, the planning horizon for this RWMP Update extends further than the expected timeline for developing these regulations – the District's current recycled water goal is based on a compliance date of 2040. Assigning low scores to these alternatives would remove them from consideration too early in the evaluation. Instead, these factors will be considered as part of the overall phasing for implementation.

Vulnerability of wastewater treatment facilities to sea level rise is also not explicitly included in the non-cost evaluation. All of the alternatives have vulnerability to sea level rise, with the exception of satellite treatment of LAVWMA effluent (LA-Chabot-10). The recently completed BACWA Nutrient Reduction Study (BACWA, 2018) concluded that SD-1 and Richmond WPCP would not be impacted by sea level rise within the next 50 years, although low-lying areas of the collection systems could be vulnerable. The other four wastewater sources considered in this evaluation (San Leandro WPCP, Pinole WPCP, West County WPCP, CCCSD WWTP) are already in the 100-year floodplain, and are vulnerable to sea level rise and other flooding conditions. This page is intentionally left blank.



Table 5-5: Potable Reuse Scoring Rubric

Oritoria	(most favorable)		Scoring Rubric		(least favorable)
Criteria	5	4	3	2	1
Environmental and Social Objectives					
Environmental justice	Project benefits entire service area.		Project benefits around half of the District service area.	Project benefits around 25% of the District service area.	Project benefits a small area (pressure zone) of the District service area.
Environmental impacts from construction	Pipeline alignment does not include any stream crossings.	Pipeline alignment includes 1 stream crossing.	Pipeline alignment includes 2 or 3 stream crossings.	Pipeline alignment includes 4 stream crossings.	Pipeline alignment includes 5+ stream crossings.
Energy use	Alternative requires <200,00 kWh/MGD/year for pumping.	Alternative requires 200,000-400,000 kWh/MGD/year for pumping.	Alternative requires 400,000-600,000 kWh/MGD/year for pumping.	Alternative requires 600,000-800,000 kWh/MGD/year for pumping.	Alternative requires >800,000 kWh/MGD/year for pumping.
Wastewater discharge	WW treatment train includes denitrification.		WW treatment train includes denitrification, but volume of reuse could impact R2 program capability (SD1 only).		
Complexity and Risk					
Institutional Complexity	Project does not require any coordination with partner agencies.	Project does not require any coordination with partner agencies but requires extensive internal coordination between departments.	Project requires coordination with 1 external partner agency.	Project requires coordination with 2-3 external partner agencies.	Project requires coordination with 4+ external partner agencies.
Regulatory Complexity	Project does not require new WDR or NPDES (treated or raw water augmentation) and has sufficient dilution flows for RO concentrate management	Project requires new WDR (groundwater augmentation)	Project requires new NPDES (reservoir augmentation) or has limited ability for RO concentrate management		Project requires new NPDES and has limited ability for RO concentrate management
Construction Risk	WWTP source has ample space for advanced treatment processes.	WWTP source has ample space for advanced treatment processes but tunneling to potable reuse target is required.	WWTP source has some space available for advanced treatment.	WWTP source has some space available for advanced treatment but tunneling to potable reuse target is required. Or WWTP source has no space currently available for advanced treatment processes.	WWTP source has no space currently available for advanced treatment processes and tunneling to potable reuse target is required. Or no satellite site identified for advanced treatment.
Operational Complexity	No impacts on existing operations	Minimal impacts to existing water operations (groundwater and reservoir augmentation utilize existing raw water sources)	Minimal impacts to existing water operations and significant impacts to hydraulics or requires MBR upgrade for secondary treatment.	Significant impacts to existing water operations (raw water or treated water augmentation introduces new water source)	Significant impacts to existing water operations and significant impacts to hydraulics or requires MBR upgrade for secondary treatment.





	Criteria										
			Envi	ronment	al and S	ocial	<u> </u>	malavit	v and Di	ak	
			()	თ	lives			mpiexit	y and Ri	SK 20	
Project	Source	Target	Environmental Justic	Environmental Impact from Construction	Energy Use	Wastewater Discharg (Nutrients)	Institutional	Regulatory	Construction	Operational Complexit	Normalized Score (out of 100)
110j000		Criteria weights	20%	5%	10%	5%	15%	15%	5%	25%	
Oro-GW	Oro Loma WPCP	Injection Wells	2	4	2	5	2	4	4	3	58
SL-Raw-1	San Leandro WPCP	USL WTP	2	4	2	5	3	5	3	2	58
SL-ResU-1	San Leandro WPCP	USL Reservoir	2	3	2	5	3	3	2	4	60
SL-Chabot-1	San Leandro WPCP	Lake Chabot	1	4	4	5	2	3	3	3	54
SL-Treat-1	San Leandro WPCP	Dunsmuir Reservoir	1	5	4	5	3	5	3	2	59
Pin-Raw-2	Pinole WPCP	Sobrante WTP	2	5	3	5	3	3	2	2	54
Pin-ResB-2	Pinole WPCP	Briones Reservoir	3	3	1	5	3	1	2	4	56
Pin-ResSP-2	Pinole WPCP	San Pablo Reservoir.	2	3	3	5	3	1	2	4	56
Pin-Treat-2	Pinole WPCP	Maloney Reservoir	1	5	3	5	3	3	2	2	50
Rich-Raw-4	Richmond WPCP	Sobrante WTP	2	3	3	5	3	3	2	2	52
Rich-ResB-4	Richmond WPCP	Briones Reservoir	3	3	1	5	3	1	2	4	56
Rich-ResSP-4	Richmond WPCP	San Pablo Reservoir	2	4	3	5	3	1	2	4	57
Rich-Treat-4	Richmond WPCP	Wildcat Aqueduct	2	5	4	5	3	3	2	2	56
WC-Raw-5	West County WPCP	Sobrante WTP	2	4	3	5	3	3	3	2	54
WC-ResB-5	West County WPCP	Briones Reservoir	3	3	1	5	3	1	3	4	57
WC-ResSP-5	West County WPCP	San Pablo Reservoir	2	4	3	5	3	1	3	4	58
WC-Treat-5	West County WPCP	Wildcat Aqueduct	2	4	4	5	3	3	3	2	56
Oro-Raw-8	Oro Loma WPCP	USL WTP	2	3	2	5	3	5	5	2	59
Oro-ResU-8	Oro Loma WPCP	USL Reservoir	2	3	2	5	3	3	4	4	62
Oro-Chabot-8	Oro Loma WPCP	Lake Chabot	1	4	4	5	2	3	5	3	56
Oro-Treat-8	Oro Loma WPCP	South Reservoir	1	5	4	5	3	5	5	1	56
CC-Raw-19	CCCSD WWTP	Mokelumne Aqueduct	5	3	2	5	3	3	5	2	65
CC-Raw-10	CCCSD WWTP	Mokelumne Aqueduct	5	3	2	5	3	5	5	2	71
CC-ResB-19	CCCSD WWTP	Briones Reservoir	3	1	1	5	3	1	5	2	47
CC-ResB-10	CCCSD WWTP	Briones Reservoir	3	1	1	5	3	3	5	2	53
SD1-Raw-30	SD1 WWTP	Orinda WTP	2	5	2	3	5	3	2	2	56
SD1-Raw-10	SD1 WWTP	Orinda WTP	2	5	2	5	5	5	2	1	59
SD1-ResU-30	SD1 WWTP	USL Reservoir	2	3	2	3	5	1	1	4	57
SD1-ResB-30	SD1 WWTP	Briones Reservoir	3	4	1	3	5	1	2	4	61
SD1-ResSP-4	SD1 WWTP	San Pablo Reservoir	2	4	3	5	5	3	2	4	69
SD1-ResU-10	SD1 WWTP	USL Reservoir	2	3	2	5	5	3	1	3	60
SD1-ResB-10	SD1 WWTP	Briones Reservoir	3	5	1	5	5	3	2	3	65
SD1-Treat-30	SD1 WWTP	Claremont Center	3	5	3	3	5	3	2	2	62
SD1-Treat-10	SD1 WWTP	Claremont Center	3	5	3	5	5	5	2	1	65
LA-Chabot-10	LAVWMA Castro Valley	Lake Chabot	1	5	5	5	2	3	1	2	50
Sat-ResSP-4	Satellite - Pt. Isabel	San Pablo Reservoir	2	5	3	5	4	1	5	3	59

Table 5-6: Potable Reuse Alternatives Non-Cost Evaluation



The normalized non-cost evaluation scores were plotted against the project unit costs to determine the highest ranking, lowest cost alternatives as presented in Figure 5-2. Due to the large number of project alternatives evaluated, only those above the median score of 57 are shown and considered as the most favorable projects for further evaluation.





Note: Marker symbology was developed to group alternatives by similar characteristics – markers are colored based on water source, sized based on relative project yield and shaped based on project type.

 \Box = groundwater augmentation, Δ = reservoir water augmentation, \circ = raw water augmentation, \diamond = treated water augmentation.



5.2 Economic Value of Recycled Water Supply

To determine a net benefit for each recycled water project, the project's cost was compared to the avoided cost, alternative supply cost, and the willingness-to-pay for recycled water. All three approaches for valuing new increments of water supply are recognized by the California Water Commission (CWC 2016) and accord with generally accepted economic principles for the valuation of water supply (Young 2005). The following presents an overview of the economic analysis of the value of recycled water for the District.

5.2.1 Avoided Cost

Avoided cost is the reduction in without-project cost that would occur as a result of a proposed project. Avoided cost can be treated as a negative cost in the calculation of a project's net unit supply cost. For example, if the unit cost of a proposed recycled water project is \$2,000/AF and implementing the project would allow the District to avoid wastewater discharge costs equal to \$400/AF for each AF of recycled water produced, the project's net unit supply cost is \$1,600/AF. For this masterplan, the avoided costs of discharge pumping or permit savings for lower volumes discharged are assumed to be de minimis. As discussed in Section 4, it is assumed that nutrient removal would be required regardless if the effluent is discharged as secondary effluent or put to beneficial reuse to meet future discharge requirements. Project costs for nutrient removal have been developed through the BACWA Nutrient Reduction Study under separate cover (BACWA, 2018). Therefore, rather than including costs for nutrient removal within the potable reuse projects and then subtracting them as avoided costs for continued discharge, the costs for nutrient removal upgrades were omitted.

5.2.2 Alternative Supply Cost

Alternative supply cost is the cost of the least-cost means of providing at least the same amount of physical water supply benefit. Alternative supply cost can vary with water year type and this needs to be considered when valuing the water from a recycled water project. For example, in years with normal water supply, a recycled water project may simply displace the use of Mokelumne River water, in which case the alternative supply cost is the incremental cost of acquiring, conveying, treating, and distributing an acre-foot of Mokelumne River water. In dry or critically dry years, on the other hand, the Mokelumne River may be in deficit and the marginal source of supply may be CVP or purchased water delivered through the Freeport Facility. In years when rationing is required, the marginal source of supply is generally going to be water saved via conservation projects. Because the cost of each of these supplies is significantly different, it is necessary to construct a weighted average cost reflecting the annual frequency in which each supply represents the alternative supply cost.

Table 5-7 sets out current alternative supply cost estimates for the District based on four marginal supply sources considered: (1) Mokelumne River water, (2) CVP contract supply, (3) dry-year purchases, and (4) average willingness-to-pay for additional water during years in which water demand is rationed. The last column of **Table 5-7** indicates the annual frequency that each source is expected to be the marginal supply. For example, while the District expects to deliver Mokelumne River water every year, Mokelumne River water is expected to be the marginal supply 78% of the time. The other 32% of the time, one of the other supplies is expected to be the marginal supply. In years with rationing, the relevant cost is what customers would be willing to pay for additional water if it could be made available.

The frequency with which a supply is expected to be the marginal source of water is derived from the W-E model simulation results for WSMP 2040 Portfolio E3 for the 2010-2040 planning horizon. The W-E model simulates the District's supplies, demands, and rationing levels through 2040 using historical hydrology spanning the years 1921 to 2003. The W-E model and WSMP 2040 portfolios are described in Appendix D of the WSMP 2040. The marginal cost of each supply is based on cost data compiled from District sources as described in the notes in **Table 5-7**. The rationing willingness-to-pay estimate is derived in the next section of this memorandum.



The alternative supply cost for new baseload supply operated continuously is the weighted-average of the four marginal supply sources. This is \$681/AF and represents the cost per acre-foot that District customers would avoid by implementing a new recycled water project that operated continuously. The alternative supply cost for new dry-year supply is the weighted-average alternative supply cost in years the District expects to operate the Freeport facility or implement demand rationing. This cost is \$1,774/AF and represents the cost per AF that District customers would avoid by implementing a new recycled water project operated in dry years only.

Marginal Supply Source	Alternative supply cost (\$/AF)	Annual Frequency ¹
Mokelumne River	\$166 ²	68%
Freeport + CVP contract supply	\$553 ³	19%
Dry-year purchases	\$1,215 ⁴	4%
Rationing Willingness-to-Pay	\$4,600 ⁵	9%
Alternative supply cost of Supply for Baseload Water	\$681 ⁶	
Alternative supply cost of Supply for Dry Year Water	\$1,774 ⁷	

Table 5-7: EBMUD Weighted-Average Alternative Water Supply Cost Estimate (2017 dollars)

Notes:

- 1. Annual frequencies based on W-E model simulation results for WSMP 2040 Portfolio E3 for 2025-2040 levels of demand.
- 2. Source of estimate is Chapter 5 of the Water Supply and Economic Modeling Report, Appendix D of WSMP 2040. The cost is comprised of: 1) Mokelumne system raw water O&M cost, 2) variable treatment cost, and 3) variable distribution cost. Pumping is required under some conditions to move raw water through the Mokelumne Aqueducts to the terminal reservoir system. WSMP 2040 assumed an average of 720 kwh/AF of water pumped through the aqueducts at a cost of \$0.115/kwh (2008 \$), or approximately \$83/AF. WSMP 2040 also assumed that water originating from the Mokelumne River is clean enough to require in-line treatment only. In-line treatment was assumed to require 25 kwh/AF of energy, \$9/AF for chemicals, and \$1/AF for disposal, for a total cost of approximately \$13/AF. WSMP 2040 assumed an average distribution energy requirement of 265 kwh/AF at a cost of \$0.153/kwh, for a total cost of approximately \$41/AF. The cost in 2008 dollars is \$137/AF. Updating to 2017 dollars, the total cost is \$166/AF.
- 3. Source of estimate is Freeport Project operation cost summaries prepared by District staff for 2014 and 2015. The cost is comprised of 1) CVP purchase cost of \$75/AF, 2) Freeport transmission costs of \$225/AF, and 3) water treatment costs of \$175/AF. Variable distribution costs are assumed to be the same as for Mokelumne River water. The cost in 2015 dollars is \$522/AF. Updating to 2017 dollars, the total cost is \$553/AF.
- 4. Source of estimate is dry-year purchase cost summaries prepared by District staff for 2015. The cost is comprised of 1) marginal purchase cost of \$700/AF for 13 TAF purchased from RD 1004 and Sycamore Mutual Water Company, 2) Freeport transmission costs of \$225/AF, and 3) water treatment costs of \$175/AF. Variable distribution costs are assumed to be the same as for Mokelumne River water. The total cost in 2015 dollars is \$1,147/AF. Updating to 2017 dollars, the estimated cost is \$1,215/AF.
- 5. Source of estimate is average willingness-to-pay in years with rationing >=10% shown in Table 4 below.
- 6. Weighted-average cost across all years.
- 7. Weighted-average cost for years with CVP contract, dry-year purchases or rationing.

5.2.3 Marginal Willingness-to-Pay

Willingness-to-pay is the dollar amount that water users would be willing to pay for the physical water supply benefit. In years when demand is unconstrained, and customers can freely purchase whatever amount of water they want, the willingness-to-pay for water at the margin is the same as the commodity charge, which currently is \$2,113/AF for potable water and \$1,647/AF for non-potable water. However, in years when rationing is required, willingness-to-pay can be several times larger. For example, estimates of willingness-to-pay for recycled water exceed \$4,500/AF at rationing levels above 20%. As with alternative supply cost, it is therefore necessary to construct a weighted average



of willingness-to-pay that reflects the expected annual frequency of different rationing levels, including years in which no rationing is needed.

To estimate willingness-to-pay in years in which rationing is required, we use the methodology developed in Griffin (1990) to estimate consumer willingness-to-pay for the increment of water forgone by water users due to restrictions on water use during a water shortage. This is a widely used methodology for valuing increments (or decrements) of water supply. For example, it provides the basis for the calculation of water supply benefits for the California Water Fix (Sunding, et al., 2013; Sunding, et al., 2015), the economic cost of the state conservation mandate (M.Cubed, et al., 2015; M.Cubed, et al., 2016), as well as numerous other statewide and regional water resources benefit-cost assessments, including the WSMP 2040 (e.g., Jenkins, et al., 1999; Jenkins, et al., 2003; Appendix D of EBMUD, 2012a).

Urban water use can be classified into several broad categories, each with a different priority of use, and the willingness-to-pay for water by utility customers depends on the intended use of each unit of water. The willingness-to-pay for water used for drinking and basic sanitation, for example, is greater than the willingness-to-pay for water used for bathing and laundry, which in turn is greater than the willingness-to-pay for water used for washing cars, for filling swimming pools, and for irrigating landscape. When faced with a water use restriction, utility customers have the choice of which types of water uses to curtail, and the framework for measuring willingness-to-pay incorporates the idea that utility customers respond to a water use restriction by eliminating less valuable water uses before eliminating more valuable water uses, for instance by reducing water used for irrigating landscape prior to reducing water used for personal hygiene.

Figure 5-3 depicts a schedule of willingness-to-pay for different units of water as a demand curve for water that orders these units from highest valued uses to lowest valued uses. Under normal conditions, a customer facing a volumetric water rate of P* demands units of water for which willingness-to-pay exceeds P*. In **Figure 5-3**, this quantity is Q* units. Units of water beyond Q* have value to the customer, but their value is less than their cost, P*, so a rational customer would choose to forego purchasing units beyond Q*. Note that when demand is unconstrained, P* represents the willingness-to-pay for an additional unit of water at the margin.

Suppose instead a quantity restriction is placed on water use so that a customer can purchase no more than Q^R units of water. The customer must forego Q^* - Q^R units of water. The value of these foregone units of water is measured by the shaded area in **Figure 5-3**. Mathematically, this shaded area is calculated as the integral of the demand function evaluated between Q^R and Q^* :

Total WTP for
$$Q^* - Q^R$$
 units of water $= \int_{Q^R}^{Q^*} D(Q) dQ$

The customer will also avoid having to directly pay for Q^* - Q^R . Thus, the customer initially avoids utility bills equal to $P^*(Q^* - Q^R)$. However, most utilities set P^* to recover both their variable operating costs and a portion of their fixed costs. Since utilities operate on a break-even basis, they will still need to recoup the fixed costs that would have been recovered by selling the Q^* - Q^R units of water. Denoting V as the portion of P* that covers the variable costs of production, the utility will still need to recover (P*-V) (Q*- QR) from the customer to cover its fixed costs. While the customer initially avoids P*(Q*- QR), the utility will seek to recover (P*-V) (Q*- QR) in the future, and the net cost avoided by the customer is therefore only V(Q*- QR).

The economic loss to the customer of foregoing Q*- Q^R units of water is therefore:

$$L(Q^*, Q^R) = \int_{Q^R}^{Q^*} D(Q) dQ - V(Q^* - Q^R)$$



Viewed in the other direction, $L(Q^*, Q^R)$ measures what the customer would be willing to pay not to forego Q*- Q^R units of water. On a per unit basis, the average willingness-to-pay for Q*- Q^R units of water is $L(Q^*, Q^R) \div (Q^* - Q^R)$ while the marginal willingness-to-pay is $D(Q^R) - V$. The marginal value corresponds to P^R-V in **Figure 5-3**.

It is convenient to represent Q^R as a multiple of Q^* . Let r be the corresponding percentage reduction in Q^* that yields Q^R . Then $Q^R = (1-r)Q^*$ and the economic loss function becomes:

$$L(Q^*, r) = \int_{(1-r)Q^*}^{Q^*} D(Q) dQ - rVQ^*$$

Operationalizing the economic loss function requires assigning a functional form to D(Q). It is conventional to use a constant elasticity of demand (CED) specification, $D(Q)=AQ^{1/e}$, where A is equal to $P^*/(Q^{*1/e})$. For example, this is the specification recommended in the California Water Commission's Water Storage Investment Program Technical Reference (CWC 2016). It is also the functional form used for the economic analyses of the California Water Fix (Sunding, et al., 2013; Sunding, 2015) and the state conservation mandate (M.Cubed, et al., 2015; M.Cubed, et al., 2016).

Figure 5-3: Consumer Willingness to Pay to Avoid a Water Shortage



Under the CED specification, the economic loss function for rationing level r is given by

CED Loss Function:
$$L(r|Q^*, P^*, e, V) = \frac{e}{1+e} P^* Q^* \left[1 - (1-r)^{\frac{1+e}{e}} \right] - rVQ^*$$

where e is the price elasticity of demand. $L(r|Q^*, P^*, e, V)$ is the monetized loss of foregoing rQ^{*} units of water.



The willingness-to-pay for additional increments of recycled water supply in years in which water is rationed is equal to the difference in the monetized loss with the additional increment of recycled water supply versus without it. This is depicted in **Figure 5-4**. Q^{R0} in **Figure 5-4** is the restricted demand level without the new increment of recycled water while Q^{R1} is the restricted level with the new increment of recycled water. The monetized shortage loss without the new increment of recycled water, the monetized shortage loss is equal to the area A+B+C+E+F-D-G in **Figure 5-4**. With the new increment of recycled water, the monetized shortage loss is equal to the area E+F-G in **Figure 5-4**. The willingness-to-pay for the new increment of recycled water is the difference between these two values, which is the area A+B+C-D. Dividing this value by the increment of new supply, Q^{R1}-Q^{R0}, yields the average per unit willingness-to-pay for the new increment recycled water.





This value can be computed using the CED loss function as:

WTP for Increment of Recycled Water

$$= \left[\frac{e}{1+e}P^*Q^*\left[(1-r_1)^{\frac{1+e}{e}} - (1-r_0)^{\frac{1+e}{e}}\right] - (r_0-r_1)VQ^*\right] \div (r_0-r_1) \cdot Q^*$$

where r₀ is the rationing percentage without the new increment of recycled water and r₁ is the rationing percentage with it.



5.2.3.1 Estimates of Willingness-to-Pay for New Increments of Recycled Water

The projected growth in new recycled water supply is taken from Table 4-1 of the District's 2015 Urban Water Management Plan. This also corresponds to the recycled water growth projection used in WSMP 2040 Portfolio E3. This projection is summarized in **Table 5-8**.

Recycled Water Capacity	2015	2020	2025	2030	2035	2040
Existing	9	9	9	9	9	9
New (post 2015)	0	2	5	8	9	11
Total	9	11	14	17	18	20

Table 5-8. Projected Growth in Recycled Water Capacity per 2015 UWMP (MGD)

The increments of new recycled water capacity in Table 2 are used with the W-E model results for WSMP 2040 Portfolio E3 to calculate the rationing percentages with (r_0) and without (r_1) the new recycled water capacity. These percentages are provided in Appendix C, Attachment 1 for each hydrologic base year in the W-E model. The values for P^{*}, Q^{*}, and *e* are given in **Table 5-9**. P^{*} is the current volumetric rate for potable water. Q^{*} is the planning level of demand used in WSMP 2040 Portfolio E3. The elasticity value of -0.2 is the average of the demand elasticities for Bay Area urban water suppliers used in the state's economic analysis of the California Water Fix (Sunding, et al., 2013; Sunding, 2015).

Table 5-9. Parameters Used to Calculate Willingness-to-Pay for Increments of New Recycled Water Capacity

Willingness-to-Pay Parameters	2020	2025	2030	2035	2040
P* (\$/AF)	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113
Q* (AF)	257,633	260,994	264,354	260,994	257,633
e	-0.2	-0.2	-0.2	-0.2	-0.2

The values in **Table 5-9** were used in combination with the shortage percentages in Appendix C, Attachment 1 to calculate the willingness-to-pay for the additional increments of recycled water capacity shown in **Table 5-8**. The willingness-to-pay estimates for each hydrologic base year in the W-E model are provided in Appendix C, Attachment 2.

Table 5-10 summarizes these estimates for potable reuse projects and non-potable recycled water projects. The only difference between potable reuse and non-potable recycled water is the willingness-to-pay in years when there is no rationing. In these years, the willingness-to-pay is equal to the volumetric rate. Currently, the District sells non-potable recycled water at a discount of close to \$500/AF to compensate customers for diminished water quality. This analysis assumes the District will maintain this discount in the future. Thus, the willingness-to-pay in years without rationing is assumed to be \$2,113/AF for potable reuse and \$1,647/AF for non-potable recycled water. The maximum willingness-to-pay ranges between \$5,100 and \$5,400/AF and corresponds to years with maximum rationing. Under WSMP 2040 Portfolio E3, the maximum rationing percentage with the new recycled water capacity is 20%; without the new capacity the maximum rationing level is 24% by 2040.

The last two rows in each section show the average willingness-to-pay for the 83 hydrologic base years in the W-E model. Across all years, these values are approximately \$2,400/AF for potable reuse and \$2,000/AF for non-potable recycled water. Across only years with rationing of 10% or more, these values are approximately \$4,600/AF. This represents what District customers would be willing to pay for additional increments of potable reuse and non-potable recycled water given current water rates and projections of future water demands and supplies as represented in WSMP 2040 Portfolio E3.



Willingness-to-Pay Estimates	2020	2025	2030	2035	2040	Approximate Average
Potable Reuse Water						
Minimum	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113	
Maximum	\$5,094	\$5,197	\$5,298	\$5,337	\$5,413	
Average for:						
All Years	\$2,366	\$2,364	\$2,402	\$2,397	\$2,401	\$2,400
Rationing >= 10%	\$4,499	\$4,529	\$4,640	\$4,604	\$4,784	\$4,600
Non-Potable Recycled Water						
Minimum	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647	
Maximum	\$5,094	\$5,197	\$5,298	\$5,337	\$5,413	
Average for:						
All Years	\$1,979	\$1,977	\$2,020	\$2,016	\$2,013	\$2,000
Rationing >= 10%	\$4,499	\$4,529	\$4,640	\$4,604	\$4,784	\$4,600

Table 5-10. Willingness-to-Pay for Increments of Potable Reuse Water (\$/AF)

5.3 Comparison of Recycled Water Project Unit Costs to Other Supplies

It is useful to compare recycled water project unit costs to the alternative supply cost and willingness-to-pay estimates. The monetized value of the recycled water produced by the project is the lesser of the alternative supply cost and the willingness-to-pay for additional water supply. The net benefit of a recycled water project is equal to its monetized value minus its net unit supply cost. **Table 5-11** provides a summary of the alternative supply costs and willingness-to-pay estimates.

Parameter	Value (\$/AF)
Alternative Supply Cost	
Dry Year Purchases (see Table 5-7)	\$1,215
Alternative supply cost for Baseload Water	\$681
Alternative supply cost for Dry Year Water	\$1,774
Willingness-to-Pay	
All years – Non-potable Reuse	\$2,000
All years – Potable Reuse	\$2,400
Dry Years	\$4,600

Table 5-11: Summar	of Economic	Evaluation	Parameters
		Eralaalon	



5.3.1 All Years (Baseload) Comparison

5.3.1.1 Non-Potable Reuse Alternatives

Figure 5-5 compares the alternative supply cost and willingness-to-pay to unit costs for the viable non-potable reuse alternatives. While all of the project unit costs exceed the alternative supply cost, four projects are less than customer willingness-to-pay for non-potable reuse and four are slightly above the customer willingness-to-pay.



Figure 5-5: Economic Comparison of Non-Potable Reuse Alternatives

Note: Marker symbology was developed to group alternatives by similar source and/or project type.

 \Box = Satellite RWTP, Δ = East Bayshore RWF, \circ = DERWA/San Ramon, \diamond = Refinery, x = CCCSD and CC Pipeline in Canal ROW.



5.3.1.2 Potable Reuse Alternatives

Figure 5-6 compares the alternative supply cost and willingness-to-pay to unit costs for the potable reuse projects with a non-cost evaluation score of 60 or greater. All of the potable reuse projects have unit costs that exceed the alternative supply cost. Five projects are less than customer willingness-to-pay for potable reuse and two are slightly above the customer willingness-to-pay.





Note: Marker symbology was developed to group alternatives by similar characteristics – markers are colored based on water source, sized based on relative project yield and shaped based on project type.

 \Box = groundwater augmentation, Δ = reservoir water augmentation, \circ = raw water augmentation, \diamond = treated water augmentation.



5.3.2 Dry Year Comparison

As noted previously, in most years (68%) there is adequate supply from the Mokelumne River to meet the Districts water demands. Therefore, potable reuse would be a marginal supply implemented only in dry years. To further evaluate the potable reuse alternatives, dry year unit costs were developed. Capital costs, labor, and maintenance were held constant while electricity and chemical costs and project life-cycle yield were reduced to 30% reflecting a usage rate of 3 out of every 10 years.

5.3.2.1 Non-Potable Reuse Alternatives

Figure 5-7 compares the dry year alternative supply cost and dry year willingness-to-pay to dry year unit costs for the non-potable reuse projects. If operated as a dry year supply, all non-potable reuse projects exceed both the alternative supply costs and only two projects are less than the willingness-to-pay.





Figure 5-7: Dry Year Economic Comparison of Non-Potable Reuse Alternatives

Note: Marker symbology was developed to group alternatives by similar source and/or project type. \Box = Satellite RWTP, Δ = East Bayshore RWF, \circ = DERWA/San Ramon, \diamond = Refinery, x = CCCSD.

5.3.2.2 Potable Reuse Alternatives

Figure 5-8 compares the dry year alternative supply cost and dry year willingness-to-pay to dry year unit costs for the potable reuse projects with a non-cost evaluation score of 60 or greater. If operated as a dry year supply, all potable reuse projects exceed both the alternative supply costs and the willingness-to-pay.





Figure 5-8: Dry Year Economic Comparison of Potable Reuse Alternatives

Note: Marker symbology was developed to group alternatives by similar characteristics – markers are colored based on water source, sized based on relative project yield and shaped based on project type.

 \Box = groundwater augmentation, Δ = reservoir water augmentation, \circ = raw water augmentation, \diamond = treated water augmentation.

5.4 Summary of Alternatives Evaluation

Based on the economic evaluation, there is no net economic benefit to implementing additional non-potable or potable reuse projects at this time as all projects exceed the alternative water supply costs. While there are a number of potable reuse options with comparable unit costs to non-potable reuse, the total capital investment needed is significantly greater as summarized in **Table 5-12**.



Project	Yield (AFY)	Unit Cost (\$/AF)	Dry Year Unit Cost (\$/AF)	Capital Cost (\$M)
Non-Potable Reuse				
DERWA/San Ramon Phase 3	800	1,900	6,300	25
DERWA/San Ramon Phase 5	300	1,600	5,400	8
DERWA/San Ramon Phase 4	300	2,900	9,700	17
East Bayshore Phase 1A only	500	2,400	7,400	16
East Bayshore Phase 1A and 1B	1,100	2,400	7,800	40
East Bayshore Phase 1A, 1B, and 2 ¹	2,900	3,000	9,400	130
Chevron/Richmond	4,300	2,600	8,600	110
Phillips 66	4,100	1,100	3,700	53
Potable Reuse				
SD-1 to Claremont Center, 10 MGD (SD1-Treat-10)	11,200	3,400	9,800	360
SD-1 to Claremont Center, 30 MGD (SD1-Treat-30)	33,600	2,300	6,700	480
SD-1 to Briones Reservoir, 10 MGD (SD1-ResB-10)	11,200	3,900	12,000	460
SD-1 to Briones Reservoir, 30 MGD (SD1-ResB-30)	33,600	2,600	7,500	690
CCCSD to Mokelumne Aq., 19 MGD (CC-Raw-19)	21,820	2,200	6,400	310
CCCSD to Mokelumne Aq., 10 MGD (CC-Raw-10)	11,200	2,300	6,500	180
Oro Loma to USL Res., 8 MGD (Oro-ResU-8)	8,960	2,800	8,600	230

Table 5-12: Highest Ranking, Lowest Cost Alternatives

The recommended plan is to continue progress to meet the District's current goal of 20 MGD by 2040 with the highestranking, lowest-cost alternatives looking at both project unit costs and total capital outlay. The recommended projects are East Bayshore Phases 1A,1B and 2, DERWA/San Ramon Phases 3 through 5, Chevron Richmond, and Phillips 66, totaling 11 MGD as shown in **Table 5-13** and **Figure 5-9**. Diablo Country Club is also included since the project is expected to go forward under a self-financing model.



Project		Yield (AFY)	Yield (MGD)
Diablo Country Club		250	0.2
DERWA/San Ramon – Phase 3		800	0.7
DERWA/San Ramon – Phase 4		300	0.3
DERWA/San Ramon – Phase 5		300	0.3
East Bayshore – Phase 21		2,900	2.4
Chevron/Richmond		4,300	3.8
P66 Rodeo Refinery		4,100	3.7
	TOTAL	12,500	11.1

Table 5-13: Recommend Non-Potable Reuse Projects to Meet 20 MGD by 2040

Notes:

1. East Bayshore Phase 2 costs include Phase 1A and 1B costs.





Figure 5-9: Recommend Recycled Water Projects



6. RECOMMENDED PROJECTS

As noted in Section 5.4, it is recommended that the District pursue a suite of non-potable reuse projects to continue working toward the current recycled water goal of 20 MGD by 2040. The primary projects include continued expansion and implementation of the DERWA/San Ramon and the East Bayshore Recycled Water Project as well as providing recycled water to the refineries in Richmond and Rodeo. The recommended projects are discussed in greater detail below and a summary of capital costs is provided.

At this time, the District can meet its potable water demand with available supplies. The capital costs for implementing potable reuse are very large and if the projects would only be operated as dry-year supply, the unit costs are greater than \$6,000/AF. While it is therefore not recommended to implement any potable reuse projects in the near-term, it is recommended that the District continue to engage in the state's efforts to develop potable reuse regulations and operator training for future long-term needs. The District should revisit the most favorable potable reuse alternatives as the treatment costs come down and the need for water supply increases in the future. To guide these future efforts, a discussion of the most promising potable reuse projects is also included in this section.

6.1 DERWA/San Ramon Valley Recycled Water Project Phases 3 through 5

Phase 1 and 2 of the DERWA/San Ramon Project are near completion, and portions of the Phase 3 recycled water pipeline and the R3000 recycled water reservoir are already constructed. While all remaining portions of the DERWA/San Ramon project scored highly in the non-cost evaluation, Phase 4 serving eastern Blackhawk scored lower and it is less cost-effective than Phase 3 or 5, and is therefore planned for implementation after Phase 3 and 5. Although they are relatively small projects, this well-established landscape irrigation program should continue to be a high priority project as long as funding and source supply are available.

6.1.1 Project Overview

The DERWA/San Ramon project requires both distribution infrastructure solely funded and constructed by the District and recycled water treatment plant expansion co-funded by the District, DSRSD, and the City of Pleasanton. **Figure 6-1** provides a summary overview of the distribution infrastructure for two phases.

Phase 3 distribution infrastructure includes the following components:

- Distribution pipelines along Dougherty Rd., Crow Canyon Rd. (optional) and Camino Tassajara corridors, ranging in size from 4 to 16 inches (partially completed);
- R3000 pump station along Dougherty Rd. (currently in planning);
- Customer retrofits.

Phase 5 distribution infrastructure builds on Phase 3 and includes the following components:

- Distribution pipelines extending along Blackhawk Rd. to the western portion of Blackhawk Country Club;
- Customer retrofits.

Phase 4 distribution infrastructure also builds on Phase 3, and includes the following components:

- Distribution pipelines extending along Blackhawk Dr. to the eastern portion of Blackhawk Country Club;
- Pump station R4000;
- Customer retrofits.



To meet these additional customer demands a number of treatment plant upgrades and expansions are also planned. In 2016, Phase 1 of the treatment plant expansion was completed adding a sixth sand filter. Phase 2 was completed in 2018 and added influents pumps, flocculation basins, UV modules and recycled water distribution pumps. This will improve the water quality of the filter feed and expand the facility's peak production capacity to 16.2 MGD. Additional expansion of the recycled water treatment facility will be dependent on timing of customers demand and source water availability.







6.1.2 Additional Considerations

DERWA has experienced peak month supply shortfalls during summer season, requiring potable water supplement. DERWA is also exploring other additional supply opportunities, including groundwater, recycled water from CCCSD, and diversion of raw wastewater from CCCSD's adjacent sewerage collection system to supplement DERWA's recycled water supply. It is important to note that wastewater supply from DSRSD is limited and the DERWA supply agreement defines recycled water service to the two-member agencies as first-come, first-served. This means that the availability of water is dependent upon how quickly the District expands its distribution system versus how quickly DSRSD develops its program. Therefore, the District should continue to expand its recycled water infrastructure while working through DERWA to secure the needed wastewater supply to meet the increasing demand.

6.2 East Bayshore Phase 2

6.2.1 Project Overview

The East Bayshore Phase 2 project (which includes expanding the customer base along existing infrastructure in Phase 1A, then adding Phase 1B and Phase 2 pipelines) is included as a recommended project because it builds on an existing District recycled water project and it serves non-potable water to a relatively large and diverse group of customers.

The District has been evaluating alternatives for East Bayshore under a separate effort. The *Final East Bayshore Recycled Water Facility Water Quality Evaluation Technical Memorandum No. 1* dated March 2018 (TM1) summarized water quality issues experienced at EBRWF and recycled water demands (BC, 2018). The initial portion of the study, concluded that the existing water quality may not be ideal for irrigating sensitive species, and is not suitable for industrial use (i.e., cooling towers and boilers) or toilet flushing. The second phase of the EBRWF Study is currently underway. Its objective is to identify short-, intermediate-, and long-term alternatives to improve recycled water quality for non-potable reuse and ultimately identify a path forward for the EBRWF recycled water program. While the preferred alternative from the study has not yet been selected, the recommended buildout plan for this program is a new 4.5-MGD MBR treatment facility at SD-1.

The MBR treatment system offers improved recycled water quality, thereby allowing for expansion of the customer base to year-round industrial customers. The MBR treatment system can be designed for a phased approach to reach buildout capacity. Annual average recycled water demand is anticipated to be around 2,867 AFY (2.6 MGD average annual demand, 4.5 MGD max month demand). An additional 0.8 MGD of in-plant reuse at SD-1 is also anticipated and included in the facility sizing, but not in the project yield. The Phase 1A (short-term) pipeline expansion would serve new customers within the existing distribution area and the Frontage Road pipeline alignment. The Phase 1B (intermediate-term) pipeline expansion would serve new customers in Oakland and Berkeley, as shown in **Figure 6-2**, while the long-term Phase 2 pipeline expansion would serve additional customers in Alameda and Albany, the UCB campus, and customers along the Powell Street and Channing Way Pipeline alignment. As discussed previously, the EBRWF Phase 2 was prioritized over the UCB Satellite Treatment Project, which only serves the demands at the UCB campus.

The new MBR treatment alternative would replace the existing tertiary membrane filters and would include screening, grit removal, activated sludge basins and membrane tanks located at SD-1. The 4.5-MGD MBR project would improve recycled water quality by treating influent flows with lower total dissolved salt concentrations (from the Adeline interceptor) and would also provide biological nutrient removal (BNR). The existing recycled water storage tank and distribution pumps have enough capacity to convey the additional flows and can be repurposed. The chlorine contactor would be demolished, and a new one would be built. Recycled water would be stored and distributed using the existing EBRWF facilities.



Figure 6-3 shows the site layout for the 4.5-MGD MBR alternative. Recycled water would be stored and distributed using the existing EBRWF facilities. This alternative would require land acquisition east of SD-1 from the California Department of Transportation (Caltrans) to construct a pump station capable of delivering influent from the Adeline Interceptor to an MBR process. The cost of land was not included in the cost estimate.

As shown in the site layout, the identified location for the EBRWF facilities has limited space and constructability is a potential issue. The current proposed location of the MBR facility is within the existing EBRWF footprint and does not provide adequate land for expansions of the MBR system if needed in the future. An alternate location at the treatment plant may be desirable to allow for expansion and/or to facilitate construction of the new MBR facilities.





Figure 6-2: East Bayshore Phase 1A, 1B and Phase 2 Project



Figure 6-3: Proposed 4.5-MGD MBR Layout at EBRWF



Note: The red line indicates the location of the existing primary effluent bypass pipeline.



6.2.2 Additional Considerations

As described in Section 4, a number of projects were evaluated using SD-1 as a supply source for a potable ruse project, including a 4-MGD reservoir augmentation project to San Pablo Reservoir, a 10-MGD reservoir augmentation project to Briones Reservoir, and a 10-MGD treated water augmentation to Claremont Center. In the future, the MBR for the EBRWF could be repurposed and/or expanded to meet potable reuse demands. Additional space would be required for advanced treatment. As shown in **Figure 6-4**, the existing distribution system could also be expanded to the San Pablo WTP for raw water augmentation or to San Pablo Reservoir for reservoir water augmentation. This would require approximately 10,000 linear feet of additional pipeline from the currently proposed pipeline expansion to serve Albany to San Pablo WTP. The East Bayshore Phase 2 pipelines in central Berkeley are preliminarily sized at 20 inches, which is adequate capacity for at least 7 MGD of flow at a maximum velocity of 5 fps.



Figure 6-4: Potential Alignment to Expand EBRWF to San Pablo WTP



6.3 Chevron Refinery/Richmond WPCP Recycled Water Project

6.3.1 Project Overview

This project would produce additional recycled water to serve the Chevron Richmond refinery from the Richmond WPCP is included as a recommended project because it builds on the District's successful partnership with Chevron on the RARE and North Richmond project. Both Chevron and the District have made significant investments to serve recycled water from the West County WPCP to the refinery for cooling tower makeup water and boiler feed systems.

Upgrading all or part of the Richmond WPCP to provide suitable water quality is one of the major components of the project. The 2016 Richmond Facilities Plan evaluated treatment upgrades at the Richmond WPCP and new pipelines to RARE, as shown in **Figure 6-5**. The proposed treatment layout, shown in **Figure 6-6**, includes split treatment using MBR for recycled water production and conventional activated sludge for the flows discharged to the Bay. The recycled water project ultimately was not the recommended master plan alternative due to the uncertainty in the need and economic feasibility of producing recycled water suitable for diversion to the District's RARE facility. In addition to the technical water quality and distribution issues, interagency coordination and financial support would be required. As stated in the 2016 Richmond Facility Plan, these challenges might be resolved when the need for recycled water becomes sufficiently critical. In the meantime, the City of Richmond can implement upgrades identified in the master plan; these upgrades would not preclude future recycled water delivery to the RARE facility. The District should engage with the City of Richmond on treatment plant upgrades as they progress, since the completion of upgrades (and their eventual performance) could have a major bearing on the timing for implementation of the recycled water project.

6.3.2 Additional Considerations

If the Richmond Refinery were to close in the future, the West County WPCP and Richmond WPCP effluent could provide advanced-treated water suitable for potable reuse in a number of ways, including reservoir water augmentation at Briones or San Pablo Reservoir, raw water augmentation at the Sobrante WTP; or treated water augmentation at the Wildcat Aqueduct. If groundwater augmentation is determined to be feasible in the area, it could be an additional potable reuse option. However, the combination of reduced demand in the Richmond area, and the location of the advanced-treated water in Richmond, would make it difficult to reach a large number of existing potable water customers without constructing significant new transmission pipelines. Converting the project from non-potable reuse to potable reuse would require significant additional investment, since very little infrastructure would be available for re-purposing (i.e., no conveyance and only a portion of the treatment system could be re-purposed).





Figure 6-5: Chevron Refinery/Richmond WPCP Pipeline to RARE



Figure 6-6: Chevron Refinery/Richmond WPCP Site Layout



Source: Adapted from (Carollo, 2016a), Figure 9.4. "Site Layout of Split Flow CAS/MBR Alternative"



6.4 Phillips 66 Refinery Recycled Water Project

6.4.1 Project Overview

The Phillips 66 Recycled Water Project is included as a recommended project because it would deliver a large amount of recycled water (up to 3.7 MGD) to a single customer, with comparatively few pipelines required due to the short distance between the sources of wastewater and the Phillips 66 Refinery.

The source of water for this project is the Pinole-Hercules WPCP (providing most of the supply) and Rodeo WPCF (providing a fraction of the supply). The combined supply is currently large enough to produce 1,340 gpm (1.9 MGD) for the boiler feed water treatment system. Remaining flow could be used to satisfy a portion of the cooling tower makeup water demand (i.e., 600 gpm or 0.86 MGD) for a total of 2.6 MGD or 2,912 AFY (Project Phase 1). In the future, if sufficient flows were available, the remaining cooling tower demand could be met (Project Phase 2). Cost estimates presented in this study include annual average recycled water delivery of up to 3.7 MGD (4,144 AFY).

As shown in **Figure 6-7**, the combined final disinfected effluent from both treatment plants would be pumped at the Rodeo Pump Station to the refinery fence line for treatment. A new pipeline would be needed to convey treated effluent from the Rodeo Pump Station to the refinery fence line and an existing pipeline would be used to convey treated effluent from the fence line to new treatment facilities. This configuration was used to develop the cost estimates for the project. Alternatively, the treatment facilities could be constructed on a vacant lot immediately outside the refinery fence line, adjacent to the Rodeo WPCF.

The new treatment facilities include membrane filtration (MF), biological aerated filter (BAF), reverse osmosis (RO) and ultraviolet (UV) disinfection. **Figure 6-8** provides a process flow diagram for this alternative. An existing (1.2 MG) tank would be used for equalization. All flows would be filtered through the MF. Following the MF, a BAF and UV disinfection would provide ammonia removal and disinfection for the cooling tower makeup water. The RO system would treat the remaining portion of the MF filtrate to produce recycled water for the boiler pre-treatment system (RO followed by mixed bed demineralizers). RO permeate would be sufficient to feed directly to the demineralizers in place of potable water. The BAF/RO blend will provide a higher quality feed water to the existing RO system. From the treatment system, the recycled water would be routed to the process areas through existing pipelines, as feasible. Allowances for new pipe segments were included in the construction cost estimates. For the cooling tower makeup water, new pipe segments were assumed to be carbon steel. For the boiler feed water, new pipe segments were assumed to be stainless steel due to the corrosive nature of the RO permeate water.

Waste streams from the recycled water treatment facility (filter backwash and RO concentrate) would be ultimately discharged to the Bay via the Refinery's permitted outfall (NPDES Permit No. CA 0005053, Order No. R2-2016-0044). The MF/BAF backwash would be discharged to the refinery's sewer system and treated at the refinery's wastewater treatment plant prior to discharge to the refinery's deep-water outfall (Discharge Point No. 002). RO concentrate would be discharged through existing pipelines to the refinery's deep-water outfall. Chemical waste streams would be routed to the existing refinery waste stream via catch basins after neutralization.

The refinery's current NPDES permit specifies a procedure for adjustment of concentration-based and mass-based effluent limits to reflect the use of recycled water, which may contain increased pollutant concentrations compared to other sources. The permit may require further modification to account for loading from the reject waste streams. Ultimately, the refinery may need to perform additional monitoring, testing and evaluations to confirm permit compliance, in particular for effluent toxicity.





Figure 6-7: P66 Refinery Alternative Site Location

6.4.2 Additional Considerations

Funding and implementation of this project could be complex because the treatment plant would be located on Refinery property. In addition, the location of the treatment plant within the refinery would likely make it difficult to adapt the treatment train for potable reuse. The project does not include any major conveyance infrastructure that could be repurposed for potable reuse (indeed, the short pipeline length is the major driver for the project's comparatively low cost), but the treated wastewater from Rodeo WPCF and/or Pinole-Hercules WPCP could be used for a potable reuse project as described in Section 4.





Figure 6-8: Phillips 66 Refinery Alternative Process Flow Diagram

6.5 Cost of Recommended Projects

Estimated capital and operating costs for the recommended project list are shown below in **Table 6-1**. Satellite treatment at Diablo Country Club is not included in the cost summary because it is expected to go forward under a self-financing model. More detailed cost estimates are included in **Appendix A**.

Project	Yield (AFY)	Capital Cost (\$M)	O&M (\$M/yr)
DERWA/San Ramon Phase 3	800	25	0.49
DERWA/San Ramon Phase 5	300	8.1	0.15
P66 Rodeo Refinery	4,100	53	2.1
DERWA/San Ramon Phase 4	300	17	0.18
East Bayshore. Phase 2	2,900	130	2.9
Chevron/Richmond	4,300	110	5.7
Total	12,700 AFY	\$ 343 M	\$ 11.5 M/yr

Table 6-1: Capital and O&M Costs of Recommended Projects

6.6 Future Potable Reuse Consideration

Due to the large capital costs and the current availability of alternate potable water sources, potable reuse is not recommended in the near-term but may become more favorable in the next 10-20 years as secondary wastewater treatment upgrades are implemented, advanced-treatment costs come down, and the need for water supply increases.

Recovery Rates: 90 percent MF, 85 percent RO and 95 percent BAF.



Potable reuse will only be cost effective when it is implemented as a large project operated in all years, not as a marginal water supply alternative operated only in dry years.

Nearly all of the alternatives for potable reuse evaluated in this report would require secondary treatment upgrades for nutrient removal as a precursor to advanced treatment. As stated previously, BACWA recently completed a Nutrient Reduction Study to quantify opportunities for removing nutrients from wastewater discharged to San Francisco Bay, either through optimizing existing operations or upgrading treatment plants (BACWA, 2018). The study provided cost estimates for upgrading each of the 37 municipal wastewater treatment plants in the Bay Area to achieve an annual or seasonal average total nitrogen concentration below 15 mg/L (corresponding to "Level 2" nitrogen removal). Of the alternatives listed below in **Table 6-2**, Oro Loma WPCP has the lowest cost for Level 2 nitrogen removal per unit of water available for potable reuse (present value of \$56M for 8 MGD of potable reuse), while SD-1 has the highest cost (present value \$2.6B for 30 MGD of potable reuse).

The timing for the nutrient regulations and associated upgrades is uncertain but is expected to be at least 10 years away (2029). Therefore, potable reuse should be revaluated during the planning and design of the WWTP upgrades. Furthermore, by including beneficial reuse of WWTP effluent, the projects may be eligible for greater funding opportunities through state and federal grant and loan programs.

The most promising potable reuse alternatives are listed below in **Table 6-2** and discussed in greater detail following the table, as a starting point for future consideration. Several of the projects were evaluated at different sizes (i.e., 10 MGD and 30 MGD) to evaluate the potential for economy of scale.

Project	Yield (AFY)	Unit Cost (\$/AF)	Dry Year Unit Cost (\$//AF)	Capital Cost (\$M)
Reservoir Augmentation				
SD-1 to Briones Reservoir, 10 MGD	11,200	3,900	12,000	510
SD-1 to Briones Reservoir, 30 MGD	33,600	2,500	7,300	690
Oro Loma to USL Reservoir, 8 MGD	8,960	2,800	8,600	230
Raw Water Augmentation				
CCCSD to Mokelumne Aqueduct, 10 MGD	11,200	2,300	6,500	180
CCCSD to Mokelumne Aqueduct, 19 MGD	21,820	2,200	6,300	310
Treated Water Augmentation				
SD-1 to Claremont Center, 10 MGD	11,200	3,400	9,800	360
SD-1 to Claremont Center, 30 MGD	33,600	2,300	6,700	480

Table 6-2: Highest Ranking, Lowest Cost Potable Reuse Alternatives

6.6.1 Surface Water Augmentation at Briones Reservoir from SD-1

Reservoir augmentation to Briones Reservoir is identified as a promising alternative for future consideration because of three main factors:

• Briones Reservoir is a more suitable site for reservoir augmentation than other District reservoirs (Upper San Leandro and San Pablo Reservoirs) because it has a large volume and a small watershed. As a result, Briones Reservoir can accept a large amount of advanced-treated water while complying with the state's newly crafted regulations for reservoir augmentation.



- Use of Briones Reservoir for reservoir augmentation is more equitable than similar projects at Upper San Leandro and San Pablo Reservoirs, which serve a smaller portion of the District's water customers.
- A large amount of water is potentially available from SD-1, although the future quality of the treated wastewater is uncertain. If SD-1 is upgraded to provide nutrient removal for the full plant flow, the effluent would become more suitable for potable reuse. Without these full-plant upgrades, a separate split treatment system (such as an MBR) would be required to treat a portion of the flow.

A project from SD-1 to Briones Reservoir could be relatively large, nearing the full capacity of SD-1. The need for water, and continued operation of Resource Recovery at SD-1 are expected to govern the sizing. Reservoir capacity is not a limiting factor.

6.6.2 Reservoir Water Augmentation at Upper San Leandro from Oro Loma WPCP

Reservoir augmentation from Oro Loma WPCP to Upper San Leandro Reservoir is identified as a promising alternative for future consideration primarily because of the large treatment facility site in comparison to SD-1, which is a highly constrained site. In addition, Oro Loma Sanitary District is already moving forward with secondary treatment upgrades for nutrient removal that are expected produce a secondary effluent quality appropriate for advanced treatment. These factors also contribute to a relatively high score for groundwater augmentation using Oro Loma WPCP, but the difficulty of using the groundwater near the extraction site lead to a relatively lower score compared to reservoir augmentation. RO concentrate disposal is not likely to pose a challenge, as the RO concentrate would be combined with other wastewater prior to being discharged through the EBDA deep water outfall.

6.6.3 Raw Water Augmentation of Mokelumne Aqueduct from CCCSD WWTP

Raw water augmentation from CCCSD WWTP to the Mokelumne Aqueduct is identified as a promising alternative for future consideration because of two main factors:

- It is an equitable project, because the advanced-treated water would enter the District's raw water system at a point where it could reach the largest possible number of the District's water customers (i.e., all of them), and
- It requires a relatively small investment in pipelines, since the CCCSD plant is located close to the Mokelumne Aqueduct in North Concord. This short pipeline alignment results in lower capital costs and environmental impacts. After entering the Mokelumne Aqueduct, the water would be conveyed to Walnut Creek WTP, Orinda WTP, or a surface water reservoir before undergoing further treatment and entering the treated water system.

CCCSD is actively pursuing other end uses for recycled water, including serving the Martinez refineries (as discussed in Section 3.3.11), which are likely to be more cost-effective than potable reuse in the near term. However, potable reuse could ultimately replace the refinery recycled water project if the refineries no longer needed the water in the future.

This project is best sized at less than the full capacity of the CCCSD plant, as RO concentrate disposal is likely to pose a challenge if the full flow is used.

6.6.4 Treated Water Augmentation at Claremont Center from SD-1

Treated water augmentation from SD-1 to Claremont Center is identified as a promising alternative because of two main factors:



- It is an equitable project compared to other treated water augmentation sources such as Richmond WPCP, Oro Loma WPCP, or West County WPCP, since the advanced-treated water would be distributed widely to West-of-Hills water customers.
- It requires the least investment in pipeline infrastructure than any other comparably sized project originating at SD-1. Claremont Center is the closest location that a large amount (~30 MGD) of water could be added to the distribution system. Smaller amounts of water could be added to the distribution system closer to SD-1 but would reach a smaller array of water customers.

The major drawback of this alternative are site constraints at SD-1.


7. IMPLEMENTATION PLAN

7.1 Project Phasing

Phasing for implementation of the recommended project list is shown below in Table 7-1.

Project	Yield (AFY)	Construction Period	Year Online	Implementation Notes
DCC Satellite Project	250	2019-2020	2020	RFP for Design-Build expected in mid-2018.
DERWA Phase 3	800	2024-2025	2025	Requires supplemental supply
DERWA Phase 5	300	2028-2029	2030	Requires supplemental supply
P66 Rodeo Refinery	4,100	2025-2030	2030+	Funding agreements with P66 Refinery are critical path. Insufficient water is available in the near term; assumes 2.6 MGD in 2030 and expansion to 3.7 MGD by 2040.
DERWA Phase 4	300	2033-2034	2034	Requires supplemental supply
East Bayshore Phase 2	2,900	2030-2035	2035	Significant capital investment required and customer outreach needed for implementation.
Chevron/Richmond	4,300	2035-2040	2040	Plant upgrades at Richmond WPCP are primary constraint.

Table 7-1: Phasing for Recommended Projects

DCC Satellite Project. This project is listed first because it is expected to be completed by 2020; an RFP for design and construction is expected in 2018. The project is proceeding under a self-financing model, and the schedule is not under District control.

DERWA Phase 3 and 5. The DERWA Phase 3 and Phase 5 expansions are listed as the first District-funded project in Table 6-1 because it is an expansion of an existing project, and because the recycled water is available on a first-come, first-serve basis. Total DERWA deliveries would expand from current deliveries of up to 0.8 MGD (existing Phase 1 and 2B) to 1.3 MGD (with Phase 2A implementation) in 2020 to about 2.3 MGD (with implementation of Phases 3 and 5) in 2030.

Phillips 66. The Phillips 66 project is listed second for implementation, to be implemented in two phases. Due to supply limitations, it is assumed that the Phillips 66 project will deliver 2.6 MGD of recycled water in 2030 and up to 3.7 MGD by 2040. Although the Phillips 66 project is not indicated for completion until after 2030, there are several tasks to complete in the intervening years, including re-initiating discussions with Phillips 66, confirming recycled water demands and recycled water quality requirements, and confirming supply from Pinole-Hercules WPCP and Rodeo WPCF.

DERWA Phase 4. The DERWA Phase 4 expansion would bring total DERWA deliveries to 2.5 MGD by 2034. This phase is listed after Phase 3 and Phase 5 because it is more expensive on a unit cost basis.

East Bayshore. The District's East Bayshore project currently delivers approximately 0.2 MGD (existing Phase 1A). With additional customer outreach, deliveries are expected to increase by an additional 0.2 MGD by 2025. Significant



capital investment and customer outreach is necessary to connect recycled water demands of 2.6 MGD (Phase 2) by 2035. Even though the East Bayshore project is shown with a later start date for construction (2030), the District can begin customer outreach and implement strategies to maximize existing assets. For example, the District could improve water quality at East Bayshore through treatment upgrades to deliver recycled water to more customers with minimal pipeline expansion. In the short-term, before treatment upgrades are constructed to address chloride water quality objectives, the District could extend select pipeline networks to serve irrigation customers that do not require significant water quality improvements.

Chevron Richmond. The proposed expansion of recycled water service to Chevron Richmond Refinery could begin by 2035. Treatment upgrades required for this recycled water project are not included in the City of Richmond's CIP. The City of Richmond is preparing for treatment upgrades that align with its long-term master plan based on current understanding of anticipated nutrient regulations. It is assumed that the recycled water project could be implemented following Level 2 nutrient upgrades (i.e., ammonia limit of 2 mg/L, total nitrogen limit of 15 mg/L) within the next 20 years. This regulatory uncertainty is the reason why this project is listed last for implementation. Nonetheless, it is recommended that the District initiate discussions with the City of Richmond to confirm timing for treatment and collection system upgrades and interest in producing recycled water for the RARE facility.

Projected recycled water deliveries for the recommended projects are shown in **Figure 7-1** for the planning horizon (2020 to 2045). Recent deliveries over the period 2013-2017 are also shown for reference. In 2016 and 2017, recycled water deliveries were significantly impacted by water supply and quality. The District's NRWRP experienced interruption of influent supply from West County due to construction shutdown. Therefore, the District had to supplement Chevron Richmond Refinery with potable water. The District's RARE recycled water demand was also supplemented with potable water due to water supply and quality issues.

In **Figure 7-1**, deliveries in 2020 include an "Existing Use" category equal to 9.2 MGD. This reflects the full capacity of each existing recycled water project, as listed in **Table 3-1**, plus an additional 0.5 MGD of capacity expected upon completion of DERWA / San Ramon Valley Phase 2A. This approach is consistent with the goal of fully utilizing all available recycled water through increased customer outreach. The "existing use" category includes 0.2 MGD of deliveries from the San Leandro WRP, even though the recycled water facility has not been operating in recent years (see Section 3.3.3 for additional details).

The total planned 2040 recycled water deliveries add up to 19.5 MGD, which is slightly below the goal of 20 MGD. This highlights the importance of customer outreach and project development to confirm the technical and economic feasibility of the recommended projects.







The effort required to implement the projects according to the proposed timeline in **Figure 7-1** is illustrated on the next page in **Figure 7-2**. The cost and effort for the two most expensive projects in the portfolio (East Bayshore and Chevron) are borne in the last 10 years of the planning period (2030-2040) rather than in the first 10 years.





Figure 7-2: Proposed Phasing of Recommended Projects

7.2 Institutional Needs and Customer Outreach

With exception of East Bayshore which would be sourced from SD-1, the other recommended projects require institutional arrangements with other wastewater providers to secure effluent for recycled water treatment and beneficial use. The East Bayshore Project has sufficient supplies for any build-out scenario, including the recommended Phase 2, and the supply from SD1 is secured.

DSRSD and the District currently work together under the DERWA Joint Powers Authority. By using operational storage in Tassajara Reservoir, the San Ramon Valley Recycled Water Project met summer demands in 2018 without the addition of potable water on peak days. However, further expansion of the San Ramon Valley Project will require supplemental supplies to meet peak summer demands due to reduced wastewater flows from water conservation practices. The District should work to secure supplies including the use of groundwater to meet peaking demands; diverting wastewater flows from Central Contra Costa Sanitary District into the DSRSD system; and engaging with the City of Livermore to discuss supply opportunities. In addition, the District should continue public and developer outreach in the region to promote the use of recycled water.

A future Chevron recycled water expansion could use up to an additional 3 MGD of recycled water. The Richmond WPCP has sufficient dry weather flow (4.4 MGD) to meet this demand. Since the District is the water purveyor in the area, the City of Richmond does not have the legal authority to purvey water unless approved by the District. Also, the City would benefit from a potential partnership with the District that would reduce treated wastewater discharge to the bay. In addition, the refinery's effluent could also be a supply.

The Phillips 66 refinery has sufficient demand (3.7 MGD) to utilize all of the available wastewater flow from the Pinole-Hercules WPCP and the Rodeo WPCF (approximately 2.7 MGD combined). The District is the water purveyor for the Pinole, Hercules, and Rodeo vicinity and the wastewater agencies do not have the legal authority to purvey water without the District's approval. The wastewater agencies would benefit from a partnership with the District that would reduce their treated wastewater discharge to the bay. The District should begin planning level evaluations now, including working with the wastewater agencies to evaluate long-term wastewater trends. Between 2005 and 2015 wastewater flows decreased by about 25%, with a portion of that reduction likely the result of water conservation practices.

7.3 Funding Opportunities

Typically, recycled water projects are financed through a combination of grants, partnerships with project beneficiaries, and at times, the Clean Water State Revolving Fund (CWSRF). Federal, state and regional funding sources are available to help implement recycled water projects. These potential funding sources include:

- USBR Title XVI Water Reuse Grant Program. This federal grant program provides funding for 25% of a project's capital cost, up to \$20M. To receive funding, projects must be authorized by Congress. None of the recommended projects in the Master Plan are currently authorized under the Title XVI program.
- USBR Water Infrastructure Improvement for the Nation (WIIN). Under the WIIN Act, USBR is providing a
 new funding opportunity for Title XVI water recycling projects. Unlike other Title XVI funding opportunities,
 projects do not need authorization by Congress to be eligible for WIIN. The opportunity may provide up to
 25% of the total cost of planning, design and/or construction, up to \$20M. Applications for the latest round of
 funding were due in July 2018 and future rounds of WIIN funding are anticipated.
- SWRCB Water Recycling Funding Program (WRFP) Construction Grants. This state grant program provides funding for 35% of a project's construction costs, up to \$15M. WRFP grants are currently

oversubscribed, but it is anticipated that interest groups and legislators will work to secure provide additional funding through a new water bond.

- SWRCB Clean Water State Revolving Fund Loan (CWSRF) Program. This state loan program provides low-interest loans (half of the General Obligation bond) for water quality infrastructure projects. This program is currently oversubscribed, and the SWRCB is holding workshops to discuss potential changes to the program and solicit input on funding priorities.
- DWR Integrated Regional Water Management (IRWM) Program. This state grant program provides funding
 for implementation of projects that are coordinated at a regional level. The recommended Master Plan projects
 are within the San Francisco Bay Area IRWM region, which has supported construction of the East Bayshore
 and the DERWA/San Ramon Valley recycled water projects. The current funding source for IRWM program
 is Proposition 1, which had the first round of solicitation for implementation projects in summer of 2018 and is
 expected to have a second round in 2019. Phase 3 of the DERWA/San Ramon Valley project is the most
 eligible among the recommended Master Plan projects for this funding source, which prioritizes constructionready projects.

The funding opportunities listed above are available to public water and wastewater agencies to fund projects with public benefits such as reduction of wastewater discharges or potable water offsets. While the refinery projects serve a single private entity, the projects may be eligible for grant funding if owned and operated by the District or another partner public agency.

Costs were summarized as part of Section 6, with unit cost for water for the recommended projects ranging from \$3,600-\$7,500/AF. As projects move forward and grants and loans become available to the District, rates and charges will be refined.

8. **REFERENCES**

- BACWA, 2018. Nutrient Reduction Study: Potential Nutrient Reduction by Treatment Optimization, Sidestream Treatment, Treatment Upgrades, and Other Means - Final Report. Prepared by HDR for BACWA, June 22, 2018. Available online at <u>https://bacwa.org/wp-</u> content/uploads/2018/06/BACWA Final_Nutrient_Reduction_Report.pdf
- BACWA, 2017. Group Annual Report. Nutrient Watershed Permit Annual Report. October 2017.
- Brezack & Associates Planning, LLC, 2013. *Diablo Country Club Satellite Recycled Water Treatment Plant Feasibility Study*. September 2013.
- Brown and Caldwell, 2007. ConocoPhillips San Francisco Refinery High-Purity Recycled Water Project Technical Study. August 2007.
- Brown and Caldwell, 2016. Richmond Refinery Recycled Water Study, Final Report. Privileged and Confidential.
- Brown and Caldwell, 2017. *East Bayshore Recycled Water Project Water Quality Improvements Study*. Draft Technical Memorandum. October 2017.
- Brown and Caldwell, 2018a. Final East Bayshore Recycled Water Facility Water Quality Evaluation Technical Memorandum No. 1 (TM1). March 2018.
- Brown and Caldwell, 2018b. East Bayshore Recycled Water Study Draft Final Report. Prepared for EBMUD, November 2018.
- California Water Commission, 2016. Water Storage Investment Program Technical Reference.
- Carollo, 2013. 2013 Contra Costa Water District Untreated Water Facilities Improvement Program Plan Update. Final. July 2013
- Carollo, 2016a. City of Richmond and Veolia Water WWTP Facility Plan. Final Draft. September 2016
- Carollo, 2016b. City of San Leandro Recycled Water Market Assessment Study. Draft. January 2016.
- CCCSD, 2013. *Recycled Water Quality and Distribution Annual Report*. Prepared by Central Contra Costa Sanitary District, February 5, 2014.
- CCCSD, 2016. *Recycled Water Wholesale Opportunities*. Prepared by RMC Water and Environment for Central Contra Costa Sanitary District. March 2016.
- CCWD, 2017. Los Vaqueros Reservoir Expansion Project Draft Supplement to the Final EIS/EIR. Prepared for the United States Bureau of Reclamation and Contra Costa Water District, June 2017.
- EBMUD, 2001. *Regional Hydrogeologic Investigation Outer Basins*. Prepared by CH2MHill for EBMUD, February 2001.
- EBMUD, 2005a. EBMUD Bayside Groundwater Project Environmental Impact Report.
- EBMUD, 2005b. Satellite Recycled Water Treatment Plant Feasibility Study. Draft.
- EBMUD, 2008a. Water Supply Engineering Statistical Report Fiscal Year Ending June 30, 2008.
- EBMUD, 2008b. Cost Estimating for Construction Engineering Standard Practice 20.3. Effective May 14, 2008.
- EBMUD, 2009. Water Supply Engineering Statistical Report Fiscal Year Ending June 30, 2009.
- EBMUD, 2010a. Water Supply Engineering Statistical Report Fiscal Year Ending June 30, 2010.
- EBMUD, 2010b. Life Cycle Cost Analysis Engineering Standard Practice 20.1. Effective August 26, 2008.

- EBMUD, 2011a. Water Supply Engineering Statistical Report Fiscal Year Ending June 30, 2011.
- EBMUD, 2011b. Average Useful Life of Water Facilities Engineering Standard Practice 462.1. Effective May 26, 2011.
- EBMUD, 2011c. East Bay Municipal Utility District Water Conservation Master Plan.
- EBMUD, 2012a. Water Supply Management Program 2040 Plan. Final April 2012.
- EBMUD, 2012b. Water Supply Engineering Statistical Report Fiscal Year Ending June 30, 2012.
- EBMUD, 2013a. Water Supply Engineering Statistical Report Fiscal Year Ending June 30, 2013.
- EBMUD, 2013b. Water Supply Assessment for the Richmond Bay Campus 2013 Long Range Development Plan. July 2013.
- EBMUD, 2014a. Water Supply Engineering Statistical Report Fiscal Year Ending June 30, 2014.
- EBMUD, 2014b. Raw Water Master Plan.
- EBMUD, 2015a. Rossmoor Golf Course. Recycled Water Use Evaluation Summary. August 2015.
- EBMUD, 2015b. Urban Water Management Plan. Final. July 2015.
- EBMUD, 2016. East Bay Municipal Utility District Urban Water Management Plan.
- EBMUD, 2016a. East Bayshore Recycled Water Project. Draft Recycled Water Quality Guidelines Summary for Cooling Towers. July 2016.
- EBMUD, 2016b. Using Water from Contra Costa Canal Preliminary Evaluation Summary. June 2016.
- EBMUD, 2017a. East Bayshore Recycled Water Expansion Study Hydraulic Analysis of Future Pipelines and Demands. July 2017.
- EBMUD, 2017ba. Fiscal Year 2016 Annual Energy Report Operations and Maintenance Department. February 2017.
- ESA, 2016. City of Richmond Terminal One Project Draft Environmental Impact Report. February 2016.
- Griffin, R. C., 1990. Valuing Urban Water Acquisitions. Water Resources Bulletin, 219-225.
- Griffin, R. C., 2006. Water Resource Economics. Cambridge: The MIT Press.
- Jenkins, M. W., & Lund, J. R., 1999. "Economic Valuation of Urban Water Use for Large-scale Modeling." Proceedings of the 26th Annual ASCE Water Resources Planning & Management Conference. Phoenix.
- Jenkins, M. W., Lund, J. R., & Howitt, R. E., 2003. "Using Economic Loss Functions to Value Urban Water Scarcity in California." *Journal AWWA*, 58-70.
- M.Cubed, ERA Economics, Roger Mann and Thomas Wegge, 2015. Executive Order B-29-15 State of Emergency Due to Severe Drought Conditions: Economic Impact Analysis. Sacramento: State Water Resources Control Board.
- M.Cubed, RMann Economics, 2016. Proposed Regulatory Framework for Extended Emergency Regulation for Urban Water Conservation: Fiscal and Economic Impact Analysis. Sacramento: State Water Resources Control Board.
- Maggiore, Bill (EBMUD). March 2016. SD-1 Direct Potable Reuse to the Potable Water Distribution System: Conceptual Feasibility Study.
- Oro Loma Sanitary District (OLSD), 2016. *Recycled Water Feasibility Study Final Report.* Completed by RMC Water and Environment and HDR for Oro Loma Sanitary District, October 2016.

- Renwick, Mary and Richard Green, 2000. "Do Residential Water Demand Side Management Policies Measure Up? An Analysis of Eight California Water Agencies." *Journal of Environmental Economics and Management*, 40, 27-55.
- RMC, 2009. Moraga Country Club. Satellite Recycled Water Treatment Plant Detailed Feasibility Study.
- RMC, 2016. Central Contra Costa Sanitary District's Recycled Water Wholesale Opportunities. Final. March 2016.
- San Francisco Bay Regional Water Quality Control Board, 2017. Water Quality Control Plan for the San Francisco Bay Basin
- Sunding, D., 2015. *CalWater Fix Economic Analysis (Draft)*. The Brattle Group. Prepared for California Natural Resources Agency, November 15, 2015.
- Sunding, D., Buck, S., Hatchett, S., & D. G., 2013. *Appendix 9.A Economic Benefits of the BDCP and Take Alternatives, BDCP Public Draft.* Sacramento: California Department of Water Resources.
- SWRCB, 2016a. Investigation on the Feasibility of Developing Uniform water Recycling Criteria for Direct Potable Reuse – Report to the Legislature. Prepared by the California State Water Resources Control Board, December 2016. Available online at https://www.waterboards.ca.gov/drinking_water/certlic/drinkingwater/ documents/rw_dpr_criteria/final_report.pdf
- SWRCB, 2016b. Expert Panel Final Report: Evaluation of the Feasibility of Developing Uniform Water Recycling Criteria for Direct Potable Reuse. Prepared for the California State Water Resources Control Board, June 2016. Available online at <u>https://www.waterboards.ca.gov/drinking_water/certlic/</u><u>drinkingwater/documents/rw_dpr_criteria/app_a_ep_rpt.pdf</u>
- SWRCB, 2016c Recommendations of the Advisory Group on the Feasibility of Developing Uniform Water Recycling Criteria for Direct Potable Reuse. Prepared for the California State Water Resources Control Board, June 2016. Available online at <u>https://www.waterboards.ca.gov/drinking_water/certlic/</u><u>drinkingwater/documents/rw_dpr_criteria/app_b_ag_rpt.pdf</u>
- SWRCB, 2017. Proposed Regulation Text Surface Water Augmentation Using Recycled Water. Available online at http://waterboards.ca.gov/drinking_water/certlic/drinkingwater/documents/swa/draft_swa_reg_text.pdf
- Tognolini, M. Personal Communication. November 8, 2017.
- Trussell et al. 2015. Demonstrating Redundancy and Monitoring to Achieve Reliable Potable Reuse. Direct Potable Reuse in California: Specialty Seminar at UC Berkeley, September 2015.
- Water Environment and Reuse Foundation. 2016. *Guidelines for Engineered Storage for Direct Potable Reuse*. Reuse-12-06. IWA Publishing.
- WateReuse, 2017. "Potable Reuse Chart," March 6, 2017. Available online at <u>https://watereuse.org/wp-content/uploads/2017/03/03-06-17-Potable-Reuse-Chart.pdf</u>
- West Yost Associates, 2017. Oakland Hills Alternative Water Supply Feasibility Study. Completed by West Yost Associates and Whitley Burchett & Associates for EBMUD, April 2017.
- Woodard and Curran, 2017a. *Expedited Purified Water Program, Ford and Coyote Facilities Plan Final Report*. Prepared for Santa Clara Valley Water District, November 2017.
- Woodard and Curran, 2017b. Expedited Purified Water Program, Direct Potable Reuse Facilities Plan Final Report. Prepared for Santa Clara Valley Water District, December 2017.
- Young, R. A., 2005. *Determining the Economic Value of Water: Concepts and Methods.* Washington, DC: Resources for the Future Press.

APPENDIX A: COST ESTIMATE DETAILS FOR NON-POTABLE REUSE ALTERNATIVES

DERWA/San Ramon Valley Recyc	led Water Proj	ect Phase 3	Cost Upd	ate		EBMUD Recy	cled Water N	Master Plan Update
Last Updated:	1-Nov-18					Discount Rate	F	Project Life
Updated by:	C. De Las Casas					3%	3	30 Years
Original Project Information								
Project Source:	DERWA/SRVRWP	Treatment and	d Distribution	Costs	September 201	8)		
Project Source CCI Date:	Ser	otember 2018				Project Source Location:	San Fr	ancisco Area
Project Source 20-Cities Avg ENR CCI:	(11,170				Location Multiplier:		1.00
December 2017 20-Cities Avg ENR CCI		10 870				Historical Multiplier:		0.97
beternber 2017 20 entes Aug Entreen		10,070		Or	iginal Project			0.57
Item	Size	Qty	Unit	01	Cost	Adjusted Project Cost	Useful Life	Salvage Value
Capital Costs								
<u>Treatment</u>								
DERWA Treatment/Supplemental Supply (E	BMUD Share)	1	I LS	\$	2,200,000	\$2,141,000	35 yrs	\$310,000
Conveyance								
SRVRWP Distribution (2.8 miles completed)		14,784 LF		\$	2,600,000	\$2,600,000	75 yrs	\$1,600,000
SRVRWP Distribution (3 miles to be comple	ted)	15,840 LF		\$	14,330,000	\$13,945,000	75 yrs	\$8,400,000
Pump Station								
Pump Station (R3000)				\$	6,958,000	\$6,771,000	50 yrs	\$2,700,000
Estimated Tatal Construction Cost includin		and Contingo				635 000 000		¢12.000.000
estimated Total Construction Cost Includin	ig implementation	and Continger	псу			\$25,000,000 Present Worth of Sa	lvage Value	\$13,000,000
O&M Costs (Annual)							wage value	\$5,500,000
General O&M								
Treatment Share			vr	ć	350,000	\$340,000		
Distribution Share			yı vr	ć	125 000	\$120,000		
			yı	ې	135,000	\$130,000		
Pump Stations		_						
Labor Costs		0	hrs	\$	135	\$0		
Electricity	100,515	kWh	kWh/yr	\$	0.15	\$15,000		
Pump Station Consumables	5%	of pump stati	on constructi	on cost	t	\$0		
			Total O&M	Costs ((\$/yr)	\$490,000		
Annualized Costs (\$ / Year)								
Annualized Capital Costs (\$/Year)	Two payments per	r year, spread o	over Project L	ife		\$1,300,000		
Annualized Salvage Value	Annualized value of	of present wor	th			-\$270,000		
Annual O&M Costs						\$490.000		
Total Annualized Cost						\$1,500,000		
Deliveries of Recycled Water	800 A	FY						
Estimated Unit Cost (\$/AF)						\$1,900		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$550		Distribution	\$1,350		
Annualized Costs (\$ / Year)	Dry Year Adjustm	ent (Supply us	ed 3/10 year:	s)				
Annualized Capital Costs (\$/Year)	Two payments per	r year, spread o	over Project L	ife		\$1,300,000	Same as cons	tant use.
Annualized Salvage Value	Annualized value of	of present work	th			-\$270,000	Same as cons	tant use.
Annual O&M Costs						\$474,500		
Total Annualized Cost						\$1,500,000		
Annual Average Deliveries of Recycled Wate	e 240 A	FY						
Estimated Unit Cost (\$/AF)						\$6,300		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$1,800		Distribution	\$4,500		

DERWA/San Ramon Valley Recycl	ed Water Pro	oject Phase 5	Cost Upd	ate		EBMUD Rec	ycled Water I	Master Plan Update
Last Updated:	1-Nov	-18				Discount Rate		Project Life
Updated by:	C. De Las Ca	sas				3%	:	30 Years
Original Project Information								
Project Source:		EBMUD/DERWA						
Project Source CCI Date:		September 2018			Projec	t Source Location:	San Fi	rancisco Area
Project Source 20-Cities Avg ENR CCI:		11,170			Lo	ocation Multiplier:		1.00
December 2017 20-Cities Avg ENR CCI:		10,870			Hi	storical Multiplier:		0.97
Item	Size	Qty	Unit	Or	iginal Project Cost	Adjusted Project Cost	Useful Life	Salvage Value
Capital Costs								
Treatment								
DERWA Treatment/Supplemental Supply (E	BMUD Share)		1 LS	\$	4,100,000	\$3,989,800	35 yrs	\$570,000
Conveyance								
SRVRWP Distribution (2.8 miles to be comp	le <mark>ted)</mark>	14,784 LF		\$	4,226,800	\$4,113,198	75 yrs	\$2,500,000
Estimated Total Construction Cost including	g Implementatio	on and Contingen	ю		D	\$8,100,000	lvago Valuo	\$3,100,000
O&M Costs (Annual)					P1	esent worth of Sa	ivage value	\$1,500,000
General O&M								
Treatment Share			vr	Ś	130.000	\$126,500		
Distribution Share			vr	Ś	20.000	\$19.500		
			Total O&N	/ Costs	(\$/yr)	\$150,000		
Annualized Costs (\$ / Year)								
Annualized Capital Costs (\$/Year)	Two payments	per year, spread	over Projec	t Life		\$410,000		
Annualized Salvage Value	Annualized val	ue of present wo	rth			-\$66,000		
Annual O&M Costs						\$150,000		
Total Annualized Cost						\$490,000		
Deliveries of Recycled Water	30	0 AFY						
Estimated Unit Cost (\$/AF)						\$1,600		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$1,000		Distribution	\$600		
Annualized Costs (\$ / Year)	Dry Year Adjus	stment (Supply u	sed 3/10 ye	ars)				
Annualized Capital Costs (\$/Year)	Two payments	per year, spread	over Projec	t Life		\$410,000	Same as con	stant use.
Annualized Salvage Value	Annualized val	ue of present wo	rth			-\$66,000	Same as con	stant use.
Annual O&M Costs						\$146,000		
Total Annualized Cost						\$490,000		
Annual Average Deliveries of Recycled Wate	er 90) AFY						
Estimated Unit Cost (\$/AF)						\$5,400		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$3,400		Distribution	\$2,000		

DERWA/San Ramon Valley Recy	cled Water Pro		EBMUD Red	ycled Water I	Master Plan Update			
Last Updated:	1-Nov-18	3				Discount Rate	Ī	Project Life
Updated by:	C. De Las Casas	5				3%	:	30 Years
Original Project Information								
Project Source:	EB	MUD/DERWA						
Project Source CCI Date:	Se	ptember 2018				Project Source Location	: San Fr	rancisco Area
Project Source 20-Cities Avg ENR CCI:		11,170				Location Multiplier		1.00
December 2017 20-Cities Avg ENR CCI:		10,870				Historical Multiplier		0.97
Item	Size	Qty	Unit	Ori	iginal Project Cost	Adjusted Project Cost	Useful Life	Salvage Value
Capital Costs								
Treatment								
DERWA Treatment Plant Expansion (EBM	UD Share)	1	LS		\$2,500,000	\$2,433,000) 35 yrs	\$350,000
Conveyance								
SRVRWP Distribution (1.4 miles to be con	npleted)	7,392 LF		\$	9,606,000	\$9,348,000) 75 yrs	\$5,600,000
Pump Station								
Pump Station (4000)		1	LS	\$	5,787,000	\$5,631,000) 50 yrs	\$2,300,000
Estimated Total Construction Cost includ	ling Implementation	and Continge	ncv			\$17,000,000	,	\$8 300 000
		i unu continge	iley			Present Worth of S	, alvage Value	\$3,400,000
O&M Costs (Annual)								
General O&M								
Treatment Share			yr	\$	130,000	\$127,000)	
Distribution Share			yr	\$	50,000	\$49,000)	
			Total O&M	Costs	(\$/yr)	\$180,000)	
Annualized Costs (\$ / Year)								
Annualized Capital Costs (\$/Year)	Two payments pe	r year, spread	over Project	Life		\$860,000)	
Annualized Salvage Value	Annualized value	of present wor	th			-\$170,000)	
Annual O&M Costs						\$180,000)	
Total Annualized Cost						\$870,000)	
Deliveries of Recycled Water	300 A	AFY						
Estimated Unit Cost (\$/AF)						\$2,900		
Breakdown of Estimated Unit Cost (\$/AF	:)	Treatment	\$400		Distribution	\$2,500		
Annualized Costs (\$ / Year)	Dry Year Adjustm	nent (Supply us	ed 3/10 yea	rs)				
Annualized Capital Costs (\$/Year)	Two payments pe	r year, spread	over Project	Life		\$860,000) Same as con	stant use.
Annualized Salvage Value	Annualized value	of present wor	th			-\$170,000) Same as con	stant use.
Annual O&M Costs						\$176,000)	
Total Annualized Cost						\$870,000)	
Annual Average Deliveries of Recycled W	at 90 A	FY						
Estimated Unit Cost (\$/AF)						\$9,700		
Breakdown of Estimated Unit Cost (\$/AF	:)	Treatment	\$1 200		Distribution	\$8 500		

Fast Bayshore Recycled Water Project P	hase 1A					EBMUD Recvo	led Water M	aster Plan Update
Last Undated	5-Dec-18					Discount Rate		Project Life
Updated by:	M. Romero					3%		30 Years
Item	Size	Qtv	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value
Capital Costs								
Treatment	Cost Estimate	s for Treatment	are based on Uni	t Costs	for Each Treat	ment Process		
RO and MF System (0.77MGD Max Month)	0.77		MGD	\$	2,031,451	\$2,000,000	35 yrs	\$290,000
IX System	0.77		MGD	\$	421,757	\$400,000	35 yrs	\$57,000
Electrical	1		LS	\$	258,886	\$260,000	35 yrs	\$37,000
Sitework	1		LS	\$	466,323	\$470,000	35 yrs	\$67,000
			Raw Constru	uction	Cost Subtotal	\$3,100,000		\$450,000
Contractor Overhead & Profit					15%	\$470,000		\$68,000
Sales Tax	50%	% of Subtotal Cos	t Applicable		9%	\$140,000		\$20,000
		Estimated Ins	talled Equipment	t Cost f	or Treatment	\$3,700,000		\$540,000
Construction Cost Including Contingency								
		I	Estimated Subtot	al Cons	truction Cost	\$3,700,000		\$540,000
Estimating Contingency					25%	\$930,000		\$140,000
		Estimated	Subtotal Cost in	cluding	Contingency	\$4,600,000		\$680,000
Capital Costs		6 8 1		10				
<u>Conveyance</u>	Lost Estimate	s for Pipelines a	in LE	JD cons	struction bia c	contingency	75	ć1 400 000
High-Density Urban Pipeline	12-in	4,000 LF	in-LF	ڊ خ	50	\$2,400,000	75 yrs	\$1,400,000
High-Density Orban Pipeline	10-10	3,000 LF	IN-LF	ې د	100.000	\$2,400,000	75 yrs	\$1,400,000
Industrial Customer Connection Costs		0	EA	ڊ خ	200,000	\$500,000	75 yrs	\$360,000
Extra pipeline allowance for industrial customers		0	EA	د خ	50,000	\$1,000,000	75 yrs	\$340,000
Extra pipeline allowance for industrial customers		0	LA	ç	30,000	\$400,000	75 yrs	\$240,000
		Estimated Su	btotal Cost (2) in	cluding	Contingency	\$7,400,000		\$4,400,000
Mobilization								
Estimated Subtotal Construction Cost including Co	ntingency					\$12,000,000		\$5,100,000
Mobilization					5%	\$600,000		
Estimated Subtotal Construction Cost including Co	ntingency and	Mobilization				\$12,600,000		\$5,100,000
Implementation								
Planning / Environmental					5%	\$630,000		
Design Cost					15%	\$1,900,000		
Project Administration and Construction Managem	ent Cost	. .:			10%	\$1,300,000		47 400 000
Estimated Total Construction Cost including Implei	mentation and	Contingency				\$16,430,000		\$5,100,000
						Present worth of Sa	vage value	\$2,100,000
O&IVI Costs (Annual) Recycled Water Treatment								
RO Electricity	0 44	615 805	kWh/vr	Ś	0 15	\$92 400		
IX Electricity	0.44	107.861	kWh/yr	ś	0.15	\$16,200		
Chemicals	0.44	- ,	MGD	\$	33,843	\$33,800		
Replacement Costs	0.77		MGD	\$	68,638	\$68,600		
Labor	1.0	1,040	hrs/MGD	\$	135	\$140,400		
Conveyance								
Annual O&M			2% of constructi	on cost	t	\$148.000		
			Total O&M Cost	s (\$/yr)	\$499,000		
Annualized Costs (\$ / Year)				- (<i>11</i> /)	<u> </u>	1		
Annualized Capital Costs (\$/Year)	Two payment	s per year, spred	nd over Project Lif	e		\$830,000		
Annualized Salvage Value	Annualized vo	alue of present w	vorth			-\$110,000		
Annual O&M Costs						\$499,000		
Total Annualized Cost						\$1,200,000		
Deliveries of Recycled Water	49	3 AFY						
Estimated Unit Cost (\$/AF)						\$2,400		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$1,300		Distribution	\$1,100		
Annualized Costs (\$ / Year)	Dry Year Adj	ustment (Supply	used 3/10 years)					
Annualized Capital Costs (\$/Year)	Two payment	s per year, spred	nd over Project Lif	e		\$830,000	Same as con:	stant use.
Annualized Salvage Value	Annualized vo	alue of present w	vorth			-\$110,000	Same as con	stant use.
Annual O&M Costs						\$389,580		
Total Annualized Cost						\$1,100,000		
Annual Average Deliveries of Recycled Water	14	8 AFY						
Estimated Unit Cost (\$/AF)						\$7,400		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$3,700		Distribution	\$3,700		

Fast Bayshore Recycled Water Project P	hase 1B					EBMUD Recvo	led Water M	aster Plan Update
Last Undated	5-Dec-18					Discount Rate		Project Life
Updated by:	M. Romero					3%		30 Years
Item	Size	Qtv	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value
Capital Costs								
Treatment	Cost Estimates	for Treatment ar	e based on Unit C	osts fo	r Each Treatm	nent Process		
MBR System 1.6 MGD (Max Month) capacity	1.6		MGD	\$	3,098,474	\$3,100,000	35 yrs	\$440,000
Diversion Pump Station	1.6		MGD	\$	1,300,000	\$1,300,000	50 yrs	\$520,000
Electrical Allowance	1		MGD	\$	485,487	\$490,000	35 yrs	\$70,000
Sitework	1		LS	\$	1,471,949	\$1,470,000	35 yrs	\$210,000
Chemicals (Storage and Use)	1		LS	\$	32,789	\$33,000	30 yrs	\$0
			Raw Constru	iction (Cost Subtotal	\$6,400,000		\$1,200,000
Contractor Overhead & Profit					15%	\$960,000		\$180,000
Sales Tax	50%	% of Subtotal Cost	Applicable		9%	\$290,000		\$54,000
		Estimated Ins	talled Equipment	Cost fo	or Treatment	\$7,700,000		\$1,400,000
Construction Cost Including Contingency								
		E	stimated Subtota	l Cons	truction Cost	\$7,700,000		\$1,400,000
Estimating Contingency					25%	\$1,900,000		\$350,000
		Estimated	Subtotal Cost inc	luding	Contingency	\$9,600,000		\$1,800,000
Capital Costs								
Conveyance	Cost Estimates	for Pipelines are	based on EBMUD	constr	uction bid cos	st + estimating contingency		
High-Density Urban Pipeline	6-in	4,900 LF	in-LF	\$	50	\$1,470,000	75 yrs	\$880,000
High-Density Urban Pipeline	8-in	13,300 LF	in-LF	\$	50	\$5,320,000	75 yrs	\$3,200,000
High-Density Urban Pipeline	12-in	4,000 LF	in-LF	\$	50	\$2,400,000	75 yrs	\$1,400,000
High-Density Urban Pipeline	16-in	3,000 LF	in-LF	\$	50	\$2,400,000	75 yrs	\$1,400,000
Irrigation Customer Connection Costs		10	EA	\$	100,000	\$1,000,000	75 yrs	\$600,000
Industrial Customer Connection Costs		13	EA	\$	200,000	\$2,600,000	75 yrs	\$1,600,000
Pipeline Crossing Allowance(Brooklyn Basin and								
Schnitzer)		1	LS	\$	2,700,000	\$2,700,000	75 yrs	\$1,600,000
Extra pipeline allowance for industrial customers		13	EA	\$	50,000	\$650,000	75 yrs	\$390,000
		Estimated Su	btotal Cost (2) inc	luding	Contingency	\$19,000,000		\$11,000,000
Mobilization								
Estimated Subtotal Construction Cost including Cor	ntingency					\$29,000,000		\$13,000,000
Mobilization	0,				5%	\$1,500,000		
Estimated Subtotal Construction Cost including Cor	tingency and M	obilization				\$31,000,000		\$13,000,000
Implementation	0,							
Planning / Environmental					5%	\$1,600,000		
Design Cost					15%	\$4,700,000		
Project Administration and Construction Manageme	ent Cost				10%	\$3,100,000		
Estimated Total Construction Cost including Implen	nentation and C	ontingency				\$40,000,000		\$13,000,000
						Present Worth of Sa	lvage Value	\$5,300,000
O&M Costs (Annual)								
Recycled Water Treatment								
MBR Electricity	0.95	778,571	kWh/yr	\$	0.15	\$116,800		
Free Chlorine	0.95		MGD	\$	11,000	\$11,000		
Replacement Costs	1.6		MGD	\$	28,075	\$28,100		
Labor	1.6	1,040	hrs/MGD	\$	135	\$224,600		
Conveyance								
Annual O&M			2% of construction	on cost		\$370,000		
Pumn Stations								
Labor Costs		0	hrs	¢	135	ŚO		
Electricity		72 623	kWh/vr	Ś	0.15	\$10,900		
Pump Station Consumables		5% of nump stat	tion construction	rost	0.15	\$65,000		
		570 OF pump star	Total O&M Costs	: (\$/vr)		\$830,000		
Annualized Costs (\$ / Year)				, (<i>Y</i> / y ·)	<u>'</u>	\$650,000		
Annualized Costs (\$7 Teal)	Two navments	ner vear spread	over Project Life			\$2,000,000		
Annualized Salvage Value	Annualized valu	ie of present wor	th			-\$270,000		
Annual O&M Costs	/ Innaunzeu vuie					\$830,000		
Total Annualized Cost						\$2,600,000		
Deliveries of Recycled Water	1.06	4 AFY				\$2,000,000		
Estimated Unit Cost (\$/AE)	2,00					\$2,400		
Breakdown of Estimated Unit Cost (\$/AE)		Treatment	\$940		Distribution	\$1,460		
Appualized Costs (\$ / Year)	Dry Yoar Adius	tmont (Sunnly us	ad 2/10 years)		Distribution	ļ1, 4 00		
Annualized Costs (\$7 fear)	Two novmonts	nor yogr coroad	eu 3/10 yeurs			¢2,000,000	Samo as con	stant uso
Annualized Capital Costs (2/ Tedi) Annualized Salvage Value	Annualized value	per year, spreau ip of procent way	th			ېدىن محدغ	Same as con	stant use
Annual O&M Costs	,uuiizeu vuit	ie oj present wor				-3210,000	June us con	stallt use.
						¢2 500 000		
Annual Average Deliveries of Recycled Water	210	AFY				\$2,300,000		
Estimated Unit Cost (\$/AE)	515					\$7.800		
Breakdown of Estimated Unit Cost (\$/AE)		Treatment	\$2.900		Distribution	\$4,900		
Si calla Still of Estimated OfficeOst (S/A)			-,			94,500		

East Bayshore Recycled Water Project Pha	ise 2					EBMUD Recy	cled Water M	aster Plan Update
Last Updated:	5-Dec-18					Discount Rate	F	Project Life
Updated by:	M. Romero					3%	3	30 Years
Item	Size	Qty	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value
Capital Costs								
Treatment	Cost Estimate	es for Treatment o	are based on Unit Co	osts fo	r Each Treatmen	t Process		
MBR System (4.5MGD MMcapacity)	4.5		MGD	Ş	5,979,853	\$6,000,000	35 yrs	\$860,000
Diversion Pump Station	4.5		MGD	Ş	1,419,005	\$1,400,000	50 yrs	\$560,000
Electrical Allowance	1		LS	Ş ¢	601,543 2 040 207	\$600,000	35 yrs	\$86,000 \$200,000
Chemicals (Storage and Lise)	1		15	ç	72 879	\$2,000,000	30 yrs	\$290,000 \$0
Chlorine Contactor	1		LS	Ś	686.063	\$700.000	35 yrs	\$100.000
			Raw Const	ructio	n Cost Subtotal	\$11,000,000		\$1,900,000
Contractor Overhead & Profit					15%	\$1,700,000		\$290,000
Sales Tax	50%	% of Subtotal Cost	Applicable		9%	\$500,000		\$86,000
		Estimated	Installed Equipmer	nt Cos	t for Treatment	\$13,000,000		\$2,300,000
Construction Cost Including Contingency								
			Estimated Subto	tal Co	nstruction Cost	\$13,000,000		\$2,300,000
Estimating Contingency		-	C		25%	\$3,300,000		\$580,000
Consided Consta (from Dide)		Estimated	Subtotal Cost (1) in	ncludi	ng Contingency	\$16,000,000		\$2,900,000
Capital Costs (from Bids)	Cost Estimate	s for Dinalinas ar	hacad on ERMUD	constr	uction hid cost u	octimating contingoncy		
<u>Conveyance</u> High-Density Urban Pineline	Cost Estimate	A 900 LE	in-I F	constr ¢	50 uction bid cost	stimuling conlingency	75 vrs	\$880.000
High-Density Urban Pipeline	8-in	58 214 LF	in-LF	Ś	50	\$23,285,600	75 yrs	\$14,000,000
High-Density Urban Pipeline	12-in	16.670 LF	in-LF	Ś	50	\$10.002.000	75 yrs	\$6.000.000
High-Density Urban Pipeline	16-in	8,000 LF	in-LF	\$	50	\$6,400,000	75 yrs	\$3,800,000
High-Density Urban Pipeline	20-in	21,285 LF	in-LF	\$	50	\$21,285,000	, 75 yrs	\$13,000,000
Irrigation Customer Connection Costs		68	EA	\$	100,000	\$6,800,000	75 yrs	\$4,100,000
Industrial Customer Connection Costs		18	EA	\$	200,000	\$3,600,000	75 yrs	\$2,200,000
Pipeline Crossing Allowance(Brooklyn Basin and								
Schnitzer)		1	LS	\$	2,700,000	\$2,700,000	75 yrs	\$1,600,000
Extra pipeline allowance for industrial customers		18	EA	\$	50,000	\$900,000	75 yrs	\$540,000
Pump Stations	Cost Estimate	es for Pump Statio	ons are based on EBI	NUD (Construction Bid	Costs		
Rump Station (Total installed HR including standby)		EA (Cost Curv	ve based on total					
Pump station (rotal installed HP, including standby)	90 HP	insta	alled HP)	\$	1,500,000	\$1,500,000	75 yrs	\$900,000
		Estimated	Subtotal Cost (2) in	ncludi	ng Contingency	\$78,000,000		\$47,000,000
Mobilization								
Estimated Subtotal Construction Cost including Conti	ngency					\$94,000,000		\$50,000,000
Mobilization					5%	\$4,700,000		
Estimated Subtotal Construction Cost including Conti	ngency and M	obilization				\$99,000,000		\$50,000,000
Implementation								
Planning / Environmental					5%	\$5,000,000		
Design Cost	Cost				15%	\$15,000,000 \$0,000,000		
Estimated Total Construction Cost including Implement	ntation and C	ontingency			10%	\$130,000,000		\$50.000.000
		ontingency				Present Worth of Sa	alvage Value	\$20,000,000
O&M Costs (Annual)								
Recycled Water Treatment								
MBR	2.5	3,024,140	kWh/yr	\$	0.15	\$453,600		
Free Chlorine	2.5		MGD	\$	52,900	\$52,900		
Replacement Costs	4.5	4 0 4 0	MGD	Ş	76,118	\$80,000		
Labor	4.5	1,040	nrs/WGD	Ş	135	\$631,800		
Conveyance								
Annual O&M			2% of construction	o cost		\$1,530,000		
Pump Stations								
Labor Costs		0	hrs	\$	135	\$0		
Electricity		347,925	kWh/yr	\$	0.15	\$52,200		
Pump Station Consumables		5% of pump s	tation construction	cost		\$70,000		
			Total O&M Costs	Ş/yr)		\$2,900,000		
Annualized Costs (\$ / Year)	T					¢c coo ooo		
Annualized Capital Costs (\$/ Year)	Appualized w	is per year, spread	over Project Lije			\$0,000,000 \$1,000,000		
Annual O&M Costs	Annaunzeu vi	and of present we				\$2,900,000		
Total Annualized Cost						\$8,500,000		
Deliveries of Recycled Water	2.8	67 AFY				\$0,500,000	•	
Estimated Unit Cost (\$/AF)	,-	-				\$3,000		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$800		Distribution	\$2,200		
Annualized Costs (\$ / Year)	Drv Year Adi	ustment (Supply)	used 3/10 vears)					
Annualized Capital Costs (\$/Year)	Two payment	ts per year, spread	d over Project Life			\$6,600,000	Same as cons	tant use.
Annualized Salvage Value	Annualized v	alue of present wo	orth			-\$1,000,000	Same as cons	tant use.
Annual O&M Costs						\$2,479,410		
Total Annualized Cost						\$8,100,000		
Annual Average Deliveries of Recycled Water	86	50 AFY				· · · · · · · · · · · · · · · · · · ·		
Estimated Unit Cost (\$/AF)						\$9,400		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$2,200		Distribution	\$7,200		

Chevron/City of Richmond WPCP Recycled Water Project Cost Update						FBMUD	Recycled Water Ma	aster Plan Undate
Last Lindated:	9-Jul-19		uute			Discount Bate	Dr	oject Life
Last opdated.						2%	20	Voars
Original Project Information	C. De Las Casas	5				570	30	Tears
Project Source	2016 City of	Richmond Faci	lity Plan					
Project Source CCI Date:	Sen	tember 2016	incy i luit			Project Source Location:	San Franci	isco Area
Project Source 20-Cities Avg ENB CCI:	566	10 403				Location Multinlier:	1 (0
December 2017 20-Cities Avg ENR CCI		10,405				Historical Multiplier:	1.0	04
								-
Item	Size	Qty	Unit	Origi	nal Project Cost	Adjusted Project Cost	Useful Life	Salvage Value
Capital Costs								
Treatment								
MBR (2016 Facilities Plan)	5.0		MGD	\$	62,600,000	\$65,000,000	35 yrs	\$9,300,000
UV Disinfection	5.0		MGD	\$	8,900,000	\$9,300,000	35 yrs	\$1,300,000
RO System	5.0		MGD	\$	19,700,000	\$21,000,000	35 yrs	\$3,000,000
RARE MF/RO expansion from 3.5 to 5 mgd	1.5		MGD	\$	2,000,000	\$2,100,000	35 yrs	\$300,000
Conveyance								. ,
Pipeline to RARE	24in	12,850 LF	in-LF	\$	9,800,000	\$10,200,000	75 yrs	\$6,100,000
Pump Station				Ŧ		÷ -		÷ -
Pump Station	200 HP	5	MGD	\$	1,600,000	\$1,700,000	50 yrs	\$680,000
Estimated Total Capital Cost including Implement	tation and Continger	ncy				\$110,000,000		\$21,000,000
	-					Present Worth o	f Salvage Value	\$8,600,000
O&M Costs (Annual)								
Recycled Water Treatment								
MBR	5.0	\$ 460.000	\$/MGD	Ś	2,300,000	\$2,300,000		
UV maintenance (2% UV const_costs)	5.0	+,	MGD	Ś	112 700	\$118,000		
BO	5.0	\$ 480,000	\$/MGD	Ś	2 400 000	\$2 400 000		
RARE MF/RO Upgrades	1.5	÷ +00,000	MGD	\$	450,000	\$470,000		
Convoyance (Dinaline and Dumn Station)								
		20/	of constructiv	an cost		\$200.000		
Allidar Galvi		278		Shicost		φ200,000		
Pump Stations								
Labor Costs (5MGD)	3475 gpm	1,000	hrs	\$	135	\$140,000		
Electricity (143,600 KWh/MGD)	4.5 MGD	646,200	kWh/yr	\$	0.15	\$97,000		
Pump Station Consumables	5%	of pump stati	on constructi	on cost		\$0		
			Total O&M	Costs (\$/yr)	\$5,700,000		
Annualized Costs (\$ / Year)								
Annualized Capital Costs (\$/Year)	Two payments per	year, spread o	ver Project Li	fe		\$5,600,000		
Annualized Salvage Value	Annualized value o	f present wort	h			-\$440,000		
Annual O&M Costs						\$5,700,000		
Total Annualized Cost						\$11,000,000		
Deliveries of Recycled Water	4,284 A	AFY						
Estimated Unit Cost (\$/AF)						\$2,600		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$2,300		Distribution	\$300		
Annualized Costs (\$ / Year)	Dry Year Adjustme	nt (Supply use	ed 3/10 years)				
Annualized Capital Costs (\$/Year)	Two payments per	year, spread o	ver Project Li	fe		\$5,600,000	Same as constant u	ise.
Annualized Salvage Value	Annualized value o	f present wort	h			-\$440,000	Same as constant u	ise.
Annual O&M Costs						\$5,657,100		
Total Annualized Cost						\$11,000,000		
Annual Average Deliveries of Recycled Water	1,285 4	AFY						
Estimated Unit Cost (\$/AF)						\$8,600		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$7,600		Distribution	\$1,000		

Phillips CC Definers Desurated Mater D	national Count Line	lata				EBAALID	Decusied Motors Me	atos Dios Lindoto
Phillips 66 Kennery Recycled Water P	roject Cost Opd	ate				EBINIOD	Recycleu water Ma	ster Plan Opuate
Last Updated:	18-May-18					Discount Rate	Pro	ject Life
Updated by:	M. Romero					3%	30	Years
Driginal Project Information	Degualed Wate	ar Tachnical C	tudu 2007					
Project Source:	Recycled Wall	er recrinical s	tudy, 2007			Draiget Course Leastion	Con Francia	A 100
Project Source CCI Date:	A					Project Source Location:	Sdil Francis	co Area
Project Source 20-Cities Average ENR CCI:		8,007				Historical Multiplier:	1.00	
December 2017 20-Cities Average Livit CCI.		10,870		0	riginal Project	riistoricar wurtiplier.	1.50)
Item	Size	Qty	Unit		Cost	Adjusted Project Cost	Useful Life	Salvage Value
Capital Costs								
Treatment								
BAF/MF/RO Treatment	1.7	1	MGD	\$	14,012,600	\$19,000,000	35 yrs	\$2,700,000
UV Treatment	1.7	1	MGD	\$	297,700	\$400,000	35 yrs	\$57,000
Electrical/Instrumentation		1	LS	\$	2,783,400	\$3,800,000	35 yrs	\$540,000
Conveyance								
Pipeline to Refinery	2"-12" Dia	3,565 LF	in-LF	\$	618,200	\$840,000	75 yrs	\$500,000
Dump Station								
Padeo Rump Station	6140			ć	1 222 744	\$1,800,000	50 yrs	\$720.000
Roued Pullip Station	OTHP		Raw Const	ې tructic	1,555,744	\$1,800,000	50 yrs	\$720,000
Contractor Overhead & Profit			Naw Cons	ucuc	15%	\$3,900,000		\$680,000
	50%	% of Subtotal			9%	\$3,500,000		\$200,000
	50/8	Fst	timated Subt	ntal Co	nstruction Cost	\$1,200,000		\$5 400 000
Construction Cost Including Contingency		23				\$51,000,000		<i>\$3,400,000</i>
		Fst	timated Subt	otal Co	Instruction Cost	\$31,000,000		\$5,400,000
Estimating Contingency		20			25%	\$7.800.000		\$1,400,000
	Estimated S	ubtotal Const	ruction Cost i	includ	ing Contingency	\$39.000.000		\$6,800,000
Mobilization					5%	\$2.000.000		+-,,
Estimated	Subtotal Construct	ion Cost inclu	ding Conting	encv a	nd Mobilization	\$41.000.000		\$6.800.000
Construction Cost Including Implementation			0.000			1 ,,.		. , , , ,
Planning / Environmental					5%	\$2,100,000		
Design Cost					15%	\$6,200,000		
Project Administration and Construction Manage	ement Cost				10%	\$4,100,000		
Estimated Total Construction Cost including Im	plementation and O	Contingency				\$53,000,000		\$6,800,000
						Present Worth o	f Salvage Value	\$2,800,000
O&M Costs (Annual)								
Recycled Water Treatment								
BAF/MF/RO/UV System Energy			EA	\$	327,400	\$613,900		
Replacement and Maintenace			EA	\$	549,756	\$746,300		
Miscellaneous, chemicals			EA	\$	175,000	\$237,600		
Labor Costs			EA	\$	133,000	\$394,700		
Duran Chatlana								
Labor Costs						ćo		
Electricity		1	E۸	ć	22.000	ېن د د ۵ مې		
Electricity Rump Station Consumables	5%	of numn stat	. EA	ç on cor	32,000	\$60,000 \$0		
	576	or pump stat	Total OSM	Costs	(¢/\/~)	ېر د 100 000		
Annualized Costs (\$ / Year)			Total Oxivi	COSIS	(\$/ ¥1)	\$2,100,000		
Annualized Costs (\$7 Teal)	Two navments ner	vear spread	over Project I	ife		\$2,700,000		
Annualized Capital Costs (3/ ICal)	Annualized value of	f nresent wor	th	ŋс		\$2,700,000 -\$140,000		
Annual O&M Costs	Annualized value o	j present wor	chi i			\$2 100 000		
Total Annualized Cost						\$4 700 000		
Deliveries of Recycled Water	4.144	AFY				\$4,700,000		
Estimated Unit Cost (\$/AF)	.,	<u></u>				\$1,100		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$1,000		Distribution	\$100		
Annualized Costs (\$ / Year)	Dry Year Adjustm	ent (Supply u	sed 3/10 year	s)				
Annualized Capital Costs (\$/Year)	Two payments per	year, spread	over Project L	ife		\$2,700,000 \$	Same as constant use	2.
Annualized Salvage Value	Annualized value o	of present wor	th			-\$140,000 \$	Same as constant use	e.
Annual O&M Costs						\$2,010,500		
Total Annualized Cost						\$4,600,000		
Annual Average Deliveries of Recycled Water	1,243 /	AFY						
Estimated Unit Cost (\$/AF)						\$3,700		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$3,500		Distribution	\$200		

Central San Regional Recycled Water Proje	ect Cost Update	2				EBMUD Rec	ycled Water I	Master Plan Update
Last Updated:	23-Mav-18	•				Discount Rate	F	Proiect Life
Updated by:	C. De Las Casas					3%	3	30 Years
Original Project Information								
Project Source:	CCCSD Comp	orehensive Waster	water Master Plan					
Project Source CCI Date:		June 2017				Project Source Location:	San Fr	ancisco Area
Project Source 20-Cities Average ENR CCI:		10,703				Location Multiplier:		1.00
December 2017 20-Cities Average ENR CCI:		10,870				Historical Multiplier:		1.02
Item	Size	Qty	Unit	Ori	iginal Project Cost	Adjusted Project Cost	Useful Life	Salvage Value
Capital Costs								
Treatment								
MBR (Aeration Basins, Membrane Tanks, Fine Screens)	23		MGD	\$	190 423 000	\$190,000,000	35 vrs	\$27 000 000
UV Disinfection	23		MGD	Ś	21 876 000	\$22,000,000	35 yrs	\$3 100 000
RO	14		MGD	Ś	56.813.000	\$58.000.000	35 yrs	\$8,300,000
Electrical Feed Upgrade	1		LS	\$	17,303,000	\$18,000,000	75 yrs	\$11,000,000
Conveyance								
Pipeline R&R			LS	\$	28,659,000	\$29,000,000	75 yrs	\$17,000,000
					Subtotal	\$320.000.000		\$66.000.000
Contractor Overhead & Profit	Apply as needed				15%	\$0		\$0
Sales Tax	50%	% of Subtotal Cost	Applicable		9%	\$0		\$0
			Estimated Subto	tal Co	nstruction Cost	\$320,000,000		\$66,000,000
Construction Cost Including Contingency								
			Estimated Subto	tal Co	nstruction Cost	\$320,000,000		\$66,000,000
Estimating Contingency	Apply as needed				25%	\$0		\$0
		Estim	ated Subtotal Cost (1) i	ncludi	ng Contingency	\$320,000,000		\$66,000,000
Construction Cost Including Implementation								
Estimated Total Construction Cost including Implemen	tation and Conting	ency				\$320,000,000		\$66,000,000
ORAA Conto (Annual)						Present Worth of Sa	ivage value	\$27,000,000
O&IVI Costs (Annual)								
Conoral O&M								
General O&M Transmost O&M (\$114 500/MGD)	20		MGD	ć	114 500	\$2,200,000		
General O&M Treatment O&M (\$114,500/MGD) Refinery O&M (\$260,600/MGD)	20		MGD MGD	\$	114,500	\$2,300,000		
<u>General O&M</u> Treatment O&M (\$114,500/MGD) Refinery O&M (\$260,600/MGD)	20 20		MGD MGD	\$ \$	114,500 260,600	\$2,300,000 \$5,200,000		
General O&M Treatment O&M (\$114,500/MGD) Refinery O&M (\$260,600/MGD) Conveyance	20 20		MGD MGD	\$ \$	114,500 260,600	\$2,300,000 \$5,200,000		
General O&M Treatment O&M (\$114,500/MGD) Refinery O&M (\$260,600/MGD) Conveyance Annual O&M	20 20 1		MGD MGD LS	\$ \$ \$	114,500 260,600 108,100	\$2,300,000 \$5,200,000 \$109,800		
General O&M Treatment O&M (\$114,500/MGD) Refinery O&M (\$260,600/MGD) Conveyance Annual O&M Pump Stations	20 20 1		MGD MGD LS	\$ \$ \$	114,500 260,600 108,100	\$2,300,000 \$5,200,000 \$109,800		
General O&M Treatment O&M (\$114,500/MGD) Refinery O&M (\$260,600/MGD) Conveyance Annual O&M Pump Stations Labor Costs	20 20 1		MGD MGD LS hrs	\$ \$ \$	114,500 260,600 108,100 135	\$2,300,000 \$5,200,000 \$109,800 \$0		
General O&M Treatment O&M (\$114,500/MGD) Refinery O&M (\$260,600/MGD) Conveyance Annual O&M Pump Stations Labor Costs Electricity	20 20 1	10,320,178	MGD MGD LS hrs kWh/yr	\$ \$ \$ \$ \$	114,500 260,600 108,100 135 0.15	\$2,300,000 \$5,200,000 \$109,800 \$0 \$1,500,000		
General O&M Treatment O&M (\$114,500/MGD) Refinery O&M (\$260,600/MGD) Conveyance Annual O&M Pump Stations Labor Costs Electricity Pump Station Consumables	20 20 1	10,320,178 5% of pump st	MGD MGD LS hrs kWh/yr ation construction cost	\$ \$ \$ \$ \$	114,500 260,600 108,100 135 0.15	\$2,300,000 \$5,200,000 \$109,800 \$1,500,000 \$0 \$0		
General O&M Treatment O&M (\$114,500/MGD) Refinery O&M (\$260,600/MGD) Conveyance Annual O&M Pump Stations Labor Costs Electricity Pump Station Consumables	20 20 1	10,320,178 5% of pump st	MGD MGD LS hrs kWh/yr ation construction cost Total 0&M Costs (\$/y	\$ \$ \$ \$ r)	114,500 260,600 108,100 135 0.15	\$2,300,000 \$5,200,000 \$109,800 \$1,500,000 \$1,500,000 \$0 \$9,100,000		
General O&M Treatment O&M (\$114,500/MGD) Refinery O&M (\$260,600/MGD) Conveyance Annual O&M Pump Stations Labor Costs Electricity Pump Station Consumables Annualized Costs (\$ / Year)	20 20 1	10,320,178 5% of pump st	MGD MGD LS hrs kWh/yr ation construction cost Total O&M Costs (\$/y	\$ \$ \$ \$ r)	114,500 260,600 108,100 135 0.15	\$2,300,000 \$5,200,000 \$109,800 \$0 \$1,500,000 \$0 \$9,100,000		
General O&M Treatment O&M (\$114,500/MGD) Refinery O&M (\$260,600/MGD) Conveyance Annual O&M Pump Stations Labor Costs Electricity Pump Station Consumables Annualized Costs (\$ / Year) Annualized Capital Costs (\$ /Year)	20 20 1 Two payments per	10,320,178 5% of pump st year, spread over	MGD MGD LS hrs kWh/yr ation construction cost Total O&M Costs (\$/y Project Life	\$ \$ \$ \$ r)	114,500 260,600 108,100 135 0.15	\$2,300,000 \$5,200,000 \$109,800 \$0 \$1,500,000 \$0 \$9,100,000 \$16,000,000		
General O&M Treatment O&M (\$114,500/MGD) Refinery O&M (\$260,600/MGD) Conveyance Annual O&M Pump Stations Labor Costs Electricity Pump Station Consumables Annualized Costs (\$ / Year) Annualized Capital Costs (\$/Year) Annualized Salvage Value	20 20 1 Two payments per Annualized value c	10,320,178 5% of pump st year, spread over f present worth	MGD MGD LS hrs kWh/yr ation construction cost Total O&M Costs (\$/y Project Life	\$ \$ \$ \$ r)	114,500 260,600 108,100 135 0.15	\$2,300,000 \$5,200,000 \$109,800 \$0 \$1,500,000 \$9,100,000 \$16,000,000 -\$1,400,000		
General O&M Treatment O&M (\$114,500/MGD) Refinery O&M (\$260,600/MGD) Conveyance Annual O&M Pump Stations Labor Costs Electricity Pump Station Consumables Annualized Costs (\$ / Year) Annualized Costs (\$ / Year) Annualized Salvage Value Annual O&M Costs	20 20 1 Two payments per Annualized value o	10,320,178 5% of pump st year, spread over f present worth	MGD MGD LS hrs kWh/yr ation construction cost Total O&M Costs (\$/y Project Life	\$ \$ \$ \$ r)	114,500 260,600 108,100 135 0.15	\$2,300,000 \$5,200,000 \$109,800 \$0 \$1,500,000 \$9,100,000 \$16,000,000 -\$1,400,000 \$9,100,000		
General O&M Treatment O&M (\$114,500/MGD) Refinery O&M (\$260,600/MGD) Conveyance Annual O&M Pump Stations Labor Costs Electricity Pump Station Consumables Annualized Costs (\$ / Year) Annualized Costs (\$ /Year) Annualized Salvage Value Annual Capital Costs (\$/Year) Annualized Salvage Value Annual Capital Costs (\$/Year) Annual Capital Costs (\$/Year)	20 20 1 Two payments per Annualized value o	10,320,178 5% of pump st year, spread over f present worth	MGD MGD LS hrs kWh/yr ation construction cost Total O&M Costs (\$/y Project Life	\$ \$ \$ \$ r)	114,500 260,600 108,100 135 0.15	\$2,300,000 \$5,200,000 \$109,800 \$1,500,000 \$9,100,000 \$16,000,000 -\$1,400,000 \$9,100,000 \$24,000,000		
General O&M Treatment O&M (\$114,500/MGD) Refinery O&M (\$260,600/MGD) Conveyance Annual O&M Pump Stations Labor Costs Electricity Pump Station Consumables Annualized Costs (\$/Year) Annualized Capital Costs (\$/Year) Annualized Costs Total Annualized Cost Deliveries of Recycled Water	20 20 1 Two payments per Annualized value of 22,40	10,320,178 5% of pump st year, spread over f present worth 0 AFY	MGD MGD LS hrs kWh/yr ation construction cost Total 0&M Costs (\$/y Project Life	\$ \$ \$ r)	114,500 260,600 108,100 135 0.15	\$2,300,000 \$5,200,000 \$109,800 \$1,500,000 \$9,100,000 \$16,000,000 -\$1,400,000 \$9,100,000 \$24,000,000		
General O&M Treatment O&M (\$114,500/MGD) Refinery O&M (\$260,600/MGD) Conveyance Annual O&M Pump Stations Labor Costs Electricity Pump Station Consumables Annualized Costs (\$/Year) Annualized Costs (\$/Year) Annualized Capital Costs (\$/Year) Annualized Capital Costs (\$/Year) Annualized Capital Costs (\$/Year) Annualized Capital Costs (\$/Year) Deliveries of Recycled Water Estimated Unit Cost (\$/AF)	20 20 1 Two payments per Annualized value of 22,40	10,320,178 5% of pump st year, spread over f present worth 0 AFY	MGD MGD LS hrs kWh/yr ation construction cost Total O&M Costs (\$/y	\$ \$ \$ r)	114,500 260,600 108,100 135 0.15	\$2,300,000 \$5,200,000 \$109,800 \$1,500,000 \$1,500,000 \$9,100,000 \$9,100,000 \$24,000,000 \$24,000,000 \$11,100		
General O&M Treatment O&M (\$114,500/MGD) Refinery O&M (\$260,600/MGD) Conveyance Annual O&M Pump Stations Labor Costs Electricity Pump Station Consumables Annualized Costs (\$/Year) Annualized Capital Costs (\$/Year) Annualized Capital Costs (\$/Year) Annualized Salvage Value Annual O&M Costs Total Annualized Cost Deliveries of Recycled Water Estimated Unit Cost (\$/AF) Breakdown of Estimated Unit Cost (\$/AF)	20 20 1 Two payments per Annualized value of 22,40	10,320,178 5% of pump st year, spread over f present worth 0 AFY Treatment	MGD MGD LS hrs kWh/yr ation construction cost Total O&M Costs (\$/y Project Life \$980	\$ \$ \$ \$ r)	114,500 260,600 108,100 135 0.15 	\$2,300,000 \$5,200,000 \$109,800 \$1,500,000 \$1,500,000 \$9,100,000 \$9,100,000 \$9,100,000 \$24,000,000 \$24,000,000 \$11,100 \$120		
General O&M Treatment O&M (\$114,500/MGD) Refinery O&M (\$260,600/MGD) Conveyance Annual O&M Pump Stations Labor Costs Electricity Pump Station Consumables Annualized Costs (\$ / Year) Annualized Costs (\$ / Year) Annualized Costs (\$ / Year) Annualized Cost Deliveries of Recycled Water Estimated Unit Cost (\$/AF) Breakdown of Estimated Unit Cost (\$/AF)	20 20 1 Two payments per Annualized value of 22,40	10,320,178 5% of pump st year, spread over f present worth 0 AFY Treatment ent (Supply used st	MGD MGD LS hrs kWh/yr ation construction cost Total O&M Costs (\$/y Project Life \$980 \$/10 years)	\$ \$ \$ r)	114,500 260,600 108,100 135 0.15 Distribution	\$2,300,000 \$5,200,000 \$109,800 \$0 \$1,500,000 \$9,100,000 \$9,100,000 \$9,100,000 \$24,000,000 \$24,000,000 \$11,100 \$120		
General O&M Treatment O&M (\$114,500/MGD) Refinery O&M (\$260,600/MGD) Conveyance Annual O&M Pump Stations Labor Costs Electricity Pump Station Consumables Annualized Costs (\$ / Year) Annualized Costs (\$ / Year) Annualized Salvage Value Annual O&M Costs Total Annualized Cost Deliveries of Recycled Water Estimated Unit Cost (\$/AF) Breakdown of Estimated Unit Cost (\$/AF] Annualized Costs (\$ / Year) Annualized Costs (\$ / Year)	20 20 1 Two payments per Annualized value of 22,40 Dry Year Adjustma Two payments per	10,320,178 5% of pump st year, spread over f present worth 0 AFY Treatment ent (Supply used 3 year, spread over	MGD MGD LS hrs kWh/yr ation construction cost Total O&M Costs (\$/y Project Life \$980 3/10 years) Project Life	\$ \$ \$ r)	114,500 260,600 108,100 135 0.15 Distribution	\$2,300,000 \$5,200,000 \$109,800 \$0 \$1,500,000 \$9,100,000 \$16,000,000 \$24,000,000 \$24,000,000 \$1,100 \$120 \$16,000,000	Same as cons	tant use.
General O&M Treatment O&M (\$114,500/MGD) Refinery O&M (\$260,600/MGD) Conveyance Annual O&M Pump Stations Labor Costs Electricity Pump Station Consumables Annualized Costs (\$/Year) Annualized Costs (\$/Year) Annualized Cost Total Annualized Cost Deliveries of Recycled Water Estimated Unit Cost (\$/AF) Breakdown of Estimated Unit Cost (\$/Year) Annualized Costs (\$/Year) Annualized Costs (\$/Year) Annualized Cost (\$/Year) Breakdown of Estimated Unit Cost (\$/AF) Annualized Costs (\$/Year)	20 20 1 Two payments per Annualized value of 22,40 Dry Year Adjustma Two payments per Annualized value of	10,320,178 5% of pump st year, spread over f present worth 0 AFY Treatment ent (Supply used a year, spread over f present worth	MGD MGD LS hrs kWh/yr ation construction cost Total O&M Costs (\$/y Project Life \$980 \$/10 years) Project Life	\$ \$ \$ r)	114,500 260,600 108,100 135 0.15 Distribution	\$2,300,000 \$5,200,000 \$109,800 \$1,500,000 \$1,500,000 \$1,400,000 \$24,000,000 \$120 \$16,000,000 \$1,400,000 \$16,000,000 \$1,400,000	Same as cons Same as cons	tant use.
General O&M Treatment O&M (\$114,500/MGD) Refinery O&M (\$260,600/MGD) Conveyance Annual O&M Pump Stations Labor Costs Electricity Pump Station Consumables Annualized Costs (\$/Year) Annualized Capital Costs (\$/Year) Annualized Salvage Value Annual O&M Costs Total Annualized Cost Deliveries of Recycled Water Estimated Unit Cost (\$/AF) Breakdown of Estimated Unit Cost (\$/AF) Annualized Costs (\$/Year) Annualiz	20 20 1 Two payments per Annualized value of 22,40 Dry Year Adjustma Two payments per Annualized value of	10,320,178 5% of pump st year, spread over f present worth 0 AFY Treatment ent (Supply used 3 year, spread over f present worth	MGD MGD LS hrs kWh/yr ation construction cost Total 0&M Costs (\$/y Project Life \$980 3/10 years) Project Life	\$ \$ \$ r)	114,500 260,600 108,100 135 0.15 Distribution	\$2,300,000 \$5,200,000 \$109,800 \$1,500,000 \$1,500,000 \$9,100,000 \$16,000,000 \$24,000,000 \$11,000 \$120 \$16,000,000 -\$1,400,000 -\$1,400,000 -\$1,400,000 -\$1,400,000 -\$1,400,000 -\$1,400,000 -\$1,400,000 -\$1,400,000 -\$1,400,000 -\$1,400,000 -\$1,400,000 -\$1,400,000 -\$1,400,000 -\$1,400,000 -\$1,400,000 -\$1,400,000 -\$1,400,000 -\$1,400,000 -\$1,000,000,000 -\$1,000,000,000 -\$1,000,000 -\$1,000,000 -\$1,0	Same as cons Same as cons	tant use. tant use.
General O&M Treatment O&M (\$114,500/MGD) Refinery O&M (\$260,600/MGD) Conveyance Annual O&M Pump Stations Labor Costs Electricity Pump Station Consumables Annualized Costs (\$/Year) Annualized Capital Costs (\$/Year) Annualized Capital Cost (\$/Year) Annualized Salvage Value Estimated Unit Cost (\$/AF) Breakdown of Estimated Unit Cost (\$/AF) Annualized Costs (\$/Year) Annualized Costs (\$/Year) Annualized Cost Annu	20 20 1 Two payments per Annualized value of 22,40 Dry Year Adjustma Two payments per Annualized value of	10,320,178 5% of pump st 5% of pump st year, spread over f present worth 0 AFY Treatment ent (Supply used 3 year, spread over f present worth	MGD MGD LS hrs kWh/yr ation costruction cost Total 0&M Costs (\$/y Project Life \$980 3/10 years) Project Life	\$ \$ \$ r)	114,500 260,600 108,100 135 0.15 Distribution	\$2,300,000 \$5,200,000 \$109,800 \$1,500,000 \$9,100,000 \$16,000,000 \$14,000,000 \$24,000,000 \$14,000,000 \$14,000,000 \$14,000,000 \$14,000,000 \$14,000,000 \$23,000,000	Same as cons Same as cons	tant use. tant use.
General O&M Treatment O&M (\$114,500/MGD) Refinery O&M (\$260,600/MGD) Conveyance Annual O&M Pump Stations Labor Costs Electricity Pump Station Consumables Annualized Capital Costs (\$/Year) Annualized Capital Costs (\$/Year) Annualized Capital Cost Deliveries of Recycled Water Estimated Unit Cost (\$/AF) Breakdown of Estimated Unit Cost (\$/AF) Annualized Costs (\$/Year) Annualized Costs (\$/Year) Annualized Costs (\$/Year) Annualized Cost (\$/Year) Annualized Cost Deliveries of Recycled Water Estimated Unit Cost (\$/AF) Breakdown of Estimated Unit Cost (\$/AF) Annualized Costs (\$/Year) Annualized Costs Total Annualized Cost Total Annualized Cost Annual Average Deliveries of Recycled Water Estimated Unit Cost (\$/AE)	20 20 1 1 Two payments per Annualized value of 22,40 Dry Year Adjustm Two payments per Annualized value of 6,720	10,320,178 5% of pump st year, spread over f present worth 0 AFY Treatment ent (Supply used 3 year, spread over f present worth	MGD MGD LS hrs kWh/yr ation construction cost Total O&M Costs (\$/y Project Life \$980 3/10 years) Project Life	\$ \$ \$ r)	114,500 260,600 108,100 135 0.15 Distribution	\$2,300,000 \$5,200,000 \$109,800 \$0 \$1,500,000 \$9,100,000 \$9,100,000 \$9,100,000 \$24,000,000 \$16,000,000 \$24,000,000 \$14,000 \$120 \$16,000,000 \$14,000,000 \$23,000,000 \$23,000,000 \$33,000	Same as cons Same as cons	tant use. tont use.

CCWD Pipeline in Canal ROW Recycled Wa	ter Project Cos	t Update				E	BMUD Recycled Water	/laster Plan Update
Last Updated:	16-May-18					Discount Rate		Project Life
Updated by:	M. Romero					3%	:	30 Years
Original Project Information								
Project Source:	2013 CCWD F	Facilities Improve	ement Plan Update					
Project Source CCI Date:		July 2013				Project Source Location:	San Franciso	o Area
Project Source 20-Cities Average ENR CCI:		9,552				Location Multiplier:	1.00	
December 2017 20-Cities Average ENR CCI:		10,870				Historical Multiplier:	1.14	
Item	Size	Qty	Unit	Or	iginal Project Cost	Adjusted Project Cost	Useful Life	Salvage Value
Capital Costs								
<u>Conveyance</u>								
Distribution Pipeline	28", 18" and 6"	97,680 LF	in-LF	\$	18,500,000	\$21,000,000	75 yrs	\$13,000,000
Pump Station								
Pump Station			LS	\$	7,500,000	\$8,500,000	50 yrs	\$3,400,000
Storage Tanks								
Welded Steel Storage Tank		1	EA	\$	500,000	\$570,000	75 yrs	\$340,000
Estimated Total Construction Cost including Impleme	ntation and Contin	gency				\$30,000,000		\$17,000,000
						Present V	Vorth of Salvage Value	\$7,000,000
O&M Costs (Annual)								
Converses								
<u>Conveyance</u>		20/ -6				ć222.000		
Annuai O&M		2% OT (construction cost			\$323,000		
Pump Stations								
Labor Costs		500) hrs	\$	135	\$67,500		
Electricity		114,669	e kWh/yr	\$	0.15	\$17,200		
Pump Station Consumables		5% of pump s	tation construction co	ost		\$372,000		
Storage Tanks								
Annual O&M		1% of	construction cost			\$5.000		
			Total O&M Costs (j/yr)		\$780.000		
Annualized Costs (Ś / Year)			,. ,			,,		
Annualized Capital Costs (\$/Year)	Two payments per	vear, spread ov	er Proiect Life			\$1,500,000		
Annualized Salvage Value	Annualized value o	of present worth	,,.			-\$360.000		
Annual O&M Costs		,,				\$780.000		
Total Annualized Cost						\$1,900.000		
Deliveries of Recycled Water	930	AFY				<u>· · · · · · · · · · · · · · · · · </u>		
Estimated Unit Cost (\$/AF)						\$2,000		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$0		Distribution	\$2,000		
Annualized Costs (\$ / Year)	Drv Year Adiustme	ent (Supply used	3/10 vears)					
Annualized Capital Costs (\$/Year)	Two payments per	year, spread ov	er Project Life			\$1,500,000	Same as constant use.	
Annualized Salvage Value	Annualized value o	of present worth				-\$360,000	Same as constant use.	
Annual O&M Costs						\$772,660		
Total Annualized Cost						\$1,900,000		
Annual Average Deliveries of Recycled Water	279	AFY						
Estimated Unit Cost (\$/AF)						\$6,800		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$0		Distribution	\$6,800		

Diablo Country Club Satellite Treatment	t Recycled Wat	er Project Co	ost Update	9		EBMUI	D Recycled Water M	aster Plan Update
Last Updated:	18-May-18	3				Discount Rate	Pr	oject Life
Updated by:	M. Romero)				3%	30) Years
Original Project Information								
Project Source:	2013	Feasibility Stud	ly					
Project Source CCI Date:	Se	eptember 2013				Project Source Location:	San Francis	sco Area
Project Source 20-Cities Average ENR CCI:		9,552				Location Multiplier:	1.00	D
December 2017 20-Cities Average ENR CCI:		10,870				Historical Multiplier:	1.14	4
Item	Size	Qty	Unit	Pro	Original piect Cost	Adjusted Project Cost	Useful Life	Salvage Value
Capital Costs					,			
Treatment								
MBR/UV/Chlorine	0.5		MGD	Ś	3.278.295	\$3,700.000	35 vrs	\$530.000
Wastewater Diversion, 12HP	1		EA	Ś	99.474	\$113.000	50 yrs	\$45.000
Wastewater Diversion Pipeline, 8" diam, 4700LF	1		EA	Ś	972,900	\$1.110.000	75 yrs	\$670.000
Disinfection for Chlorine Residual	1		FA	Ś	15 431	\$18,000	35 yrs	\$2,600
Beturn Solids Pump Station	1		FΔ	Ś	34 937	\$40,000	50 yrs	\$16,000
Return Solids Force Main, 5 0001 F	1		EA	ć	945.000	\$1,090,000	75 yrs	\$650,000
Return Solids Force Main, 5,000Er	1		LA	Ļ	945,000	\$1,080,000	7 J YI S	\$050,000
<u>Conveyance</u>								
Non-urban Pipeline		1,000 LF	in-LF	\$	189,000	\$220,000	75 yrs	\$130,000
Pump Station								
Pump Station				\$	20,833	\$24,000	50 yrs	\$9,600
Storage Tanks								
Welded Steel Storage Tank		1	EA	\$	52,848	\$60,000	75 yrs	\$36,000
-					Subtotal	\$6,400,000		\$2,100,000
		Estimate	d Subtotal C	onstru	uction Cost	\$6,400,000		\$2,100,000
Construction Cost Including Contingency								
		Estimate	d Subtotal C	onstru	uction Cost	\$6,400,000		\$2,100,000
Estimating Contingency					25%	\$1,600,000		\$530,000
	Es	timated Subtota	al Cost includ	ling Co	ontingency	\$8.000.000		\$2,600,000
Mobilization					8%	\$640.000		+_,,
	Es	timated Subtot	al Cost includ	ling Co	ontingency	\$8.600.000		\$2.600.000
Construction Cost Including Implementation				0	,	1.,,		. , ,
Planning / Environmental					5%	\$430.000		
Design Cost					15%	\$1,300,000		
Project Administration and Construction Managem	ent Cost				10%	\$860.000		
Estimated Total Construction Cost including Imple	mentation and Co	ntingency				\$11.000.000		\$2.600.000
		0. 1				Present Worth	of Salvage Value	\$1,100,000
O&M Costs (Annual)								
Recycled Water Treatment								
MRR LIV chloring disinfection			MCD	ć	222 020	¢270.000		
Floctricity	401 500		WGD	ې د	522,620	\$570,000		
Electricity	401,500		K VVII/ yI	Ş	<i>39,</i> 422	\$00,000		
Conveyance								
Annual O&M		1,000 LF		\$	18,782	\$21,000		
			Total O&M	Costs	(\$/yr)	\$450,000		
Annualized Costs (\$ / Year)								
Annualized Capital Costs (\$/Year)	Two payments pe	r year, spread o	ver Project Li	fe		\$560,000		
Annualized Salvage Value	Annualized value	of present worth	h			-\$56,000		
Annual O&M Costs						\$450.000		
Total Annualized Cost						\$950.000		
Deliveries of Recycled Water	246	AFY				<i>+•••,000</i>		
Estimated Unit Cost (\$/AE)						\$3.900		
Breakdown of Estimated Unit Cost (\$/AE)		Treatment	\$3.600		istribution	\$300		
Annualized Costs (\$ / Year)	Dry Year Adjuster	ent (Supply use	od 3/10 years	, ,	Stribution	4000		
Annualized Capital Costs (\$ /Vear)	Two novments no	r vear spread o	ver Project Li	fo		¢E60.000.0	ame as constant	
Annualized Salvage Value	Annualized value	of present worth	h			200,000 3 _\$E6 000 9	ame as constant use	
		oj present worth	,			-200,000 3	unie us constant use	•
						2403,000 ¢010.000		
Annual Average Deliveries of Recycled Water	74.4	LEV.				\$910,000		
Estimated Unit Cost (\$ (AF)	74 F					¢12.000		
Estimated Unit Cost (\$/AF)		T	644.000		distanting of	\$12,000		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$11,000	D	istribution	\$1,000		

Moraga Country Club Satellite Recycle	d Water Projec	t Cost Upd	ate			EBMUD Recy	cled Water N	/laster Plan Update
Last Updated:	24-Jan-18					Discount Rate	ŀ	Project Life
Updated by:	M. Romero					3%	3	30 Years
Original Project Information								
Project Source:	MCC SF	RWTP Detailed	d FS					
Project Source CCI Date:	Sep	tember 2009				Project Source Location:	San Fr	ancisco Area
Project Source 20-Cities Average ENR CCI:		8,586				Location Multiplier:		1.00
December 2017 20-Cities Average ENR CCI:		10,870				Historical Multiplier:		1.27
Item	Size	Otv	Unit	Ori	iginal Project	Adjusted Project Cost	Liseful Life	Salvage Value
	5120				Cost	Aujusteu Project cost		Survage value
Capital Costs								
Ireatment	0.5		MCD	ć	7 000 000	¢10,000,000	25	ć1 400 000
MBR/UV/Chlorine	0.5		MGD	Ş	7,900,000	\$10,000,000	35 yrs	\$1,400,000
Construction Cost including Contingency		Fatimate d C		-leading	- C	ć10.000.000		ć1 400 000
		Estimated S	ubtotal Cost in	ciuair	ig Contingency	\$10,000,000		\$1,400,000
Nobilization	Fatiments of Contra-			5%	\$500,000		ć1 400 000	
Construction Control and a local production	Estimated Subto	tal Cost inclu	ding Contingei	ncy an	d Wobilization	\$11,000,000		\$1,400,000
Construction Cost including implementation					50/	6550 000		
Planning / Environmental					5%	\$550,000		
Design Cost					15%	\$1,700,000		
Project Administration and Construction Manager					10%	\$1,100,000		61 400 000
Estimated Total Construction Cost including impl	ementation					\$14,000,000 Decemble of Co		\$1,400,000
						Present worth of Sa	lvage value	\$570,000
Daw Costs (Annual)								
	0.5		MCD	ć	56 057	¢71.000		
Energy Chemical Lise	0.5		MGD	ې د	1 100	\$71,000		
Chemical Use	0.5		MGD	ې د	1,190	\$1,500		
IN Classic Replacement Cost	0.5		MGD	Ş	25,000	\$32,000 ¢5 200		
UV Lamp Replacement Cost	0.5		MGD	Ş	4,160	\$5,300		
Labor Costs	0.5		MGD	ې د د د د	33,800	\$43,000		
			Total O&IVI 0	LOSTS ((\$/yr)	\$150,000		
Annualized Costs (\$ / Year)	- .			· c		6740.000		
Annualized Capital Costs (\$/Year)	I wo payments per	r year, spreaa	over Project L	ije		\$710,000		
	Annualizea value	of present wo	rtn			-\$29,000		
						\$150,000		
Total Annualized Cost	170 4	ΓV				\$830,000		
	1/9 A	FT				ć4 coo		
Estimated Unit Cost (\$/AF)		T	¢4.000		Distribution	\$4,600		
Annualized Costs (\$ / Year)	Due Vege Adjuster	Treatment	\$4,600		Distribution	ŞU		
Annualized Costs (\$ / fear)	Dry Year Aajustin	ent (Supply u	sea 3/10 years	if a		¢710.000	Campa as son	tant uso
Annualized Capital Costs (\$/Tear)	Appualized value	yeur, spread	over Project L	уe		\$710,000	Samo as cons	stant use.
Annual OSM Costs		j present WO	rul -			->23,000	Sume us cons	stuill use.
						\$152,800		
Appual Average Deliveries of Recycled Water	E / A /	v				\$830,000		
Fatimated Unit Cost (\$ (55)	54 Ar					¢15.000		
Estimated Unit Cost (\$/AF)		Turatura	¢15.000	_	Distribust	\$15,000		
breakdown of Estimated Unit Cost (S/AF)		Treatment	315.000		Distribution	50		

Moraga Area Satellite Treatment Recycled Water Project Cost						EBMUD Recycled Water Master Plan Update			
Last Updated:	20-May-18				[Discount Rate	P	roject Life	
Updated by:	M. Romero				3	3%	3	0 Years	
Item	Size	Qty	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value	
Capital Costs									
Treatment	Cost Estimat	es for Treatment	are based on Unit C	osts f	for Each Treatr	nent Process	25	<i></i>	
MBR/UV/Chlorine	0.5		MGD	Ş	17,700,000	\$8,900,000	35 yrs	\$1,300,000	
Constant and Constant of Constitution					Subtotal	\$8,900,000		\$1,300,000	
Contractor Overnead & Profit	50%				15%	\$1,335,000		\$200,000 ¢50,000	
Sales Tax	50%	5 OF SUDIOLAL COS	t Applicable	ost fo	970	\$401,000		\$39,000	
Construction Cost Including Contingency		LStimateu ins	caneu Equipment C	USUIC	of freatment	\$11,000,000		\$1,000,000	
construction cost including contingency			Estimated Subtotal	Const	truction Cost	\$11.000.000		\$1.600.000	
Estimating Contingency					25%	\$2,750,000		\$400.000	
		Estimated Su	btotal Cost (1) inclu	ding	Contingency	\$14,000,000		\$2,000,000	
Capital Costs (from Bids)					<u> </u>	, ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,		1 ,,	
Conveyance	Cost Estimat	es for Pipelines a	re based on EBMUD	cons	truction bid co	st + estimating contin	gency		
Low-Density Urban Pipeline (<12")	2-in	3,100 LF	in-LF	\$	40	\$248,000	75 yrs	\$150,000	
Low-Density Urban Pipeline (<12")	3-in	5,150 LF	in-LF	\$	40	\$618,000	75 yrs	\$370,000	
Low-Density Urban Pipeline (<12")	4-in	4,600 LF	in-LF	\$	40	\$736,000	75 yrs	\$440,000	
Dumn Stations	Cost Estimat	oc for Dump Stati	ions are based on El		Construction	Rid Costs		. ,	
Pump Stations	COST ESTIMAT	es jor Pump stati 2	ons are based on EE	siviUD ¢	17 500	652 500	2E virc	¢7 500	
Distribution Pumps (2+1, 20 hp)		5 EA (Cost Cur	va bacad on total	Ş	17,500	\$52,500	35 yrs	\$7,500	
Pump Station (Total installed HP, including standby)	CO 11D	EA (COST CUP		ć	1 200 000	¢1 200 000	F.O. 1. 199	¢520.000	
Starage Tarks	OU HP	Insta an far Charman Ta	alleu HP)	ې ۱/۳۳/۱	1,300,000	\$1,300,000	50 yrs	\$520,000	
Storage Taliks		es joi storage ra		yuıı د خ	2	¢60.000	7E vrc	¢26.000	
Welded Steel Storage Tank (<40 MG)	0.03 MG	1		ې د	2	\$00,000 ¢70,000	75 yrs	\$30,000	
Welded Steel Storage Tank (<40 MG)	0.04 MG	1		ې د	2	\$70,000	75 yrs	\$42,000 \$72,000	
Welded Steel Storage Tallk (<40 Mid)	0.06 1910	Estimated Su	htotal Cost (2) inclu	ç naina	Contingoncy	\$120,000	75 yis	\$72,000	
Mabilization		Estimateu Su	biolai cost (2) ilicit	uing	contingency	\$5,200,000		\$1,600,000	
Estimated Subtotal Construction Cost including Cor	tingoncy					\$17,000,000		\$2,600,000	
Mobilization	lingency				5%	\$850,000		\$3,000,000	
Estimated Subtotal Construction Cost including Cor	tingency and	Mobilization			570	\$18,000,000		\$3 600 000	
Implementation	reingency and	Wobilization				\$10,000,000		\$3,000,000	
Planning / Environmental					5%	\$900.000			
Design Cost					15%	\$2,700,000			
Project Administration and Construction Manageme	ent Cost				10%	\$1,800,000			
Estimated Total Construction Cost including Implem	nentation and	Contingency			10/0	\$23,000,000		\$3,600,000	
		e eentingener				Present Worth	of Salvage Value	\$1,500,000	
O&M Costs (Annual)									
Recycled Water Treatment									
MBR/UV (chemical, power, maintenance)	0.2		MGD	\$	424,000	\$93,000			
Labor	0.5	1,040	hrs/MGD	\$	135	\$70,000			
Electricity	0.2	1,130,000	kWh/yr	\$	0.15	\$37,300			
Conveyance									
Annual O&M			2% of construction	l cost		\$32,000			
Ruma Stations									
Labor Costs		0	brc	ć	125	ćo			
Labor Costs		125 506	llis kWb/vr	ې د	155	پر مود مدغ			
Rump Station Concumpbion		133,390	NVVII/ yi	ې act	0.15	\$20,300 ¢0			
Pump station consumables		5% of pump st		USL		ŞU			
Storage Tanks									
Annual O&M		1% of c	onstruction cost			\$2,500			
			Total O&M Costs	\$/yr)		\$260,000			
Annualized Costs (\$ / Year)									
Annualized Capital Costs (\$/Year)	Two paymen	ts per year, sprea	nd over Project Life			\$1,200,000			
Annualized Salvage Value	Annualized v	alue of present w	vorth			-\$76,000			
Annual O&M Costs						\$260,000			
Total Annualized Cost					_	\$1,400,000			
Deliveries of Recycled Water	2	46 AFY							
Estimated Unit Cost (\$/AF)						\$5,700			
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$4,500		Distribution	\$1,200			
Annualized Costs (\$ / Year)	Dry Year Ad	ustment (Supply	used 3/10 years)						
Annualized Capital Costs (\$/Year)	Two paymen	ts per year, spred	nd over Project Life			\$1,200,000 Sc	ame as constant us	е.	
Annualized Salvage Value	Annualized v	alue of present w	vorth			-\$76,000 Se	ame as constant us	е.	
Annual O&M Costs						\$214,780			
Total Annualized Cost						\$1,300,000			
Annual Average Deliveries of Recycled Water	7	4 AFY			-				
Estimated Unit Cost (\$/AF)						\$18,000			
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$15,000		Distribution	\$3,000			

Oakland Hills Satellite Treatment Re	cycled Water Pr	oject Cost	Update			EBMU	D Recycled Water M	aster Plan Update		
Last Updated:	20-May-18	·	•			Discount Rate	Pr	Project Life		
Updated by:	M. Romero					3%	30 Years			
Original Project Information										
Project Source:	Oakla	nd Hills FS, 20	017							
Project Source CCI Date:		April 2017			Proje	ect Source Location:	San Franci	sco Area		
Project Source 20-Cities Average ENR CCI:		10,678			l	Location Multiplier:	1.00			
December 2017 20-Cities Average ENR CCI:		10,870		-	H	istorical Multiplier:	1.0	2		
Item	Size	Qty	Unit	U	Cost	Cost	Useful Life	Salvage Value		
Capital Costs										
<u>Treatment</u>										
MBR System (0.5 MGD capacity)	0.5		MGD	\$	5,401,000	\$5,500,000	35 yrs	\$790,000		
UV System	0.5		MGD	\$	426,000	\$430,000	35 yrs	\$61,000		
Diversion Structure	1		EA	\$	168,000	\$170,000	50 yrs	\$68,000		
Diversion Pumps	1		EA	\$	126,000	\$130,000	50 yrs	\$52,000		
Diversion Pipeline (12" diam, 6,800LF)	1		EA	Ş	2,568,000	\$2,600,000	75 yrs	\$1,600,000		
Electrical Infrastructure	1		EA	Ş	420,000	\$430,000	75 yrs	\$260,000		
Conveyance										
Low-Density Urban Pipeline	4" to 8"	14,250 LF	in-LF	\$	2,364,000	\$2,400,000	75 yrs	\$1,400,000		
Pump Station										
Pump Station (n=1+1, 1.5HP each)	1.5HP	2	HP	\$	7,000	\$7,100	50 yrs	\$2,800		
Storage Tanks										
Welded Steel Storage Tank	0.5 MG	1	EA	Ś	1 200 000	\$1,200,000	75 vrs	\$720.000		
Construction Cost Including Contingency	0.5 1110		271	Ŷ	1,200,000	\$1,200,000	75 915	\$720,000		
<u> </u>		Es	timated Subt	otal Co	onstruction Cost	\$13,000,000		\$5,000,000		
Estimating Contingency	Apply as needed				25%	\$3,300,000		\$1,300,000		
		Estimated S	Subtotal Cost	includi	ing Contingency	\$16,000,000		\$6,300,000		
Mobilization					5%	\$800,000				
	Estimated Subt	otal Cost inclu	uding Conting	ency a	nd Mobilization	\$17,000,000		\$6,300,000		
Construction Cost Including Implementation										
Planning / Environmental					5%	\$850,000				
Design Cost					15%	\$2,600,000				
Project Administration and Construction Manage	gement Cost				10%	\$1,700,000				
Estimated Total Construction Cost including Im	plementation					\$22,000,000 Brosont Worth	of Salvago Valuo	\$6,300,000		
O&M Costs (Annual)						Present Worth	of Salvage value	\$2,800,000		
Recycled Water Treatment										
MBR/UV/Chlorine (Maintenance/Energy)	0.5		MGD	Ś	82,000	\$83.000				
Operation Costs	0.5		MGD	\$	66,000	\$67,000				
Labor Costs				ć		ćo				
Electricity				ç	-	ېن 16 000				
Pump Station Consumables	5%	of nump stat	ion constructi	ې مە دەد	10,000	910,000 02				
	570		Total O&M	Costs	(Ś/vr)	\$170.000				
Annualized Costs (\$ / Year)					(+/)-/	+=: =/===				
Annualized Capital Costs (\$/Year)	Two payments per	year, spread	over Project L	ife		\$1,100,000				
Annualized Salvage Value	Annualized value of	of present wor	rth			-\$130,000				
Annual O&M Costs						\$170,000				
Total Annualized Cost						\$1,100,000				
Deliveries of Recycled Water	347 A	FY								
Estimated Unit Cost (\$/AF)						\$3,200				
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$2,400		Distribution	\$800				
Annualized Costs (\$ / Year)	Dry Year Adjustm	ent (Supply us	sed 3/10 year	s)						
Annualized Capital Costs (\$/Year)	Two payments per	r year, spread	over Project L	ife		\$1,100,000	Same as constant us	е.		
Annualized Salvage Value	Annualized value o	of present wor	rth			-\$130,000	Same as constant us	е.		
Annual O&M Costs						\$154,800				
Iotal Annualized Cost	104 4	EV			-	\$1,100,000				
Estimated Unit Cost (\$ /AF)	104 A	. 1				¢11.000				
Breakdown of Estimated Unit Cost (\$ /AF)		Treatment	Ś8 100		Distribution	\$11,000				
Dicakaowii of Estimated Offic Cost (S/AF)		neament	90,100		Distribution	92,500				

UCB Satellite Treatment Recycled Wate	r Project C	ost				EBMUD Recyc	led Water Ma	aster Plan Update
Last Updated:	20-May-18	-				Discount Rate	F	Project Life
Updated by:	M. Romero					3%	3	0 Years
Item	Size	Qty	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value
Capital Costs	Cost Estimate	o for Troatmont	are based on Unit Ca	octo d	for Each Troat	mant Brocass		
MBR System (1 5-mgd capacity)	1 5	es jor meatment	MGD	sis j	4 652 300	\$4 652 300	35 vrs	\$660,000
UV system	1.5		MGD	\$	1,212,000	\$1,212,000	35 yrs	\$170,000
On-site hypochlorite generation system	1.5		MGD	\$	1,350,000	\$1,350,000	35 yrs	\$190,000
Diversion Structure (pumps, structure)	1		EA	\$	458,000	\$458,000	50 yrs	\$180,000
Diversion Pipeline (100LF, 10" diameter)	1		EA	\$	50,000	\$50,000	75 yrs	\$30,000
Site Work and electrical (includes piping)	1		EA	\$	1,325,802	\$1,325,802	50 yrs	\$530,000
UV/MCC/Blowers Building	1		EA	\$	3,000,000	\$3,000,000	50 yrs	\$1,200,000
Contractor Oracle and C. Des Cl					Subtotal	\$12,000,000		\$3,000,000
Contractor Overnead & Profit	50%	% of Subtotal Cost	Applicable		15% 9%	\$1,800,000		\$450,000 \$140,000
Sales Tax	50%	Estimated Ins	talled Equipment Co	ost fo	or Treatment	\$14,000,000		\$3.600.000
Construction Cost Including Contingency						, ,,		,,
		E	stimated Subtotal C	Cons	truction Cost	\$14,000,000		\$3,600,000
Estimating Contingency					25%	\$3,500,000		\$900,000
Constant Consta		Estimated Sul	btotal Cost (1) inclu	ding	Contingency	\$18,000,000		\$4,500,000
Capital Costs	Cost Estimate	os for Pinalinas ar	a bacad on ERMUD	conc	truction hid co	st + actimating contingancy		
Ligh-Density Urban Pineline	15-in	5 000 LE	in-I F	Ś	50	\$3 750 000	75 vrs	\$2 300 000
		6 8 6 7		¥		ç5), 56,666	75 115	\$2,500,000
Pump Stations	Cost Estimate	es for Pump Static	ons are based on EBI	NUL	Construction	Bid Costs		
Pump Station (Total installed HP, including standby)	450 HP	EA (COSI CUIV	lled HP)	Ś	2 100 000	\$2 100 000	50 vrs	\$840.000
Storage Tanks	Cost Estimate	es for Storage Tar	nks are based on EBN	лUD	Construction	Bid Costs		<i>t</i> - · · <i>j</i> - · · <i>j</i>
Bro strossod Constrato Storago Tank (<20 MG)			EA (Cost Curve by					
Fre-stressed concrete storage rank (<20 MG)	1.5 MG	1	Volume)	\$	2,400,000	\$2,400,000	100 yrs	\$1,700,000
Pre-stressed Concrete Tank Allowances			3% of Line Abo	ove +	\$0.15M	\$220,000		
		Estimated Sul	btotal Cost (2) inclu	ding	Contingency	\$8,500,000		\$4,800,000
Mobilization						407.000.000		40.000.000
Estimated Subtotal Construction Cost including Con	ntingency				F0/	\$27,000,000		\$9,300,000
NODIIIZATION	ntingency and	Mobilization			5%	\$1,400,000		\$9 300 000
Implementation	ingency and	Widdinzation				\$28,000,000		\$5,500,000
Planning / Environmental					5%	\$1,400,000		
Design Cost					15%	\$4,200,000		
Project Administration and Construction Manageme	nt Cost				10%	\$2,800,000		
Estimated Total Construction Cost including Impler	nentation and	I Contingency				\$36,000,000	hanna Malua	\$9,300,000
O&M Costs (Annual)						Present Worth of 3a	vage value	\$5,800,000
Recycled Water Treatment								
MBR	0.8	1130000	kWh/MGD	\$	0.15	\$135,600		
UV	0.8	162,500	kWh/MGD	\$	0.15	\$19,013		
WW Diversion Pumping	28,667	20/ of MDD/	kWh	Ş	0.15	\$4,300		
Free Chlorine	0.8	Z% OF IVIBR/ U	MGD	¢	49 000	\$117,286 \$39,000		
Chemicals	0.8		MGD	ś	120.000	\$96.000		
Labor	1.6	1,040	hrs/MGD	\$	135	\$220,000		
Conveyance								
Annual O&M			2% of construction	cost		\$75,000		
Pump Stations								
Labor Costs (1.6 MGD/1,100 gpm)		500	hrs	\$	135	\$68,000		
Electricity		298,138	kWh/yr	\$	0.15	\$45,000		
Pump Station Consumables		5% of pump sta	ation construction co	ost		\$0		
Storage Tanks								
Annual O&M		1% of co	onstruction cost			\$24,000		
			Total O&M Costs (\$	\$/yr)		\$840,000		
Annualized Costs (\$ / Year)								
Annualized Capital Costs (\$/Year)	Two paymen	ts per year, sprea	d over Project Life			\$1,800,000		
Annualized Salvage Value	Annualized v	alue of present w	orth			-\$190,000		
						\$840,000		
Deliveries of Recycled Water	87	74 AFY			-	\$2,500,000		
Estimated Unit Cost (\$/AF)						\$2,900		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$1,900		Distribution	\$1,000		
Annualized Costs (\$ / Year)	Dry Year Adj	ustment (Supply	used 3/10 years)					
Annualized Capital Costs (\$/Year)	Two paymen	ts per year, sprea	d over Project Life			\$1,800,000	Same as cons	tant use.
Annualized Salvage Value	Annualized v	alue of present w	orth			-\$190,000	Same as cons	tant use.
Annual O&M Costs						\$714,189		
Annual Average Deliveries of Recycled Water	26	52 AFY			-	ş2,300,000		
Estimated Unit Cost (\$/AF)						\$8,800		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$6,100		Distribution	\$2,700		

APPENDIX B: COST ESTIMATE DETAILS FOR POTABLE REUSE ALTERNATIVES

Alternative Oro-GW							EBMUD Recyc	led Water Master Plan Update
Last Updated:	30-May-18					Discount Rate	l l	Project Life
Updated by:	I. Jaffe					3%	3	30 Years
CCI (20 City, Dec 2017): 10870.06								
Item	Size	Qty	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value
Capital Costs								
Treatment	Cost Estimates	for Treatment are	e based on Unit Cos	ts for l	Each Treatmen	nt Process	20 μmg	ćo
	0		MGD	ç	335 300	30 \$0	30 yrs	30 \$0
BAC	0		MGD	ŝ	300,900	\$0 \$0	30 yrs	50
MF/UF system	8		MGD	Ş	1,225,000	\$9,800,000	30 yrs	\$0
RO System	8		MGD	\$	1,475,000	\$12,000,000	30 yrs	\$0
Advanced Oxidation and Disinfection	8		MGD	\$	437,500	\$3,500,000	30 yrs	\$0
Chemicals (Storage and Use)	8			\$	125,000	\$1,000,000	30 yrs	\$0
Sitework/Piping/Structures	8		MGD	Ş	3,187,500	\$26,000,000	30 yrs	\$0
Groundwater Wells								
Injection Well	1 MG	8		\$	1,679,000	\$13,000,000	50 yrs	\$5,200,000
Extraction Well (includes wellhead treatment)	2 MG	4		Ş	4,068,000	\$16,000,000	50 yrs	\$6,400,000
			Raw Constru	iction	Cost Subtotal	\$81,000,000		\$12,000,000
Salos Tax	E0%	% of Subtotal Cost	Applicable		15%	\$12,000,000		\$1,800,000
30163 103	50%	76 OF SUBLOCAL COST	Estimated Subtota		struction Cost	\$3,000,000		\$14,000
			Estimated Subtota	ii cons	struction cost	\$37,000,000		\$14,000,000
Construction Cost Including Contingency								
<u></u>			Estimated Subtota	l Cons	struction Cost	\$97.000.000		\$14.000.000
Estimating Contingency					25%	\$24,000,000		\$3,500,000
		Estimated S	ubtotal Cost (1) inc	luding	g Contingency	\$120,000,000		\$18,000,000
Capital Costs (from Bids)								
Conveyance	Cost Estimates	for Pipelines are b	pased on EBMUD co	onstru	ction bid cost	+ estimating contingency		
Non-urban Pipeline	24-in	100 LF	in-LF	\$	30	\$72,000	75 yrs	\$43,000
Low-Density Urban Pipeline	24-in	35,100 LF	in-LF	Ş	40	\$33,696,000	/5 yrs	\$20,000,000
Tranchless Crossings	24-IN	6,000 LF	IN-LF	Ş	50	\$7,200,000	75 yrs	\$4,300,000
HDD (All Inclusive)		1 500	IF	¢	2 200	¢3 300 000	75 vrc	\$2,000,000
		1,500		ڊ 	2,200	\$3,300,000	7 J YI S	\$2,000,000
Pump Stations	Cost Estimates	for Pump Station	s are based on EBM	UD Ca	onstruction Bid	Costs		
Pump Station - Treatment to Injection (Total Installed	1 500 HD	EA (Cost Curv	o based on total	ć	F 200 000	ćE 200.000	FOurs	¢2 100 000
Pump Station - Extraction to Distribution (Total	1,500 HP	insta	illed HP)	Ş	5,500,000	\$5,500,000	50 yis	\$2,100,000
installed HP, including standby)	750 HP	1130	linearin y	Ś	2.800.000	\$2.800.000	50 vrs	\$1,100.000
Storage Tanks	Cost Estimates	for Storage Tanks	are based on EBM	UD Co	nstruction Bid	Costs		+-,,
Storage Tanks (at extraction well site)		,	EA (Cost Curve by	,				
(material determined during pre-design)	0.1 MG	1	Volume)	\$	2,200,000	\$2,200,000	75 yrs	\$1,300,000
		Estimated S	ubtotal Cost (2) inc	luding	Contingency	\$55,000,000		\$31 000 000
Estimated Subtotal Construction Cost including Contin	ngencv	Lotindicu o	ubtotu: 000t (1)		, contingency	\$180.000.000		\$49.000.000
Mobilization	0. 1				5%	\$9,000,000		\$2,500,000
Estimated Subtotal Construction Cost including Conti	ngency and Mo	bilization				\$190,000,000		\$52,000,000
Implementation								
Planning / Environmental					5%	\$9,500,000		\$2,600,000
Design Cost					15%	\$29,000,000		\$7,800,000
Project Administration and Construction Management	Cost	-			10%	\$19,000,000		\$5,200,000
Estimated Total Capital Cost including implementatio	in and Continge	псу				\$250,000,000 Present Worth of Sa	lvage Value	\$68,000,000
O&M Costs (Annual)								+==,===,===
Advanced Water Treatment								
RO System	8		MGD	\$	480,000	\$3,800,000		
Advanced Oxidation and Disinfection	8		MGD	\$	49,000	\$390,000		
Free Chlorine	8		MGD	Ş	32,000	\$260,000		
Chemicais	8	2 090	MGD brs/MCD	Ş	120,000	\$960,000		
Electricity	0	7 240 000	kWh/vr	ç	0.15	\$1,200,000		
Monitoring		7,240,000	\$/vear	ś	100.000	\$100.000		
Groundwater Wells			+//	*		+===)===		
Labor Costs		1,456	hrs	\$	135	\$200,000		
Electricity		5,733,000	kWh/yr	\$	0.15	\$860,000		
Well Consumables		0.5% of wel	l construction cost			\$150,000		
Conveyance								
Annual O&M			2% of construction	n cost		\$890,000		
Pump Stations								
Labor Costs		1,000	hrs	\$	135	\$140,000		
Electricity		8,930,000	kWh/yr	\$	0.15	\$1,300,000		
Pump Station Consumables		5% of pump sta	tion construction c	ost		\$410,000		
Storage Tanks								
Annual O&M		1% of co	nstruction cost			\$22.000		
			Total O&M Costs	(\$/yr)		\$15,000,000		
Annualized Costs (\$ / Year)								
Annualized Capital Costs (\$/Year)	Two payments	per year, spread o	over Project Life			\$13,000,000		
Annualized Salvage Value	Annualized val	ue of present wort	th			-\$1,400,000		
Annual O&M Costs						\$15,000,000		
Total Annualized Cost		CO. 1.5V				\$27,000,000		
Estimated Unit Cost (\$ /AE)	8,0	DU AFT				62 200		
Broakdown of Estimated Unit Cost (\$/AF)		Troatment	\$2.100	_	Distribution	\$3,300		
Dreakdown of Estimated Unit Cost (\$/AF)		reatment	\$2,100		Distribution	\$1,200		
Annualized Costs (\$ / Year)	Dry Year Adjus	stment (Supply us	ed 3/10 years)				<i>c</i>	
Annualized Capital Costs (\$/Year)	I wo payments	per year, spread o	over Project Life			\$13,000,000	Same as cons	tant use.
Annualized Salvage Value	Annualized val	ue oj present wort	.11			-\$1,400,000 \$12,176,000	same as cons	tunt use.
Total Annualized Cost						\$12,176,000		
Annual Average Deliveries of Recycled Water	2.4	18 AFY						
Estimated Unit Cost (\$/AF)	2,4					\$9,900		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$6,200		Distribution	\$3,700		

Alternative SL-Raw-1						E	BMUD Recycled Water	Master Plan Update
Last Updated:	30-May-18					Discount Rate	Project Life	
Updated by:	I. Jaffe					3%	30 Years	
CCI (20 City, Dec 2017): 10870.06								
Item	Size	Qty	Unit		Unit Cost	Total Cost	Useful Life Sa	lvage Value
Capital Costs				<i>.</i> -				
Ireatment ME/UE system	Cost Estimates	for Treatment are	basea on Unit Cost	s for E	ach Treatment	¢1 700 000	20 μm	ćo
RO System	1		MGD	ç	1,223,000	\$1,700,000	30 yrs	90 \$0
Advanced Oxidation and Disinfection	1		MGD	Ś	437 500	\$610,000	30 yrs	90 \$0
Chemicals (Storage and Use)	1		MIGD	Ś	125 000	\$180,000	30 yrs	\$0 \$0
Sitework/Piping/Structures	1		MGD	\$	3,187,500	\$4,500,000	30 yrs	\$0
			Raw Constru	ction (Cost Subtotal	\$9,100,000		\$0
Contractor Overhead & Profit					15%	\$1,400,000		\$0
Sales Tax	50%	% of Subtotal Cost	Applicable	-	9%	\$410,000		\$0
			Estimated Subtota	l Cons	truction Cost	\$11,000,000		\$0
Construction Cost Including Contingency			Estimated Cubtote	Come	truction Cost	¢11 000 000		ćo
Estimating Contingency			Estimated Subtota	Cons	25%	\$11,000,000		30 \$0
Estimating contingency		Estimated S	ubtotal Cost (1) incl	luding	Contingency	\$14.000.000		\$0
Capital Costs (from Bids)						Ţ_ ,,,		
Conveyance	Cost Estimates	for Pipelines are b	ased on EBMUD cor	nstruc	tion bid cost +	estimating contingency		
Non-urban Pipeline	10-in	300 LF	in-LF	\$	30	\$90,000	75 yrs	\$54,000
Low-Density Urban Pipeline	10-in	18,600 LF	in-LF	\$	40	\$7,440,000	75 yrs	\$4,500,000
High-Density Urban Pipeline	10-in	9,600 LF	in-LF	\$	50	\$4,800,000	75 yrs	\$2,900,000
Trenchless Crossings								
HDD (All Inclusive)		3,350	LF	Ş	2,200	\$7,400,000	75 yrs	\$4,400,000
Pump Stations	Cost Estimates	for Pump Stations	are based on EBML	JD Cor	nstruction Bid	Costs		
Pump Station (Total installed HP, including standby)		EA (Cost Curv	e based on total					
rump station (rotal installed fir, including standby)	225 HP	insta	lled HP)	\$	1,600,000	\$1,600,000	50 yrs	\$640,000
Storage Tanks	Cost Estimates	for Storage Tanks	are based on EBMU	ID Cor	struction Bid	Costs		
Storage Tanks			EA (Cost Curve by					
(material determined during pre-design)	0.2 MG	3	Volume)	Ş	2,500,000	\$7,500,000	75 yrs	\$4,500,000
		Estimated S	ubtotal Cost (2) incl	luding	Contingency	\$29,000,000		\$17,000,000
Estimated Subtotal Construction Cost including Cont	ingency					\$43,000,000		\$17,000,000
Mobilization				-	5%	\$2,200,000		\$850,000
Estimated Subtotal Construction Cost including Cont	ingency and Mo	bilization				\$45,000,000		\$18,000,000
Implementation					E0/	¢2 200 000		\$000,000
Design Cost					15%	\$2,300,000		\$300,000
Project Administration and Construction Managemen	t Cost				10%	\$4,500,000		\$1,800,000
Estimated Total Capital Cost including Implementation	on and Continge	ency				\$59,000,000		\$23,000,000
	_	-				Present Worth of Sal	vage Value	\$9,400,000
O&M Costs (Annual)								
Advanced Water Treatment			MCD	÷	225.000	¢460.000		
NIF/UF system	1		MGD	ې د	325,000	\$460,000		
Advanced Oxidation and Disinfection	1		MGD	ڊ د	480,000	\$69,000		
Free Chlorine	1		MGD	\$	32,000	\$45,000		
Chemicals	1		MGD	\$	120,000	\$170,000		
Labor	1	2,080	hrs/MGD	\$	135	\$390,000		
Electricity		1,267,000	kWh/yr	\$	0.15	\$190,000		
Monitoring			\$/year	\$	500,000	\$500,000		
Conveyance								
Annual O&M			2% of construction	cost		\$390,000		
Pump Stations								
Labor Costs		500	hrs	\$	135	\$68,000		
Electricity		823,000	kWh/yr	\$	0.15	\$120,000		
Pump Station Consumables		5% of pump sta	tion construction co	ost		\$80,000		
Storage Tanks								
Annual O&M		1% of co	nstruction cost			\$75.000		
			Total O&M Costs (\$/yr)		\$3,200,000		
Annualized Costs (\$ / Year)								
Annualized Capital Costs (\$/Year)	Two payments	per year, spread o	over Project Life			\$3,000,000		
Annualized Salvage Value	Annualized val	ue of present wort	h			-\$480,000		
Annual O&M Costs						\$3,200,000		
Iotal Annualized Lost	4					\$5,700,000		
Deliveries of Recycled Water	1,5	/U AFY				¢2 C00		
Breakdown of Estimated Unit Cost (\$/AE)		Treatment	\$2.200		Distribution	\$3,600		
		Treatment	32,200		Distribution	Ş1,400		
Annualized Costs (\$ / Year)	Dry Year Adjus	tment (Supply us	ed 3/10 years)			¢2.000.000.	amo as constant	
Annualized Capital Costs (\$/Year)	I wo payments	per year, spread o	iver Project Life			\$3,000,000 \$	ame as constant use.	
Annual O&M Costs	Annuunzea Van	ie oj present wort				-2400,000 3 \$7 500 500	ume us constant use.	
Total Annualized Cost						\$5,000,000		
Annual Average Deliveries of Recycled Water	47	1 AFY				\$5,000,000		
Estimated Unit Cost (\$/AF)						\$11,000		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$5,900		Distribution	\$5,100		

Alternative SL-Resul-1							FBMUD Recycle	d Water Master Plan Undate
Last Lindated:	30-May-18					Discount Rate	Pro	iert Life
Updated by:	I. Jaffe					3%	30	Years
CCI (20 City, Dec 2017): 10870.06								
Item	Size	Qty	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value
Capital Costs								
Treatment	Cost Estimates	for Treatment are	e based on Unit Co	osts for E	ach Treatmen	t Process	20	ć0.
NF/UF System	1		MGD	Ş	1,225,000	\$1,700,000	30 yrs	\$U ¢0
Advanced Ovidation and Disinfection	1		MGD	ڊ د	1,473,000	\$2,100,000	30 yrs	0Ę \$0
Chemicals (Storage and Use)	1		MGD	ç ç	125 000	\$180,000	30 yrs	30 \$0
Sitework/Piping/Structures	1		MGD	ś	3.187.500	\$4,500,000	30 yrs	\$0 \$0
			Raw Const	ruction	Cost Subtotal	\$9,100,000		\$0
Contractor Overhead & Profit					15%	\$1,400,000		\$0
Sales Tax	50%	% of Subtotal Cost	t Applicable		9%	\$410,000		\$0
			Estimated Subto	tal Cons	truction Cost	\$11,000,000		\$0
Construction Cost Including Contingonau								
Construction Cost including Contingency			Estimated Subto	tal Cons	truction Cost	\$11.000.000		\$0
Estimating Contingency					25%	\$2,800,000		\$0
		Estimated S	Subtotal Cost (1) in	ncluding	Contingency	\$14,000,000		\$0
Capital Costs (from Bids)								
Conveyance	Cost Estimates	for Pipelines are	based on EBMUD	construc	tion bid cost +	estimating contingency	75	
Non-urban Pipeline	10-in	1,800 LF	in-LF	Ş	30	\$540,000	75 yrs	\$320,000
Low-Density Urban Pipeline	10-in	20,900 LF	in-LF	\$	40	\$8,360,000	75 yrs	\$5,000,000
	10-111	5,000 LF	III-LF	Ş	50	\$2,500,000	75 yrs	\$1,500,000
Microtuppel Xings (1 jack & 1 receiv. nit)		9	FΔ	Ś	620,000	\$5,600,000		
Microtunnel Pine		7 000	LA	Ś	2 800	\$20,000,000	75 vrs	\$12,000,000
HDD (All Inclusive)		3,200	LF	ś	2,200	\$7.000.000	75 yrs	\$4,200,000
Dumu Chatlana	Cont Entire store	fan Duran Chatian			_,	(+ ',,
Pump stations	Cost Estimates	FA (Cost Cur	s are based on EBI	VIUD CO	Istruction Bia	COSTS		
Pump Station (Total installed HP, including standby)	300 HP	EA (COSt Cur	alled HP)	Ś	1 800 000	\$1,800,000	50 yrs	\$720.000
	500 111	5.1			2,000,000	\$1,000,000	50 (15	\$7.20,000
Estimated Subtatel Construction Cost including Cost	ingener	Estimated	subtotal Cost (2) II	ncluding	Contingency	\$46,000,000		\$24,000,000
Estimated Subtotal Construction Cost Including Cont	ingency				E 0/	\$60,000,000		\$24,000,000
Estimated Subtotal Construction Cost including Cont	ingency and Mo	hilization			370	\$5,000,000		\$1,200,000
Implementation	ingener and me	2 mzatron				<i>\$66,666,666</i>		<i><i><i>q</i>=0,000,000</i></i>
Environmental Documentation & Permits					5%	\$3,200,000		\$1,300,000
Design Cost					15%	\$9,500,000		\$3,800,000
Project Administration and Construction Managemer	nt Cost				10%	\$6,300,000		\$2,500,000
Estimated Total Capital Cost including Implementati	on and Conting	ency				\$82,000,000		\$33,000,000
						Present Worth of Sa	Ivage Value	\$14,000,000
O&M Costs (Annual)								
MF/UF system	1		MGD	Ś	325 000	\$460,000		
RO System	1		MGD	\$	480,000	\$670,000		
Advanced Oxidation and Disinfection	1		MGD	\$	49,000	\$69,000		
Free Chlorine	1		MGD	\$	32,000	\$45,000		
Chemicals	1		MGD	\$	120,000	\$170,000		
Labor	1	2,080	hrs/MGD	\$	135	\$390,000		
Electricity		1,267,000	kWh/yr	Ş	0.15	\$190,000		
Conveyance								
Annual O&M			2% of construction	on cost		\$880,000		
Pump Stations								
Labor Costs		500	hrs	\$	135	\$68,000		
Electricity		963,000	kWh/yr	\$	0.15	\$140,000		
Pump Station Consumables		5% of pump sta	ation construction	cost		\$90,000		
Annualized Costs (\$ / Year)			Total O&IVI Costs	s (\$/yr)		\$3,200,000		
Annualized Costs (\$7 Tear)	Two navments	ner vear spread	over Proiect Life			\$4,200,000		
Annualized Salvage Value	Annualized val	ue of present wor	th			-\$710,000		
Annual O&M Costs						\$3,200,000		
Total Annualized Cost			_			\$6,700,000		
Deliveries of Recycled Water	1,5	70 AFY						
Estimated Unit Cost (\$/AF)						\$4,300		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$1,900		Distribution	\$2,400		
Annualized Costs (\$ / Year)	Dry Year Adjus	tment (Supply us	ed 3/10 years)					
Annualized Capital Costs (\$/Year)	Two payments	per year, spread	over Project Life			\$4,200,000	same as constar	nt use.
Annualized Salvage Value	Annualized val	ue of present wor	tn			-\$/10,000	same as constar	it use.
Annual U&IVI COSTS						\$2,790,500 \$6,200,000		
Annual Average Deliveries of Recycled Water	47	1 AFY				ş0,500,000		
Estimated Unit Cost (\$/AF)	47					\$13.000		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$5,700		Distribution	\$7,300		

Alternative SL-Chabot-1							EBMUD Recycle	d Water Master Plan Update
Last Updated:	30-May-18					Discount Rate	Pro	piect Life
Updated by:	I. Jaffe					3%	30	Years
CCI (20 City, Dec 2017): 10870.06								
Item	Size	Qty	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value
Capital Costs	6 1 F 11 1	(- , , ,						
ME/UE system	Lost Estimates	jor Treatment are	MGD	sts jor i ¢	1 225 000	\$1 700 000	30 yrs	ŚŊ
RO System	1		MGD	ç ¢	1,223,000	\$1,700,000	30 yrs	30 \$0
Advanced Oxidation and Disinfection	1		MGD	ć	437 500	\$610,000	30 yrs	\$0 \$0
Chemicals (Storage and Use)	1		WIGD	Ś	125.000	\$180.000	30 yrs	\$0
Sitework/Piping/Structures	1		MGD	ś	3.187.500	\$4,500,000	30 yrs	\$0 \$0
			Raw Constr	ruction	Cost Subtotal	\$9,100,000		\$0
Contractor Overhead & Profit					15%	\$1,400,000		\$0
Sales Tax	50%	% of Subtotal Cost	Applicable		9%	\$410,000		\$0
			Estimated Subtot	tal Con	struction Cost	\$11,000,000		\$0
Construction Cost Including Contingency			Fatimated Subtat	al Can	struction Cost	¢11.000.000		ćo
Estimating Contingency			Estimated Subtor	Lai Con	25%	\$11,000,000		50
		Estimated S	ubtotal Cost (1) in	cludin	g Contingency	\$14,000,000		\$0
Capital Costs (from Bids)								
Conveyance	Cost Estimates	for Pipelines are b	ased on EBMUD c	onstru	ction bid cost +	estimating contingency		
Non-urban Pipeline	10-in	4,400 LF	in-LF	\$	30	\$1,320,000	75 yrs	\$790,000
Low-Density Urban Pipeline	10-in	14,500 LF	in-LF	\$	40	\$5,800,000	75 yrs	\$3,500,000
High-Density Urban Pipeline	10-in	7,500 LF	in-LF	\$	50	\$3,750,000	75 yrs	\$2,300,000
Trenchless Crossings								
HDD (All Inclusive)		2,600	LF	Ş	2,200	\$5,700,000	75 yrs	\$3,400,000
Pump Stations	Cost Estimates	for Pump Stations	s are based on EBN	AUD Co	nstruction Bid	Costs		
Pump Station (Total installed HP including standby)		EA (Cost Curv	e based on total					
runp station (rotal installed fir, including standby)	120 HP	insta	illed HP)	\$	1,400,000	\$1,400,000	50 yrs	\$560,000
Storage Tanks	Cost Estimates	for Storage Tanks	are based on EBN	1UD Co	nstruction Bid	Costs		
Storage Tanks			EA (Cost Curve b	y .				
(material determined during pre-design)	0.0 MG	0	Volume)	Ş	2,300,000	ŞO	75 yrs	Ş0
		Estimated S	ubtotal Cost (2) in	cludin	g Contingency	\$18,000,000		\$11,000,000
Estimated Subtotal Construction Cost including Cont	ingency					\$32,000,000		\$11,000,000
Mobilization					5%	\$1,600,000		\$550,000
Estimated Subtotal Construction Cost including Cont	ingency and Mo	bilization				\$34,000,000		\$12,000,000
Implementation								
Environmental Documentation & Permits					5%	\$1,700,000		\$600,000
Design Cost Design Administration and Construction Managemen	t Cost				15%	\$5,100,000		\$1,800,000
Estimated Total Canital Cost including Implementati	on and Conting	ancy			10%	\$3,400,000		\$1,200,000
	on and continge	licy				Present Worth of Sa	alvage Value	\$6,500,000
O&M Costs (Annual)								
Advanced Water Treatment								
MF/UF system	1		MGD	\$	325,000	\$460,000		
RO System	1		MGD	\$	480,000	\$670,000		
Advanced Oxidation and Disinfection	1		MGD	Ş	49,000	\$69,000		
Chomicals	1		MGD	Ş ¢	32,000	\$45,000 \$170,000		
Labor	1	2 080	brs/MGD	ç	120,000	\$170,000		
Electricity	-	1 267 000	kWh/vr	Ś	0.15	\$190,000		
		1,207,000		Ŷ	0.125	<i>\$136,666</i>		
<u>Conveyance</u>			2% of constructio	n cost		¢220.000		
Annual Oxivi			2% of constructio	in cost		\$550,000		
Pump Stations								
Labor Costs		500	hrs	Ş	135	\$68,000		
Electricity		482,000	kWh/yr	Ş	0.15	\$72,000		
Pump Station Consumables		5% of pump sta	Total OSM Costs	cost		\$70,000		
Annualized Casta (\$ / Year)			Total O&IVI Costs	(\$/yr)		\$2,500,000		
Annualized Costs (\$ / Tear)		par year spread a	war Project Life			\$2,200,000		
Annualized Capital Costs (5) Tear)	Annualized val	ue of present wor	h			-\$330,000		
Annual Q&M Costs	, innuanzea van					\$2,500,000		
Total Annualized Cost						\$4,400,000		
Deliveries of Recycled Water	78	0 AFY			-	, , ,		
Estimated Unit Cost (\$/AF)						\$5,600		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$3,800		Distribution	\$1,800		
Annualized Costs (\$ / Year)	Dry Year Adius	tment (Supply us	ed 3/10 years)					
Annualized Capital Costs (\$/Year)	Two payments	per year, spread o	over Project Life			\$2,200,000	Same as consta	nt use.
Annualized Salvage Value	Annualized val	ue of present worl	th			-\$330,000	Same as consta	nt use.
Annual O&M Costs						\$2,200,100		
Total Annualized Cost						\$4,100,000		
Annual Average Deliveries of Recycled Water	23	4 AFY						
Estimated Unit Cost (\$/AF)						\$18,000		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$12,000		Distribution	\$6,000		

Link Undersite in Part Link Part Part Part Part Part Part Part Part	Alternative SL-Treat-1							EBMUD Recycled Water	Master Plan Update
Under 	Last Updated:	30-May-18				Di	scount Rate	Project Life	2
City Die Core 2017: 10070 06 Los Core 2017: 10070 06 <thlos 06<="" 10070="" 2017:="" core="" th=""> <thlos 20<="" core="" td=""><td>Updated by:</td><td>I. Jaffe</td><td></td><td></td><td></td><td>3%</td><td>6</td><td>30 Years</td><td></td></thlos></thlos>	Updated by:	I. Jaffe				3%	6	30 Years	
Inter Op/ Op/ </td <td>CCI (20 City, Dec 2017): 10870.06</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>	CCI (20 City, Dec 2017): 10870.06								
Cartering Control Control factory for Proteomed Protect Proteomed Protect And Control 1 Mode 5 35.300 50 mm 50 And Control 1 Mode 5 35.300 50 mm 50 And Control 1 Mode 5 35.300 80 mm 50 And Control 1 Mode 5 12.20.00 80 mm 50 And Control 1 Mode 5 12.20.00 80 mm 50 And Control 1 Mode 5 12.20.00 80 mm 50 Stands All Profile 1 Mode 5 12.20.00 80 mm 50 Stands All Profile 1 Mode 5 12.20.00 90 mm 50 Stands All Profile 1 Mode 5 12.20.00 90 mm 50 Stands All Profile 1 Mode 5 12.20.00 90 mm 50 Stands All Profile 1 12.20.00 7	Item	Size	Qty	Unit		Unit Cost	Total Cost	Useful Life Sa	alvage Value
Internation Code Data Park Park Park Park Park Park Park Par	Capital Costs						-		
Dir. 1 Mith 2 20200 201000 201000 201000 2010000 2010000 20	Treatment	Cost Estimates	for Treatment are	e based on Unit Co.	sts for .	Each Treatment F	rocess	20 μm	¢ο
Dir 1 MCG 5 1.7.2.00.00 1.00.00 <th1.00.00< th=""> 1.00.00.00 <th1.00.00< th=""></th1.00.00<></th1.00.00<>	BAC	1		MGD	ç	335,300	\$470,000	30 yrs	\$0 \$0
no. syncem 1 Mod 5 1.772.000 52.120.000 Born 50 Central Solutions and Dirichteison 1 Mod 5 1.272.000 52.100.000 Born 50 Central Solutions and Dirichteison 1 Mod 5 1.272.000 55.100.000 Born 50 Contract Central AF Inft 1 Mod 5 1.272.000 55.100 55.000 50 Contract Central AF Inft 10 50 1.072.000 50 50 Contract Central AF Inft 100 1.000.000 50 1.000.000 70 50 Contract Central AF Inft 100 1.000.000 1.000.000 70 70 50	ME/UE system	1		MGD	Ś	1.225.000	\$1,700.000	30 yrs	\$0 \$0
Advance of Divide Cost 1 Moto 5 17.500 5500.000 10 yr 60 Diverse Virging Construction 1 Note 17.500 5500.000 10 yr 60 Diverse Virging Construction 15 510.000.000 10 yr 60 50	RO System	1		MGD	\$	1,475,000	\$2,100,000	30 yrs	\$0
Chandled, Kinger, and Liej 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Advanced Oxidation and Disinfection	1		MGD	\$	437,500	\$610,000	30 yrs	\$0
Site work Program Structures 1 Mol 3 3, 217,000 9, 200,000 0 pm 90 Site Site To 10 0.000 5, 200,000 0 pm 90 Site Site To 100 5, 200,000 0 pm 90 Site Site To 100 5, 555,000 0 pm 90 Site Site To 100 5, 555,000 0 pm 90 Site Site Site Site Site Site Site Site	Chemicals (Storage and Use)	1			\$	125,000	\$180,000	30 yrs	\$0
Base Construction Cost Method 91,000,000 95 Site Is 2011 State Cost Method 91,000,000 95 Construction Cost Method 131,000,000 95 Construction Cost Method 100,000,000 95 Construction Cost Method 100,000,000 95,000,000 95 Construction Cost Method 100,000,000 95,000,000 75 yrs 53,000,000 Migh Detastion Parities 100,000,000 95 yrs 53,000,000 75 yrs 53,000,000 Migh Detastion Parities 100,000,000 100,000,000 100,000,000 100,000,000 100,000,000 Migh Detastion Parities 100,000,000,000,000,000,000,000,000,000	Sitework/Piping/Structures	1		MGD	\$	3,187,500	\$4,500,000	30 yrs	\$0
Cale Set 1.420,000 0.00 Set is if satisfies on parality 1.82,000,000 1.9				Raw Constru	uction	Cost Subtotal	\$10,000,000		\$0
Initia 100 2011 In 00.00001 211.000.0000 201 Center tracking Contingency Estimated Subtact Construction Cost 211.000.000 50 Center tracking Contingency Estimated Subtact Construction Cost 211.000.000 50 Center tracking Contingency Estimated Subtact Cost (1) Including Contingency 513.000.000 70 yrt 513.000 70 yrt 513.0000 513.0000 513.0000	Contractor Overhead & Profit	E 09/	% of Subtotal Cost	Applicable		15%	\$1,500,000		\$0 ¢0
Calculate Contraction Cost and Contraction Cost at 512,000,000 25 Estimated Solution Contraction Cost at 512,000,000 Sign at 20,000,00	Sales Tax	50%	% of Subtotal Cost	Estimated Subtot	al Cons	9%	\$450,000		50 \$0
Set interest of solutional contingency Set is				Estimated Subtot		cruction cost	\$12,000,000		φ¢
Extended Solution So	Construction Cost Including Contingency								
Estimate consignery 25% 53.000.00 59 Construction Constructions of Lip ModuLing Constructions and Distributions of Di				Estimated Subtot	al Cons	truction Cost	\$12,000,000		\$0
Cale Allower and the Allower and Care (Linkeling Contingency Properties and Care at estimating and Care at estimat	Estimating Contingency					25%	\$3,000,000		\$0
Calible Calibratic for Papering one based on TSBUC Calibration bill call residuality contingency " Status Convertised 10% 50000 7 mml 10% 50000 7 mml 10% 510000 Convertised 10% 50000 7 mml 10% 510000 7 mml 10% 510000 7 mml 10% 510000 7 mml 10% 510000 7 mml 10% 5100000			Estimated S	ubtotal Cost (1) in	cluding	Contingency	\$15,000,000		\$0
Cale of Data Statute Cale of Data Statute Statute Statute Statute Statute Big Definition Doin 5,000 F Try to 5,000 D Try to 5,000 D Hig-Definition Doin 5,000 F Try to 5,200 DO Try to 5,200 DO Hig-Definition Doin 5,000 F Try to 5,200 DO Try to 5,200 DO Hig-Definition Cale Gateries for Pamp Stations Cale Gateries for Pamp Stations 5,400 DO Try to 5,400 DO Partial Statute Cale Gateries for Pamp Stations S 2,500 DO Try to 5,400 DO Partial Statute Cale Gateries for Pamp Stations S 2,500 DO Try to 5,400 DO Statute Statute Statute Statute for Pamp Statute Statute for Pamp Statute Statute Pamp Pamp Pamp Pamp Pamp Pamp Pamp Pamp	Capital Costs (from Bids)	Cont Entire to a	for Disaling and						
non-bendpit prisme length 10m 6.000 if nu-f 5 4.00 52,700,000 75 yrs 52,700,000 Trendbest Cosnigs 10m 9,000 if in-f 5 50 54,700,000 75 yrs 52,700,000 Trendbest Cosnigs 100 (Mintelwe) 2,420 us 5 2,000 53,900,000 75 yrs 52,700,000 Pump States Cost Cosnee and Cost Cost Cost Cost	Conveyance	Lost Estimates	for Pipelines are I	in LE	constru	ction bid cost + es	stimating contingency	7E virc	¢180.000
Bit-Dimension Dom 9,0012 m.4 f S 500 \$4,700,000 75 yrs \$2,800,000 Prenches Crossing Cost Extended For Jang Station Cloal Installed HP, Including Extended HP, Including For Jang Station Cloal Installed HP, Including Contingency Tarks or based on EBMAD Construction Bit Close Station P, Station Cloal Installed HP, Including Contingency Tarks or based on EBMAD Construction Bit Close Station Bit	Low-Density Urban Pineline	10-in	6,900 LF	in-LF	ڊ خ	40	\$2 760 000	75 yrs	\$180,000
Three is a cosing in the second of	High-Density Urban Pipeline	10-in	9.400 LF	in-LF	Ś	50	\$4,700,000	75 yrs	\$2,800.000
Index decision Cost Entrong for hum Sotion to solve de EMU Contruction Bill Costs: Section Section Name Section Cost Estimates for Sarange Tanks are based or EMU Contruction Bill Costs: Section	Trenchless Crossings		2, 30 2.		Ý	50	<i>ϕ</i> ,,, cc,000	,	÷2,000,000
Particip Cost & Estimates (or hump Station or blased in DEMUID Construction Bill Cats: Pump Station (Total Installed HP) including standby) 20, IP S 3, 400,000 50 yrs \$ 556,000 Strange Tanis Cost Estimates for hump Station or blased in DEMUID Construction Bill Cats: S 3, 400,000 50 yrs \$ 556,000 Strange Tanis Cost Estimates for hump Station S 3, 500,000 \$ 57,000,000 \$ 51,000,000 \$ 51,000,000 \$ 59,000 \$ 51,000,000 \$ 59,000 \$ 51,000,000 \$ 59,000<	HDD (All Inclusive)		2,420	LF	\$	2,200	\$5,300,000	75 yrs	\$3,200,000
Part Station (Total installed HP), including standb) Different Station (Total installed HP) S 1, 400,000 S 1, 40	Pump Stations	Cost Estimates	for Pump Station	s are based on FRA	NUDCO	onstruction Bid Co	osts		
Pump Dation (101a) (101a) installed HP) I 100 (100 a) S 1,400,000		Loumates	EA (Cost Curv	ve based on total					
Storage Tanks (name of examp) Cost Estimates for Storage Tanks are based and RBMLD Construction RBI Costs Volume) S S 150,000 75 yrs S 45,00,000 Estimated Subtoral Const. Including Contingency Mobilization S12,000,000 75 yrs S13,000,000 S14,000,000 S1	Pump Station (Total installed HP, including standby)	120 HP	insta	alled HP)	\$	1,400,000	\$1,400,000	50 yrs	\$560,000
Storage Tariks interent determined durg ner design interent design	Storage Tanks	Cost Estimates	for Storage Tanks	s are based on EBN	ЛUD Co	nstruction Bid Co	sts		
Instended atomage damage dam	Storage Tanks			EA (Cost Curve b	y				
Estimated Subtoal Cost (2) including Contingency 532,000,000 531,000,000 Mobilization 5% 531,000,000 551,000,000 Stimated Subtoal Construction Cost including Contingency and Mobilization 5% 531,000,000 551,000,000 Indemnetation 5% 531,000,000 551,000,000 551,000,000 Design Cost 15% 559,000,000 521,000,000 521,000,000 Design Cost 15% 559,000,000 521,000,000 521,000,000 Design Cost 75% 551,000,000 52	(material determined during pre-design)	0.2 MG	3	Volume)	\$	2,500,000	\$7,500,000	75 yrs	\$4,500,000
Estimate Subtrail Construction Cost including Contingency and Mobilization 5% 53,000,000 5553,000 Estimate Subtrail Construction Cost including Contingency and Mobilization 5% 53,000,000 531,000,000 Design Cost 5% 53,000,000 531,000,000 531,000,000 Design Cost 5% 53,000,000 531,000,000 531,000,000 Design Cost 15% 53,000,000 531,000,000 531,000,000 Design Cost 10% 53,000,000 531,000,000 531,000,000 Design Cost Standon Cost Standon Cost Standon Cost Standon Cost Ender Matter Treatment. 10% S 125,000 Standon Cost Standon Cost Conce 1 MGD S 125,000 Standon Cost St			Estimated S	ubtotal Cost (2) in	cluding	Contingency	\$22,000,000		\$13,000,000
Mobilization 5% \$1,000,000 \$560,000 implementation 5% \$28,000,000 \$57,000,000 invisonmental focumentation & Permits 5% \$2,000,000 \$21,000,000 Decign Cost 15% \$3,000,000 \$21,000,000 \$21,000,000 Decign Cost 15% \$3,000,000 \$21,000,000	Estimated Subtotal Construction Cost including Cont	tingency					\$37,000,000		\$13,000,000
Estimate Subtrait Construction Cost including Contingency and Mobilization \$39,000,000 \$31,000,000 \$31,000,000 Environmental Documentation & Permits 5% \$2,000,000 \$21,000,000 \$21,000,000 Design Cost 10% \$3,000,000 \$21,000,000 \$	Mobilization					5%	\$1,900,000		\$650,000
Implementation Decision cost 5% \$2,000,000 \$700,000 Decision cost 15% \$2,000,000 \$21,000,000 Strike Cost 15% \$2,000,000 \$21,000,000 Strike Cost 55,000,000 \$21,000,000 \$21,000,000 Strike Cost 55,000,000 \$21,000,000 \$21,000,000 Strike Cost Strike Cost \$51,000,000 \$21,000,000 Cost Strike Cost \$20,000,000 \$21,000,000 Cost Strike Cost \$20,000,000 \$21,000,000 Cost 1 MGD \$12,000 \$160,000 MG/UF system 1 MGD \$22,000 \$170,000 Advanced Didation and Disinfection 1 MGD \$23,000 \$54,000 Feet Choinine 1 MGD \$23,000 \$54,000 \$57,000 Advanced Didation and Disinfection 1 MGD \$23,000 \$1,000,000 Strike Water Treatment \$20,000 \$1,000,000 \$1,000,000 \$1,000,000 Strike Water Treatment \$	Estimated Subtotal Construction Cost including Cont	tingency and M	obilization				\$39,000,000		\$14,000,000
Design Cost Design Cost Design Cost S ± 500,000 S ± 1,000,000 Bit mated Total Capital Cost including Implementation and Contingency S ± 0,000,000 S ± 0,000,000 GRM Costs (Annual) S ± 0,000,000 S ± 0,000,000 S ± 0,000,000 GRM Costs (Annual) S ± 0,000,000 S ± 0,000,000 S ± 0,000,000 GRM Costs (Annual) M GE S ± 7,000 S ± 0,000,000 S ± 0,000,000 MAxing (M star (Testment) M GE S ± 25,000 S ± 0,000,000 S ± 0,000,000 MK (JF system 1 M GE S ± 232,000 S ± 0,000,000 S ± 0,000,000 MK (JF system) 1 M GE S ± 232,000 S ± 0,000 S ± 0,000 Advanced Valciation and Disinfection 1 M GE S ± 120,000 S ± 0,000 S ± 0,000 Chemicals 1 2,000 htts/M MG S ± 135 S ± 0,000 S ± 0,000 S WT CO&M (Conta) S ± 11 M G S 70 S ± 0,000 S ± 0,000 S WT CO&M (Conta) S ± 0 M K of construction cost S ± 0,000 S ± 0,000 M = 0,000 S treat contal S ± 0	Implementation Environmental Documentation & Permits					5%	\$2,000,000		\$700.000
Dip 10% \$ 390,000 \$ 31,00,000 Estimated total Capital Cost including implementation and Contingency 98,000 98,000 98,000 ORM Costs function 1 MGD \$ 57,000,000 \$ 58,000,000 Oran ore 1 MGD \$ 57,000 \$ 58,000,000 BAC 1 MGD \$ 52,000,000 \$ 58,000,000 MFUF System 1 MGD \$ 23,2000 \$ 54,000,000 RAS concerned Vater Treatment 1 MGD \$ 440,000 \$ 56,0000 RAS system 1 MGD \$ 32,000 \$ 56,000 RAS system 1 MGD \$ 32,000 \$ 51,000,000 Labor 1 Z,000 \$ 51,000,000 \$ 50,000 Chemicals 1 Z,000 \$ 51,000,000 \$ 50,000 Montoring 5,11 MG \$ 70 \$ 36,000 Conversare \$ 2,000 \$ 1,000,000 \$ 51,000,000 \$ 50,000 Montoring \$ 2,700 \$ 1,700,000 \$ 50,000 \$ 50,000	Design Cost					15%	\$5,900,000		\$700,000
Estimated Total Capital Cost including Implementation and Contingency \$51.000.000 Present Worth of Salvage Value \$51.800.000 SPresent Worth of Salvage Value \$51.800.000 SPresent Worth of Salvage Value \$51.000.000 \$51.800.000 Octore BAC 1 MGD \$57.000 \$80.000 \$16.000 </td <td>Project Administration and Construction Managemen</td> <td>nt Cost</td> <td></td> <td></td> <td></td> <td>10%</td> <td>\$3,900,000</td> <td></td> <td>\$1,400,000</td>	Project Administration and Construction Managemen	nt Cost				10%	\$3,900,000		\$1,400,000
CRAW Costs (Annual) Present Worth of Salvage Value \$7,400,000 Advanced Water Treatment. Conne 1 MGD \$57,000 \$80,000 <td>Estimated Total Capital Cost including Implementati</td> <td>on and Conting</td> <td>ency</td> <td></td> <td></td> <td></td> <td>\$51,000,000</td> <td></td> <td>\$18,000,000</td>	Estimated Total Capital Cost including Implementati	on and Conting	ency				\$51,000,000		\$18,000,000
ObsM costs (Annual) Advanced Water Treatment. Ozone 1 MGD \$ 57,000 \$ \$80,000 MAC 1 MGD \$ 325,000 \$ \$ \$60,000 MF/U System 1 MGD \$ 325,000 \$ \$ \$60,000 Advanced Vater Treatment 1 MGD \$ 480,000 \$ \$57,000 Advanced Vater Treatment 1 MGD \$ 120,000 \$ \$17,000 Chemicals 1 MGD \$ 120,000 \$ \$17,000 Labor 1 2,080 hrs/MGD \$ 135 \$ \$260,000 Monitoring 1,740,200 kWh/yr \$ 0.15 \$ \$260,000 Swrt O&M (Onida) 511 MG \$ 70 \$ \$ 36,000 Conveyance 2% of construction cost \$ \$ \$ \$ \$ \$ 260,000 \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$							Present Worth of Sa	alvage Value	\$7,400,000
Advanced Water Treatment Statument Done 1 MGD \$ \$7,000 \$80,000 BAC 1 MGD \$ \$25,000 \$560,000 RO System 1 MGD \$ \$40,000 \$570,000 RO System 1 MGD \$ \$40,000 \$570,000 RO System 1 MGD \$ \$22,000 \$45,000 Free Chlorine 1 MGD \$ \$22,000 \$45,000 Chemicals 1 MGD \$ \$20,000 \$170,000 Labor 1,20,80 hrs/MGD \$ 135 \$3390,000 Electricity 1,740,200 kWh/yr \$ 0.15 \$2560,000 Monitoring SVT O&M (Orinda) \$11 MG \$ 70 \$36,000 SWT O&M (Orinda) \$11 MG \$ 70 \$36,000 Pump Station Consumables 2½ of construction cost \$260,000 Pump Station Consumables 5% of pump station construction cost \$70,000 Storage Tank \$39,000 \$39,000 Annual O&M 1% of construction cost \$75,000 Annuallo &M mualled Cost (\$/Yer) \$39,900,00	O&M Costs (Annual)								
Date 1 mbd 3 31,000 \$26,000 Mf/L F system 1 Mbd 5 315,000 \$346,000 MS ystem 1 Mbd 5 325,000 \$460,000 Advanced Oxidation and Disinfection 1 Mbd \$49,000 \$569,000 Advanced Oxidation and Disinfection 1 Mbd \$12,000 \$170,000 Chemicals 1 Mbd \$12,000 \$170,000 Labor 1,740,200 kWM/yr \$0.15 \$260,000 Switt O&M (Orinda) 511 MG \$70 \$36,000 Conveyance	Advanced Water Treatment	1		MGD	ć	E7 000	¢90.000		
MF (J) F yatem 1 MGD \$ 32,000 \$ 546,000 RO System 1 MGD \$ 480,000 \$ 577,000 Advanced Oxidation and Disinfection 1 MGD \$ 480,000 \$ 577,000 Tree Chorine 1 MGD \$ 32,000 \$ 545,000 Chemicais 1 MGD \$ 32,000 \$ 5170,000 Labor 1 2,080 hrs/MGD \$ 135 \$ 5380,000 Motoring 1,740,200 hrs/MGD \$ 135 \$ 5360,000 Montoring Sylvear \$ 1,000,000 \$ 1,000,000 \$ 1,000,000 Swifee Water Treatment Sylvear \$ 1,000,000 \$ 546,000 Swifee Water Treatment Store \$ 2560,000 \$ 576,000 Swifee Water Treatment Store \$ 500 hrs< \$ 135	BAC	1		MGD	ŝ	116 000	\$160,000		
RC System 1 MGD \$ 49,000 \$\$70,000 Advanced Oxidation and Disinfection 1 MGD \$ 32,000 \$\$45,000 Chemicals 1 MGD \$ 120,000 \$\$170,000 Labor 1,740,200 hr/mKD \$ 135 \$330,000 Electricity 1,740,200 kWh/rr \$ 0.15 \$260,000 SWT O&M (Orinda) 511 MG \$ 70 \$36,000 Conveyance Sylvera \$ 1,000,000 \$36,000 Pump Station S Stor 0 mr s \$ 135 \$566,000 Pump Station Consumables 5% of pump station construction cost \$ 570,000 Annual O&M 2% of construction cost \$ 570,000 Storage Tanks Stor 0 mr s \$ 135 \$ 580,000 Annual O&M 1% of construction cost \$ 570,000 \$ 570,000 Annual O&M 1% of construction cost \$ 570,000 \$ 300,000 Annual O&M 1% of construction cost \$ 570,000 \$ 300,000 Annual O&M 1% of construction cost \$ 570,000 <td>MF/UF system</td> <td>1</td> <td></td> <td>MGD</td> <td>Ś</td> <td>325.000</td> <td>\$460.000</td> <td></td> <td></td>	MF/UF system	1		MGD	Ś	325.000	\$460.000		
Advanced Oxidation and Disinfection 1 MGD \$ 49,000 \$69,000 Free Chlorine 1 MGD \$ 120,000 \$170,000 Labor 1 2,080 hrs/MGD \$ 135 \$539,000 Electricity 1,740,200 KWh/Yr \$ 0.15 \$5260,000 Montoring SVT 0&M (trinda) \$11 MG \$ 70 \$36,000 Conveyant Annual 0&M 2% of construction cost \$260,000 \$260,000 Annual 0&M 2% of construction cost \$260,000 \$260,000 Pump Stations 2% of construction cost \$260,000 \$270,000 Pump Station Consumables 5% of pump station construction cost \$77,000 \$75,000 Pump Station Consumables 5% of pump station construction cost \$75,000 \$75,000 Annual 0AM 1% of construction cost \$53,00,000 \$380,000 \$380,000 Annual 0AM 1% of construction cost \$55,000,000 \$390,000 \$390,000 Annual 0AM 1% of construction cost \$57,000 \$380,000 \$380,000 Annual 0AM 1% of construction cost \$5,000,000 </td <td>RO System</td> <td>1</td> <td></td> <td>MGD</td> <td>\$</td> <td>480,000</td> <td>\$670,000</td> <td></td> <td></td>	RO System	1		MGD	\$	480,000	\$670,000		
Free Chlorine 1 MGD \$ 32,000 \$ \$45,000 Chemicals 1 2,080 hrs/MGD \$ 125 \$ \$380,000 Labor 1 2,080 hrs/MGD \$ 135 \$ \$380,000 Electricity 1,740,200 kWh/yr \$ 0.15 \$ \$266,000 Monitoring S/year \$ 1,000,000 \$ \$1,000,000 SWT 0&M (forida) \$ \$11 MG \$ 70 \$ \$36,000 Conveyance	Advanced Oxidation and Disinfection	1		MGD	\$	49,000	\$69,000		
Chemicals 1 MGD S 120,000 \$170,000 Labor 1 2,080 hrs/MGD \$ 135 \$3390,000 Electricity 1,740,200 k/Wh/yr \$ 0.000 \$1,000,000 Surface Water Treatment Stronge Treatment Stronge Treatment \$ 1,000,000 Swr D&M (Grinda) \$11 MG \$ 70 \$36,000 Omeverace	Free Chlorine	1		MGD	\$	32,000	\$45,000		
Labor 1 2,080 hrs/McD 5 135 5,390,000 Monitoring 1,740,200 k/W/W/y 5 0.15 5,260,000 Monitoring S/Y O&M (Orinda) 511 MG 5 70 \$36,000 Convergance Annual 0&M 2% of construction cost \$260,000 \$200,000 Monitoring S/Y O&M (Orinda) 511 MG \$70 \$36,000 Convergance Site Site \$260,000 Pump Station Consumables \$260,000 Pump Station Consumables 500 hrs \$135 \$569,000 Pump Station Consumables 5% of pump station construction cost \$77,000 \$77,000 Storage Tanks Total 0&M Costs (\$/yrar) \$79,000 \$3,900,000 Annualized Costs (\$/yrar) \$3,900,000 Annualized Costs (\$/Year) Two poyments per year, spread over Project Life \$3,800,000 \$3,800,000 Annualized Salvage Value Annualized value of present worth \$3,800,000 \$3,800,000 Annualized Capital Costs (\$/Year) Two poyments per year, spread over Pr	Chemicals	1		MGD	\$	120,000	\$170,000		
LifeCirclity 1,740,200 kWn/yr S 0.15 S.260,000 SWT G&M (Orinda) S11 MG S 70 S36,000 Conveyance Annual O&M 2% of construction cost \$260,000 Pump Stations S 135 \$68,000 Labor Costs 500 hrs S 135 \$68,000 Pump Station Consumables 5% of pump station construction cost \$70,000 \$70,000 Storage Tanks Total O&M Costs (\$/yra) \$3,900,000 \$33,900,000 Annual C&M 1% of construction cost \$75,000 Annual C&M 1% of construction cost \$33,900,000 Annual C&M Costs (\$ /Year) Two payments per year, spread over Project Life \$33,900,000 Total Annualized Costs (\$ /Year) Two payments per year, spread over Project Life \$33,900,000	Labor	1	2,080	hrs/MGD	Ş	135	\$390,000		
Midlating Sylear S 1,000,000 Surface Water Treatment SWT 0&M (Orinda) 511 MG S 70 \$36,000 Conveyance	Electricity		1,740,200	kWh/yr	Ş	0.15	\$260,000		
SWT 0 & Mich (Prinda) 511 MG \$ 70 \$36,000 Conveyance Annal 0 & M 2% of construction cost \$260,000 Pump Stations Iabor Costs 500 hrs \$ 135 \$66,000 Electricity 462,000 kWt N/yr \$ 0.15 \$66,000 Pump Station Consumables 5% of pump station construction cost \$770,000 \$770,000 Storage Tanks Total 0 & M Costs (\$/Year) \$75,000 \$3,300,000 Annual 0& 1% of construction cost \$75,000 \$3,300,000 Annualized Costs (\$/Year) Two payments per year, spread over Project Life \$2,600,000 Annualized Costs (\$/Year) Two payments per year, spread over Project Life \$3,300,000 Annual 0 & Costs \$75,000 \$3,300,000 Total Annualized Value of tots (\$/AF) \$5,100,000 \$3,300,000 Costs \$70 Payments per year, spread over Project Life \$3,900,000 Costs \$70 Payments per year, spread over Project Life \$3,900,000 Costs \$70 Payments per year, spread over Project Life \$3,900,000	Ivionitoring			\$/year	Ş	1,000,000	\$1,000,000		
Conveyance Conveyance Annual 0&M 2% of construction cost \$260,000 Pump Stations Image: Cost of the state of the st	SWT O&M (Orinda)	511		MG	Ś	70	\$36.000		
Conversion Station Station Labor Costs 500 hrs \$ 135 \$68,000 Pump Station S 500 hrs \$ 135 \$68,000 Pump Station Consumables 500 hrs \$ 135 \$68,000 Pump Station Consumables 5% of pump station construction cost \$70,000 \$70,000 Storage Tanks Total O&M Costs (\$/yr) \$3,900,000 \$3,900,000 Annual Costs (\$ / Year) Two payments per year, spread over Project Life \$2,600,000 \$3,900,000 Annual Costs (\$/Year) Two payments per year, spread over Project Life \$3,900,000 \$3,900,000 Annual Cost (\$/Year) Two payments per year, spread over Project Life \$2,600,000 \$3,900,000 Annual Cost (\$/Year) Two payments per year, spread over Project Life \$3,900,000 \$3,900,000 Cost (\$/Year) Two payments per year, spread over Project Life \$3,900,000 \$3,900,000 Deliveries of Recycled Water 1,570 AFY \$3,900,000 \$3,900,000 \$3,900,000 Annualized Costs (\$/Year) Dry Year Adjustiment (Supply used 3/10 years) \$3,900,000	Comunication	511		ine	Ŷ	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	\$30,000		
Pump Station 2.4 of Construction Cost 25.0 of Construction Cost Pump Station Costs 500 hrs \$ 135 \$68,000 Electricity 462,000 kWh/yr \$ 0.15 \$69,000 Pump Station Consumables 5% of pump station construction cost \$70,000 Storage Tanks \$75,000 \$70,000 Annual O&M 1% of construction cost \$75,000 Annual Costs (\$ / Year) \$3,900,000 Annualized Costs (\$ / Year) \$3,900,000 Annualized Capital Costs (\$ / Year) \$3,900,000 Annualized Capital Costs (\$ / Year) \$3,900,000 Annualized Cost (\$ / Year) \$3,900,000 Annualized Cost (\$ / Year) Two payments per year, spread over Project Life \$2,600,000 Annualized Cost (\$ / Year) \$3,900,000 \$3,900,000 Cost (\$ / Year) Two payments per year, spread over Project Life \$3,900,000 Deliveries of Recycled Water 1,570 AFY \$3,900 Estimated Unit Cost (\$ / Ker) Treatment \$2,700 Distribution Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Costs (\$ / Year) Treatment \$2,700 Distribution Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$2,600,000 Annualized Costs (\$ / Year) Two paymen	Conveyance Appual O&M			2% of construction	n cost		\$260,000		
Prump stations Labor Costs 500 hrs \$ 135 \$ 68,000 Pump Station Consumables 5% of pump station construction cost \$ 70,000 Storage Tanks \$ 70,000 \$ 775,000 Annual 0&M 1% of construction cost \$ 75,000 Annual 0&M 1% of construction cost \$ 52,600,000 Annual 0&M Annual costs (\$/Year) Two payments per year, spread over Project Life \$ 52,600,000 Annual 0&M Costs \$ 75,000 \$ 53,900,000 \$ 53,900,000 Total Annualized Cost \$ 75,000 \$ 53,900,000 Total Annualized Cost \$ 76 \$ 53,900,000 Total Annualized Cost \$ 76 \$ 53,900,000 Breakdown of Estimated Unit Cost (\$/AF) Treatment \$ 2,700 Distribution \$ 1,200 Annualized Costs (\$/Year) Dry payments per year, spread over Pro				270 or constructio	0050		\$200,000		
Lectricity300ins51.53506,000Pump Station Consumables5% of pump station construction cost\$70,000Storage TanksAnnual 0&M1% of construction cost\$75,000Total 0&M Costs (\$/Year)Storage ValueAnnual 0&M1% of construction cost\$75,000Annual 0&M Costs (\$/Year)Total 0&M Costs (\$/Year)Total 0&M Costs (\$/Year)Two payments per year, spread over Project Life\$2,600,000Annualized Costs (\$/Year)Two payments per year, spread over Project Life\$2,600,000Annualized Costs (\$/Year)Two payments per year, spread over Project Life\$2,600,000Colspan="2">Colspan="2"Colspan	Labor Costs		500	hrc	ć	100	650 000		
Pump Station Consumables 5% of pump station construction cost \$70,000 Storage Tanks 1% of construction cost \$75,000 Annual 0&M 1% of construction cost \$75,000 Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$2,600,000 Annualized Costs (\$ / Year) Annualized value of present worth -\$380,000 Annualized Costs \$3,900,000 \$3,900,000 Total Annualized Cost \$3,900,000 \$3,900,000 Deliveries of Recycled Water 1,570 AFY \$3,900 Estimated Unit Cost (\$/AF) Treatment \$2,700 Distribution \$1,200 Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Costs (\$ / Year) Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$2,600,000 Same as constant use. Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$3,800,000 Same as constant use. Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$3,800,000 Same as constant use. Annualized Costs (\$ / Year) Two payments per year, spread over Project	Electricity		462 000	his kwb/yr	ې د	135	\$08,000 \$69,000		
Storage Tanks 1% of construction cost \$75,000 Annual 0&M 1% of construction cost \$75,000 Annualized Costs (\$ / Year) 33,900,000 Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$2,600,000 Annualized Salvage Value Annualized value of present worth -\$380,000 Annual 0&M Costs \$3,900,000 \$3,900,000 Total Annualized Cost \$3,900,000 \$3,900,000 Annual 0&M Costs \$3,900,000 \$3,900,000 Deliveries of Recycled Water 1,570 AFY \$3,900 Estimated Unit Cost (\$/AF) Treatment \$2,700 Distribution \$1,200 Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) \$3,800,000 Same as constant use. Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$2,600,000 Same as constant use. Annualized Costs (\$ /Year) Two payments per year, spread over Project Life \$3,800,000 Same as constant use. Annualized Costs (\$ /Year) Two payments per year, spread over Project Life \$2,600,000 Same as constant use. Annualized Costs (\$ /Year) Two payments per year, spread over Project Life \$2,500,000 Same as constant use.	Pump Station Consumables		5% of pump sta	ation construction	cost	0.15	\$70.000		
Storage Tanks Annual 0&M 1% of construction cost \$75,000 Total 0&M Costs (\$/Year) \$3,900,000 Annualized Costs (\$/Year) Two payments per year, spread over Project Life \$2,600,000 Annualized Costs (\$/Year) Two payments per year, spread over Project Life \$2,600,000 Annualized Salvage Value Annualized value of present worth -\$380,000 Annualized Costs \$3,900,000 \$3,900,000 Total Annualized Value of present worth -\$380,000 Annualized Cost \$3,900,000 Total Annualized Value of present worth -\$380,000 Total Annualized Cost \$3,900,000 Total Annualized Cost \$3,900,000 Deliveries of Recycled Water 1,570 AFY Estimated Unit Cost (\$/AF) Treatment \$2,700 Distribution \$1,200 Annualized Costs (\$ Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Costs (\$ / Year) Dry Year Adjustment Super do ver Project Life \$2,600,000 Same as constant use. Annualized Costs (\$ / Year) Dry Year Adjustment Super do ver Project Life \$2,600,000 Same as constant							<i></i>		
Annual V&M L's of Construction Cost \$7,000 Total O&M Costs (\$/Year) Annualized Costs (\$/Year) Annualized Costs (\$/Year) Annualized Salvage Value Annualized Value Annualized Value Annualized Value Deliveries of Recycled Water Estimated Unit Cost (\$/AF) Annualized Costs (\$/Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Salvage Value Annualized Salvage Value Annualized Costs (\$/Year) Annualized Costs S1,200 Distribution S1,200 Same as constant use. Annualized Salvage Value Annualized Costs (\$/Year) Annualized Costs (\$/AF) Breakdown of Estimated Unit Cost (\$/AF) Breakdown of Estimated Unit Cost (\$/AF] Areatment \$7,000 Distribution Stant Unit Cost (\$/AF]	Storage Tanks		401 - 5				ć75.000		
Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$2,600,000 Annualized Capital Costs (\$/Year) Two payments per year, spread over Project Life \$2,600,000 Annualized Salvage Value Annualized value of present worth -\$380,000 Annualized Costs \$3,900,000 \$3,900,000 Cotal Annualized Cost \$3,900,000 \$6,100,000 Deliveries of Recycled Water 1,570 AFY \$56,100,000 Breakdown of Estimated Unit Cost (\$/AF) Treatment \$2,700 Distribution \$1,200 Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Costs (\$ / Year) Annualized value of present worth -\$380,000 Same as constant use. Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$2,600,000 Same as constant use. Annualized Costs (\$ / Year) Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$2,600,000 Same as constant use. -\$380,000 Same as constant use. Annualized Costs \$2,776,000 Same as constant use. -\$380,000 Same as constant use. Annualized Cost \$2,776,000 Annual O&M Costs \$2,706,000 Same as constant use. \$5,000,000 Same as constant use. S5,000,000	Annuai O&IVI		1% 01 00	Total OSM Cost	(¢ /)		\$75,000		
Annualized Capital Costs (\$/Year) Two payments per year, spread over Project Life \$2,600,000 Annualized Salvage Value Annualized value of present worth -\$380,000 Annualized Costs Annualized Cost \$3,900,000 Deliveries of Recycled Water 1,570 AFY Estimated Unit Cost (\$/AF) Treatment \$2,700 Distribution \$1,200 Annualized Costs (\$/Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Costs (\$/Year) Two payments per year, spread over Project Life \$2,600,000 Annualized Costs (\$/Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Costs (\$/Year) Two payments per year, spread over Project Life \$2,700,000 Same as constant use. Annualized Costs (\$/Year) Two payments per year, spread over Project Life \$2,700,000 Same as constant use. Annualized Costs (\$/Year) Two payments per year, spread over Project Life \$2,776,000 Total Annualized Cost \$2,776,000 Total Annualized Cost \$2,776,000 Total Annualized Cost \$5,000,000 Same as constant use. Annual O&M Costs \$2,776,000 Total Annualized Cost \$5,000,000 Same as constant use. Annual O&M Costs \$2,776,000 Total Annualized Cost \$2,776,000 Total Annualized Cost \$2,776,000 Total Annualized Cost \$2,776,000 Freatment \$7,000 Distribution \$4,000	Annualized Costs (\$ / Year)			1018100810100313	(2/ 91)		\$3,900,000		
Annualized Salvage Value Annualized value of present worth -\$380,000 Annual O&M Costs \$3,900,000 Total Annualized Cost \$3,900,000 Deliveries of Recycled Water 1,570 AFY Estimated Unit Cost (\$/AF) \$3,900 Breakdown of Estimated Unit Cost (\$/AF) Treatment Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$2,600,000 Same as constant use. Annualized Salvage Value Annualized value of present worth -\$380,000 Same as constant use. Annual O&M Costs \$2,776,000 \$2,776,000 Total Annualized Cost (\$/AF) Treatment \$1,000 Breakdown of Estimated Unit Cost (\$/AF) Treatment \$7,000 Distribution	Annualized Capital Costs (\$/Year)	Two payments	per year, spread	over Project Life			\$2,600,000		
Annual 0&M Costs \$3,900,000 Total Annualized Cost \$6,100,000 Deliveries of Recycled Water 1,570 AFY Estimated Unit Cost (\$/AF) \$3,900 Breakdown of Estimated Unit Cost (\$/AF) Treatment \$2,700 Distribution \$1,200 Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Costs (\$ / Year) Two payments per year, spread over Project Life Annualized Salvage Value Annualized value of present worth Annualized Costs \$2,776,000 Annualized Cost \$2,776,000 Annualized Cost \$2,776,000 Annualized Cost \$2,776,000 Shout Cost (\$/AF) Treatment \$7,000 Distribution \$4,000	Annualized Salvage Value	Annualized val	ue of present wor	th			-\$380,000		
Total Annualized Cost \$6,100,000 Deliveries of Recycled Water 1,570 AFY Estimated Unit Cost (\$/AF) \$3,900 Breakdown of Estimated Unit Cost (\$/AF) Treatment \$2,700 Distribution Annualized Capital Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Capital Costs (\$ / Year) Annualized Capital Costs (\$ / Year) Two payments per year, spread over Project Life \$2,600,000 Same as constant use. Annualized Salvage Value Annualized value of present worth -\$380,000 Same as constant use. Annual O&M Costs \$2,776,000 \$2,776,000 Total Annualized Cost \$2,776,000 \$5,000,000 Stimated Unit Cost (\$/AF) Treatment \$7,000 \$11,000 Breakdown of Estimated Unit Cost (\$/AF) Treatment \$7,000 Distribution	Annual O&M Costs						\$3,900,000		
Deliveries of Recycled Water 1,570 AFY Estimated Unit Cost (\$/AF) Treatment \$2,700 Distribution \$1,200 Annualized Costs (\$/ Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Costs (\$/ Year) Second over Project Life \$2,600,000 Same as constant use. Annualized Salvage Value Annualized value of present worth -\$380,000 Same as constant use. Annual O&M Costs \$2,776,000 \$2,776,000 Total Annualized Cost \$2,776,000 \$5,000,000 Annual Verage Deliveries of Recycled Water 471 AFY \$11,000 Breakdown of Estimated Unit Cost (\$/AF) Treatment \$7,000 Distribution	Total Annualized Cost						\$6,100,000		
Istimated Unit Cost (\$/AF) Treatment \$3,900 Breakdown of Estimated Unit Cost (\$/AF) Treatment \$2,700 Distribution \$1,200 Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$2,600,000 Same as constant use. Annualized Salvage Value Annualized value of present worth -\$380,000 Same as constant use. Annual Q&M Costs \$2,776,000 \$2,776,000 Total Annualized Cost \$5,000,000 Same as constant use. Annual Verage Deliveries of Recycled Water 471 AFY \$11,000 Breakdown of Estimated Unit Cost (\$/AF) Treatment \$7,000 Distribution \$4,000	Deliveries of Recycled Water	1,5	70 AFY				A		
Breakdown of tstimated Unit Cost (\$/AF) Treatment \$2,700 Distribution \$1,200 Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years)	Estimated Unit Cost (\$/AF)			40.000			\$3,900		
Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Capital Costs (\$ / Year) Two payments per year, spread over Project Life \$2,600,000 Same as constant use. Annualized Salvage Value Annualized value of present worth -\$380,000 Same as constant use. Annual Xerge Deliveries of Recycled Water 471 AFY \$5,000,000 Estimated Unit Cost (\$/AF) Treatment \$7,000 Distribution Standown of Estimated Unit Cost (\$/AF) Treatment \$7,000 Distribution	Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$2,700		Distribution	\$1,200		
Annualized Capital Costs (\$/Year) Two payments per year, spread over Project Life \$2,600,000 Same as constant use. Annualized Salvage Value Annualized value of present worth -\$380,000 Same as constant use. Annual 0&M Costs \$2,776,000 Same as constant use. \$2,776,000 Total Annualized Cost \$2,776,000 Same as constant use. Annual Average Deliveries of Recycled Water 471 AFY	Annualized Costs (\$ / Year)	Dry Year Adjus	tment (Supply us	ed 3/10 years)					
Annualized salvage Value Annualized value of present worth -\$380,000 Same as constant use. Annual 0&M Costs \$2,776,000 Total Annualized Cost \$5,000,000 Annual Average Deliveries of Recycled Water 471 AFY Estimated Unit Cost (\$/AF) \$11,000 Breakdown of Estimated Unit Cost (\$/AF) Treatment \$7,000 Distribution	Annualized Capital Costs (\$/Year)	Two payments	per year, spread	over Project Life			\$2,600,000	Same as constant use.	
Annual Josin Costs \$2,776,000 Total Annualized Cost \$5,000,000 Annual Average Deliveries of Recycled Water 471 AFY Estimated Unit Cost (\$/AF) \$11,000 Breakdown of Estimated Unit Cost (\$/AF) Treatment \$7,000 Distribution	Annualized Salvage Value	Annualized val	ue of present wor	th			-\$380,000	Same as constant use.	
Annual Average Deliveries of Recycled Water 471 AFY Estimated Unit Cost (\$/AF) Breakdown of Estimated Unit Cost (\$/AF) Treatment \$7,000 Distribution \$4,000	Annual U&M Costs						\$2,776,000 \$5,000,000		
Estimated Unit Cost (\$/AF) \$11,000 Breakdown of Estimated Unit Cost (\$/AF) Treatment \$7,000 Distribution \$4,000	Annual Average Deliveries of Recycled Water	/7	1 AFY				\$5,000,000		
Breakdown of Estimated Unit Cost (\$/AF) Treatment \$7,000 Distribution \$4,000	Estimated Unit Cost (\$/AF)	47					\$11.000		
	Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$7,000		Distribution	\$4,000		

Alternative Pin-Raw-2						E	BMUD Recycled Water	Master Plan Update
Last Updated:	30-May-18					Discount Rate	Project Life	
Updated by:	I. Jaffe					3%	30 Years	
CCI (20 City, Dec 2017): 10870.06	Sizo	Otv	Unit		Init Cost	Total Cost	Licoful Lifo S:	lvage Value
Capital Costs	5120	Q(y	Onit		onit cost	Total Cost		ivage value
Treatment	Cost Estimates	for Treatment are	based on Unit Cost	s for E	ach Treatmen	nt Process		
MF/UF system	2		MGD	\$	1,225,000	\$2,100,000	30 yrs	\$0
RO System	2		MGD	\$	1,475,000	\$2,500,000	30 yrs	\$0
Advanced Oxidation and Disinfection	2		MGD	Ş	437,500	\$740,000	30 yrs	\$0 ¢0
Sitework/Pining/Structures	2		MGD	Ş Ç	3 187 500	\$210,000	30 yrs	\$0 \$0
Sitework/ riping/sit detailes	2		Raw Constru	ction (Cost Subtotal	\$11,000,000	30 413	\$0
Contractor Overhead & Profit					15%	\$1,700,000		\$0
Sales Tax	50%	% of Subtotal Cost	Applicable		9%	\$500,000		\$0
			Estimated Subtota	l Cons	truction Cost	\$13,000,000		\$0
Construction Cost Including Contingency								
construction cost metalling contingency			Estimated Subtota	l Cons	truction Cost	\$13,000,000		\$0
Estimating Contingency					25%	\$3,300,000		\$0
		Estimated S	ubtotal Cost (1) incl	luding	Contingency	\$16,000,000		\$0
Capital Costs (from Bids)	Cost Estimates	fan Dinalinaa ana k						
Conveyance Non-urban Pineline	Lost Estimates	for Pipelines are t	in-LE	nstruci ¢	ion bia cost +	estimating contingency	75 yrs	\$1,200,000
Low-Density Urban Pipeline	10-in	22.100 LF	in-LF	Ś	40	\$8.840.000	75 yrs	\$5,300,000
High-Density Urban Pipeline	10-in	400 LF	in-LF	\$	50	\$200,000	75 yrs	\$120,000
Trenchless Crossings								
HDD (All Inclusive)		1,100	LF	\$	2,200	\$2,400,000	75 yrs	\$1,400,000
Pump Stations	Cost Estimates	for Pump Stations	are based on EBML	JD Cor	struction Bid	Costs		
Pump Station (Total installed HP, including standby)		EA (Cost Curv	e based on total					
	180 HP	insta	lled HP)	\$	1,600,000	\$1,600,000	50 yrs	\$640,000
Storage Tanks	Cost Estimates	for Storage Tanks	are based on EBMU	JD Con	struction Bid	Costs		
Storage Tanks	0.2 MG	2	EA (Cost Curve by Volume)	ć	2 500 000	\$7,500,000	75 yrs	\$4 500 000
	0.2 1010	5		,	2,300,000	\$7,500,000	75 915	\$4,500,000
Ectimated Subtatal Construction Cost including Cont	ingongy	Estimated S	ubtotal Cost (2) incl	luding	Contingency	\$23,000,000		\$13,000,000 \$13,000,000
Mobilization	ingency				5%	\$35,000,000		\$13,000,000
Estimated Subtotal Construction Cost including Cont	ingency and Mo	bilization			•,-	\$41,000,000		\$14,000,000
Implementation								
Environmental Documentation & Permits					5%	\$2,100,000		\$700,000
Design Cost	+ C+				15%	\$6,200,000		\$2,100,000
Project Administration and Construction Managemen	n and Conting	ancy			10%	\$4,100,000		\$1,400,000
	on una continge	licy				Present Worth of Sa	lvage Value	\$7,400,000
O&M Costs (Annual)								
Advanced Water Treatment	2		MCD	ć	225.000	Ć550.000		
RO System	2		MGD	Ş Ş	325,000 480.000	\$550,000		
Advanced Oxidation and Disinfection	2		MGD	Ş	49,000	\$83,000		
Free Chlorine	2		MGD	\$	32,000	\$54,000		
Chemicals	2		MGD	\$	120,000	\$200,000		
Labor	2	2,080	hrs/MGD	Ş	135	\$480,000		
Monitoring		1,538,500	KWN/Yr Ś/wear	Ş ¢	500.000	\$230,000		
C			ə/ year	Ŷ	500,000	\$500,000		
Conveyance Appual Q&M			2% of construction	cost		\$270.000		
			270 01 001301 00001	COSC		\$270,000		
Labor Costs		500	brs	ć	125	\$68,000		
Electricity		658.000	kWh/vr	ŝ	0.15	\$99,000		
Pump Station Consumables		5% of pump sta	tion construction co	ost		\$80,000		
Storage Tanks								
Annual O&M		1% of co	nstruction cost			\$75.000		
			Total O&M Costs (\$/yr)		\$3,500,000		
Annualized Costs (\$ / Year)								
Annualized Capital Costs (\$/Year)	Two payments	per year, spread o	over Project Life			\$2,700,000		
Annualized Salvage Value	Annualizea vai	ue of present wort	n			-\$380,000		
Total Annualized Cost						\$5,800,000		
Deliveries of Recycled Water	1,9	00 AFY						
Estimated Unit Cost (\$/AF)						\$3,100		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$2,100		Distribution	\$1,000		
Annualized Costs (\$ / Year)	Dry Year Adjus	stment (Supply us	ed 3/10 years)					
Annualized Capital Costs (\$/Year)	Two payments	per year, spread o	over Project Life			\$2,700,000	Same as constant use.	
Annualized Salvage Value	Annualized val	ue of present wort	'n			-\$380,000	Same as constant use.	
Annuai U&M Costs						\$2,750,900 \$5,100,000		
Annual Average Deliveries of Recycled Water	57	0 AFY				\$3,100,000		
Estimated Unit Cost (\$/AF)						\$8,900		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$5,800		Distribution	\$3,100		
Alternative Bin BosB 2								Water Master Plan Undate
--	-----------------	---------------------	----------------------------	-----------	---------------------	-----------------------	------------------	----------------------------
Arternative Fin-Resp-2	20 May 10				D	in a numb Data	EBINIOD Recycled	i water waster Plan Opuate
Last Updated:	30-May-18				D	iscount Rate	Pro	ect Life
Updated by:	I. Jatte				35	%	30 1	rears
CCI (20 City, Dec 2017): 10870.06	<u>.</u>	0						
Item	Size	Qty	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value
Capital Costs								
<u>Treatment</u>	Cost Estimates	for Treatment are	e based on Unit Co	osts for	Each Treatment F	Process		
MF/UF system	2		MGD	Ş	1,225,000	\$2,100,000	30 yrs	\$0
RO System	2		MGD	Ş	1,475,000	\$2,500,000	30 yrs	\$0
Advanced Oxidation and Disinfection	2		MGD	Ş	437,500	\$740,000	30 yrs	\$0
Chemicals (Storage and Use)	2			\$	125,000	\$210,000	30 yrs	\$0
Sitework/Piping/Structures	2		MGD	\$	3,187,500	\$5,400,000	30 yrs	\$0
			Raw Consti	ruction	Cost Subtotal	\$11,000,000		\$0
Contractor Overhead & Profit					15%	\$1,700,000		\$0
Sales Tax	50%	% of Subtotal Cost	Applicable		9%	\$500,000		\$0
			Estimated Subto	tal Con	struction Cost	\$13,000,000		\$0
Construction Cost Including Contingency								
			Estimated Subto	tal Con	struction Cost	\$13,000,000		\$0
Estimating Contingency					25%	\$3,300,000		\$0
		Estimated S	ubtotal Cost (1) ir	ncludin	g Contingency	\$16,000,000		\$0
Capital Costs (from Bids)								
Conveyance	Cost Estimates	for Pipelines are l	pased on EBMUD c	constru	ction bid cost + es	stimating contingency		
Non-urban Pipeline	10-in	62,900 LF	in-LF	\$	30	\$18,870,000	75 yrs	\$11,000,000
Low-Density Urban Pipeline	10-in	14,100 LF	in-LF	\$	40	\$5,640,000	75 yrs	\$3,400,000
High-Density Urban Pipeline	10-in	800 LF	in-LF	\$	50	\$400,000	75 yrs	\$240,000
Trenchless Crossings								
HDD (All Inclusive)		1,700	LF	\$	2,200	\$3,700,000	75 yrs	\$2,200,000
Dump Stations	Cost Estimatos	for Dump Station	are bacad on FDA		notruction Rid Co	acto		
Pump stations	Cost Estimates	Joi Punip Station	s ure bused on EBN	VIUD CC		1515		
Pump Station (Total installed HP, including standby)	275 110	EA (COST CUIN	/e based on total	ć	1 000 000	¢1 000 000	FOurs	¢760.000
	375 HP	Insta	lileu HP)	Ş	1,900,000	\$1,900,000	50 yrs	\$760,000
		Estimated S	ubtotal Cost (2) ir	ncludin	g Contingency	\$31,000,000		\$18,000,000
Estimated Subtotal Construction Cost including Cont	ingency					\$47,000,000		\$18,000,000
Mobilization					5%	\$2,400,000		\$900,000
Estimated Subtotal Construction Cost including Cont	ingency and Mo	bilization				\$49,000,000		\$19,000,000
Implementation								
Environmental Documentation & Permits					5%	\$2,500,000		\$950,000
Design Cost					15%	\$7,400,000		\$2,900,000
Project Administration and Construction Managemer	nt Cost				10%	\$4,900,000		\$1,900,000
Estimated Total Capital Cost including Implementati	on and Continge	ency				\$64,000,000		\$25,000,000
						Present Worth of Sa	alvage Value	\$10,000,000
O&M Costs (Annual)								
Advanced Water Treatment								
MF/UF system	2		MGD	\$	325,000	\$550,000		
RO System	2		MGD	\$	480,000	\$820,000		
Advanced Oxidation and Disinfection	2		MGD	\$	49,000	\$83,000		
Free Chlorine	2		MGD	Ş	32,000	\$54,000		
Chemicals	2		MGD	Ş	120,000	\$200,000		
Labor	2	2,080	hrs/MGD	\$	135	\$480,000		
Electricity		1,538,500	kWh/yr	\$	0.15	\$230,000		
Surface Water Treatment								
SWT O&M (Orinda)	621		MG	\$	70	\$43,000		
Conveyance								
Annual O&M			2% of construction	on cost		\$570,000		
Dump Stations								
Pump Stations		500	har	ć	125	¢.co. 000		
Labor Costs		500	nrs	Ş	135	\$68,000		
Electricity		1,559,000	kwn/yr	\$	0.15	\$230,000		
Pump Station Consumables		5% of pump sta	Total Office of the second	cost		\$95,000		
			Total O&IVI Costs	s (\$/yr)		\$3,400,000		
Annualized Costs (\$ / Year)						44 444 444		
Annualized Capital Costs (\$/Year)	Two payments	per year, spread o	over Project Life			\$3,300,000		
Annualized Salvage Value	Annualized val	ue of present wor	th			-\$510,000		
Annual O&M Costs						\$3,400,000		
Iotal Annualized Cost		0.451			_	\$6,200,000		
Deliveries of Recycled Water	1,90	JO AFY						
Estimated Unit Cost (\$/AF)						\$3,300		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$1,900		Distribution	\$1,400		
Annualized Costs (\$ / Year)	Dry Year Adius	tment (Supply us	ed 3/10 years)					
Annualized Capital Costs (\$/Year)	Two payments	per year, spread o	over Project Life			\$3,300,000	Same as constan	t use.
Annualized Salvage Value	Annualized val	ue of present wor	th			-\$510.000	Same as constan	t use.
Annual O&M Costs						\$2,893,100		
Total Annualized Cost						\$5,700,000		
Annual Average Deliveries of Recycled Water	57	0 AFY			-	\$3,7 88,000		
Estimated Unit Cost (\$/AF)	57					\$10,000		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$5,600	-	Distribution	\$4,400		
						¥ ., .VV		

Alternative Pin-ResSP-2							EBMUD Recycled	Water Master Plan Lindate
Lact Indated:	20 May 18					Discount Pata	Diviob Recycled	oct Life
Last Opualeu.	50-ividy-16					20/	20 8	loars
CCI (20 City, Dec 2017): 10870.06	i. Jaile					370	501	ears
Item	Size	Otv	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value
Capital Costs		~~~						
Treatment	Cost Estimates	for Treatment are	based on Unit Co	sts for l	Each Treatmen	t Process		
MF/UF system	2		MGD	\$	1,225,000	\$2,100,000	30 yrs	\$0
RO System	2		MGD	\$	1,475,000	\$2,500,000	30 yrs	\$0
Advanced Oxidation and Disinfection	2		MGD	\$	437,500	\$740,000	30 yrs	\$0
Chemicals (Storage and Use)	2			\$	125,000	\$210,000	30 yrs	\$0
Sitework/Piping/Structures	2		MGD	\$	3,187,500	\$5,400,000	30 yrs	\$0
			Raw Consti	ruction	Cost Subtotal	\$11,000,000		\$0
Contractor Overhead & Profit					15%	\$1,700,000		\$0
Sales Tax	50%	% of Subtotal Cost	Applicable		9%	\$500,000		\$0 \$0
			Estimated Subto	tai Con	struction Cost	\$13,000,000		ŞU
Construction Cost Including Contingonsy								
construction cost metuding contingency			Estimated Subto	tal Con	struction Cost	\$13,000,000		ŚŊ
Estimating Contingency			Estimated Subto	cui con	25%	\$3,300.000		\$0 \$0
		Estimated S	ubtotal Cost (1) ir	ncludin	g Contingency	\$16,000,000		\$0
Capital Costs (from Bids)					5 5 7			
Conveyance	Cost Estimates	for Pipelines are b	pased on EBMUD c	constru	ction bid cost +	estimating contingency		
Non-urban Pipeline	10-in	21,800 LF	in-LF	\$	30	\$6,540,000	75 yrs	\$3,900,000
Low-Density Urban Pipeline	10-in	20,800 LF	in-LF	\$	40	\$8,320,000	75 yrs	\$5,000,000
Trenchless Crossings								
HDD (All Inclusive)		1,700	LF	\$	2,200	\$3,700,000	75 yrs	\$2,200,000
Pump Stations	Cost Estimates	for Pump Stations	s are based on EBN	ЛUD Co	onstruction Bid	Costs		
		EA (Cost Curv	ve based on total					
Pump Station (Total installed HP, including standby)	225 HP	insta	illed HP)	\$	1,600,000	\$1,600,000	50 yrs	\$640,000
		Estimated S	ubtotal Cost (2) ir	cludin	Contingency	\$20,000,000		\$12,000,000
Estimated Subtotal Construction Cost including Cont	ingency	Estimated S		leidain	5 contingency	\$36,000,000		\$12,000,000
Mobilization					5%	\$1.800.000		\$600.000
Estimated Subtotal Construction Cost including Cont	ingency and Mo	bilization				\$38,000,000		\$13,000,000
Implementation								
Environmental Documentation & Permits					5%	\$1,900,000		\$650,000
Design Cost					15%	\$5,700,000		\$2,000,000
Project Administration and Construction Managemen	t Cost				10%	\$3,800,000		\$1,300,000
Estimated Total Capital Cost including Implementation	on and Continge	ency				\$49,000,000		\$17,000,000
						Present Worth of Sa	alvage Value	\$7,000,000
O&M Costs (Annual) Advanced Water Treatment								
ME/LIE system	2		MGD	¢	325 000	\$550.000		
RO System	2		MGD	Ş	480,000	\$820,000		
Advanced Oxidation and Disinfection	2		MGD	\$	49,000	\$83,000		
Free Chlorine	2		MGD	\$	32,000	\$54,000		
Chemicals	2		MGD	\$	120,000	\$200,000		
Labor	2	2,080	hrs/MGD	\$	135	\$480,000		
Electricity		1,538,500	kWh/yr	Ş	0.15	\$230,000		
Surface Water Treatment	621		MC	ć	254	¢160.000		
SWI O&W (Sobrante WIP)	621		IVIG	Ş	254	\$160,000		
Conveyance								
Annual O&M			2% of construction	on cost		\$370,000		
Pump Stations								
Labor Costs		500	hrs	\$	135	\$68,000		
Electricity		853,000	kWh/yr	\$	0.15	\$130,000		
Pump Station Consumables		5% of pump sta	tion construction	cost		\$80,000		
			Total O&M Costs	: (\$/yr)		\$3,200,000		
Annualized Costs (\$ / Year)	T		unan Dua ia at Lifa			¢3 500 000		
Annualized Capital Costs (\$/Year)	I wo payments	per year, spread c	over Project Life			\$2,500,000		
	Annuunzeu vui	ue oj present wort	.11			\$3 200,000		
						\$5,200,000		
Deliveries of Recycled Water	1.9	DO AFY						
Estimated Unit Cost (S/AF)						\$2.800		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$1,900		Distribution	\$900		
	Due Ve au Adies		- d 2/40					
Annualized Costs (\$ / Tear)	Two novments	ner vegr spread a	over Project Life			\$2 500 000	Same as constan	t lise
Annualized Salvage Value	Annualized val	ve of present wort	h			-\$360,000	Same as constant	t use.
Annual O&M Costs	, innuunzeu vun	ac of present wort				\$2,683,200	constant	
Total Annualized Cost						\$4,800.000		
Annual Average Deliveries of Recycled Water	57	0 AFY				. ,,	-	
Estimated Unit Cost (\$/AF)						\$8,400		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$5.600		Distribution	\$2,800		

Alternative Pin-Treat-2							EBMUD Recyc	led Water Master Plan Update
Last Updated:	30-May-18				[Discount Rate	P	Project Life
Updated by:	I. Jaffe				З	3%	3	30 Years
CCI (20 City, Dec 2017): 10870.06								
Item	Size	Qty	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value
Treatment	Cost Estimates	for Treatment are	phased on Unit Co	sts for F	ach Treatment	Process		
Ozone	2	joi meatment are	MGD	\$	335,300	\$570,000	30 yrs	\$0
BAC	2		MGD	\$	300,900	\$510,000	30 yrs	\$0
MF/UF system	2		MGD	\$	1,225,000	\$2,100,000	30 yrs	\$0
RO System	2		MGD	\$	1,475,000	\$2,500,000	30 yrs	\$0
Advanced Oxidation and Disinfection	2		MGD	Ş	437,500	\$740,000	30 yrs	\$0 \$0
Sitework/Pining/Structures	2		MGD	ڊ خ	3 187 500	\$210,000	30 yrs	ېن د ۲
Site work, riping, set detailes	-		Raw Const	ruction	Cost Subtotal	\$12,000,000		\$0
Contractor Overhead & Profit					15%	\$1,800,000		\$0
Sales Tax	50%	% of Subtotal Cost	Applicable		9%	\$540,000		\$0
			Estimated Subto	tal Cons	truction Cost	\$14,000,000		\$0
Construction Cost Including Contingency								
construction cost meldung contingency			Estimated Subto	tal Cons	truction Cost	\$14,000,000		\$0
Estimating Contingency					25%	\$3,500,000		\$0
		Estimated S	iubtotal Cost (1) ii	ncluding	Contingency	\$18,000,000		\$0
Capital Costs (from Bids)		(D : /: /	1 500 (110					
Conveyance Non-urban Pineline	Lost Estimates	for Pipelines are b	in-LE	construct s	tion bid cost + e 30	stimating contingency \$480,000	75 vrs	\$290.000
Low-Density Urban Pipeline	10-in 10-in	1,000 LF	in-LF	ŝ	40	\$440,000	75 yrs	\$260.000
Trenchless Crossings		_,				• • • • • • • • • • • • • • • • • • •	,	+,
HDD (All Inclusive)		550	LF	\$	2,200	\$1,200,000	75 yrs	\$720,000
Pump Stations	Cost Estimates	for Pump Stations	s are based on EBN	MUD Cor	nstruction Bid C	osts		
Pump Station (Total installed HP, including standby)		EA (Cost Curv	ve based on total					
Pump station (Total installed HP, including standby)	225 HP	insta	alled HP)	\$	1,600,000	\$1,600,000	50 yrs	\$640,000
Storage Tanks	Cost Estimates	for Storage Tanks	are based on EBN	ЛUD Cor	nstruction Bid Co	osts		
Storage Tanks	0.2 MG	2	EA (Cost Curve t	oy ¢	2 500 000	\$7 E00 000	7E vrc	¢4 500 000
	0.2 1010		voluille)	ې 	2,300,000	\$7,500,000	75 yis	\$4,500,000
Estimated Subtatal Construction Cost including Cont	ingongy	Estimated S	Subtotal Cost (2) ii	ncluding	Contingency	\$11,000,000		\$6,400,000
Mobilization	ingency				5%	\$25,000,000		\$0,400,000
Estimated Subtotal Construction Cost including Cont	ingency and Mc	bilization			•/-	\$31,000,000		\$6,700,000
Implementation								
Environmental Documentation & Permits					5%	\$1,600,000		\$340,000
Design Cost	t Cast				15%	\$4,700,000		\$1,000,000
Estimated Total Capital Cost including Implementation	on and Continge	ncv			10%	\$3,100,000		\$8,700,000
		, ,				Present Worth of Sa	lvage Value	\$3,600,000
O&M Costs (Annual)								
Advanced Water Treatment	2		MGD	ć	E7 000	¢07.000		
BAC	2		MGD	ş S	116.000	\$200.000		
MF/UF system	2		MGD	\$	325,000	\$550,000		
RO System	2		MGD	\$	480,000	\$820,000		
Advanced Oxidation and Disinfection	2		MGD	Ş	49,000	\$83,000		
Chemicals	2		MGD	\$	120,000	\$200,000		
Labor	2	2,080	hrs/MGD	\$	135	\$480,000		
Electricity		2,113,100	kWh/yr	\$	0.15	\$320,000		
Monitoring			\$/year	\$	1,000,000	\$1,000,000		
<u>Conveyance</u>								
Annual O&M			2% of construction	on cost		\$42,000		
Pump Stations								
Labor Costs		500	hrs	\$	135	\$68,000		
Electricity		780,000	kWh/yr	Ş	0.15	\$120,000		
Pump station consumables		5% of pump sta	ition construction	COSL		\$80,000		
Storage Tanks								
Annual O&M		1% of co	Total OSM Cost	- (¢ /µr)		\$75,000		
Annualized Costs (\$ / Year)			Total Oxfor Costs	5 (<i>Ş</i> / ¥I)		\$4,200,000		
Annualized Capital Costs (\$/Year)	Two payments	per year, spread o	over Project Life			\$2,000,000		
Annualized Salvage Value	Annualized val	ue of present wort	th			-\$180,000		
Annual O&M Costs						\$4,200,000		
Total Annualized Cost	1.0	00 451			-	\$6,000,000		
Deliveries of Recycled Water	1,9	UU AFY				\$2.200		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$2.600		Distribution	\$3,200		
Annualized Costs (A (March	Dury Ma		-12/10-		Jistinution	2000		
Annualized Costs (\$ / Year)	Two novments	ner year spread	ea 3/10 years)			\$2,000,000	Same as core	tant use
Annualized Salvage Value	Annualized val	ue of present work	th			-\$180.000	Same as cons	tant use.
Annual O&M Costs		, present work	-			\$3,003,200		
Total Annualized Cost					_	\$4,800,000		
Annual Average Deliveries of Recycled Water	57	0 AFY						
Estimated Unit Cost (\$/AF)			46.000			\$8,400		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$6,800		Distribution	\$1,600		

Alternative Rich-Raw-4							EBMUD Recycle	ed Water Master Plan Update
Last Updated:	30-May-18					Discount Rate	Pr	oject Life
Updated by:	I. Jaffe					3%	30) Years
CCI (20 City, Dec 2017): 10870.06								
Item	Size	Qty	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value
Capital Costs	Cont Entirenter	(T + +	hand on Unit Con			4 Dec		
ME/UE system	Cost Estimates	jor Treatment are	MGD	ts jor i ¢	1 225 000	\$4,400,000	30 yrs	ŚO
RO System	4		MGD	ç	1,225,000	\$5,300,000	30 yrs	90 \$0
Advanced Oxidation and Disinfection	4		MGD	Ś	437.500	\$1,600.000	30 yrs	\$0
Chemicals (Storage and Use)	4			Ş	125,000	\$450,000	30 yrs	\$0
Sitework/Piping/Structures	4		MGD	\$	3,187,500	\$11,000,000	30 yrs	\$0
Other (Specify)		0	EA	\$	500	\$0	30 yrs	\$0
			Raw Constru	uction	Cost Subtotal	\$23,000,000	1	\$0
Contractor Overhead & Profit					15%	\$3,500,000	1	\$0
Sales Tax	50%	% of Subtotal Cost	Applicable		9%	\$1,000,000		\$0
			Estimated Subtota	al Con	struction Cost	\$28,000,000		\$0
Construction Cost Including Contingency								
			Estimated Subtota	al Con	struction Cost	\$28,000,000	1	\$0
Estimating Contingency					25%	\$7,000,000		\$0
		Estimated S	ubtotal Cost (1) inc	ludin	g Contingency	\$35,000,000	1	\$0
Capital Costs (from Bids)		C - 1 1						
Conveyance	Cost Estimates	for Pipelines are b	ased on EBMUD co	onstru	ction bid cost +	estimating contingency	75	ća 200.000
Non-urban Pipeline	16-IN 16 in	11,400 LF	IN-LF	Ş	30	\$5,472,000	75 yrs	\$3,300,000
Low-Density Urban Pipeline	10-III 16-in	7 700 LF	in-LF	ç ç	40	\$19,264,000	75 yrs	\$12,000,000
Trenchless Crossings	10-111	7,700 EI	11-51	Ŷ	50	\$0,100,000	75 yrs	\$5,700,000
HDD (All Inclusive)		2,050	LF	\$	2,200	\$4,500,000	75 yrs	\$2,700,000
Dump Stations	Cost Estimatos	for Dump Stations	are bacad on EPM		nstruction Pid	Costs		
Pump Stations	Cost Estimates	FA (Cost Curv	e based on total	UD CC	instruction biu	COSIS		
Pump Station (Total installed HP, including standby)	375 HP	insta	lled HP)	Ś	1.900.000	\$1,900.000	50 vrs	\$760.000
Storage Tanks	Cost Estimates	for Storage Tanks	are based on EBM	UD Co	nstruction Bid	Costs	,	+ · · · · · · · · · · · · · · · · · · ·
Storage Tanks		, ,	EA (Cost Curve by	/				
(material determined during pre-design)	0.3 MG	3	Volume)	\$	2,600,000	\$7,800,000	75 yrs	\$4,700,000
		Estimated S	ubtotal Cost (2) inc	ludin	g Contingency	\$45,000,000		\$27,000,000
Estimated Subtotal Construction Cost including Cont	ingency		.,			\$80,000,000		\$27,000,000
Mobilization					5%	\$4,000,000		\$1,400,000
Estimated Subtotal Construction Cost including Cont	ingency and Mo	bilization				\$84,000,000		\$28,000,000
Implementation								
Environmental Documentation & Permits					5%	\$4,200,000		\$1,400,000
Design Cost Project Administration and Construction Managemen	t Cost				15%	\$13,000,000		\$4,200,000
Estimated Total Capital Cost including Implementati	on and Conting	ency			10%	\$3,400,000		\$2,800,000
	en una continge					Present Worth of S	alvage Value	\$15,000,000
O&M Costs (Annual)								
Advanced Water Treatment								
MF/UF system	4		MGD	Ş	325,000	\$1,200,000		
RU System Advanced Oxidation and Disinfection	4		MGD	Ş ¢	480,000	\$1,700,000		
Free Chlorine	4		MGD	ś	32.000	\$120.000		
Chemicals	4		MGD	\$	120,000	\$430,000	1	
Labor	4	2,080	hrs/MGD	\$	135	\$1,000,000	1	
Electricity		3,258,000	kWh/yr	\$	0.15	\$490,000	1	
Monitoring			\$/year	\$	500,000	\$500,000	1	
Conveyance								
Annual O&M			2% of construction	n cost		\$710,000	1	
Pump Stations								
Labor Costs		500	hrs	\$	135	\$68,000	1	
Electricity		1,445,000	kWh/yr	\$	0.15	\$220,000	1	
Pump Station Consumables		5% of pump sta	tion construction c	ost		\$95,000		
Storage Tanks								
Annual O&M		1% of co	nstruction cost			\$78,000		
			Total O&M Costs	(\$/yr)		\$6,800,000		
Annualized Costs (\$ / Year)	- .		D 1 111			Å5 600 000		
Annualized Capital Costs (\$/Year)	I wo payments	per year, spread o	iver Project Life			\$5,600,000		
Annual O&M Costs	Annuunzeu vui	ue oj present wort	11			-5700,000 \$6,800,000		
Total Annualized Cost						\$12,000,000		
Deliveries of Recycled Water	4,0	30 AFY					-	
Estimated Unit Cost (\$/AF)						\$3,000		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$2,000		Distribution	\$1,000		
Annualized Costs (\$ / Year)	Dry Year Adius	stment (Supply us	ed 3/10 vears)					
Annualized Capital Costs (\$/Year)	Two payments	per year, spread o	ver Project Life			\$5,600,000	Same as consta	ant use.
Annualized Salvage Value	Annualized val	ue of present wort	h			-\$760,000	Same as consta	ant use.
Annual O&M Costs						\$5,559,000		
Total Annualized Cost				_		\$10,000,000		
Annual Average Deliveries of Recycled Water	1,2	09 AFY				-		
Estimated Unit Cost (\$/AF)			45 300	_		\$8,300		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$5,700		Distribution	\$2,600		

Alternative Rich-ResB-4							EBMUD Recycled	Water Master Plan Lindate
Lact Indated:	20 May 18					Niccount Pata	EDIVIOD Recycled	oct Life
Last opuated.	J laffe				2	%	30 V	ect Life
CCI (20 City, Dec 2017): 10870.06	1. Jane				5	70	501	cars
Item	Size	Qtv	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value
Capital Costs								
Treatment	Cost Estimates	for Treatment are	e based on Unit Co	sts for l	Each Treatment	Process		
MF/UF system	4		MGD	\$	1,225,000	\$4,400,000	30 yrs	\$0
RO System	4		MGD	\$	1,475,000	\$5,300,000	30 yrs	\$0
Advanced Oxidation and Disinfection	4		MGD	\$	437,500	\$1,600,000	30 yrs	\$0
Chemicals (Storage and Use)	4			\$	125,000	\$450,000	30 yrs	\$0
Sitework/Piping/Structures	4		MGD	\$	3,187,500	\$11,000,000	30 yrs	\$0
			Raw Const	ruction	Cost Subtotal	\$23,000,000		\$0
Contractor Overhead & Profit					15%	\$3,500,000		\$0
Sales Tax	50%	% of Subtotal Cost	t Applicable		9%	\$1,000,000		\$0
			Estimated Subto	tal Con	struction Cost	\$28,000,000		\$0
Construction Cost Including Contingency								
			Estimated Subto	tal Con	struction Cost	\$28,000,000		\$0
Estimating Contingency					25%	\$7,000,000		\$0
		Estimated S	Subtotal Cost (1) ir	ncludin	g Contingency	\$35,000,000		\$0
Capital Costs (from Bids)								
Conveyance	Cost Estimates	for Pipelines are	based on EBMUD o	constru	ction bid cost + e	stimating contingency		
Non-urban Pipeline	16-in	38,800 LF	in-LF	\$	30	\$18,624,000	75 yrs	\$11,000,000
Low-Density Urban Pipeline	16-in	18,000 LF	in-LF	\$	40	\$11,520,000	75 yrs	\$6,900,000
High-Density Urban Pipeline	16-in	9,100 LF	in-LF	\$	50	\$7,280,000	75 yrs	\$4,400,000
Trenchless Crossings								
HDD (All Inclusive)		2,450	LF	\$	2,200	\$5,400,000	75 yrs	\$3,200,000
Pump Stations	Cost Estimates	for Pump Station	s are based on FBN		Instruction Bid Co	osts		
	cost Estimates	FA (Cost Cur	ve based on total					
Pump Station (Total installed HP, including standby)	900 HP	insta	alled HP)	Ś	3.300.000	\$3,300,000	50 yrs	\$1,300.000
						+-,		+-,,
		Estimated S	Subtotal Cost (2) in	ncludin	g Contingency	\$46,000,000		\$27,000,000
Estimated Subtotal Construction Cost including Cont	ingency				50/	\$81,000,000		\$27,000,000
					5%	\$4,100,000		\$1,400,000
Estimated Subtotal Construction Cost including Cont	ingency and ivio	bilization				\$85,000,000		\$28,000,000
Implementation					50/	ć4 200 000		ć1 400 000
Environmental Documentation & Permits					5%	\$4,300,000		\$1,400,000
Design Cost	t Cost				15%	\$15,000,000		\$4,200,000
Project Administration and Construction Managemen	i Cusi	-			10%	\$6,500,000		\$2,800,000
Estimated Total Capital Cost Including Implementation	on and continge	ency				\$110,000,000 Present Worth of S	alvage Value	\$36,000,000
O&M Costs (Annual)							and ge tulue	<i><i><i></i></i></i>
Advanced Water Treatment								
MF/UF system	4		MGD	\$	325,000	\$1,200,000		
RO System	4		MGD	\$	480,000	\$1,700,000		
Advanced Oxidation and Disinfection	4		MGD	\$	49,000	\$180,000		
Free Chlorine	4		MGD	\$	32,000	\$120,000		
Chemicals	4		MGD	\$	120,000	\$430,000		
Labor	4	2,080	hrs/MGD	\$	135	\$1,000,000		
Electricity		3,258,000	kWh/yr	\$	0.15	\$490,000		
Surface Water Treatment								
SWT O&M (Orinda)	1,314		MG	\$	70	\$92,000		
Conveyance								
Annual O&M			2% of construction	on cost		\$860,000		
Pump Stations								
Labor Costs		500	hrs	Ś	135	\$68 000		
Electricity		3 199 000	kWh/vr	Ś	0.15	\$480,000		
Pump Station Consumables		5% of nump st	ation construction	cost	0.125	\$170,000		
		570 of pump st	Total O&M Costs	s (\$/vr)		\$6.800.000		
Annualized Costs (\$ / Year)				. (+/ 1-/		++,,		
Annualized Capital Costs (S/Year)	Two payments	per year. spread	over Proiect Life			\$5.600.000		
Annualized Salvage Value	Annualized val	ue of present wor	th			-\$760,000		
Annual O&M Costs						\$6,800,000		
Total Annualized Cost						\$12,000,000		
Deliveries of Recycled Water	4,03	30 AFY			_			
Estimated Unit Cost (\$/AF)						\$3,000		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$1,900		Distribution	\$1,100		
Annualized Costs (\$ / Year)	Dry Veer Adin	tment (Sunnly	ad 3/10 years					
Annualized Costs (\$ / Tear)	Two novments	ner vegr spread	over Project Life			\$5 600 000	Same as constant	t lise
Annualized Salvage Value	Annualized val	up of present wor	th			-\$760 000	Same as constant	use.
Annual O&M Costs		ac of present wor				\$5 661 600	22.ne as constant	
Total Annualized Cost						\$11.000 000		
Annual Average Deliveries of Recycled Water	1.20	09 AFY			-	+_1,000,000		
Estimated Unit Cost (\$/AF)						\$9,100		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$5.600		Distribution	\$3.500		

Alternative Rich-ResSP-4							EBMUD Recycled	Water Master Plan Lindate
Last Lindated:	30-May-18				Di	iscount Pate	EDIVIOD Recycled	act Life
Last opuated.	J laffe				30		20 V	ect Life
CCI (20 City, Dec 2017): 10870.06	i. Jane				57	⁷⁰	501	cars
Item	Size	Qtv	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value
Capital Costs								
Treatment	Cost Estimates	for Treatment are	e based on Unit Co	sts for	Each Treatment P	Process		
MF/UF system	4		MGD	\$	1,225,000	\$4,400,000	30 yrs	\$0
RO System	4		MGD	\$	1,475,000	\$5,300,000	30 yrs	\$0
Advanced Oxidation and Disinfection	4		MGD	\$	437,500	\$1,600,000	30 yrs	\$0
Chemicals (Storage and Use)	4			\$	125,000	\$450,000	30 yrs	\$0
Sitework/Piping/Structures	4		MGD	\$	3,187,500	\$11,000,000	30 yrs	\$0
			Raw Const	ruction	Cost Subtotal	\$23,000,000		\$0
Contractor Overhead & Profit					15%	\$3,500,000		\$0
Sales Tax	50%	% of Subtotal Cost	t Applicable		9%	\$1,000,000		\$0
			Estimated Subto	tal Con	struction Cost	\$28,000,000		\$0
Construction Cost Including Contingency								
			Estimated Subto	tal Con	struction Cost	\$28,000,000		\$0
Estimating Contingency					25%	\$7,000,000		\$0
		Estimated S	Subtotal Cost (1) ir	ncludin	g Contingency	\$35,000,000		\$0
Capital Costs (from Bids)								
Conveyance	Cost Estimates	for Pipelines are	based on EBMUD o	constru	ction bid cost + es	stimating contingency	75	<u> </u>
Non-urban Pipeline	16-in	7,700 LF	IN-LF	Ş	30	\$3,696,000	75 yrs	\$2,200,000
Low-Density Urban Pipeline	16-in	19,200 LF	in-LF	Ş	40	\$12,288,000	75 yrs	\$7,400,000
High-Density Urban Pipeline	16-IN	9,100 LF	in-LF	Ş	50	\$7,280,000	75 yrs	\$4,400,000
Irenchless Crossings		2.050	15	~	2 200	Ć4 500 000	75	ća 700.000
HDD (All Inclusive)		2,050	LF	Ş	2,200	\$4,500,000	75 yrs	\$2,700,000
Pump Stations	Cost Estimates	for Pump Station	s are based on EBN	NUD Са	onstruction Bid Co	osts		
Pump Station (Total installed HP, including standby)		EA (Cost Cur	ve based on total					
rump station (rotal installed fir, including statuby)	450 HP	insta	alled HP)	\$	2,100,000	\$2,100,000	50 yrs	\$840,000
		Estimated S	Subtotal Cost (2) ir	ncludin	g Contingency	\$30.000.000		\$18.000.000
Estimated Subtotal Construction Cost including Cont	ingency				· · · · · · · · · · · · · · · · · · ·	\$65,000,000		\$18,000,000
Mobilization	0,				5%	\$3,300,000		\$900,000
Estimated Subtotal Construction Cost including Cont	ingency and Mo	bilization				\$68,000,000		\$19,000,000
Implementation								
Environmental Documentation & Permits					5%	\$3,400,000		\$950,000
Design Cost					15%	\$10,000,000		\$2,900,000
Project Administration and Construction Managemen			10%	\$6,800,000		\$1,900,000		
Estimated Total Capital Cost including Implementation	on and Continge	ency				\$88,000,000		\$25,000,000
						Present Worth of S	alvage Value	\$10,000,000
O&M Costs (Annual)								
Advanced Water Treatment					225 222	ć4 200 000		
NF/UF system	4		MGD	Ş	325,000	\$1,200,000		
Advanced Oxidation and Disinfection	4		MGD	ç ç	480,000	\$1,700,000		
Free Chlorine	4		MGD	ŝ	32 000	\$120,000		
Chemicals	4		MGD	Ś	120.000	\$430.000		
Labor	4	2.080	hrs/MGD	Ś	135	\$1.000.000		
Electricity		3,258,000	kWh/yr	\$	0.15	\$490,000		
Surface Water Treatment			.,					
SWT O&M (Sobrante WTP)	1,314		MG	\$	254	\$330,000		
Conveyance								
Annual O&M			2% of construction	on cost		\$560.000		
Dump Stations						\$300,000		
Pump Stations		500	h ar	÷	125	¢60.000		
Labor Costs		500	nrs	Ş	135	\$68,000		
Electricity		1,703,000	KVVN/yr	\$	0.15	\$260,000		
Pump Station Consumables		5% of pump sta	Total OSM Cost	COST		\$110,000		
Annualized Costs (\$ / Year)			Total Oxfor Costs	s (ə/ yı)		\$0,400,000		
Annualized Costs (\$ / Tear)	Two navments	ner vear spread	over Project Life			\$4 500 000		
Annualized Capital Costs (5/ Tear)	Annualized valu	ue of present wor	th			-\$510,000		
Annual O&M Costs		ac of present wor				\$6.400 000		
Total Annualized Cost						\$10.000.000		
Deliveries of Recycled Water	4.03	30 AFY			_	,,	-	
Estimated Unit Cost (\$/AF)						\$2,500		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$2,000		Distribution	\$500		
Annualized Costs (\$ / Yoar)	Dry Yoar Adim	tment / Cur - lu	ad 3/10 years)					
Annualized Costs (\$ / Teal)	Two normants	ner vegr. coroca	over Project Life			\$1 500 000	Same as constant	t lise
Annualized Capital Costs (2/ Teal)	Annualized val	per year, spread i	th			¢510 ۵۵۰-	Same as constant	use.
		ae oj present wor				¢ς 302 υυ	Same as constant	
Total Annualized Cost						\$3,307,000 \$9,207,000		
Annual Average Deliveries of Recycled Water	1 20	09 AFY			_	ç9,300,000	-	
Estimated Unit Cost (\$/AF)	2,2					\$7,700		
Breakdown of Estimated Unit Cost (S/AF)		Treatment	\$5.700		Distribution	\$2.000		

Alternative Rich-Treat-4							FBMUD Recvo	led Water Mast	er Plan Undate
Last Lindated:	30-May-18					Discount Rate	P	roject Life	er i lan opuate
Updated by:	L Jaffe					3%		0 Years	
CCI (20 City, Dec 2017): 10870.06							-		
Item	Size	Qty	Unit		Unit Cost	Total Cost	Useful Life	Salvage	Value
Capital Costs									
<u>Treatment</u>	Cost Estimates	for Treatment are	based on Unit Cos	ts for E	ach Treatmen	t Process			
Ozone	4		MGD	\$	335,300	\$1,200,000	30 yrs		\$0
BAC	4		MGD	Ş	300,900	\$1,100,000	30 yrs		\$0
MF/UF system	4		MGD	Ş	1,225,000	\$4,400,000	30 yrs		\$0 \$0
RU System	4		MGD	Ş	1,475,000	\$5,300,000	30 yrs		\$0 ¢0
Chemicals (Storage and Lise)	4		WGD	ې د	437,500	\$1,000,000	30 yrs		\$0 \$0
Sitework/Pining/Structures	4		MGD	ڊ خ	3 187 500	\$430,000	30 yrs		30 \$0
Sitework, riping, Structures	4		Raw Constru	uction	Cost Subtotal	\$25,000,000	30 913		\$0 \$0
Contractor Overhead & Profit			nuw constru		15%	\$3.800.000			\$0 \$0
Sales Tax	50%	% of Subtotal Cost	Applicable		9%	\$1,100,000			\$0
			Estimated Subtota	al Cons	struction Cost	\$30,000,000	1		\$0
Construction Cost Including Contingency									
			Estimated Subtota	al Cons	struction Cost	\$30,000,000			\$0
Estimating Contingency					25%	\$7,500,000			\$0
		Estimated S	ubtotal Cost (1) ind	cluding	contingency	\$38,000,000			\$0
Capital Costs (from Bids)		<i>.</i>							
Conveyance	Cost Estimates	for Pipelines are b	ased on EBMUD co	onstruc	tion bid cost +	estimating contingency	75		¢460.000
Non-urban Pipeline	16-in 16 in	1,600 LF	in-LF	Ş	30	\$768,000	75 yrs		\$460,000 ¢0
Low-Density Urban Pipeline	16-IN 16 in		IN-LF	Ş ¢	40	ŞU	75 yrs		\$U \$0
ingi-bensity orban ripenne	10-111	θEi	111-61	Ļ	50	ŲÇ	75 y13		ŲÇ
Pump Stations	Cost Estimates	for Pump Stations	are based on EBM	UD Co	nstruction Bid	Costs			
Pump Station (Total installed HP, including standby)	200.115	EA (Cost Curv	e based on total	<u>,</u>	4 000 000	<u> </u>	50		4720.000
Channes Taulus	300 HP	insta	lied HP)	Ş	1,800,000	\$1,800,000	50 yrs		\$720,000
Storage Tanks	Cost Estimates	for Storage Tanks	are based on EBINI	UD COI	nstruction Bia	Costs			
Storage Lanks	0.2 MG	2	Volume)	/ ¢	2 600 000	¢7 000 000	75 vrc		\$4 700 000
	0.3 Mid	3	volume)	Ş	2,000,000	\$7,800,000	73 yrs		\$4,700,000
		Estimated S	ubtotal Cost (2) inc	luding	g Contingency	\$10,000,000			\$5,900,000
Estimated Subtotal Construction Cost including Cont	ingency					\$48,000,000			\$5,900,000
Mobilization	ingona cand Ma	hilization			5%	\$2,400,000			\$300,000
Estimated Subtotal Construction Cost including Cont	ingency and ivid	obilization				\$50,000,000			\$6,200,000
Environmental Documentation & Permits					5%	\$2,500,000			\$310.000
Design Cost					15%	\$7,500,000			\$930.000
Project Administration and Construction Managemen	t Cost				10%	\$5,000,000			\$620,000
Estimated Total Capital Cost including Implementation	on and Continge	ency				\$65,000,000			\$8,100,000
		-				Present Worth of S	alvage Value		\$3,300,000
O&M Costs (Annual)									
Advanced Water Treatment									
Ozone	4		MGD	Ş	57,000	\$210,000			
BAC	4		MGD	ې د	225,000	\$420,000 \$1,200,000			
RO System	4		MGD	ŝ	480.000	\$1,200,000			
Advanced Oxidation and Disinfection	4		MGD	\$	49,000	\$180,000			
Free Chlorine	4		MGD	\$	32,000	\$120,000			
Chemicals	4		MGD	\$	120,000	\$430,000			
Labor	4	2,080	hrs/MGD	\$	135	\$1,000,000			
Electricity		4,474,800	kWh/yr	\$	0.15	\$670,000			
Monitoring			\$/year	\$	1,000,000	\$1,000,000			
Conveyance									
Annual O&M			2% of construction	n cost		\$15,400			
Pump Stations									
Labor Costs		500	hrs	\$	135	\$68,000			
Electricity		980,000	kWh/yr	\$	0.15	\$150,000			
Pump Station Consumables		5% of pump sta	tion construction c	ost		\$90,000			
Storage Tanks									
Annual Q&M		1% of co	nstruction cost			\$78.000			
, and oan		1/0 01 00	Total O&M Costs	(\$/vr)		\$7.300.000			
Annualized Costs (\$ / Year)						, ,,			
Annualized Capital Costs (\$/Year)	Two payments	per year, spread o	over Project Life			\$3,300,000			
Annualized Salvage Value	Annualized val	ue of present wort	:h			-\$170,000			
Annual O&M Costs						\$7,300,000			
Total Annualized Cost						\$10,000,000	_		
Deliveries of Recycled Water	4,0	30 AFY							
Estimated Unit Cost (\$/AF)						\$2,500			
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$2,400		Distribution	\$100			
Annualized Costs (\$ / Year)	Dry Year Adjus	stment (Supply us	ed 3/10 years)						
Annualized Capital Costs (\$/Year)	Two payments	per year, spread o	over Project Life			\$3,300,000	Same as const	ant use.	
Annualized Salvage Value	Annualized val	ue of present wort	h -			-\$170,000	Same as const	ant use.	
Annual O&M Costs						\$5,672,400			
Total Annualized Cost						\$8,800,000			
Annual Average Deliveries of Recycled Water	1,2	09 AFY							
Estimated Unit Cost (\$/AF)						\$7,300			
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$6,600		Distribution	\$700			

Alternative WC-Raw-5						El	3MUD Recycled Water	Master Plan Update
Last Updated:	30-May-18				Dis	count Rate	Project Life	
Updated by:	I. Jaffe				3%		30 Years	
CCI (20 City, Dec 2017): 10870.06	C :	01	11-14			T-1-1 C1	1	
Capital Costs	Size	Qty	Unit		Unit Cost	Total Cost	Jsetul Lite Sa	livage value
Treatment	Cost Estimates	for Treatment are	e based on Unit Cost	s for E	ach Treatment Pr	ocess		
MF/UF system	5		MGD	\$	1,225,000	\$5,800,000	30 yrs	\$0
RO System	5		MGD	\$	1,475,000	\$6,900,000	30 yrs	\$0
Advanced Oxidation and Disinfection	5		MGD	\$	437,500	\$2,100,000	30 yrs	\$0
Chemicals (Storage and Use)	5		MCD	Ş	125,000	\$590,000	30 yrs	\$0 \$0
Sitework/Fiping/Structures	J		Raw Constru	, ction (Cost Subtotal	\$13,000,000	30 yis	30 \$0
Contractor Overhead & Profit					15%	\$4,500,000		\$0
Sales Tax	50%	% of Subtotal Cost	Applicable		9%	\$1,400,000		\$0
			Estimated Subtota	l Cons	truction Cost	\$36,000,000		\$0
Construction Cost Including Contingency								
construction cost merulang contingency			Estimated Subtota	l Cons	truction Cost	\$36.000.000		\$0
Estimating Contingency					25%	\$9,000,000		\$0
		Estimated S	ubtotal Cost (1) inc	luding	Contingency	\$45,000,000		\$0
Capital Costs (from Bids)	0 1 E II 1	(D ; /; /	1 501 (110					
Conveyance Non-urban Pipeline	Cost Estimates	for Pipelines are t	in-LE	nstruc ¢	tion bid cost + est	imating contingency	75 yrs	\$1 800 000
Low-Density Urban Pipeline	20-in	44.100 LF	in-LF	Ś	40	\$35,280,000	75 yrs	\$21,000,000
High-Density Urban Pipeline	20-in	7,400 LF	in-LF	\$	50	\$7,400,000	75 yrs	\$4,400,000
Trenchless Crossings								
HDD (All Inclusive)		3,400	LF	\$	2,200	\$7,500,000	75 yrs	\$4,500,000
Pump Stations	Cost Estimates	for Pump Stations	s are based on EBML	JD Cor	nstruction Bid Cos	ts		
Pump Station (Total installed HP including standby)		EA (Cost Curv	e based on total					
	600 HP	insta	illed HP)	\$	2,400,000	\$2,400,000	50 yrs	\$960,000
Storage Tanks	Cost Estimates	for Storage Tanks	are based on EBML	ID Cor	struction Bid Cos	ts		
Storage Tanks	0.4 MG	2	EA (Cost Curve by Volume)	ć	2 700 000	\$8 100 000	75 vrs	\$4 900 000
	0.4 Mid	5		, <u>,</u>	2,700,000	\$3,100,000	75 ¥13	\$4,500,000
Estimated Subtatal Construction Cast including Cont	ingongy	Estimated S	ubtotal Cost (2) inc	luding	Contingency	\$64,000,000		\$38,000,000
Mobilization	ingency				5%	\$110,000,000		\$1,900,000
Estimated Subtotal Construction Cost including Cont	ingency and Mo	bilization				\$120,000,000		\$40,000,000
Implementation								
Environmental Documentation & Permits					5%	\$6,000,000		\$2,000,000
Design Cost	+ C+				15%	\$18,000,000		\$6,000,000
Project Administration and Construction Managemen	n and Conting	ancy			10%	\$12,000,000		\$4,000,000
	en una continge					Present Worth of Salv	/age Value	\$21,000,000
O&M Costs (Annual)								
Advanced Water Treatment	-		MCD	ć	225 000	¢1 500 000		
RO System	5		MGD	Ş Ş	480.000	\$1,500,000		
Advanced Oxidation and Disinfection	5		MGD	\$	49,000	\$230,000		
Free Chlorine	5		MGD	\$	32,000	\$150,000		
Chemicals	5		MGD	\$	120,000	\$560,000		
Labor	5	2,080	hrs/MGD	Ş	135	\$1,300,000		
Monitoring		4,255,500	Ś/year	ې د	500.000	\$540,000		
Companyo			<i>oy</i> year	Ŷ	500,000	\$500,000		
Annual O&M			2% of construction	cost		\$1,060,000		
Pump Stations			270 01 001511401011	cost		\$1,000,000		
Labor Costs		500	hrs	Ś	135	\$68,000		
Electricity		2,021,000	kWh/yr	Ş	0.15	\$300,000		
Pump Station Consumables		5% of pump sta	tion construction co	ost		\$120,000		
Storage Tanks								
Annual O&M		1% of co	instruction cost			\$81,000		
			Total O&M Costs (\$/yr)		\$8,800,000		
Annualized Costs (\$ / Year)								
Annualized Capital Costs (\$/Year)	I WO payments	per year, spread o	over Project Life			\$8,100,000 _\$1,100,000		
Annual Q&M Costs	Annadiized vai	ie oj present word				\$8.800.000		
Total Annualized Cost						\$16,000,000		
Deliveries of Recycled Water	5,2	50 AFY						
Estimated Unit Cost (\$/AF)						\$3,000		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$1,900		Distribution	\$1,100		
Annualized Costs (\$ / Year)	Two normants	tment (Supply us	ea 3/10 years)			\$ <u>8</u> 100 000 5	ame as constant use	
Annualized Salvage Value	Annualized val	year year, spread of ue of present work	th			-\$1.100.000 \$	ame as constant use.	
Annual O&M Costs						\$7,304,000		
Total Annualized Cost						\$14,000,000		
Annual Average Deliveries of Recycled Water	1,5	78 AFY						
Estimated Unit Cost (\$/AF)						\$8,900		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$5,700		Distribution	\$3,200		

Alternative W/C-BesB-5								d Water Master Plan Lindate
Last Lindated:	30-May-18					Discount Pate	EDIVIOD Recycles	iect Life
Lindated by:	J laffe					3%	30	Vears
CCI (20 City, Dec 2017): 10870.06	1. June					370	50	Tears
Item	Size	Otv	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value
Canital Costs	0.20							ourrage value
Treatment	Cost Estimates	for Treatment are	based on Unit Co	sts for	Each Treatmer	nt Process		
ME/LIE system	5	jor meannent are	MGD	sts jon i S	1 225 000	\$5,800,000	30 yrs	\$0
BO System	5		MGD	Ś	1 475 000	\$6,900,000	30 yrs	\$0
Advanced Oxidation and Disinfection	5		MGD	ś	437 500	\$2,100,000	30 yrs	\$0 \$0
Chemicals (Storage and Lise)	5		WIGD	ç	125 000	\$2,100,000	30 yrs	0Ç \$0
Sitework/Pining/Structures	5		MGD	ś	3 187 500	\$15,000,000	30 yrs	\$0 \$0
Site worky riping/structures	5		Raw Const	ruction	Cost Subtotal	\$30,000,000	50 413	\$0 \$0
Contractor Overhead & Profit			nuw const	uction	15%	\$4 500,000		\$0 \$0
Sales Tay	50%	% of Subtotal Cost	Applicable		9%	\$4,500,000		0Ç \$0
	50%	76 01 Subtotal Cost	Estimated Subto	tal Con	struction Cost	\$1,400,000		90 \$0
			Estimated Subto	tui con	struction cost	\$30,000,000		ΨŪ
Construction Cost Including Contingency								
construction cost menuing contingency			Estimated Subto	tal Con	struction Cost	\$36,000,000		\$0
Estimating Contingency					25%	\$9.000.000		\$0
Loting contingency		Estimated S	ubtotal Cost (1) ir	ncludin	g Contingency	\$45.000.000		\$0
Capital Costs (from Bids)		Lotinated o			g contingency	<i> </i>		γu
Conveyance	Cost Estimates	for Pinelines are h	ased on FRMUD (onstru	ction hid cost +	estimatina contingency		
Non-urban Pipeline	20-in	60.600 LF	in-LF	Ś	30	\$36.360.000	75 vrs	\$22,000,000
Low-Density Urban Pineline	20-in	33 000 LF	in-LF	ć	40	\$26,400,000	75 yrs	\$16,000,000
High-Density Urban Pineline	20-in	6 300 LE	in-LF	ć	50	\$6 300 000	75 yrs	\$3,800,000
Trenchless Crossings	20-111	0,300 LF	ill=LF	Ş	50	ş0,500,000	75 915	şs,ouu,000
San Pablo Tunnel Rehab		17 600	IF	ć	3 500	¢63 000 000	75 vrs	¢27 000 000
HDD (All Inclusive)		2 200	16	ç	2 200	\$4,800,000	75 yrs	\$2,000,000
(All inclusive)		2,200	u	Ŷ	2,200	\$4,000,000	75 913	\$2,500,000
Pump Stations	Cost Estimates	for Pump Stations	are based on EBN	NUD Ca	onstruction Bid	Costs		
Pump Station (Total installed HP including standby)		EA (Cost Curv	e based on total					
	1,200 HP	insta	lled HP)	\$	4,200,000	\$4,200,000	50 yrs	\$1,700,000
		Estimated S	ubtotal Cost (2) ir	ncludin	g Contingency	\$140,000,000		\$83,000,000
Estimated Subtotal Construction Cost including Cont	ingency					\$190,000,000		\$83,000,000
Mobilization					5%	\$9,500,000		\$4,200,000
Estimated Subtotal Construction Cost including Cont	ingency and Mo	bilization				\$200.000.000		\$87.000.000
Implementation	0 7							
Environmental Documentation & Permits					5%	\$10,000,000		\$4,400,000
Design Cost					15%	\$30.000.000		\$13.000.000
Project Administration and Construction Managemer	nt Cost				10%	\$20.000.000		\$8,700,000
Estimated Total Capital Cost including Implementati	on and Conting	encv				\$260.000.000		\$110.000.000
		,				Present Worth of Sa	alvage Value	\$45,000,000
O&M Costs (Annual)							-	
Advanced Water Treatment								
MF/UF system	5		MGD	\$	325,000	\$1,500,000		
RO System	5		MGD	\$	480,000	\$2,300,000		
Advanced Oxidation and Disinfection	5		MGD	\$	49,000	\$230,000		
Free Chlorine	5		MGD	\$	32,000	\$150,000		
Chemicals	5		MGD	\$	120,000	\$560,000		
Labor	5	2,080	hrs/MGD	\$	135	\$1,300,000		
Electricity		4,253,500	kWh/yr	\$	0.15	\$640,000		
Surface Water Treatment								
SWT O&M (Orinda)	1,716		MG	\$	70	\$120,000		
Conveyance								
Annual O&M			2% of construction	on cost		\$2,700.000		
Burner Chatland								
Pump Stations		500			105	<u> </u>		
Labor Costs		500	nrs	Ş	135	\$68,000		
Electricity		4,514,000	kWh/yr	Ş	0.15	\$680,000		
Pump Station Consumables		5% of pump sta	tion construction	cost		\$210,000		
			Total O&M Costs	5 (Ş/yr)		\$10,000,000		
Annualized Costs (\$ / Year)								
Annualized Capital Costs (\$/Year)	Two payments	per year, spread o	over Project Life			\$13,000,000		
Annualized Salvage Value	Annualized val	ue of present wort	h			-\$2,300,000		
Annual O&M Costs						\$10,000,000		
Total Annualized Cost						\$21,000,000	-	
Deliveries of Recycled Water	5,2	60 AFY						
Estimated Unit Cost (\$/AF)						\$4,000		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$1,900		Distribution	\$2,100		
Annualized Costs (\$ / Year)	Drv Year Adius	stment (Supply us	ed 3/10 vears)					
Annualized Capital Costs (\$/Year)	Two navments	per year. spread o	ver Proiect Life			\$13.000.000	Same as constan	nt use.
Annualized Salvage Value	Annualized val	ue of present wort	h			-\$2.300 000	Same as constar	nt use.
Annual O&M Costs						\$8,953,000	us constar	
Total Annualized Cost						\$20,353,000 \$20,000,000		
Annual Average Deliveries of Recycled Water	15	78 AFY				\$20,000,000	-	
Estimated Unit Cost (\$/AE)	1,5					\$13,000		
Proakdown of Estimated Unit Cast (\$ /AE)		Trootment	\$E 600	_	Dictribution	\$13,000		
Dieakuowii oi Estilliateu onit Cost (S/AF)		rieaument	J J,000		DISTRIBUTION	\$7,400		

Alternative WC-BesSB-5								Water Master Plan Lindate
Arternative wc-kessr-s	20 May 18					Discount Pate	EBINIOD Recycles	iost Life
Last Opualed.	Julaffe					2%	30.1	Ject Life Vears
CCI (20 City, Dec 2017): 10870.06	1. Jane					570	50	rears
Item	Size	Qty	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value
Capital Costs								
Treatment	Cost Estimates	for Treatment are	e based on Unit Co	sts for	Each Treatmen	t Process		
MF/UF system	5		MGD	\$	1,225,000	\$5,800,000	30 yrs	\$0
RO System	5		MGD	\$	1,475,000	\$6,900,000	30 yrs	\$0
Advanced Oxidation and Disinfection	5		MGD	\$	437,500	\$2,100,000	30 yrs	\$0
Chemicals (Storage and Use)	5			\$	125,000	\$590,000	30 yrs	\$0
Sitework/Piping/Structures	5		MGD	\$	3,187,500	\$15,000,000	30 yrs	\$0
			Raw Const	ruction	Cost Subtotal	\$30,000,000		\$0
Contractor Overhead & Profit					15%	\$4,500,000		\$0
Sales Tax	50%	% of Subtotal Cost	Applicable		9%	\$1,400,000		\$0
			Estimated Subto	tal Con	struction Cost	\$36,000,000		\$0
Construction Cost Including Contingency								
			Estimated Subto	tal Con	struction Cost	\$36,000,000		\$0
Estimating Contingency					25%	\$9,000,000		\$0 \$0
		Estimated S	ubtotal Cost (1) ir	ncludin	g Contingency	\$45,000,000		\$0
Capital Costs (from Bids)	6 1 F 11 1	(a: /: /	1 501 (110					
Conveyance	Cost Estimates	Jor Pipelines are L	in LE	constru	ction bia cost +	estimating contingency	75	¢7,700,000
Non-urban Pipeline	20-III 20 in	21,400 LF	III-LF	ې د	30	\$12,640,000	75 yrs	\$7,700,000
Low-Density Urban Pipeline	20-III 20 in	22,500 LF	III-LF	ې د	40	\$18,000,000 \$6,600,000	75 yrs	\$11,000,000
Tranchloss Crossings	20-111	0,000 LF	III-LF	Ş	50	\$0,000,000	75 yrs	\$4,000,000
San Dablo Tunnel Rebab		17 600	15	ć	3 500	\$62,000,000	75 vrs	\$37,000,000
HDD (All Inclusive)		2 000	LF	ś	2 200	\$4,400,000	75 yrs	\$2,600,000
		2,000		Ý	2,200	ç+,+00,000	75 yrs	\$2,000,000
Pump Stations	Cost Estimates	for Pump Stations	s are based on EBN	ИUD Са	onstruction Bid	Costs		
Pump Station (Total installed HP, including standby)		EA (Cost Curv	e based on total					
· ····································	600 HP	insta	illed HP)	\$	2,400,000	\$2,400,000	50 yrs	\$960,000
		Estimated S	ubtotal Cost (2) ir	ncludin	g Contingency	\$110,000,000		\$63,000,000
Estimated Subtotal Construction Cost including Cont	ingency					\$160,000,000		\$63,000,000
Mobilization					5%	\$8,000,000		\$3,200,000
Estimated Subtotal Construction Cost including Cont	ingency and Mo	bilization				\$170,000,000		\$66,000,000
Implementation								
Environmental Documentation & Permits					5%	\$8,500,000		\$3,300,000
Design Cost					15%	\$26,000,000		\$9,900,000
Project Administration and Construction Managemer	nt Cost				10%	\$17,000,000		\$6,600,000
Estimated Total Capital Cost including Implementati	on and Continge	ency				\$220,000,000		\$86,000,000
						Present Worth of S	alvage Value	\$35,000,000
O&M Costs (Annual)								
ME/LE system	5		MGD	ć	325.000	\$1 500 000		
RO System	5		MGD	¢ ¢	480.000	\$1,500,000		
Advanced Oxidation and Disinfection	5		MGD	Ś	49,000	\$230.000		
Free Chlorine	5		MGD	\$	32,000	\$150,000		
Chemicals	5		MGD	\$	120,000	\$560,000		
Labor	5	2,080	hrs/MGD	\$	135	\$1,300,000		
Electricity		4,253,500	kWh/yr	\$	0.15	\$640,000		
Surface Water Treatment								
SWT O&M (Sobrante WTP)	1,716		MG	\$	254	\$440,000		
Conveyance								
Annual O&M			2% of construction	on cost		\$2,100,000		
Ruma Stations						.,,,		
Labor Costs		500	brc	ć	125	\$68 000		
Electricity		2 358 000	kWb/yr	ڊ خ	0.15	\$08,000		
Pump Station Consumables		2,538,000	tion construction	çost	0.15	\$330,000		
rump station consumables		570 01 pump 3ta	Total O&M Costs	(\$/wr)		\$120,000		
Annualized Costs (\$ / Year)			Total Odivi Costs	, (<i>၃/ ۹</i> - /		\$5,800,000		
Annualized Costs (\$ / Tear)	Two navments	ner vegr spread a	wer Project Life			\$11,000,000		
Annualized Capital Costs (5) Tear)	Annualized val	ve of present wort	h			-\$1 800 000		
Annual O&M Costs	, and an 200 Parts					\$9.800.000		
Total Annualized Cost						\$19.000.000		
Deliveries of Recycled Water	5.2	50 AFY				+	-	
Estimated Unit Cost (\$/AF)	-,-					\$3.600		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$1.900		Distribution	\$1,700		
			+-,			+-/		
Annualized Costs (\$ / Year)	Dry Year Adjus	tment (Supply us	ea 3/10 years)			¢44 000 000	Camo '	t.uco
Annualized Capital Costs (\$/Year)	i wo payments	per year, spread o	over Project Life			\$11,000,000	Same as constar	it use.
Annualized Salvage Value	Annualized val	ue of present wort	:n			-\$1,800,000	same as constar	it use.
						\$8,20U,UUU		
Annual Average Deliveries of Recycled Water	1 5	78 AFV				\$17,000,000	-	
Estimated Unit Cost (\$/AE)	1,5					\$11.000		
Breakdown of Estimated Unit Cest (\$/AF)		Treatment	\$5 700		Distribution	\$11,000		
Diculture of Lotinated Onit Cost (5/AF)		neatment	÷3,700		Distribution	JJ,300		

Alternative WC-Treat-5							EBMUD Recyc	led Water Master Plan Update
Last Updated:	30-May-18				[Discount Rate		Project Life
Updated by:	I. Jaffe				3	3%		30 Years
CCI (20 City, Dec 2017): 10870.06	Cino	011	11-14		Unit Cost	Total Cast	Lineful Life	Columno Volum
Capital Costs	5120	Qty	Onit		Unit Cost	Total Cost	Userur Lite	Salvage value
Treatment	Cost Estimates	for Treatment ar	e based on Unit Co	osts for	Each Treatmen	t Process		
Ozone	5		MGD	\$	335,300	\$1,600,000	30 yrs	\$0
BAC	5		MGD	\$	300,900	\$1,400,000	30 yrs	\$0
RO System	5		MGD	s ¢	1,225,000	\$5,800,000	30 yrs 30 yrs	\$U \$0
Advanced Oxidation and Disinfection	5		MGD	Ş	437,500	\$2,100,000	30 yrs	\$0 \$0
Chemicals (Storage and Use)	5			\$	125,000	\$590,000	30 yrs	\$0
Sitework/Piping/Structures	5		MGD	\$	3,187,500	\$15,000,000	30 yrs	\$0
Contractor Overboad & Brofit			Raw Constr	ruction	Lost Subtotal	\$33,000,000		\$0
Sales Tax	50%	% of Subtotal Cost	Applicable		9%	\$1,500,000		\$0
			Estimated Subtot	tal Cons	truction Cost	\$40,000,000		\$0
Construction Cost Including Contingency			Estimated Subtot	tal Cons	truction Cost	\$40,000,000		\$0
Estimating Contingency					25%	\$10,000,000		\$0
		Estimated S	ubtotal Cost (1) in	cluding	Contingency	\$50,000,000		\$0
Capital Costs (from Bids)	Cost Estimates	for Pinelines are	hased on ERMUD	constru	ction hid cost +	estimating contingency		
Non-urban Pipeline	20-in	1,100 LF	in-LF	\$	30	\$660,000	75 yrs	\$400,000
Low-Density Urban Pipeline	20-in	5,400 LF	in-LF	\$	40	\$4,320,000	75 yrs	\$2,600,000
High-Density Urban Pipeline	20-in	4,800 LF	in-LF	\$	50	\$4,800,000	75 yrs	\$2,900,000
Trenchless Crossings		4 200				40.000 000		Å4 COD 000
HDD (All Inclusive)		1,200	LF	Ş	2,200	\$2,600,000	75 yrs	\$1,600,000
Pump Stations	Cost Estimates	for Pump Station	s are based on EB	MUD Co	onstruction Bid	Costs		
Pump Station (Total installed HP, including standby)	375 HP	EA (COST CUR	lled HP)	Ś	1 900 000	\$1 900 000	50 vrs	\$760.000
Storage Tanks	Cost Estimates	for Storage Tank	s are based on EBI	MUD Ca	onstruction Bid	Costs	50 (15	<i>\$1</i> 00,000
Storage Tanks			EA (Cost Curve b	ру				
(material determined during pre-design)	0.4 MG	3	Volume)	\$	2,700,000	\$8,100,000	75 yrs	\$4,900,000
		Estimated S	ubtotal Cost (2) in	ncluding	Contingency	\$22,000,000		\$13,000,000
Estimated Subtotal Construction Cost including Con	tingency				50/	\$72,000,000		\$13,000,000
Finishing Construction Cost including Con	tingency and M	obilization			5%	\$3,600,000		\$650,000
Implementation	ungener una m	o pinization				<i>\$10,000,000</i>		¢1,000,000
Environmental Documentation & Permits					5%	\$3,800,000		\$700,000
Design Cost					15%	\$11,000,000		\$2,100,000
Project Administration and Construction Managemen	ion and Conting	ency			10%	\$7,600,000		\$1,400,000
						Present Worth of Sa	alvage Value	\$7,400,000
O&M Costs (Annual)								
Advanced Water Treatment	E		MGD	ć	E7 000	\$270.000		
BAC	5		MGD	ş Ş	116.000	\$550.000		
MF/UF system	5		MGD	\$	325,000	\$1,500,000		
RO System	5		MGD	\$	480,000	\$2,300,000		
Advanced Oxidation and Disinfection	5		MGD	ş	49,000 32,000	\$230,000		
Chemicals	5		MGD	Ş	120,000	\$560,000		
Labor	5	2,080	hrs/MGD	\$	135	\$1,300,000		
Electricity		5,842,100	kWh/yr	\$	0.15	\$880,000		
Monitoring			\$/year	Ş	1,000,000	\$1,000,000		
Conveyance			2% of construction	on cost		\$2E0.000		
Aminual Oxivi			2/6 OF CONStruction	JII COSL		\$250,000		
Labor Costs		500	hrs	Ś	135	\$68.000		
Electricity		1,415,000	kWh/yr	\$	0.15	\$210,000		
Pump Station Consumables		5% of pump sta	tion construction	cost		\$95,000		
Storage Tanks								
Annual O&M		1% of co	nstruction cost			\$81,000		
· · · · · · · · · · · · · · · · · · ·			Total O&M Costs	s (\$/yr)		\$9,400,000		
Annualized Costs (\$ / Year)	Two navments	ner vear spread	over Project Life			\$5,000,000		
Annualized Salvage Value	Annualized val	ue of present wor	th			-\$380,000		
Annual O&M Costs						\$9,400,000		
Total Annualized Cost					-	\$14,000,000		
Deliveries of Recycled Water	5,2	50 AFY				ća 700		
Estimated Unit Cost (\$/AF) Breakdown of Estimated Unit Cost (\$/AE)		Treatment	\$2 300		Distribution	\$2,700		
	Dry Years Ad	tmont /Current	ad 2/10		Listingution	φ τ ου		
Annualized Costs (\$ / Year) Annualized Capital Costs (\$/Year)	Two payments	per year snread	over Project Life			\$5.000.000	Same as con	stant use.
Annualized Salvage Value	Annualized val	ue of present wor	th			-\$380,000	Same as cons	stant use.
Annual O&M Costs						\$7,484,000		
Total Annualized Cost					-	\$12,000,000		
Annual Average Deliveries of Recycled Water	1,5	O AFY				\$7 600		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$6,300		Distribution	\$1,300		

Alternative Oro-Raw-8						E	BMUD Recycled Water	Master Plan Update
Last Updated:	30-May-18					Discount Rate	Project Life	
Updated by:	I. Jaffe					3%	30 Years	
CCI (20 City, Dec 2017): 10870.06								
Item	Size	Qty	Unit		Unit Cost	Total Cost	Useful Life Sa	lvage Value
Capital Costs								
Treatment	Cost Estimates	for Treatment are	based on Unit Cost	s for E	ach Treatmen	t Process	20	60
MF/UF system	8		MGD	Ş	1,225,000	\$9,800,000	30 yrs	\$0
RU System	8		MGD	Ş	1,475,000	\$12,000,000	30 yrs	\$0 ¢0
Chemicals (Storage and Use)	8		MGD	Ş	437,500	\$3,500,000	30 yrs	\$0 ¢0
Citework / Diping / Structures	8		MCD	ې د	2 1 2 7 5 00	\$1,000,000	30 yrs	\$0 \$0
Sitework/riping/Situationes	0		Raw Constru	ction	Cost Subtotal	\$20,000,000	50 y13	50 \$0
Contractor Overhead & Profit			nuw constru	cuon	15%	\$7,800,000		\$0
Sales Tax	50%	% of Subtotal Cost	Applicable		9%	\$2,300,000		\$0
			Estimated Subtota	l Cons	truction Cost	\$62,000,000		\$0
Construction Cost Including Contingency								
			Estimated Subtota	l Cons	truction Cost	\$62,000,000		\$0
Estimating Contingency					25%	\$16,000,000		\$0
		Estimated S	ubtotal Cost (1) inc	luding	Contingency	\$78,000,000		\$0
Capital Costs (from Bids)								
Conveyance	Cost Estimates	for Pipelines are b	ased on EBMUD co	nstruc	tion bid cost +	estimating contingency		40.000.000
Non-urban Pipeline	24-in	7,900 LF	in-LF	Ş	30	\$5,688,000	75 yrs	\$3,400,000
Low-Density Urban Pipeline	24-in	29,000 LF	in-LF	Ş	40	\$27,840,000	75 yrs	\$17,000,000
Tranchless Crossings	24-IN	3,500 LF	IN-LF	Ş	50	\$4,200,000	75 yrs	\$2,500,000
HDD (All loclusive)		3 750	IE	ć	2 200	\$8 300 000	75 vrs	\$5,000,000
HDD (All Inclusive)		3,730	LF	Ş	2,200	\$8,500,000	75 yis	\$3,000,000
Pump Stations	Cost Estimates	for Pump Stations	are based on EBMU	JD Col	nstruction Bid	Costs		
Pump Station (Total installed HP, including standby)		EA (Cost Curv	e based on total					
	1,200 HP	insta	lled HP)	Ş	4,200,000	\$4,200,000	50 yrs	\$1,700,000
Storage Tanks	Cost Estimates	for Storage Tanks	are based on EBML	JD Cor	struction Bid	Costs		
Storage Tanks	0.7.146	2	EA (Cost Curve by	~	2 000 000	ćo 000 000	75	ĆF 400 000
(material determined during pre-design)	0.7 MG	3	volume)	Ş	3,000,000	\$9,000,000	75 yrs	\$5,400,000
		Estimated S	ubtotal Cost (2) inc	luding	Contingency	\$59,000,000		\$35,000,000
Estimated Subtotal Construction Cost including Cont	ingency					\$140,000,000		\$35,000,000
Mobilization					5%	\$7,000,000		\$1,800,000
Estimated Subtotal Construction Cost including Cont	ingency and Mo	bilization				\$150,000,000		\$37,000,000
Implementation					E0/	¢7 500 000		¢1,000,000
Design Cost					15%	\$7,300,000		\$1,900,000
Project Administration and Construction Managemen	t Cost				10%	\$15.000.000		\$3,700.000
Estimated Total Capital Cost including Implementation	on and Continge	ncv				\$200.000.000		\$48.000.000
······································						Present Worth of Sal	vage Value	\$20,000,000
O&M Costs (Annual)								
Advanced Water Treatment								
MF/UF system	8		MGD	\$	325,000	\$2,600,000		
RO System	8		MGD	Ş	480,000	\$3,800,000		
Advanced Oxidation and Disinfection	8		MGD	Ş	49,000	\$390,000		
Free Chlorine Chamicals	8		MGD	Ş	32,000	\$260,000		
Chemicals	8	2 090	MGD hrs/MCD	Ş	120,000	\$960,000		
Electricity	0	2,080	hwb /w	ې د	0.15	\$2,200,000		
Monitoring		7,240,000	Ś/vear	ç	500.000	\$1,100,000		
-			ə/ year	Ŷ	500,000	\$500,000		
Conveyance			201 5 1 1			6000.000		
Annual O&M			2% of construction	cost		\$920,000		
Pump Stations								
Labor Costs		500	hrs	\$	135	\$68,000		
Electricity		4,816,000	kWh/yr	\$	0.15	\$720,000		
Pump Station Consumables		5% of pump sta	tion construction co	ost		\$210,000		
Storage Tanks								
Annual O&M		1% of co	nstruction cost			\$90,000		
			Total O&M Costs (\$/yr)		\$14,000,000		
Annualized Costs (\$ / Year)								
Annualized Capital Costs (\$/Year)	Two payments	per year, spread o	over Project Life			\$10,000,000		
Annualized Salvage Value	Annualized val	ie of present wort	h			-\$1,000,000		
Annual O&M Costs						\$14,000,000		
Deliveries of Resycled Water	0.00					\$23,000,000		
Deliveries of Recycled Water	8,9	50 AFY				¢2.000		
Estimated Unit Cost (\$/AF)			<u>ć1 000</u>		Distribution	\$2,600		
Breakdown of Estimated Unit Cost (\$/AF)		reatment	\$1,900		Distribution	\$700		
Annualized Costs (\$ / Year)	Dry Year Adjus	tment (Supply use	ed 3/10 years)					
Annualized Capital Costs (\$/Year)	Two payments	per year, spread o	ver Project Life			\$10,000,000 \$	ame as constant use.	
Annualized Salvage Value	Annualized val	ue of present wort	h			-\$1,000,000	Same as constant use.	
Annual O&M Costs						\$11,340,000		
Total Annualized Cost						\$20,000,000		
Annual Average Deliveries of Recycled Water	2,68	SS AFY				*		
Estimated Unit Cost (\$/AF)				_		\$7,400		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$5,600		Distribution	\$1,800		

Alternative Oro-Resil-8							EBMUD Recycled	Water Master Plan Lindate
Last Lindated:	30-May-18					Discount Bate	Proi	ect Life
Updated by:	I. Jaffe					3%	30 Y	'ears
CCI (20 City, Dec 2017): 10870.06								
Item	Size	Qty	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value
Capital Costs								
Treatment	Cost Estimates	for Treatment are	e based on Unit Co	osts for E	ach Treatment	t Process		
MF/UF system	8		MGD	\$	1,225,000	\$9,800,000	30 yrs	ŞU
Advanced Ovidation and Disinfection	0 8		MGD	ç	1,475,000	\$12,000,000	30 yrs	30 \$0
Chemicals (Storage and Use)	8		MGD	ç ç	125 000	\$3,300,000	30 yrs	30 \$0
Sitework/Piping/Structures	8		MGD	Ś	3.187.500	\$26.000.000	30 yrs	\$0 \$0
			Raw Const	ruction	Cost Subtotal	\$52,000,000	1	\$0
Contractor Overhead & Profit					15%	\$7,800,000		\$0
Sales Tax	50%	% of Subtotal Cost	Applicable		9%	\$2,300,000		\$0
			Estimated Subto	tal Cons	truction Cost	\$62,000,000		\$0
Construction Cost Including Contingency			Ectimated Subto	tal Conc	truction Cost	\$62.000.000		¢0
Estimating Contingency			Estimated Subto		25%	\$16,000,000		30 \$0
Estimating contingency		Estimated S	ubtotal Cost (1) i	ncluding	Contingency	\$78,000,000		\$0
Capital Costs (from Bids)								
Conveyance	Cost Estimates	for Pipelines are	based on EBMUD	construc	tion bid cost +	estimating contingency		
Non-urban Pipeline	24-in	25,300 LF	in-LF	\$	30	\$18,216,000	75 yrs	\$11,000,000
Low-Density Urban Pipeline	24-in	29,000 LF	in-LF	\$	40	\$27,840,000	75 yrs	\$17,000,000
High-Density Urban Pipeline	24-in	3,500 LF	in-LF	\$	50	\$4,200,000	75 yrs	\$2,500,000
Trenchless Crossings		0			c20.000	¢5, 600, 000		
Microtunnel Xings (1 jack & 1 receiv. pit)		9	EA	Ş	620,000	\$5,600,000	75	ć12 000 000
HDD (All Inclusive)		7,000		Ş ¢	2,800	\$20,000,000	75 yrs	\$12,000,000
(All inclusive)		3,430		Ş	2,200	\$7,000,000	75 yis	\$4,000,000
Pump Stations	Cost Estimates	for Pump Station	s are based on EBI	MUD Coi	nstruction Bid	Costs		
Pump Station (Total installed HP, including standby)	1 500 UD	EA (Cost Cur	/e based on total	ć	F 200 000	ÉF 200.000	F0 yes	ć2 100 000
	1,500 HP	IIISta		Ş	5,500,000	\$5,500,000	50 yrs	\$2,100,000
		Estimated S	ubtotal Cost (2) in	ncluding	Contingency	\$89,000,000		\$49,000,000
Estimated Subtotal Construction Cost including Cont	ingency				50/	\$170,000,000		\$49,000,000
Retirected Subtotal Construction Cost including Cost	ingongy and Ma	hilization			5%	\$8,500,000		\$2,500,000
Implementation	ingency and wid	omzation				\$180,000,000		\$32,000,000
Environmental Documentation & Permits					5%	\$9.000.000		\$2,600,000
Design Cost					15%	\$27,000,000		\$7,800,000
Project Administration and Construction Managemer	nt Cost				10%	\$18,000,000		\$5,200,000
Estimated Total Capital Cost including Implementati	on and Continge	ency				\$230,000,000		\$68,000,000
						Present Worth of Sa	alvage Value	\$28,000,000
O&M Costs (Annual)								
ME/LIE system	8		MGD	s	325 000	\$2,600,000		
RO System	8		MGD	Ş	480,000	\$3,800,000		
Advanced Oxidation and Disinfection	8		MGD	\$	49,000	\$390,000		
Free Chlorine	8		MGD	\$	32,000	\$260,000		
Chemicals	8		MGD	\$	120,000	\$960,000		
Labor	8	2,080	hrs/MGD	Ş	135	\$2,200,000		
Electricity		7,240,000	kwn/yr	Ş	0.15	\$1,100,000		
Conveyance								
Annual O&M			2% of construction	on cost		\$1,670,000		
Pump Stations								
Labor Costs		500	hrs	\$	135	\$68,000		
Electricity		5,848,000	kWh/yr	\$	0.15	\$880,000		
Pump Station Consumables		5% of pump sta	Total OSM Cost	cost		\$270,000		
Annualized Costs (\$ / Year)			Total Oxivi Costs	s (\$/ ¥I)		\$14,000,000		
Annualized Costs (\$7 red)	Two payments	per year, spread	over Proiect Life			\$12.000.000		
Annualized Salvage Value	Annualized val	ue of present wor	th			-\$1,400,000		
Annual O&M Costs						\$14,000,000		
Total Annualized Cost						\$25,000,000		
Deliveries of Recycled Water	8,9	60 AFY						
Estimated Unit Cost (\$/AF)						\$2,800		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$1,900		Distribution	\$900		
Annualized Costs (\$ / Year)	Dry Year Adjus	tment (Supply us	ed 3/10 years)					
Annualized Capital Costs (\$/Year)	Two payments	per year, spread	over Project Life			\$12,000,000	Same as constant	t use.
Annualized Salvage Value	Annualized val	ue of present wor	th			-\$1,400,000	Same as constan	t use.
Annual U&M Costs						\$11,958,000		
Annual Average Deliveries of Recycled Water	2.6	88 AFY			-	\$25,000,000		
Estimated Unit Cost (\$/AF)	2,0					\$8,600		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$5.600		Distribution	\$3.000		

Alternative Over Chahat 8								
Alternative Oro-Chabot-8							EBIVIOD Recycled	water Master Plan Opdate
Last Updated:	30-May-18				l	Discount Rate	Proje	ect Life
Updated by:	I. Jatte				2	3%	30 Ye	ears
CCI (20 City, Dec 2017): 10870.06								
ltem	Size	Qty	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value
Capital Costs								
Treatment	Cost Estimates	for Treatment are	e based on Unit Co	sts for l	Each Treatment	Process		
MF/UF system	8		MGD	Ş	1,225,000	\$9,800,000	30 yrs	\$0
RO System	8		MGD	\$	1,475,000	\$12,000,000	30 yrs	\$0
Advanced Oxidation and Disinfection	8		MGD	\$	437,500	\$3,500,000	30 yrs	\$0
Chemicals (Storage and Use)	8			\$	125,000	\$1,000,000	30 yrs	\$0
Sitework/Piping/Structures	8		MGD	\$	3,187,500	\$26,000,000	30 yrs	\$0
			Raw Consti	ruction	Cost Subtotal	\$52,000,000		\$0
Contractor Overhead & Profit					15%	\$7,800,000		\$0
Sales Tax	50%	% of Subtotal Cost	Applicable		9%	\$2,300,000		\$0
			Estimated Subto	tal Con	struction Cost	\$62,000,000		\$0
Construction Cost Including Contingency								
			Estimated Subto	tal Con	struction Cost	\$62,000,000		\$0
Estimating Contingency					25%	\$16,000,000		\$0
5 5 7		Estimated S	ubtotal Cost (1) ir	ncludina	z Contingency	\$78.000.000		\$0
Capital Costs (from Bids)					,,	÷••,•••		7-
Conveyance	Cost Estimates	for Pinelines are l	hased on FRMUD o	onstrue	tion hid cost + i	estimatina continaency		
Non-urban Pineline	24-in	7 700 L F	in-IF	\$	30	\$5 544 000	75 vrs	\$3 300 000
Low-Density Urban Pineline	24 in 24-in	14 200 LE	in-LE	ć	40	\$13 632 000	75 yrs	\$8,200,000
High Density Urban Pipeline	24-in 24 in	E 600 I E	in LE	ç	40	\$15,052,000 \$6 730,000	75 yrs	\$3,200,000
Transhlass Crassings	24-111	3,000 LF	III-LF	ç	50	30,720,000	75 915	\$4,000,000
IIDD (All Inclusive)		2 200	15	ć	2 200	ćr 100.000	75	¢2 100 000
HDD (All Inclusive)		2,500	LF	Ş	2,200	\$5,100,000	75 yrs	\$5,100,000
Pump Stations	Cost Estimates	for Pump Station	s are based on EBN	ЛUD Co	nstruction Bid C	Costs		
Duran Chattan (Tabal tastallad UD, taskadina atau dhu)		EA (Cost Curv	ve based on total					
Pump Station (Total Installed HP, Including standby)	750 HP	insta	alled HP)	\$	2,800,000	\$2,800,000	50 yrs	\$1,100,000
		Estimated 6	whetetal Cost (2) in	مايطامه	Contingonau	624 000 000		¢20,000,000
		Estimated S	Subtotal Cost (2) in	iciuain	g Contingency	\$34,000,000		\$20,000,000
Estimated Subtotal Construction Cost including Cont	ingency				50/	\$110,000,000		\$20,000,000
Mobilization					5%	\$5,500,000		\$1,000,000
Estimated Subtotal Construction Cost including Cont	ingency and Mo	bilization				\$120,000,000		\$21,000,000
Implementation								
Environmental Documentation & Permits					5%	\$6,000,000		\$1,100,000
Design Cost					15%	\$18,000,000		\$3,200,000
Project Administration and Construction Managemer	nt Cost				10%	\$12,000,000		\$2,100,000
Estimated Total Capital Cost including Implementati	on and Continge	ency				\$160,000,000		\$27,000,000
						Present Worth of Sa	alvage Value	\$11,000,000
O&M Costs (Annual)								
Advanced Water Treatment				<u>,</u>	225 222	<u> </u>		
MF/UF system	8		MGD	Ş	325,000	\$2,600,000		
RO System	8		MGD	Ş	480,000	\$3,800,000		
Advanced Oxidation and Disinfection	8		MGD	Ş	49,000	\$390,000		
Free Chlorine Chamicala	ð		MGD	Ş	32,000	\$260,000		
	0	2 000	IVIGD	Ş	120,000	\$960,000		
Labor	8	2,080	nrs/MGD	Ş	135	\$2,200,000		
Electricity		7,240,000	kWh/yr	Ş	0.15	\$1,100,000		
Conveyance								
Annual O&M			2% of construction	on cost		\$620,000		
Pump Stations								
Labor Costr		E00	brc	ć	125	¢69.000		
Electricity		2 752 000	hins kwb/wr	ڊ د	0.15	\$08,000		
Electricity		2,752,000	KVV1/yi	ڊ 	0.15	\$410,000		
		5% of pump sta		LOSI		\$140,000		
			Total O&IVI Costs	5 (Ş/yr)		\$13,000,000		
Annualized Costs (\$ / Year)	- ,		0.1.1.1.1			<u> </u>		
Annualized Capital Costs (\$/Year)	I wo payments	per year, spreaa o	over Project Life			\$8,100,000		
Annualized Salvage Value	Annualized val	ie of present wor	tn			-\$560,000		
Annual O&M Costs						\$13,000,000		
Total Annualized Cost					-	\$21,000,000		
Deliveries of Recycled Water	8,9	50 AFY						
Estimated Unit Cost (\$/AF)						\$2,300		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$1,900		Distribution	\$400		
Annualized Costs (\$ / Year)	Dry Year Adius	tment (Sunnly us	ed 3/10 years)					
Annualized Capital Costs (\$/Vear)	Two novments	ner vear spread	over Project Life			\$8 100 000	Same as constant	use.
Annualized Salvage Value	Annualized val	le of present wor	th			-\$560.000	Same as constant	use.
Annual O&M Costs		.c oj present won				\$10 637 000	same as constant	
Total Annualized Cost						\$10,037,000 \$10,000 000		
Annual Average Deliveries of Recycled Water	2 6	R8 AFV			-	\$10,000,000		
Estimated Unit Cost (\$ /AE)	2,00					¢c 700		
Estimated Unit Cost (\$/AF)		Turatura	¢5 600	_	Distail 1	\$0,700 61.000		
preakuown of Estimated Unit Cost (\$/AF)		rreatment	33.000		Distribution	51.100		

Alternative Oro-Treat-8							EBMUD Recycl	led Water Master Plan Update
Last Updated:	30-May-18					Discount Rate	F	roject Life
Updated by:	I. Jaffe					3%	3	30 Years
CCI (20 City, Dec 2017): 10870.06	Cine	011	11		Linit Cost	Total Cast	Heeful Life	Colugge Volue
Capital Costs	5120	Qty	Onit		Unit Cost	Total Cost	Userur Lite	Salvage value
Treatment	Cost Estimates	for Treatment ar	e based on Unit Co	osts for	Each Treatmer	nt Process		
Ozone	8		MGD	\$	335,300	\$2,700,000	30 yrs	\$0
BAC	8		MGD	Ş	300,900	\$2,400,000	30 yrs	\$0
RO System	8		MGD	Ş ¢	1,225,000	\$9,800,000	30 yrs	\$U \$0
Advanced Oxidation and Disinfection	8		MGD	\$	437,500	\$3,500,000	30 yrs	\$0 \$0
Chemicals (Storage and Use)	8			\$	125,000	\$1,000,000	30 yrs	\$0
Sitework/Piping/Structures	8		MGD	\$	3,187,500	\$26,000,000	30 yrs	\$0
Contractor Overhead & Profit			Raw Constr	ruction	Lost Subtotal	\$57,000,000 \$8,600,000		\$0 \$0
Sales Tax	50%	% of Subtotal Cost	Applicable		9%	\$2,600,000		\$0
			Estimated Subto	tal Cons	struction Cost	\$68,000,000		\$0
Construction Cost Including Contingency			Estimated Subtot	tal Cons	struction Cost	\$68,000,000		\$0
Estimating Contingency					25%	\$17,000,000		\$0
		Estimated S	ubtotal Cost (1) in	cluding	Contingency	\$85,000,000		\$0
Capital Costs (from Bids)	Cost Estimates	for Pinelines are	hased on ERMUD	constru	uction hid cost 4	estimating contingency		
Non-urban Pipeline	24-in	700 LF	in-LF	\$	30	\$504,000	75 yrs	\$300,000
Low-Density Urban Pipeline	24-in	8,100 LF	in-LF	\$	40	\$7,776,000	75 yrs	\$4,700,000
High-Density Urban Pipeline	24-in	1,400 LF	in-LF	\$	50	\$1,680,000	75 yrs	\$1,000,000
Trenchless Crossings		200		~	2 200	¢660.000	75	¢400.000
HDD (All Inclusive)		300	LF	\$	2,200	\$660,000	75 yrs	\$400,000
Pump Stations	Cost Estimates	for Pump Station	is are based on EB	MUD C	onstruction Bid	Costs		
Pump Station (Total installed HP, including standby)	750 HP	EA (COST CUIV	alled HP)	Ś	2.800.000	\$2.800.000	50 vrs	\$1,100.000
Storage Tanks	Cost Estimates	for Storage Tank	s are based on EBI	MUD Ca	onstruction Bid	Costs	50 115	\$1,100,000
Storage Tanks			EA (Cost Curve b	ру				
(material determined during pre-design)	0.7 MG	3	Volume)	\$	3,000,000	\$9,000,000	75 yrs	\$5,400,000
		Estimated S	ubtotal Cost (2) in	cluding	Contingency	\$22,000,000		\$13,000,000
Estimated Subtotal Construction Cost including Con	tingency				F0/	\$110,000,000		\$13,000,000
Estimated Subtotal Construction Cost including Con	tingency and M	obilization			5%	\$3,500,000 \$120.000.000		\$050,000
Implementation								
Environmental Documentation & Permits					5%	\$6,000,000		\$700,000
Design Cost	at Cost				15%	\$18,000,000		\$2,100,000
Estimated Total Capital Cost including Implementati	ion and Conting	encv			10%	\$12,000,000		\$1,400,000
······································						Present Worth of Sa	alvage Value	\$7,400,000
O&M Costs (Annual)								
Ozone	8		MGD	Ś	57.000	\$460.000		
BAC	8		MGD	Ş	116,000	\$930,000		
MF/UF system	8		MGD	\$	325,000	\$2,600,000		
RO System	8		MGD	Ş	480,000	\$3,800,000		
Free Chlorine	8		MGD	ŝ	32,000	\$260,000		
Chemicals	8		MGD	\$	120,000	\$960,000		
Labor	8	2,080	hrs/MGD	\$	135	\$2,200,000		
Electricity		9,944,000	kWh/yr \$/year	ş	0.15	\$1,500,000		
Convoyance			Ş∕ yeai	Ŷ	1,000,000	\$1,000,000		
Annual O&M			2% of construction	on cost		\$210,000		
Pump Stations								
Labor Costs		500	hrs	\$	135	\$68,000		
Electricity		2,523,000	kWh/yr	\$	0.15	\$380,000		
Pump Station Consumables		5% of pump sta	ation construction	cost		\$140,000		
Storage Tanks								
Annual O&M		1% of co	Total OSM Cost	- (¢ /)		\$90,000		
Annualized Costs (\$ / Year)			Total Oxivi Costs	s (ş/yr)		\$15,000,000		
Annualized Capital Costs (\$/Year)	Two payments	per year, spread	over Project Life			\$8,100,000		
Annualized Salvage Value	Annualized val	ue of present wor	-th			-\$380,000		
Annual O&M Costs						\$15,000,000		
Deliveries of Recycled Water	8.9	50 AFY			-	\$23,000,000	-	
Estimated Unit Cost (\$/AF)	0,54					\$2,600		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$2,200		Distribution	\$400		
Annualized Costs (\$ / Year)	Dry Year Adjus	stment (Supply us	ed 3/10 years)					
Annualized Capital Costs (\$/Year)	Two payments	per year, spread	over Project Life			\$8,100,000	Same as cons	tant use.
Annualized Salvage Value	Annualized val	ue of present wor	-th			-\$380,000	Same as cons	tant use.
Annual O&M Costs						\$12,118,000		
Annual Average Deliveries of Recycled Water	2.6	38 AFY			-	\$20,000,000	-	
Estimated Unit Cost (\$/AF)	2,00					\$7,400		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$6,700		Distribution	\$700		

Alternative CC-Raw-19							EBMUD Recycled Wat	er Master Plan Update
Last Updated:	30-May-18					Discount Rate	Project Li	fe
Updated by:	I. Jaffe					3%	30 Years	
CCI (20 City, Dec 2017): 10870.06	c :	<u></u>						
Capital Costs	Size	Qty	Unit		Unit Cost	Total Cost	Useful Life	Salvage value
Treatment	Cost Estimates	for Treatment are	based on Unit Cost	ts for E	ach Treatmen	t Process		
MF/UF system	19		MGD	\$	1,225,000	\$23,000,000	30 yrs	\$0
RO System	19		MGD	\$	1,475,000	\$28,000,000	30 yrs	\$0
Advanced Oxidation and Disinfection	19		MGD	\$	437,500	\$8,300,000	30 yrs	\$0
Chemicals (Storage and Use)	19			\$	125,000	\$2,400,000	30 yrs	\$0
Sitework/Piping/Structures	19		MGD	\$	3,187,500	\$61,000,000	30 yrs	\$0
			Raw Constru	iction	Cost Subtotal	\$120,000,000		\$0
Contractor Overhead & Profit	500/	M . (C. http://c.o.)	A P M.		15%	\$18,000,000		\$0 ¢0
sales lax	50%	% of Subtotal Cost	Applicable Estimated Subtota	l Cons	9% struction Cost	\$5,400,000		\$0 \$0
			Estimated Subtota	ii cons	struction cost	Ş1 4 0,000,000		ÇÇ.
Construction Cost Including Contingency								
			Estimated Subtota	al Cons	struction Cost	\$140,000,000		\$0
Estimating Contingency					25%	\$35,000,000		\$0
Constant Consta (forum Dista)		Estimated S	ubtotal Cost (1) inc	luding	contingency	\$180,000,000		\$0
Capital Costs (from Bids)	Cost Estimator	for Dipolinos ara k	acad on ERMUD ca	netrue	tion hid cost 4	actimating contingancy		
Non-urban Pipeline	36-in	15 300 LF	in-I F	s Ś	30	\$16 524 000	75 vrs	\$9 900 000
High-Density Urban Pipeline	36-in	3.200 LF	in-LF	ŝ	50	\$5,760,000	75 yrs	\$3,500,000
Trenchless Crossings		-,		·		,,	- / -	, ,
Microtunnel Xings (1 jack & 1 receiv. pit)		3	EA	\$	620,000	\$1,900,000		
Microtunnel Pipe		2,500	LF	\$	2,800	\$7,000,000	75 yrs	\$4,200,000
Pump Stations	Cost Estimates	for Pump Stations	are based on EBM	UD Co	nstruction Bid	Costs		
		EA (Cost Curv	e based on total					
Pump Station (Total installed HP, including standby)	2,400 HP	insta	lled HP)	\$	9,500,000	\$9,500,000	50 yrs	\$3,800,000
Storage Tanks	Cost Estimates	for Storage Tanks	are based on EBM	UD Col	nstruction Bid	Costs		
Storage Tanks			EA (Cost Curve by	,				
(material determined during pre-design)	1.6 MG	3	Volume)	\$	4,000,000	\$12,000,000	75 yrs	\$7,200,000
		Estimated S	ubtotal Cost (2) inc	luding	contingency	\$53,000,000		\$29,000,000
Estimated Subtotal Construction Cost including Cont	ingency					\$230,000,000		\$29,000,000
Mobilization					5%	\$12,000,000		\$1,500,000
Estimated Subtotal Construction Cost including Cont	ingency and Mo	obilization				\$240,000,000		\$31,000,000
Implementation Environmental Decumentation & Permits					E 9/	\$12,000,000		\$1 600 000
Design Cost					15%	\$12,000,000		\$1,000,000
Project Administration and Construction Managemer	nt Cost				10%	\$24,000,000		\$3,100,000
Estimated Total Capital Cost including Implementati	on and Conting	ency				\$310,000,000		\$40,000,000
						Present Worth of S	alvage Value	\$16,000,000
O&M Costs (Annual)								
Advanced Water Treatment	10		MGD	ć	225 000	\$6 200 000		
RO System	19		MGD	ŝ	480.000	\$9,100,000		
Advanced Oxidation and Disinfection	19		MGD	\$	49,000	\$930,000		
Free Chlorine	19		MGD	\$	32,000	\$610,000		
Chemicals	19		MGD	\$	120,000	\$2,300,000		
Labor	19	2,080	hrs/MGD	Ş	135	\$5,300,000		
Electricity		17,195,000	kwn/yr	Ş	0.15	\$2,600,000		
Surface Water Treatment			ə/year	Ş	300,000	\$300,000		
SWT O&M (Walnut Creek)	6,935		MG	\$	76	\$530,000		
Conveyance	-,					,		
Annual O&M			2% of construction	n cost		\$620,000		
			2/0 01 001150 00010			<i>\$</i> 020,000		
Labor Costs		500	hrc	ć	125	¢60.000		
Electricity		10 621 000	kWh/vr	ڊ د	0.15	\$08,000		
Pump Station Consumables		5% of pump sta	tion construction c	ost	0.15	\$480.000		
						,,		
Storage Tanks		10/ - 6				¢120.000		
Annual O&M		1% Of CO	Total OSM Costs /	¢ /um		\$120,000		
Annualized Costs (\$ / Year)			Total Odivi Costs (\$51,000,000		
Annualized Capital Costs (\$/Year)	Two payments	per vear, spread o	over Proiect Life			\$16.000.000		
Annualized Salvage Value	Annualized val	ue of present wort	:h			-\$810,000		
Annual O&M Costs						\$31,000,000		
Total Annualized Cost						\$46,000,000		
Deliveries of Recycled Water	21,2	280 AFY						
Estimated Unit Cost (\$/AF)						\$2,200		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$1,900		Distribution	\$300		
Annualized Costs (\$ / Year)	Dry Year Adju	stment (Supply us	ed 3/10 years)					
Annualized Capital Costs (\$/Year)	Two payments	per year, spread o	over Project Life			\$16,000,000	Same as constant use.	
			h			-\$810,000	Same as constant use.	
Annualized Salvage Value	Annualized val	ue of present wort				60F 050 000		
Annualized Salvage Value Annual O&M Costs	Annualized val	ue of present wort				\$25,260,000		
Annualized Salvage Value Annual O&M Costs Total Annualized Cost Annual Average Deliveries of Recycled Water	Annualized val	84 AFY				\$25,260,000 \$40,000,000	-	
Annualized Salvage Value Annual O&M Costs Total Annualized Cost Annual Average Deliveries of Recycled Water Estimated Unit Cost (5/AF)	Annualized val	84 AFY				\$25,260,000 \$40,000,000 \$6,300		

Alternative CC-Raw-10						E	BMUD Recycle	ed Water Master Plan Update
Last Updated:	30-May-18					Discount Rate	Pr	oject Life
Updated by:	I. Jaffe					3%	30) Years
CCI (20 City, Dec 2017): 10870.06								
Item	Size	Qty	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value
Capital Costs								
Treatment	Cost Estimates	for Treatment are	based on Unit Cost	s for E	ach Treatmen	t Process	20	ćo
NF/UF system	10		MGD	Ş	1,225,000	\$12,000,000	30 yrs	\$0 ¢0
RO System	10		MGD	Ş	1,475,000	\$15,000,000	30 yrs	\$0
Advanced Oxidation and Disinfection	10		MGD	Ş	437,500	\$4,400,000	30 yrs	\$0
Chemicals (Storage and Use)	10		MCD	Ş	125,000	\$1,300,000	30 yrs	\$U ¢0
sitework/Piping/structures	10		NIGD Bow Constru	<u>ې</u>	5,167,500	\$32,000,000	50 yrs	\$0 \$0
Contractor Overhead & Brofit			Raw Construc	cuon	1 50/	\$0,000,000		30
	50%	% of Subtotal Cost	Applicable		13%	\$3,800,000		\$0 \$0
Sales Tax	50%	% OF SUDIOLAT COST	Estimated Subtota	Con	570 struction Cost	\$2,900,000		30 \$0
			Linated Subtota	r cons	struction cost	\$78,000,000		υÇ
Construction Cost Including Contingency								
<u></u>			Estimated Subtota	l Con	struction Cost	\$78,000,000		\$0
Estimating Contingency					25%	\$20,000,000		\$0
		Estimated S	ubtotal Cost (1) incl	luding	contingency	\$98,000,000		\$0
Capital Costs (from Bids)								
<u>Conveyance</u>	Cost Estimates	for Pipelines are b	ased on EBMUD co	nstruc	tion bid cost +	estimating contingency		
Non-urban Pipeline	24-in	15,300 LF	in-LF	\$	30	\$11,016,000	75 yrs	\$6,600,000
High-Density Urban Pipeline	24-in	3,200 LF	in-LF	\$	50	\$3,840,000	75 yrs	\$2,300,000
Trenchless Crossings								
HDD (All Inclusive)		2,500	LF	\$	2,200	\$5,500,000	75 yrs	\$3,300,000
Pump Stations	Cost Estimates	for Pump Stations	are based on EBML	JD Co	nstruction Bid	Costs		
Dump Station (Total installed UD including standbu)		EA (Cost Curv	e based on total					
Pump Station (Total Installed HP, Including standby)	1,500 HP	insta	lled HP)	\$	5,300,000	\$5,300,000	50 yrs	\$2,100,000
Storage Tanks	Cost Estimates	for Storage Tanks	are based on EBMU	JD Co	nstruction Bid	Costs		
Storage Tanks			EA (Cost Curve by					
(material determined during pre-design)	0.9 MG	3	Volume)	\$	3,200,000	\$9,600,000	75 yrs	\$5,800,000
		Estimated S	ubtotal Cost (2) incl	luding	Contingency	\$35,000,000		\$20,000,000
Estimated Subtotal Construction Cost including Cont	ingency					\$130,000,000		\$20,000,000
Mobilization					5%	\$6,500,000		\$1,000,000
Estimated Subtotal Construction Cost including Cont	ingency and Mo	bilization				\$140,000,000		\$21,000,000
Implementation								
Environmental Documentation & Permits					5%	\$7,000,000		\$1,100,000
Design Cost					15%	\$21,000,000		\$3,200,000
Project Administration and Construction Managemen	t Cost				10%	\$14,000,000		\$2,100,000
Estimated Total Capital Cost including Implementation	on and Continge	ency				\$180,000,000		\$27,000,000
						Present Worth of Sa	Ivage Value	\$11,000,000
O&M Costs (Annual) Advanced Water Treatment								
ME/IIE system	10		MCD	ć	225 000	\$2,200,000		
RO System	10		MGD	ŝ	480,000	\$4,800,000		
Advanced Oxidation and Disinfection	10		MGD	Ś	49.000	\$490.000		
Free Chlorine	10		MGD	\$	32,000	\$320,000		
Chemicals	10		MGD	\$	120,000	\$1,200,000		
Labor	10	2,080	hrs/MGD	\$	135	\$2,800,000		
Electricity		9,050,000	kWh/yr	\$	0.15	\$1,400,000		
Monitoring			\$/year	\$	500,000	\$500,000		
Surface Water Treatment								
SWT O&M (Walnut Creek)	3,650		MG	\$	76	\$280,000		
Conveyance								
Annual O&M			2% of construction	cost		\$410,000		
Pump Stations								
Labor Costs		500	hrs	¢	135	\$68,000		
Electricity		5 590 000	kWb/yr	ç	0.15	\$840,000		
Pump Station Consumables		5% of nump sta	tion construction co	ې het	0.15	\$270,000		
		570 01 pump sta		550		\$270,000		
Storage Tanks								
Annual O&M		1% of co	nstruction cost			\$96,000		
			Total O&M Costs (Ş/yr)		\$17,000,000		
Annualized Costs (\$ / Year)		par yoar spraad a	war Brojact Lifa			¢0 100 000		
	Annualized value	ie of present wort	h			-\$560,000		
	rannaunzeu van	ic of present wort				\$17 000 000		
Total Annualized Cost						\$26 000 000		
Deliveries of Recycled Water	11.2	00 AFY				\$20,000,000		
Estimated Unit Cost (\$/AF)	,2					\$2,300		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$2,000		Distribution	\$300		
Annualized Costs (\$ / Year)	Dry Voor Adi	tmont (Sumalu	ad 2/10 years					
Annualized Costs (\$ / Tear)	Two part Adjus	nent (Supply us	eu 3/10 years)			¢0 100 000	Same as const	antuse
Annualized Capital Costs (\$/Year)	Appualized	per yeur, spredd o	wei Project Life			\$9,100,000 ·	Same as const	ant use.
Annual O&M Costs	Annuulizea Vall	ie oj present wort				->>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>	Juille us coliste	ant use.
Total Annualized Cost						\$33 UUU UUU \$13,596,000		
Annual Average Deliveries of Recycled Water	2 21	50 AFY				ېدکې		
Estimated Unit Cost (\$/AF)	3,51					\$6,500		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$5.700		Distribution	\$800		

Alternative CC-ResB-19							EBMUD Recycle	d Water Master Plan Lindate
Last Lindated:	30-May-18					Discount Rate	DWOD Recycle	viect Life
Lindated by:	J laffe					3%	30	Years
CCI (20 City, Dec 2017): 10870.06	i. suric					570	50	
Item	Size	Qty	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value
Capital Costs								
Treatment	Cost Estimates	for Treatment are	e based on Unit Cost	s for E	ach Treatmen	t Process		
MF/UF system	19		MGD	\$	1,225,000	\$23,000,000	30 yrs	\$0
RO System	19		MGD	\$	1,475,000	\$28,000,000	30 yrs	\$0
Advanced Oxidation and Disinfection	19		MGD	\$	437,500	\$8,300,000	30 yrs	\$0
Chemicals (Storage and Use)	19			\$	125,000	\$2,400,000	30 yrs	\$0
Sitework/Piping/Structures	19		MGD	\$	3,187,500	\$61,000,000	30 yrs	\$0
			Raw Constru	ction	Cost Subtotal	\$120,000,000		\$0
Contractor Overhead & Profit					15%	\$18,000,000		\$0
Sales Tax	50%	% of Subtotal Cost	Applicable		9%	\$5,400,000		\$0
			Estimated Subtota	I Cons	truction Cost	\$140,000,000		\$0
Construction Cost Including Contingency								
			Estimated Subtota	I Cons	truction Cost	\$140,000,000		\$0
Estimating Contingency					25%	\$35,000,000		\$0
		Estimated S	ubtotal Cost (1) inc	luding	Contingency	\$180,000,000		\$0
Capital Costs (from Bids)								
Conveyance	Cost Estimates	for Pipelines are l	pased on EBMUD co	nstruc	tion bid cost +	estimating contingency		
Non-urban Pipeline	36-in	27,600 LF	in-LF	\$	30	\$29,808,000	75 yrs	\$18,000,000
Low-Density Urban Pipeline	36-in	27,200 LF	in-LF	\$	40	\$39,168,000	75 yrs	\$24,000,000
Trenchless Crossings								
Microtunnel Xings (1 jack & 1 receiv. pit)		8	EA	\$	620,000	\$5,000,000		
Microtunnel Pipe		3,300	LF	\$	2,800	\$9,200,000	75 yrs	\$5,500,000
Pump Stations	Cost Estimates	for Pump Station	s are based on FRMI		nstruction Rid	Costs		
	cost Estimates	EA (Cost Curv	e based on total	00 00.		0000		
Pump Station (Total installed HP, including standby)	3.600 HP	insta	illed HP)	Ś	17.000.000	\$17.000.000	50 vrs	\$6.800.000
Storage Tanks	Cost Estimates	for Storaae Tanks	are based on EBML	JD Coi	nstruction Bid	Costs		
Storage Tanks		,	EA (Cost Curve by					
(material determined during pre-design)	0.0 MG	0	Volume)	\$	2,300,000	\$0	75 yrs	\$0
		Estimated 6	whether Cost (2) in a	متاميا	Contingonau	¢100.000.000		¢54.000.000
Ectimated Subtotal Construction Cost including Cont	Ingonau	Estimated 5	ubtotal Cost (2) Inc	luaing	contingency	\$100,000,000		\$54,000,000
Mobilization	ingency				E 9/	\$280,000,000		\$34,000,000
Estimated Subtotal Construction Cost including Cont	ingency and Mo	hilization			570	\$14,000,000		\$57,000,000
Implementation	ingency and mo	Sinzution				\$250,000,000		\$37,000,000
Environmental Documentation & Permits					5%	\$15,000,000		\$2,900,000
Design Cost					15%	\$13,000,000		\$8,600,000
Project Administration and Construction Managemer	t Cost				10%	\$29,000,000		\$5,700,000
Estimated Total Capital Cost including Implementati	on and Continge	ncv			1076	\$380,000,000		\$3,700,000
	en una continge					Present Worth of Sa	lvage Value	\$30,000,000
O&M Costs (Annual)								
Advanced Water Treatment								
MF/UF system	19		MGD	\$	325,000	\$6,200,000		
RO System	19		MGD	\$	480,000	\$9,100,000		
Advanced Oxidation and Disinfection	19		MGD	\$	49,000	\$930,000		
Free Chlorine	19		MGD	\$	32,000	\$610,000		
Chemicals	19		MGD	\$	120,000	\$2,300,000		
Labor	19	2,080	hrs/MGD	\$	135	\$5,300,000		
Electricity		17,195,000	kWh/yr	\$	0.15	\$2,600,000		
Surface Water Treatment								
SWT O&M (Orinda)	6,935		MG	\$	70	\$490,000		
Convevance								
Annual O&M			2% of construction	l cost		\$1,660,000		
Pump Stations								
Labor Costs		1000	brc	ć	125	\$140.000		
		16 612 000	lins	ç	133	\$140,000		
Electricity		10,013,000	KVVII/yi	Ş	0.15	\$2,500,000		
Pump station consumables		5% of pump sta	Total OSM Costs (¢ /url		000,000		
Annualized Costs (\$ / Year)			10131 00010 00313 (<i>, , , , ,</i>		\$55,000,000		
Annualized Costs (\$7 Fear)	Two payments	ner vear spread (over Project Life			\$19.000.000		
Annualized Salvage Value	Annualized valu	ie of present wor	h			-\$1,500,000		
	/	ie of present work				\$33,000,000		
Total Annualized Cost						\$51,000,000		
Deliveries of Recycled Water	21.2	80 AFY				\$51,000,000		
Estimated Unit Cost (\$/AF)						\$2.400		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$1.900		Distribution	\$500		
Annualized Costs (C. (March)	Day Very All	two outs / Courses 1	ad 2/10					
Annualized Costs (\$ / Year)	Dry Year Adjus	tment (Supply us	ea 3/10 years)			¢10.000.000	Sama as senst-	nt uco
Annualized Capital Costs (\$/Year)	Appualized	per yeur, spreda d	wer Project Lije			\$19,000,000	Some as consta	nt use.
Annualized Salvage Value	Annualized valu	ie oj present worl	.11			-\$1,500,000	same as consta	nt use.
Annual U&M Costs						\$26,730,000		
Annual Average Doliveries of Desured at Mater	C 24					\$44,000,000		
Ectimated Unit Cost (\$ (45)	6,32	D4 AFT				¢C 000		
Estimated Unit Cost (\$/AF)		Treatment	¢5 500	_	Distributio	\$6,900		
Dreakdown of Estimated Unit Cost (\$/AF)		rreatment	\$5,500		Distribution	\$1,400		

Alternative CC-ResB-10							FBMUD Recycle	d Water Master Plan Lindate
Last Lindated:	30-May-18					Discount Pate	Pro	iect Life
Lindated by:	L laffe					3%	30	Vears
CCL (20 City, Dec 2017): 10870.06	1. June					570	50	rears
Item	Size	Qty	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value
Capital Costs								
Treatment	Cost Estimates	for Treatment are	based on Unit Co	sts for l	ach Treatmer	nt Process		
MF/UF system	10		MGD	\$	1,225,000	\$12,000,000	30 yrs	\$0
RO System	10		MGD	\$	1,475,000	\$15,000,000	30 yrs	\$0
Advanced Oxidation and Disinfection	10		MGD	\$	437,500	\$4,400,000	30 yrs	\$0
Chemicals (Storage and Use)	10			\$	125,000	\$1,300,000	30 yrs	\$0
Sitework/Piping/Structures	10		MGD	<u></u>	3,187,500	\$32,000,000	30 yrs	\$0
Contractor Querhand & Drafit			Raw Constr	ruction	Lost Subtotal	\$65,000,000		\$U ¢0
	E 0%/	% of Subtotal Cost	Applicable		15%	\$9,800,000		50 ¢0
Sales Tax	50%	% OF SUBIOLAL COST	Estimated Subtot	tal Con	struction Cost	\$2,900,000		50 \$0
						<i>\$10,000,000</i>		γu
Construction Cost Including Contingency								
			Estimated Subtor	tal Con	struction Cost	\$78,000,000		\$0
Estimating Contingency					25%	\$20,000,000		\$0
		Estimated S	ubtotal Cost (1) in	ncluding	g Contingency	\$98,000,000		\$0
Capital Costs (from Bids)								
<u>Conveyance</u>	Cost Estimates	for Pipelines are b	ased on EBMUD c	construc	tion bid cost +	estimating contingency		
Non-urban Pipeline	24-in	27,600 LF	in-LF	\$	30	\$19,872,000	75 yrs	\$12,000,000
Low-Density Urban Pipeline	24-in	27,200 LF	in-LF	\$	40	\$26,112,000	75 yrs	\$16,000,000
Trenchless Crossings								
HDD (All Inclusive)		3,300	LF	Ş	2,200	\$7,300,000	75 yrs	\$4,400,000
Pump Stations	Cost Estimates	for Pump Stations	are based on EBN	ЛUD Co	nstruction Bid	Costs		
Pump Station (Total installed HP including standby)		EA (Cost Curv	e based on total					
and station (rotal instance in , including statiosy)	2,000 HP	insta	lled HP)	\$	7,500,000	\$7,500,000	50 yrs	\$3,000,000
		Estimated S	ubtotal Cost (2) in	ncluding	g Contingency	\$61,000,000		\$35,000,000
Estimated Subtotal Construction Cost including Cont	ingency					\$160,000,000		\$35,000,000
Mobilization					5%	\$8,000,000		\$1,800,000
Estimated Subtotal Construction Cost including Cont	ingency and Mo	bilization				\$170,000,000		\$37,000,000
Implementation								
Environmental Documentation & Permits					5%	\$8,500,000		\$1,900,000
Design Cost					15%	\$26,000,000		\$5,600,000
Project Administration and Construction Managemen	it Cost				10%	\$17,000,000		\$3,700,000
Estimated Total Capital Cost including Implementation	on and Continge	ency				\$220,000,000 Present Worth of S	alvago Valuo	\$48,000,000
O&M Costs (Annual)							intuge value	<i><i><i></i></i></i>
Advanced Water Treatment								
MF/UF system	10		MGD	\$	325,000	\$3,300,000		
RO System	10		MGD	\$	480,000	\$4,800,000		
Advanced Oxidation and Disinfection	10		MGD	\$	49,000	\$490,000		
Free Chlorine	10		MGD	Ş	32,000	\$320,000		
Chemicals	10	2 090	MGD brs/MCD	Ş	120,000	\$1,200,000		
Electricity	10	2,080	kWb/vr	ç ç	135	\$2,800,000		
Surface Water Treatment		5,050,000	KVVII/ yi	ç	0.15	\$1,400,000		
SWT Q&M (Orinda)	3 650		MG	Ś	70	\$260,000		
	3,050		ine	Ŷ		<i>\$200,000</i>		
Conveyance Appual ORM			2% of construction	n cost		\$1.070.000		
Annual Galvi			2/0 OF CONStructio	JII COSL		\$1,070,000		
Pump Stations						*** ***		
Labor Costs		500	hrs	Ş	135	\$68,000		
Electricity		8,743,000	KVVN/yr	Ş	0.15	\$1,300,000		
Pump station consumables		5% of pump sta	Total O&M Costs	(\$/wr)		\$360,000		
Annualized Costs (\$ / Year)				· (\$17,000,000		
Annualized Capital Costs (\$/Year)	Two payments	per vear, spread o	over Proiect Life			\$11.000.000		
Annualized Salvage Value	Annualized val	ie of present wort	:h			-\$1,000,000		
Annual O&M Costs						\$17,000,000		
Total Annualized Cost						\$27,000,000		
Deliveries of Recycled Water	11,2	00 AFY						
Estimated Unit Cost (\$/AF)						\$2,400		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$1,900		Distribution	\$500		
Annualized Costs (\$ / Year)	Dry Year Adjus	tment (Supply us	ed 3/10 years)					
Annualized Capital Costs (\$/Year)	Two payments	per year, spread o	over Project Life			\$11,000,000	Same as constar	nt use.
Annualized Salvage Value	Annualized val	ue of present wort	h			-\$1,000,000	Same as constar	nt use.
Annual O&M Costs						\$14,252,000		
Total Annualized Cost						\$24,000,000		
Annual Average Deliveries of Recycled Water	3,30	50 AFY				-		
Estimated Unit Cost (\$/AF)			4			\$7,100		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$5,700		Distribution	S1.400		

Alternative SD1-Raw-30							EBMUD Recycled	d Water Master Plan Update
Last Updated:	30-May-18					Discount Rate	Pro	ject Life
Updated by:	I. Jaffe					3%	30	Years
CCI (20 City, Dec 2017): 10870.06								
Item	Size	Qty	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value
Capital Costs	Cost Estimates	for Treatment are	hased on Unit Cost	s for F	Each Treatme	at Process		
MF/UF system	30	jor meatment are	MGD	s jui e Ś	1.225.000	\$37.000.000	30 vrs	\$0
RO System	30		MGD	ś	1.475.000	\$44.000.000	30 yrs	\$0
Advanced Oxidation and Disinfection	30		MGD	\$	437,500	\$13,000,000	30 yrs	\$0
Chemicals (Storage and Use)	30			\$	125,000	\$3,800,000	30 yrs	\$0
Sitework/Piping/Structures	30		MGD	\$	3,187,500	\$96,000,000	30 yrs	\$0
			Raw Constru	ction	Cost Subtotal	\$190,000,000)	\$0
Contractor Overhead & Profit					15%	\$29,000,000		\$0
Sales Tax	50%	% of Subtotal Cost	Applicable		9%	\$8,600,000		\$0
			Estimated Subtota	I Cons	struction Cost	\$230,000,000		\$0
Construction Cost Including Contingency								
contraction cost mentaling contingency			Estimated Subtota	l Cons	struction Cost	\$230,000,000		\$0
Estimating Contingency					25%	\$58,000,000	1	\$0
		Estimated S	ubtotal Cost (1) incl	luding	g Contingency	\$290,000,000)	\$0
Capital Costs (from Bids)								
Conveyance	Cost Estimates	for Pipelines are b	based on EBMUD cor	nstruc	tion bid cost -	+ estimating contingency	75	¢20,000,000
Non-urban Pipeline	42-in 42-in	26,700 LF	in-LF	Ş	30	\$33,642,000	75 yrs	\$20,000,000
Low-Density Urban Pipeline	42-111 42 in	19,500 LF	in LE	ې د	40 E0	\$32,700,000	75 yrs	\$20,000,000
Trenchless Crossings	42-111	12,700 LF	III-LF	Ş	30	\$20,070,000	, 75 yis	\$10,000,000
San Pablo Tunnel Rehab		17.600	LF	Ś	3.500	\$62,000,000	75 vrs	\$37.000.000
Microtunnel Xings (1 iack & 1 receiv. pit)		4	EA	\$	620.000	\$2,500,000	1	<i>\$31,000,000</i>
Microtunnel Pipe		2,200	LF	\$	2,800	\$6,200,000	75 yrs	\$3,700,000
Pump Stations	Cost Estimates	for Pump Station	are based on FRMI		nstruction Bio	Costs	•	
	cost Estimates	FA (Cost Curv	e hased on total			1 COSIS		
Pump Station (Total installed HP, including standby)	3.600 HP	insta	lled HP)	Ś	17.000.000	\$17.000.000	50 yrs	\$6.800.000
Storage Tanks	Cost Estimates	for Storaae Tanks	are based on EBMU	ID Col	nstruction Bid	Costs	,	+ • / • • • • • • • • •
Storage Tanks		,	EA (Cost Curve by					
(material determined during pre-design)	2.5 MG	3	Volume)	\$	4,500,000	\$14,000,000	75 yrs	\$8,400,000
		Estimated S	ubtotal Cost (2) incl	luding	Contingency	\$190,000,000	1	\$110,000,000
Estimated Subtotal Construction Cost including Cont	ingency	Littinuccu s		aame	contingency	\$480,000,000	, 	\$110,000,000
Mobilization	5- 7				5%	\$24,000,000	1	\$5,500,000
Estimated Subtotal Construction Cost including Cont	ingency and Mo	bilization				\$500,000,000)	\$120,000,000
Implementation								
Environmental Documentation & Permits					5%	\$25,000,000	1	\$6,000,000
Design Cost					15%	\$75,000,000)	\$18,000,000
Project Administration and Construction Managemen	t Cost				10%	\$50,000,000		\$12,000,000
Estimated Total Capital Cost including Implementation	on and Continge	ency				\$650,000,000 Brocopt Worth of S	alvago Valuo	\$160,000,000 \$65,000,000
O&M Costs (Annual)						Present Worth of 5	alvage value	\$05,000,000
Advanced Water Treatment								
MF/UF system	30		MGD	\$	325,000	\$9,800,000	1	
RO System	30		MGD	\$	480,000	\$14,000,000	1	
Advanced Oxidation and Disinfection	30		MGD	\$	49,000	\$1,500,000		
Free Chlorine Chamicals	30		MGD	Ş	32,000	\$960,000		
Labor	30	2 080	hrs/MGD	ڊ خ	120,000	\$3,000,000		
Electricity	50	27 150 000	kWh/vr	ŝ	0.15	\$4,100,000		
Monitoring		27,1200,000	\$/year	Ş	500,000	\$500,000		
Conveyance								
Annual Q&M			2% of construction	cost		\$3,300.000	1	
			270 01 001150 0000	0050		<i>\$3,500,000</i>		
Labor Costs		1000	brc	ć	125	\$140.000		
Electricity		18 060 000	his kwb/wr	ې د	155	\$140,000		
Pump Station Consumables		5% of nump sta	tion construction co	, st	0.15	\$850.000		
		ove of pump sta				<i>ç</i> 050,000		
Storage Tanks		10/ 5				<u>.</u>		
Annual O&M		1% of co	nstruction cost	÷ /		\$140,000		
Annualized Costs (\$ / Year)				<i>>/ yı)</i>		\$50,000,000	,	
Annualized Capital Costs (\$/Year)	Two payments	per vear, spread o	over Project Life			\$33.000.000)	
Annualized Salvage Value	Annualized val	ue of present wort	h			-\$3,300,000	1	
Annual O&M Costs						\$50,000,000)	
Total Annualized Cost						\$80,000,000	<u> </u>	
Deliveries of Recycled Water	33,6	00 AFY						
Estimated Unit Cost (\$/AF)						\$2,400		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$1,900		Distribution	\$500		
Annualized Costs (\$ / Year)	Dry Year Adjus	tment (Supply us	ed 3/10 years)					
Annualized Capital Costs (\$/Year)	Two payments	per year, spread o	over Project Life			\$33,000,000	Same as constan	it use.
Annualized Salvage Value	Annualized val	ue of present wort	h			-\$3,300,000	Same as constan	it use.
Annual O&M Costs						\$41,688,000	1	
Total Annualized Cost		00 4 51				\$71,000,000	-	
Annual Average Deliveries of Recycled Water	10,0	OU AFT				67.000		
Breakdown of Estimated Unit Cost (\$ /AF)		Treatment	\$5.600	_	Distribution	\$7,000		
Dieakuowii of Estimateu Offit Cost (\$/AF)		neatment	J J,000		Distribution	ş1,400		

Alternative SD1-Raw-10							EBMUD Recycled Water	· Master Plan Update
Last Updated:	30-May-18					Discount Rate	Project Life	2
Updated by:	I. Jaffe					3%	30 Years	
CCI (20 City, Dec 2017): 10870.06	<u>Ciao</u>	01	11mit		Unit Cost	Total Cast	Lisoful Life C	alvaga Valua
Capital Costs	Size	Qty	Unit		Unit Cost	Total Cost	Oserui Lite S	alvage value
Treatment	Cost Estimates	for Treatment are	based on Unit Cost	s for E	ach Treatmen	nt Process		
MBR	10		MGD	\$	11,100,000	\$110,000,000	30 yrs	\$0
RO System	10		MGD	\$	1,475,000	\$15,000,000	30 yrs	\$0
Advanced Oxidation and Disinfection	10		MGD	\$	437,500	\$4,400,000	30 yrs	\$0
Chemicals (Storage and Use)	10			\$	125,000	\$1,300,000	30 yrs	\$0
Sitework/Piping/Structures	10		MGD	\$	3,187,500	\$32,000,000	30 yrs	\$0
Construction Occursticated & Deafit			Raw Constru	ction	Cost Subtotal	\$160,000,000		\$0 ¢0
Contractor Overnead & Profit	500/	0/ of Cubbobol Cost	Applicable		15%	\$24,000,000		\$0 ¢0
Sales Tax	50%		Estimated Subtota		9%	\$1,200,000		\$0 \$0
			Lotimated Subtota	Cons	cost cost	\$150,000,000		20
Construction Cost Including Contingency								
			Estimated Subtota	l Cons	struction Cost	\$190,000,000		\$0
Estimating Contingency					25%	\$48,000,000		\$0
		Estimated S	ubtotal Cost (1) inc	luding	contingency	\$240,000,000		\$0
Capital Costs (from Bids)	Cost Estimatos	for Dinalinas are k	acad on FRMUD an	notrue	tion hid cost i	actimating contingonau		
Non-urban Pipeline	24-in	26 700 LE	in-I F	ristruc ¢	20 superior 20	\$19 224 000	75 vrs	\$12,000,000
Low-Density Urban Pipeline	24-in	19.500 LF	in-LF	Ś	40	\$18,720,000	75 yrs	\$11,000,000
High-Density Urban Pipeline	24-in	12,700 LF	in-LF	\$	50	\$15,240,000	75 yrs	\$9,100,000
Trenchless Crossings		,		·		, .,		,,
San Pablo Tunnel Rehab		17,600	LF	\$	3,500	\$62,000,000	75 yrs	\$37,000,000
HDD (All Inclusive)		2,200	LF	\$	2,200	\$4,800,000	75 yrs	\$2,900,000
Pump Stations	Cost Estimates	for Pump Stations	are based on EBML	JD Co	nstruction Bid	Costs		
		EA (Cost Curv	e based on total					
Pump Station (Total Installed HP, including standby)	1,500 HP	insta	lled HP)	\$	5,300,000	\$5,300,000	50 yrs	\$2,100,000
Storage Tanks	Cost Estimates	for Storage Tanks	are based on EBMU	JD Col	nstruction Bid	Costs		
Storage Tanks			EA (Cost Curve by					
(material determined during pre-design)	0.9 MG	3	Volume)	Ş	3,200,000	\$9,600,000	75 yrs	\$5,800,000
		Estimated S	ubtotal Cost (2) inc	luding	contingency	\$130,000,000		\$80,000,000
Estimated Subtotal Construction Cost including Cont	ingency					\$370,000,000		\$80,000,000
Mobilization					5%	\$19,000,000		\$4,000,000
Estimated Subtotal Construction Cost including Cont	ingency and Mo	bilization				\$390,000,000		\$84,000,000
Implementation					E 9/	¢20,000,000		\$4 200 000
Design Cost					15%	\$20,000,000		\$4,200,000
Project Administration and Construction Managemen	t Cost				10%	\$39.000.000		\$8,400,000
Estimated Total Capital Cost including Implementation	on and Continge	ency				\$510,000,000		\$110,000,000
						Present Worth of Sa	Ivage Value	\$45,000,000
O&M Costs (Annual)								
Advanced Water Treatment	10		MCD	ć	202.000	¢2,000,000		
RO System	10		MGD	ç ç	480.000	\$2,900,000		
Advanced Oxidation and Disinfection	10		MGD	Ş	49,000	\$490,000		
Free Chlorine	10		MGD	\$	32,000	\$320,000		
Chemicals	10		MGD	\$	120,000	\$1,200,000		
Labor	10	2,080	hrs/MGD	\$	135	\$2,800,000		
Electricity		19,230,000	kWh/yr	\$	0.15	\$2,900,000		
Monitoring			\$/year	Ş	500,000	\$500,000		
<u>Conveyance</u>								
Annual O&M			2% of construction	cost		\$2,400,000		
Pump Stations								
Labor Costs		500	hrs	\$	135	\$68,000		
Electricity		6,020,000	kWh/yr	\$	0.15	\$900,000		
Pump Station Consumables		5% of pump sta	tion construction co	ost		\$270,000		
Storage Tanks								
Annual O&M		1% of co	nstruction cost			\$96,000		
			Total O&M Costs (\$/yr)		\$20,000,000		
Annualized Costs (\$ / Year)		-						
Annualized Capital Costs (\$/Year)	Two payments	per year, spread c	ver Project Life			\$26,000,000		
Annualized Salvage Value	Annualized vali	ie of present wort	n			-\$2,300,000		
Total Annualized Cost						\$20,000,000 \$44 000 000		
Deliveries of Recycled Water	11.2	00 AFY				<u> </u>		
Estimated Unit Cost (\$/AF)	,					\$3,900		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$2,900		Distribution	\$1,000		
Annualized Costs (\$ / Year)	Dry Year Adius	tment (Sunnly us	ad 3/10 years)					
Annualized Capital Costs (\$/Year)	Two navments	per year. spread o	ver Proiect Life			\$26.000.000	Same as constant use	
Annualized Salvage Value	Annualized val	ie of present wort	h			-\$2,300.000	Same as constant use.	
Annual O&M Costs		.,,				\$15,570,000		
Total Annualized Cost						\$39,000,000		
Annual Average Deliveries of Recycled Water	3,3	50 AFY						
Estimated Unit Cost (\$/AF)						\$12,000		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$8,300		Distribution	\$3,700		

Alternative SD1-Best -30								d Water Master Plan Lindate
Last Lindated:	30-May-18					Discount Pate	EDIVIOD Recycles	iect Life
Lindated by:	J laffe				1		30	Vears
CCI (20 City, Dec 2017): 10870.06	insurie					575	50	
Item	Size	Qty	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value
Capital Costs								
Treatment	Cost Estimates	for Treatment are	based on Unit Co	sts for l	Each Treatment	Process		
MF/UF system	30		MGD	\$	1,225,000	\$37,000,000	30 yrs	\$0
RO System	30		MGD	\$	1,475,000	\$44,000,000	30 yrs	\$0
Advanced Oxidation and Disinfection	30		MGD	\$	437,500	\$13,000,000	30 yrs	\$0
Chemicals (Storage and Use)	30			\$	125,000	\$3,800,000	30 yrs	\$0
Sitework/Piping/Structures	30		MGD	\$	3,187,500	\$96,000,000	30 yrs	\$0
			Raw Consti	ruction	Cost Subtotal	\$190,000,000		\$0
Contractor Overnead & Profit	500/		A P b .		15%	\$29,000,000		\$0 ¢0
sales Tax	50%	% of Subtotal Cost	Applicable	tal Con	9%	\$8,600,000		ېں دە
			Estimated Subto		struction cost	\$230,000,000		ΟÇ
Construction Cost Including Contingency								
t			Estimated Subto	tal Con	struction Cost	\$230,000,000		\$0
Estimating Contingency					25%	\$58,000,000		\$0
		Estimated S	ubtotal Cost (1) ir	ncludin	g Contingency	\$290,000,000		\$0
Capital Costs (from Bids)								
Conveyance	Cost Estimates	for Pipelines are b	ased on EBMUD o	constru	ction bid cost + e	estimating contingency		
Non-urban Pipeline	42-in	17,800 LF	in-LF	\$	30	\$22,428,000	75 yrs	\$13,000,000
Low-Density Urban Pipeline	42-in	26,800 LF	in-LF	\$	40	\$45,024,000	75 yrs	\$27,000,000
High-Density Urban Pipeline	42-in	3,800 LF	in-LF	\$	50	\$7,980,000	75 yrs	\$4,800,000
Trenchless Crossings		_						
Microtunnel Xings (1 jack & 1 receiv. pit)		/	EA	Ş	620,000	\$4,300,000	75	640 000 000
Microtunnel Pipe		10,550	LF	Ş	2,800	\$30,000,000	75 yrs	\$18,000,000
Pump Stations	Cost Estimates	for Pump Stations	are based on EBN	ИUD Ca	onstruction Bid C	Costs		
Pump Station (Total installed HP including standby)		EA (Cost Curv	e based on total					
	4,200 HP	insta	lled HP)	\$	22,000,000	\$22,000,000	50 yrs	\$8,800,000
		Estimated S	ubtotal Cost (2) ir	ncludin	g Contingency	\$130,000,000		\$72,000,000
Estimated Subtotal Construction Cost including Cont	ingency				_	\$420,000,000		\$72,000,000
Mobilization					5%	\$21,000,000		\$3,600,000
Estimated Subtotal Construction Cost including Cont	ingency and Mo	bilization				\$440,000,000		\$76,000,000
Implementation								
Environmental Documentation & Permits					5%	\$22,000,000		\$3,800,000
Design Cost					15%	\$66,000,000		\$11,000,000
Project Administration and Construction Managemen	t Cost				10%	\$44,000,000		\$7,600,000
Estimated Total Capital Cost Including Implementation	on and continge	incy				\$570,000,000 Present Worth of Sa	alvage Value	\$98,000,000
O&M Costs (Annual)							intuge tunue	<i> </i>
Advanced Water Treatment								
MF/UF system	30		MGD	\$	325,000	\$9,800,000		
RO System	30		MGD	\$	480,000	\$14,000,000		
Advanced Oxidation and Disinfection	30		MGD	\$	49,000	\$1,500,000		
Free Chlorine	30		MGD	Ş	32,000	\$960,000		
Chemicals	30	2 000	MGD	\$	120,000	\$3,600,000		
Labor	30	2,080	nrs/IviGD	Ş	135	\$8,400,000		
Electricity		27,150,000	KVV11/yr	Ş	0.15	\$4,100,000		
Conveyance								
Annual O&M			2% of construction	on cost		\$2,200,000		
Pump Stations								
Labor Costs		1000	hrs	\$	135	\$140,000		
Electricity		21,500,000	kWh/yr	\$	0.15	\$3,200,000		
Pump Station Consumables		5% of pump sta	tion construction	cost		\$1,100,000		
			Total O&M Costs	5 (\$/yr)		\$49,000,000		
Annualized Costs (\$ / Year)			0 1 1 1 1			¢20,000,000		
Annualized Capital Costs (\$/Year)	I wo payments	per year, spreaa c	over Project Life			\$29,000,000		
Annualized Salvage Value	Annualized vali	ie of present wort	'n			-\$2,000,000		
						\$49,000,000		
Deliveries of Recycled Water	33.6	00 AFY			-	\$70,000,000		
Estimated Unit Cost (\$/AF)						\$2.300		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$1,800		Distribution	\$500		
	Due Ve en Adier							
Annualized Costs (\$ / Year)	Two normants	ner year coroca	ver Project Life			¢20.000.000	Same as consta	at use
Annualized Salvage Value	Annualized val	ie of present wort	h			-\$2 000,000	Same as constar	nt use.
Annual Q&M Costs	, unidunzeu vun	e oj present wort				\$40.698.000	- and us constan	
Total Annualized Cost						\$68.000.000		
Annual Average Deliveries of Recycled Water	10,0	80 AFY			-	,		
Estimated Unit Cost (\$/AF)						\$6,700		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$5,600		Distribution	\$1,100		

Alternative SD1-ResB-30							EBMUD Recycled	Water Master Plan Update
Last Updated:	30-May-18					Discount Rate	Proj	ect Life
Updated by:	I. Jaffe					3%	30 \	'ears
CCI (20 City, Dec 2017): 10870.06								
Item	Size	Qty	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value
Capital Costs	Cost Estimatos	for Tractmont are	hacad on Unit Co	etc for	Each Troatmar	at Process		
ME/LIE system	30	jor rreatment are	MGD	sis jui ¢	1 225 000	\$37 000 000	30 yrs	ŚO
BO System	30		MGD	Ś	1 475 000	\$44,000,000	30 yrs	\$0
Advanced Oxidation and Disinfection	30		MGD	Ś	437.500	\$13,000,000	30 yrs	\$0
Chemicals (Storage and Use)	30			Ś	125.000	\$3.800.000	30 yrs	\$0
Sitework/Piping/Structures	30		MGD	\$	3,187,500	\$96,000,000	30 yrs	\$0
			Raw Const	ruction	Cost Subtotal	\$190,000,000		\$0
Contractor Overhead & Profit					15%	\$29,000,000		\$0
Sales Tax	50%	% of Subtotal Cost	Applicable		9%	\$8,600,000		\$0
			Estimated Subto	tal Con	struction Cost	\$230,000,000		\$0
Construction Cost Including Contingency			Estimated Subto	tal Con	struction Cost	\$230,000,000		ŚO
Estimating Contingency			Estimated Subto		25%	\$58.000.000		\$0 \$0
		Estimated S	ubtotal Cost (1) ir	ncludin	g Contingency	\$290,000,000		\$0
Capital Costs (from Bids)								
Conveyance	Cost Estimates	for Pipelines are b	ased on EBMUD o	constru	ction bid cost +	estimating contingency		
Non-urban Pipeline	42-in	36,400 LF	in-LF	\$	30	\$45,864,000	75 yrs	\$28,000,000
Low-Density Urban Pipeline	42-in	19,600 LF	in-LF	\$	40	\$32,928,000	75 yrs	\$20,000,000
High-Density Urban Pipeline	42-in	11,800 LF	in-LF	\$	50	\$24,780,000	75 yrs	\$15,000,000
Trenchless Crossings								
San Pablo Tunnel Rehab		17,600	LF	\$	3,500	\$62,000,000	75 yrs	\$37,000,000
Microtunnel Xings (1 jack & 1 receiv. pit)		4	EA	\$	620,000	\$2,500,000		
Microtunnel Pipe		2,400	LF	\$	2,800	\$6,700,000	75 yrs	\$4,000,000
Pump Stations	Cost Estimates	for Pump Stations	are based on EBN	NUD Ca	onstruction Bid	Costs		
Pump Station (Total installed HP, including standby)		EA (Cost Curv	e based on total					
rump station (rotal installed fir, including statuby)	5,400 HP	insta	lled HP)	\$	33,000,000	\$33,000,000	50 yrs	\$13,000,000
		Estimated S	ubtotal Cost (2) ir	ncludin	g Contingency	\$210,000,000		\$120,000,000
Estimated Subtotal Construction Cost including Cont	ingency		.,			\$500,000,000		\$120,000,000
Mobilization					5%	\$25,000,000		\$6,000,000
Estimated Subtotal Construction Cost including Cont	ingency and Mo	bilization				\$530,000,000		\$130,000,000
Implementation								
Environmental Documentation & Permits					5%	\$27,000,000		\$6,500,000
Design Cost					15%	\$80,000,000		\$20,000,000
Project Administration and Construction Managemen	t Cost				10%	\$53,000,000		\$13,000,000
Estimated Total Capital Cost including Implementation	on and Continge	ency				\$690,000,000		\$170,000,000
OSM Costs (Annual)						Present Worth of S	alvage Value	\$70,000,000
Advanced Water Treatment								
ME/UE system	30		MGD	Ś	325.000	\$9.800.000		
RO System	30		MGD	\$	480,000	\$14,000,000		
Advanced Oxidation and Disinfection	30		MGD	\$	49,000	\$1,500,000		
Free Chlorine	30		MGD	\$	32,000	\$960,000		
Chemicals	30		MGD	\$	120,000	\$3,600,000		
Labor	30	2,080	hrs/MGD	\$	135	\$8,400,000		
Electricity		27,150,000	kWh/yr	\$	0.15	\$4,100,000		
Surface Water Treatment								
SWT O&M (Orinda)	10,950		MG	\$	70	\$770,000		
Conveyance								
Annual O&M			2% of construction	on cost		\$3,500,000		
Pump Stations								
Labor Costs		1000	hrs	\$	135	\$140,000		
Electricity		27,090,000	kWh/yr	\$	0.15	\$4,100,000		
Pump Station Consumables		5% of pump sta	tion construction	cost		\$1,700,000		
			Total O&M Costs	s (\$/yr)		\$53,000,000		
Annualized Costs (\$ / Year)								
Annualized Capital Costs (\$/Year)	Two payments	per year, spread o	over Project Life			\$35,000,000		
Annualized Salvage Value	Annualized val	ue of present wort	h			-\$3,600,000		
Annual O&M Costs						\$53,000,000		
Total Annualized Cost	22.0	00 4 51				\$84,000,000	-	
Deliveries of Recycled Water	33,0	OU AFY				¢3 500		
Estimated Unit Cost (\$/AF) Proakdown of Estimated Unit Cost (\$ (AE)		Trootmont	¢1 000		Distribution	\$2,500		
Breakdown of Estimated Onit Cost (\$/AF)		Treatment	\$1,900		Distribution	3000		
Annualized Costs (\$ / Year)	Dry Year Adjus	tment (Supply us	ed 3/10 years)			12	6	
Annualized Capital Costs (\$/Year)	Two payments	per year, spread o	over Project Life			\$35,000,000	Same as constan	t use.
Annualized Salvage Value	Annualized val	ue of present wort	'n			-\$3,600,000	same as constan	t use.
Annual U&M Costs						\$43,099,000		
Annual Average Deliveries of Posyclod Water	10.0					\$74,000,000	-	
Estimated Unit Cost (\$/AE)	10,0					\$7 200		
Breakdown of Estimated Unit Cost (\$ /AE)		Treatment	\$5 700		Distribution	\$1,500		
Dieakuowii of Estimateu Offit Cost (\$/AF)		neatment	4J,700		Distribution	\$1,000		

Alternative SD1-ResSP-4							EBMUD Recycle	d Water Master Plan Update
Last Lindated:	30-May-18					Discount Bate	Pro	niect Life
Updated by:	I. Jaffe					3%	30	Years
CCI (20 City, Dec 2017): 10870.06								
Item	Size	Qty	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value
Capital Costs								
Treatment	Cost Estimates	for Treatment are	e based on Unit Co	sts for	Each Treatme	nt Process		
MBR	4		MGD	\$	16,600,000	\$66,000,000	30 yrs	\$0
RO System	4		MGD	\$	1,475,000	\$5,900,000	30 yrs	\$0
Advanced Oxidation and Disinfection	4		MGD	\$	437,500	\$1,800,000	30 yrs	\$0
Chemicals (Storage and Use)	4			\$	125,000	\$500,000	30 yrs	\$0
Sitework/Piping/Structures	4		MGD	\$	3,187,500	\$13,000,000	30 yrs	\$0
			Raw Const	ruction	Cost Subtota	l \$87,000,000		\$0
Contractor Overhead & Profit					15%	\$13,000,000		\$0
Sales Tax	50%	% of Subtotal Cost	Applicable		9%	\$3,900,000		\$0
			Estimated Subto	tal Con	struction Cos	t \$100,000,000		\$0
Construction Cost Including Contingency			Estimated Subta	tal Car	struction Cos	¢100.000.000		ćo
Estimating Contingency			Estimated Subto	tai con	25%	\$25,000,000		30 \$0
Estimating contingency		Estimated S	ubtotal Cost (1) in		a Contingenci	\$23,000,000		30 \$0
Capital Costs (from Bids)		Estimateu s		Iciuum	g contingenc	y \$130,000,000		ŞU
Conveyance	Cost Estimates	for Pinelines are l	ased on FRMUD (onstru	ction hid cost	+ estimating contingency		
Non-urban Pipeline	16-in	5.300 LF	in-LF	Ś	30	\$2,544.000	75 vrs	\$1,500.000
Low-Density Urban Pipeline	16-in	18.300 LF	in-LF	Ś	40	\$11,712,000	75 vrs	\$7.000.000
High-Density Urban Pipeline	16-in	11.800 LF	in-LF	Ś	50	\$9,440,000	75 vrs	\$5,700,000
Trenchless Crossings		,		Ŧ	50	<i>41,13,000</i>	- ,	<i>+-,--)000</i>
San Pablo Tunnel Rehab		17,600	LF	\$	3,500	\$62,000,000	75 yrs	\$37,000,000
HDD (All Inclusive)		2,400	LF	\$	2,200	\$5,300,000	75 yrs	\$3,200,000
Duma Stations	Cost Estimator	for Dump Station	are based on FRA		natruction Di	d Costs		
Pump Stations	Cost Estimates	FA (Cost Cur	s are based on EBN	NUDC	onstruction Bil	Costs		
Pump Station (Total installed HP, including standby)	600 HB	EA (COST CUIT		ć	2 400 000	¢2,400,000	FOurc	\$060.000
	000 HF	IIIsta	lileu HF)	Ş	2,400,000	\$2,400,000	50 yrs	\$900,000
		Estimated S	ubtotal Cost (2) ir	ncludin	g Contingenc	\$93,000,000		\$55,000,000
Estimated Subtotal Construction Cost including Cont	ingency					\$220,000,000		\$55,000,000
Mobilization					5%	\$11,000,000		\$2,800,000
Estimated Subtotal Construction Cost including Cont	ingency and Mo	obilization				\$230,000,000		\$58,000,000
Implementation					50/	¢12.000.000		ća 000 000
Environmental Documentation & Permits					5%	\$12,000,000		\$2,900,000
Design Cost					15%	\$35,000,000		\$8,700,000
Project Administration and Construction Managemen	n cost				10%	\$23,000,000		\$5,800,000
Estimated Total Capital Cost including implementati	on and conting	ency				Present Worth of Sa	alvage Value	\$31,000,000
O&M Costs (Annual)								,
Advanced Water Treatment								
MBR	4		MGD	\$	402,000	\$1,600,000		
RO System	4		MGD	\$	480,000	\$1,900,000		
Advanced Oxidation and Disinfection	4		MGD	\$	49,000	\$200,000		
Free Chlorine	4		MGD	Ş	32,000	\$130,000		
Chemicals	4	2,000	MGD	\$	120,000	\$480,000		
	4	2,080	nrs/WGD	Ş	135	\$1,100,000		
Electricity		7,692,000	KVVN/yr	Ş	0.15	\$1,200,000		
SW/T O.8.M (Sobranto W/TR)	1 460		MG	ć	254	\$270.000		
Swi odwi (Sobiance wiri)	1,400		WIG	Ļ	234	\$370,000		
Conveyance								
Annual O&M			2% of construction	on cost		\$1,820,000		
Pump Stations								
Labor Costs		500	hrs	\$	135	\$68,000		
Electricity		1,949,000	kWh/yr	\$	0.15	\$290,000		
Pump Station Consumables		5% of pump sta	tion construction	cost		\$120,000		
			Total O&M Costs	s (\$/yr)		\$9,300,000		
Annualized Costs (\$ / Year)								
Annualized Capital Costs (\$/Year)	Two payments	per year, spread o	over Project Life			\$15,000,000		
Annualized Salvage Value	Annualized val	ue of present worl	th			-\$1,600,000		
Annual O&M Costs						\$9,300,000		
Total Annualized Cost						\$23,000,000		
Deliveries of Recycled Water	4,4	80 AFY				45 400		
Estimated Unit Cost (\$/AF)			<u> </u>		N N N	\$5,100		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$3,600		Distribution	n \$1,500		
Annualized Costs (\$ / Year)	Dry Year Adju	stment (Supply us	ed 3/10 years)					
Annualized Capital Costs (\$/Year)	Two payments	per year, spread o	over Project Life			\$15,000,000	Same as consta	nt use.
Annualized Salvage Value	Annualized val	ue of present worl	th			-\$1,600,000	Same as consta	nt use.
Annual O&M Costs						\$7,549,000		
Total Annualized Cost						\$21,000,000		
Annual Average Deliveries of Recycled Water	1,3	44 AFY		_				
Estimated Unit Cost (\$/AF)						\$16,000		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$11,000		Distribution	n \$5,000		

Alternative SD1 Result 10								Water Master Plan Lindate
Alternative SD1-Reso-10	20.14 10						ЕБІЙОД Кесусіес	water Master Plan Opdate
Last Updated:	30-IVIay-18				l	Discount Rate	Proj	ect Life
Updated by:	I. Jarre				-	3%	30 1	ears
Item	Sizo	Otv	Unit		Unit Cost	Total Cost	Licoful Life	Salvago Valuo
Capital Costs	5120	Qty	Onit		Onit Cost	Total Cost	Oserur Ene	Jaivage Value
Treatment	Cost Estimates	for Treatment are	hased on Unit Co	sts for	Each Treatment	Process		
MBR	10	joi meatment are	MGD	\$ \$	11 100 000	\$110,000,000	30 yrs	\$0
BO System	10		MGD	Ś	1 475 000	\$15,000,000	30 yrs	\$0
Advanced Oxidation and Disinfection	10		MGD	Ś	437 500	\$4 400 000	30 yrs	\$0
Chemicals (Storage and Lise)	10		WIGD	ç	125 000	\$1,300,000	30 yrs	\$0 \$0
Sitework/Pining/Structures	10		MGD	ś	3 187 500	\$32,000,000	30 yrs	\$0 \$0
	10		Raw Const	ruction	Cost Subtotal	\$160,000,000	50 115	¢0
Contractor Overhead & Profit					15%	\$24,000,000		\$0
Sales Tax	50%	% of Subtotal Cost	Annlicable		9%	\$7 200 000		\$0
	5070		Estimated Subto	tal Con	struction Cost	\$190.000.000		\$0 \$0
						+,,		
Construction Cost Including Contingency								
			Estimated Subto	tal Con	struction Cost	\$190,000,000		\$0
Estimating Contingency					25%	\$48,000,000		\$0
		Estimated S	ubtotal Cost (1) ir	ncludin	g Contingency	\$240,000,000		\$0
Capital Costs (from Bids)								
Conveyance	Cost Estimates	for Pipelines are b	based on EBMUD o	constru	ction bid cost + e	estimating contingency		
Non-urban Pipeline	24-in	28,800 LF	in-LF	\$	30	\$20,736,000	75 yrs	\$12,000,000
Low-Density Urban Pipeline	24-in	26,800 LF	in-LF	\$	40	\$25,728,000	75 yrs	\$15,000,000
High-Density Urban Pipeline	24-in	3,800 LF	in-LF	\$	50	\$4,560,000	75 yrs	\$2,700,000
Trenchless Crossings								
HDD (All Inclusive)		3,550	LF	\$	2,200	\$7,800,000	75 yrs	\$4,700,000
Dump Stations	Cost Estimatos	for Dump Station	are based on FRA		netruction Rid (acts.		
Pump stations	Cost Estimates	FA (Cost Curr	s ure bused on total	NUDCL		.0515		
Pump Station (Total installed HP, including standby)	1 200 HD	EA (COST CUN		ć	6 500 000	\$6 E00 000	FOver	\$2,600,000
	1,000 HP	IIIsta	lileu HF)	ç	0,300,000	\$0,500,000	30 yrs	\$2,000,000
		Estimated S	ubtotal Cost (2) ir	ncludin	g Contingency	\$65,000,000		\$37,000,000
Estimated Subtotal Construction Cost including Cont	tingency					\$310,000,000		\$37,000,000
Mobilization					5%	\$16,000,000		\$1,900,000
Estimated Subtotal Construction Cost including Cont	ingency and Mo	bilization				\$330,000,000		\$39,000,000
Implementation								
Environmental Documentation & Permits					5%	\$17,000,000		\$2,000,000
Design Cost					15%	\$50,000,000		\$5,900,000
Project Administration and Construction Managemer	nt Cost				10%	\$33,000,000		\$3,900,000
Estimated Total Capital Cost including Implementati	on and Continge	ency				\$430,000,000		\$51,000,000
						Present worth of Sa	livage value	\$21,000,000
O&IVI Costs (Annual)								
MBB	10		MGD	Ś	292 000	\$2,900,000		
BO System	10		MGD	Ś	480.000	\$4,800,000		
Advanced Oxidation and Disinfection	10		MGD	Ś	49,000	\$490.000		
Free Chlorine	10		MGD	\$	32,000	\$320,000		
Chemicals	10		MGD	\$	120,000	\$1,200,000		
Labor	10	2,080	hrs/MGD	\$	135	\$2,800,000		
Electricity		19,230,000	kWh/yr	\$	0.15	\$2,900,000		
Conveyance								
Annual Q&M			2% of construction	on cost		\$1 180 000		
			2,0 01 0010010000			<i>φ</i> 2)200)000		
Pump Stations		500			405			
Labor Costs		500	nrs	Ş	135	\$68,000		
Electricity		7,310,000	KVVN/yr	\$ 	0.15	\$1,100,000		
Pump Station Consumables		5% of pump sta	Total OS M Cost	cost		\$330,000		
Annualized Cente (\$ / Year)			Total Oxivi Costs	s (\$/ yr)		\$18,000,000		
Annualized Costs (\$ / Tear)		par year caread	war Project Life			\$22,000,000		
Annualized Capital Costs (37 real)	Appuglized val	per yeur, spreuu c	h			-\$1 100 000		
Annual O&M Costs	Annuunzeu van	ae oj present won				-31,100,000 ¢18 000 000		
Total Annualized Cost						\$10,000,000		
Deliveries of Recycled Water	11 7	00 AFY			-	\$35,000,000		
Estimated Unit Cost (\$/AF)	11,2					\$3 500		
Breakdown of Estimated Unit Cost (\$/AE)		Treatment	\$2,800		Distribution	\$700		
breakdown of Estimated Onit Cost (3/Ar)		Treatment	\$2,000		Distribution	Ş700		
Annualized Costs (\$ / Year)	Dry Year Adjus	tment (Supply us	ed 3/10 years)				<u> </u>	
Annualized Capital Costs (\$/Year)	I wo payments	per year, spread o	over Project Life			\$22,000,000	same as constan	t use.
Annualized Salvage Value	Annualized val	ue of present worl	n			-\$1,100,000	same as constan	t use.
Annual U&M Costs						\$14,224,000		
					-	\$35,000,000		
Entimated Unit Cost (\$ (AE)	3,3	JU AFT				¢10.000		
Estimated Unit Cost (\$/AF)		Turatura	<u>ća 200</u>		Distall	\$10,000		
Breakdown of Estimated Unit Cost (\$/AF)		reatment	\$8,300		Distribution	\$1,700		

Alternative SD1-ResB-10							EBMUD Recycle	d Water Master Plan Update	
Last Lindated:	30-May-18					Discount Bate	Pro	Project Life	
Updated by:	L laffe					3%	30 Years		
CCI (20 City, Dec 2017): 10870.06									
Item	Size	Qty	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value	
Capital Costs									
Treatment	Cost Estimates	for Treatment are	based on Unit Co	sts for	Each Treatme	nt Process			
MBR	10		MGD	\$	11,100,000	\$110,000,000	30 yrs	\$0	
RO System	10		MGD	\$	1,475,000	\$15,000,000	30 yrs	\$0	
Advanced Oxidation and Disinfection	10		MGD	\$	437,500	\$4,400,000	30 yrs	\$0	
Chemicals (Storage and Use)	10			\$	125,000	\$1,300,000	30 yrs	\$0	
Sitework/Piping/Structures	10		MGD	\$	3,187,500	\$32,000,000	30 yrs	\$0	
			Raw Constr	ruction	Cost Subtota	l \$160,000,000		\$0	
Contractor Overhead & Profit					15%	\$24,000,000		\$0	
Sales Tax	50%	% of Subtotal Cost	Applicable		9%	\$7,200,000		\$0	
			Estimated Subton	tal Con	struction Cos	t \$190,000,000		\$0	
Construction Cost Including Contingency			Estimated Subtat		struction Cos	¢100.000.000		ćo	
Estimating Contingency			Estimated Subton	Lai Cor	25%	\$190,000,000		30 \$0	
Estimating contingency		Ectimated S	ubtotal Cost (1) in		2.3%	\$48,000,000		30 \$0	
Capital Costs (from Bids)		Estimateu 5	ubtotal Cost (1) II	iciuum	g contingency	, şz40,000,000		ŞU	
Conveyance	Cost Estimates	for Pinelines are h	ased on FRMUD c	onstru	ction hid cost	+ estimatina contingency			
Non-urban Pipeline	24-in	36.400 LF	in-LF	Ś	30	\$26.208.000	75 vrs	\$16.000.000	
Low-Density Urban Pipeline	24-in	19.600 LF	in-LF	Ś	40	\$18.816.000	75 vrs	\$11.000.000	
High-Density Urban Pipeline	24-in	11.800 LF	in-LF	Ś	50	\$14,160,000	75 vrs	\$8,500,000	
Trenchless Crossings		,				, , , , , , , , , , , , , , , , , , , ,	- / -	, ,	
San Pablo Tunnel Rehab		17,600	LF	\$	3,500	\$62,000,000	75 yrs	\$37,000,000	
HDD (All Inclusive)		2,400	LF	\$	2,200	\$5,300,000	75 yrs	\$3,200,000	
Duma Stations	Cost Estimatos	for Dump Station	are baced on FRA	AUD C	netruction Di	d Casta	-		
Pump Stations	Cost Estimates	Jor Pump Stations	are based on EBN	NUD C	DISTRUCTION BIO	COSTS			
Pump Station (Total installed HP, including standby)	2 000 HP	EA (COSt Curv		ć	7 500 000	\$7 500 000	50 yrs	\$3,000,000	
	2,000 HF	IIIsta	lieu HF)	ç	7,300,000	\$7,500,000	30 yrs	\$5,000,000	
		Estimated S	ubtotal Cost (2) in	ncludin	g Contingency	\$130,000,000		\$79,000,000	
Estimated Subtotal Construction Cost including Cont	ingency					\$370,000,000		\$79,000,000	
Mobilization					5%	\$19,000,000		\$4,000,000	
Estimated Subtotal Construction Cost including Cont	ingency and Mo	bilization				\$390,000,000		\$83,000,000	
Implementation					50/	¢20.000.000		ć 4 200 000	
Environmental Documentation & Permits					5%	\$20,000,000		\$4,200,000	
Design Cost Design Cost	t Cost				15%	\$59,000,000		\$12,000,000	
Project Administration and Construction Managemen	n cost				10%	\$39,000,000		\$8,300,000	
Estimated Total Capital Cost including implementati	on and continge	ency				Present Worth of Sa	alvage Value	\$45,000,000	
O&M Costs (Annual)								,,.	
Advanced Water Treatment									
MBR	10		MGD	\$	292,000	\$2,900,000			
RO System	10		MGD	\$	480,000	\$4,800,000			
Advanced Oxidation and Disinfection	10		MGD	\$	49,000	\$490,000			
Free Chlorine	10		MGD	Ş	32,000	\$320,000			
Chemicals	10	2 000	MGD	\$	120,000	\$1,200,000			
	10	2,080	nrs/IviGD	Ş	135	\$2,800,000			
Electricity		19,230,000	kvvn/yr	Ş	0.15	\$2,900,000			
SWT O&M (Orinda)	3 650		MG	ć	70	\$260,000			
Swi Odivi (Offica)	3,050		NIG	Ļ	70	\$200,000			
Conveyance									
Annual O&M			2% of constructio	on cost		\$2,500,000			
Pump Stations									
Labor Costs		500	hrs	\$	135	\$68,000			
Electricity		9,030,000	kWh/yr	\$	0.15	\$1,400,000			
Pump Station Consumables		5% of pump sta	tion construction	cost		\$380,000			
			Total O&M Costs	; (\$/yr)		\$20,000,000			
Annualized Costs (\$ / Year)									
Annualized Capital Costs (\$/Year)	Two payments	per year, spread o	over Project Life			\$26,000,000			
Annualized Salvage Value	Annualized val	ue of present wort	'n			-\$2,300,000			
Annual O&M Costs						\$20,000,000			
Total Annualized Cost						\$44,000,000			
Deliveries of Recycled Water	11,2	200 AFY				40.000			
Estimated Unit Cost (\$/AF)			<u> </u>		D ¹ · · · · · ·	\$3,900			
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$2,900		Distribution	n \$1,000			
Annualized Costs (\$ / Year)	Dry Year Adjus	stment (Supply us	ed 3/10 years)						
Annualized Capital Costs (\$/Year)	Two payments	per year, spread o	over Project Life			\$26,000,000	Same as consta	nt use.	
Annualized Salvage Value	Annualized val	ue of present wort	h			-\$2,300,000	Same as consta	nt use.	
Annual O&M Costs						\$15,762,000			
Total Annualized Cost						\$39,000,000			
Annual Average Deliveries of Recycled Water	3,3	60 AFY		_					
Estimated Unit Cost (\$/AF)						\$12,000			
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$8,300		Distribution	n \$3,700			

Alternative SD1-Treat-30							EBMUD Recycled \	Nater Master Plan Update
Last Updated:	30-May-18					Discount Bate	Proie	ect Life
Updated by:	I. Jaffe					3%	30 Ye	ears
CCI (20 City, Dec 2017): 10870.06								
Item	Size	Qty	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value
Capital Costs								
<u>Treatment</u>	Cost Estimates	for Treatment ar	e based on Unit Co	sts for l	Each Treatmer	nt Process		
Ozone	30		MGD	\$	335,300	\$10,000,000	30 yrs	\$0
BAC	30		MGD	\$	300,900	\$9,000,000	30 yrs	\$0
MF/UF system	30		MGD	\$	1,225,000	\$37,000,000	30 yrs	\$0
RO System	30		MGD	Ş	1,475,000	\$44,000,000	30 yrs	\$0
Advanced Oxidation and Disinfection	30		MGD	Ş	437,500	\$13,000,000	30 yrs	\$0 ¢0
Chemicals (Storage and Use)	30		MCD	Ş	125,000	\$3,800,000	30 yrs	\$U ¢0
Sitework/Piping/Structures	30		Paw Constru		3,187,500	\$96,000,000	30 yrs	50 \$0
Contractor Overhead & Profit			Raw Constru	uction	15%	\$210,000,000		90 ¢0
	50%	% of Subtotal Cost	Applicable		9%	\$52,000,000		50 \$0
	50%	/0 01 50510101 C031	Estimated Subtot	al Cons	truction Cost	\$250,000,000		\$0 \$0
			Lotimated oubtot			\$250,000,000		ý.
Construction Cost Including Contingency								
			Estimated Subtot	al Cons	truction Cost	\$250,000,000		\$0
Estimating Contingency					25%	\$63,000,000		\$0
		Estimated S	ubtotal Cost (1) in	cluding	Contingency	\$310,000,000		\$0
Capital Costs (from Bids)								
Conveyance	Cost Estimates	for Pipelines are	based on EBMUD c	construc	tion bid cost +	estimating contingency		
Non-urban Pipeline	42-in	2,600 LF	in-LF	\$	30	\$3,276,000	75 yrs	\$2,000,000
Low-Density Urban Pipeline	42-in	1,200 LF	in-LF	\$	40	\$2,016,000	75 yrs	\$1,200,000
High-Density Urban Pipeline	42-in	2,500 LF	in-LF	\$	50	\$5,250,000	75 yrs	\$3,200,000
Trenchless Crossings								
Microtunnel Xings (1 jack & 1 receiv. pit)		3	EA	\$	620,000	\$1,900,000	75	40 500
Microtunnel Pipe		1,500	LF	Ş	2,800	\$4,200,000	75 yrs	\$2,500,000
Pump Stations	Cost Estimates	for Pump Station	s are based on EBN	MUD Ca	nstruction Bid	Costs		
Pump Station (Total installed HP including standby)		EA (Cost Curv	ve based on total					
rump station (rotal installed fir, including statuby)	3,000 HP	insta	alled HP)	\$	13,000,000	\$13,000,000	50 yrs	\$5,200,000
Storage Tanks	Cost Estimates	for Storage Tank	s are based on EBN	ЛUD Co	nstruction Bid	Costs		
Storage Tanks			EA (Cost Curve b	iy .				
(material determined during pre-design)	2.5 MG	3	Volume)	\$	4,500,000	\$14,000,000	75 yrs	\$8,400,000
		Estimated S	ubtotal Cost (2) in	cluding	Contingency	\$44,000,000		\$23,000,000
Estimated Subtotal Construction Cost including Con	tingency					\$350,000,000		\$23,000,000
Mobilization					5%	\$18,000,000		\$1,200,000
Estimated Subtotal Construction Cost including Con	tingency and M	obilization				\$370,000,000		\$24,000,000
Implementation								
Environmental Documentation & Permits					5%	\$19,000,000		\$1,200,000
Design Cost					15%	\$56,000,000		\$3,600,000
Project Administration and Construction Managemen	nt Cost				10%	\$37,000,000		\$2,400,000
Estimated Total Capital Cost including implementation	on and Conting	ency				\$480,000,000 Present Worth of S	alvago Valuo	\$31,000,000
O&M Costs (Annual)						Tresent Worth of S	alvage value	\$15,000,000
Advanced Water Treatment								
Ozone	30		MGD	\$	57,000	\$1,700,000		
BAC	30		MGD	\$	116,000	\$3,500,000		
MF/UF system	30		MGD	\$	325,000	\$9,800,000		
RO System	30		MGD	\$	480,000	\$14,000,000		
Advanced Oxidation and Disinfection	30		MGD	\$	49,000	\$1,500,000		
Free Chlorine	30		MGD	\$	32,000	\$960,000		
Chemicals	30		MGD	Ş	120,000	\$3,600,000		
Labor	30	2,080	hrs/MGD	Ş	135	\$8,400,000		
Electricity		37,290,000	kWh/yr	Ş	0.15	\$5,600,000		
Monitoring			\$/year	Ş	1,000,000	\$1,000,000		
SWT ORM (Origon)	10.050		MG	ć	70	¢770.000		
SWI O&W (Offica)	10,950		NIG	Ş	70	\$770,000		
Conveyance								
Annual O&M			2% of constructio	on cost		\$330,000		
Pump Stations								
Labor Costs		1000	hrs	\$	135	\$140,000		
Electricity		14,620,000	kWh/yr	\$	0.15	\$2,200,000		
Pump Station Consumables		5% of pump sta	tion construction	cost		\$650,000		
Storage Tanks								
Annual O&M		1% of co	instruction cost			\$140,000		
			Total O&M Costs	; (\$/yr)		\$54,000,000		
Annualized Costs (\$ / Year)								
Annualized Capital Costs (\$/Year)	Two payments	per year, spread	over Project Life			\$24,000,000		
Annualized Salvage Value	Annualized val	ue of present wor	th			-\$660,000		
Annual O&M Costs						\$54,000,000		
Total Annualized Cost						\$77,000,000	-	
Deliveries of Recycled Water	33,6	00 AFY						
Estimated Unit Cost (\$/AF)						\$2,300		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$2,100		Distribution	\$200		
Annualized Costs (\$ / Year)	Dry Year Adjus	tment (Supply us	ed 3/10 years)					
Annualized Capital Costs (\$/Year)	Two payments	per year, spread	over Project Life			\$24,000,000	Same as constant	use.
Annualized Salvage Value	Annualized val	ue of present wor	th			-\$660,000	Same as constant	use.
Annual O&M Costs						\$44,399,000		
Total Annualized Cost						\$68,000,000		
Annual Average Deliveries of Recycled Water	10,0	80 AFY						
Estimated Unit Cost (\$/AF)						\$6,700		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$6,300		Distribution	\$400		

Alternative SD1-Treat-10							EBMUD Recycled Water	Master Plan Update
Last Updated:	30-May-18					Discount Rate	Project Life	2
Updated by:	I. Jaffe					3%	30 Years	
CCI (20 City, Dec 2017): 10870.06	C'	0	11		11-12-01-12	Tables	11-st-11/t- c	
Item Capital Costs	Size	Qty	Unit		Unit Cost	Total Cost	Userul Lite Si	alvage value
Treatment	Cost Estimates	for Treatment an	e based on Unit Cos	ts for	Each Treatmen	t Process		
MBR	10		MGD	\$	11,100,000	\$110,000,000	30 yrs	\$0
Ozone	10		MGD	\$	335,300	\$3,400,000	30 yrs	\$0
BAC BO System	10		MGD	ş	300,900	\$3,000,000	30 yrs	\$0 \$0
Advanced Oxidation and Disinfection	10		MGD	Ś	437.500	\$1,000,000	30 yrs	\$0 \$0
Chemicals (Storage and Use)	10			\$	125,000	\$1,300,000	30 yrs	\$0
Sitework/Piping/Structures	10		MGD	\$	3,187,500	\$32,000,000	30 yrs	\$0
Contractor Quark and & Drafit			Raw Constru	iction	Cost Subtotal	\$170,000,000		\$0 ¢0
Sales Tax	50%	% of Subtotal Cost	Applicable		9%	\$26,000,000 \$7,700,000		\$0 \$0
	50/0		Estimated Subtota	al Con	struction Cost	\$200,000,000		\$0
Construction Cost Including Contingency			Faller and California	1.6		ć200.000.000		ća
Estimating Contingency			Estimated Subtota	ii Con	25%	\$200,000,000		\$0 \$0
Estimating contingency		Estimated S	ubtotal Cost (1) inc	luding	g Contingency	\$250,000,000		\$0
Capital Costs (from Bids)								
Conveyance	Cost Estimates	for Pipelines are	based on EBMUD c	onstru	ction bid cost +	estimating contingency		
Non-urban Pipeline	24-in 24-in	2,600 LF	in-LF	Ş	30	\$1,872,000	75 yrs	\$1,100,000
High-Density Urban Pipeline	24-in 24-in	2.500 LF	in-LF	ŝ	40 50	\$3,000,000	75 yrs	\$1,800,000
Trenchless Crossings		_,		+		+-//		+_,,
HDD (All Inclusive)		1,500	LF	\$	2,200	\$3,300,000	75 yrs	\$2,000,000
Pump Stations	Cost Estimates	for Pump Station	s are based on EBN	1UD C	onstruction Bid	Costs		
Pump Station (Total installed HP including standby)		EA (Cost Curv	ve based on total					
	1,200 HP	insta	illed HP)	\$	4,200,000	\$4,200,000	50 yrs	\$1,700,000
Storage Tanks	Cost Estimates	for Storage Tank	FA (Cost Curve by	IUD Ca	onstruction Bid	Costs		
(material determined during pre-design)	0.9 MG	3	Volume)	Ś	3,200,000	\$9,600,000	75 yrs	\$5,800,000
		Estimated S	ubtotal Cost (2) inc	luding	Contingency	\$23,000,000		\$13,000,000
Estimated Subtotal Construction Cost including Cont	tingency	Lotinated o	abtotal 000t (1)		, contingency	\$270,000,000		\$13,000,000
Mobilization					5%	\$14,000,000		\$650,000
Estimated Subtotal Construction Cost including Cont	tingency and M	obilization				\$280,000,000		\$14,000,000
Implementation Environmental Documentation & Permits					5%	\$14,000,000		\$700.000
Design Cost					15%	\$42,000,000		\$2,100,000
Project Administration and Construction Managemen	nt Cost				10%	\$28,000,000		\$1,400,000
Estimated Total Capital Cost including Implementati	on and Conting	ency				\$360,000,000	aluana Malua	\$18,000,000
O&M Costs (Annual)						Present worth of S	aivage value	\$7,400,000
Advanced Water Treatment								
MBR	10		MGD	\$	292,000	\$2,900,000		
Ozone	10		MGD	Ş	57,000	\$570,000		
RO System	10		MGD	\$	480,000	\$4,800,000		
Advanced Oxidation and Disinfection	10		MGD	\$	49,000	\$490,000		
Free Chlorine	10		MGD	\$	32,000	\$320,000		
Chemicals	10	2 080	MGD brs/MGD	ş	120,000	\$1,200,000		
Electricity	10	22,610.000	kWh/vr	Ś	0.15	\$3,400,000		
Monitoring		,,	\$/year	\$	1,000,000	\$1,000,000		
Surface Water Treatment								
SWT O&M (Orinda)	3,650		MG	\$	70	\$260,000		
Conveyance								
Annual O&M			2% of construction	n cost		\$186,000		
Pump Stations		500				450.000		
Labor Costs Electricity		4 873 000	nrs kWb/yr	Ş ¢	135	\$68,000 \$730,000		
Pump Station Consumables		5% of pump sta	tion construction of	ost	0.15	\$210,000		
Storen Tarka								
Annual Q&M		1% of cc	instruction cost			\$96.000		
			Total O&M Costs	(\$/yr)		\$20,000,000		
Annualized Costs (\$ / Year)								
Annualized Capital Costs (\$/Year)	Two payments	per year, spread	over Project Life			\$18,000,000		
	Annuulizeu vui	ue oj present wor	un .			\$20.000.000		
Total Annualized Cost						\$38,000,000	_	
Deliveries of Recycled Water	11,2	00 AFY					-	
Estimated Unit Cost (\$/AF)						\$3,400		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$3,200		Distribution	\$200		
Annualized Costs (\$ / Year)	Dry Year Adjus	stment (Supply us	ed 3/10 years)					
Annualized Capital Costs (\$/Year)	Two payments	per year, spread	over Project Life			\$18,000,000	Same as constant use.	
Annualized Salvage Value	Annualized val	ue oj present wor	LTI			-\$380,000 \$15 393 000	sume as constant use.	
Total Annualized Cost						\$33,000,000		
Annual Average Deliveries of Recycled Water	3,30	50 AFY					-	
Estimated Unit Cost (\$/AF)						\$9,800		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$9,500		Distribution	\$300		

Distant bit Distant bit <thdistant bit<="" th=""> <thdistant bit<="" th=""></thdistant></thdistant>	Alternative LA-Chabot-10							EBMUD Recycled V	Vater Master Plan Update
Update by: Jote is the set of	Last Updated:	30-May-18					Discount Rate	Project Life	
Cit D2 00, 09, 2073, 12870,06 Sote Ont Unit Ont Cost Unit Cost	Updated by:	I. Jaffe				3	3%	30 Ye	ars
thm top Op Unit Unit Col Top I op I	CCI (20 City, Dec 2017): 10870.06								
General Construction of a process of a proces of a process of a process of a process of a process of	Item	Size	Qty	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value
Tatalana Cost Damate for Preciment are based on UNE Cost Data for Preciment Precision Support Preci	Capital Costs								
Model 3: 1.242,000 3.99,0000 2.0100 2.0100 3.99,0000 2.0100 2.0100 3.99,0000 2.0100 2.0100 3.99,0000 2.01000 2.01000 2.01000 2.01000 2.01000 2.01000 2.01000 2.01000 2.01000 2.01000 2.01000 2.0000 2.0	Treatment	Cost Estimates	for Treatment are	e based on Unit Co	osts for E	Each Treatment	Process		
0.00000000000000000000000000000000000	MF/UF system	8		MGD	Ş	1,225,000	\$9,800,000	30 yrs	\$0
number of barby number of barby number of barby Status Stat	RU System	8		MGD	Ş	1,475,000	\$12,000,000	30 yrs	\$U ¢0
States Meto S	Chemicals (Storage and Use)	8		MGD	Ş	437,500	\$3,500,000	30 yrs	\$U ¢0
Cathor of product statustic Statusic <thstatustic< td=""><td>Sitework/Pining/Structures</td><td>8</td><td></td><td>MGD</td><td>ڊ ک</td><td>3 187 500</td><td>\$1,000,000</td><td>30 yrs</td><td>50 \$0</td></thstatustic<>	Sitework/Pining/Structures	8		MGD	ڊ ک	3 187 500	\$1,000,000	30 yrs	50 \$0
Contraction Contraction Contraction Cont 975 5200,000 975 State Tax State State Subtrated Contraction Cont 520,000,000 97 Contraction Cont Including Contingency Estimated Subtrated Contraction Cont 520,000,000 90 Contraction Cont 520,000,000 90 90 90 Contraction Cont 520,000,000 79 yrs 90,000 90 Contraction Cont 520,000,000 79 yrs 95,000,000 95,91	Siteworky riping/structures	0		Raw Const	ruction	Cost Subtotal	\$52,000,000	30 y13	50 \$0
Sale 1 a. Str. V. Stranuel Cost Application Y. V. Stranuel Cost Application Y. Stranuel Cost Application <thy. application<="" cost="" stranuel="" th=""> Y. Stranuel</thy.>	Contractor Overhead & Profit			nuw const	luction	15%	\$7,800,000		\$0 \$0
Leinated Subtoral Construction Cost 542,000,000 59 Construction Cost Including Contingency Estimated Subtoral Construction Cost 542,000,000 59 Capital Cost Including Contingency Estimated Subtoral Construction Cost 542,000,000 59 Capital Cost Inform Biol Cost Estimated Subtoral Construction Biol Costs 515,120,000 75 yr 551,000 Construction Part Part Part Part Part Part Part Part	Sales Tax	50%	% of Subtotal Cost	Applicable		9%	\$2,300,000		\$0
Sectionated Solutional Contingency Situated Solutional Contingency Solutional Contingency Situated Solutional Contingency<				Estimated Subto	tal Con	struction Cost	\$62,000,000		\$0
Statustical Cost Including Contingency Estimated Subtrait Continuction Cost \$52,000,000 \$50,000,00 \$50,000,00 \$50,000,00 \$50,000,00 \$50,000,00 \$50,000,00 \$50,000,00 \$50,000,00 \$50,000,00 \$50,000,00 \$50,000,00 \$50,000,00 \$50,000,00 \$50,000,00 \$50,000,00 <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>									
Limitandian Construction Cost 93.000000 93 Estimated Subtral Construction Add cost retineming construction bid cost retineming construction cost including construction c	Construction Cost Including Contingency								
Estimated Estimated Subtral Cost [Unidadie Contingency S78,000.000 S0 Capital Cost from Bid) Cost Estimated Subtral Cost [Unidadie Contingency S78,000.000 75 yrs S70,000.00 Non-mbank Pipeline 24-m 2,100 l m ir l S 30 S11,120.00 75 yrs S50,000 Non-mbank Victure Pipeline 24-m 2,100 l m ir l S 30 S11,200.00 75 yrs S50,000 Non-mbank Victure Pipeline 24-m 2,100 l m ir l S 50 S14,000.00 S0 yrs S56,000 S9 yrs S66,000 S9 yrs S66,000 S9 yrs S56,000 S9 yrs S66,000 S9 yrs S66,000 S9 yrs<				Estimated Subto	tal Con	struction Cost	\$62,000,000		\$0
Capital Costs (from Bids) Estimated Subtral Cost (finducing contingency S78,000,000 S9 Concernance Cost Estimates (for Pincines are based on EBMUD construction bid cost + estimating contingency S78,000,000 Ty vir S50,000,000 S51,000,000 S51,	Estimating Contingency					25%	\$16,000,000		\$0
Laniset Construction Bioly Cost Estimates for Pipelines are based on EBMUD construction bid cost + estimating contingency Status			Estimated S	Subtotal Cost (1) in	ncluding	g Contingency	\$78,000,000		\$0
Camera and construction construction and construction const (\$4,0,0,0,0,0,0,0,0,0,0,0,0,0,0,0,0,0,0,0	Capital Costs (from Bids)	Cont Entimeter	fan Dinalinaa anal						
Character Construction Construction <td>Conveyance Non-urban Bingling</td> <td>Cost Estimates</td> <td>2 100 LE</td> <td>in LE</td> <td>construc c</td> <td>ction bid cost + i</td> <td>é1 E12 000</td> <td>7E vrc</td> <td>\$010.000</td>	Conveyance Non-urban Bingling	Cost Estimates	2 100 LE	in LE	construc c	ction bid cost + i	é1 E12 000	7E vrc	\$010.000
Cart of control 2 + in 2 + in 2 + in 3 + in< <	I ow-Density I Irban Pineline	24-III 24-in	9 700 LF	in-LF	ç ç	3U 30	\$1,312,000 \$1,312,000	75 yrs	¢2 KUU VUU \$310,000
Instruction	High-Density Urban Pipeline	24-in	1 300 LF	in-LF	ڊ خ	40 50	\$3,312,000 \$1 560 000	75 yrs	\$3,000,000 \$940 000
Pump Station (Total installed PP), including standby) Cost Estimates (P Lump Station and Desire (Lump Based on ESMUZO construction Bid Costs) S 1,600,000 S 1,600,000 S 0 yrs S 64,000 Estimates Subtotal Construction cost including Contingency S 25,000 S 3,100,000	ingi-bensity orban ripenne	24-111	1,500 Ei		Ŷ	50	Ş1,500,000	7.5 yis	\$540,000
Pump Station (Total installed HP, including standby) 225 HP Installed HP \$ 1,600,000 \$ 50 yrs \$ 564,000 Etimated Subtotal Cost (2) Including Contingency \$ 1,600,000 \$ 50 yrs \$ 564,000 Model Subtotal Costruction Cost Including Contingency and Mobilization \$ 5% \$ 4,600,000 \$ \$ 81,000,000 Model Subtotal Costruction Cost Including Contingency and Mobilization \$ 5% \$ 4,500,000 \$ \$ 31,000,000 Bitmated Subtotal Costruction Cost Including Contingency and Mobilization \$ 5% \$ \$ 4,500,000 \$ \$ 31,000,000 Review Management Cost 15% \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	Pump Stations	Cost Estimates	for Pump Station	s are based on EBI	MUD Co	instruction Bid C	Costs		
Let / IP manual (H7) 3 1,000,000 3,1,000,000 3,1,000,000 3,1,000,000 5,100,000 <th< td=""><td>Pump Station (Total installed HP, including standby)</td><td>225 U.D</td><td>EA (COST CUIN</td><td>ve based on total</td><td>ć</td><td>1 600 000</td><td>\$1 600 000</td><td>EQ yrs</td><td>\$640,000</td></th<>	Pump Station (Total installed HP, including standby)	225 U.D	EA (COST CUIN	ve based on total	ć	1 600 000	\$1 600 000	EQ yrs	\$640,000
Estimated Subtral Construction Cost including Contingency Sit Mode One Sit Mode One <t< td=""><td></td><td>223 HF</td><td>IIIsta</td><td>alleu HF)</td><td>Ş</td><td>1,000,000</td><td>\$1,000,000</td><td>50 yis</td><td>\$040,000</td></t<>		223 HF	IIIsta	alleu HF)	Ş	1,000,000	\$1,000,000	50 yis	\$040,000
Estimate Subtration Cost including Contrigency and Mobilization 5% 54,000,000 5410,000 Estimate Subtration Construction Cost including Contrigency and Mobilization 5% 54,000,000 5410,000 Environmental Documentation & Permits 5% 54,000,000 5430,000 Design Cost 15% 515,000,000 513,300,000 513,000,000 Stimate Administration and Construction Management Cost 10% 537,000,000 541,000,000 Stimate Total Capital Cost including Implementation and Contingency 10% 537,000,000 541,000,000 Stimate Total Capital Cost including Implementation and Contingency 250,000 54,000,000 541,000,000 Stimate Total Capital Cost including Implementation and Contingency 322,000 52,600,000 54,000,000 RO System 8 MGD 5 322,000 52,600,000 RO System 8 22,000 52,000,000 54,000,000 Advanced Oxidation and Disinfection 8 22,000 58,000,000 Labor 8 23,000 133 52,200,000 Conner Station Construction cost 520			Estimated S	Subtotal Cost (2) in	ncluding	g Contingency _	\$14,000,000		\$8,100,000
SN 3/1.000.000 3/	Estimated Subtotal Construction Cost including Cont	ingency				50/	\$92,000,000		\$8,100,000
Calminest Subtract Construction Cost Including Unitingent / and industriation Site	NODIIIZation	ingoncy and Mo	hilization			5%	\$4,600,000		\$410,000
Bit Mathematication SM S4, 900,000 S430,000 S430,000 Design Cost 15% S15,000,000 S13,000,000 S13,000,000 S131,000,000 S130,000,000 S120,000 S120,000 S120,000,000 <	Implementation	ingency and ivio	Dilization				\$97,000,000		\$8,500,000
Design Cost 15% \$15,000,000 \$1,300,000 Project Administration and Construction Management Cost 15% \$15,000,000 \$830,000 Design Cost \$130,000,000 Present Worth of Salvage Value \$4,300,000 Design Cost \$130,000,000 Present Worth of Salvage Value \$4,300,000 Design Cost \$130,000,000 Present Worth of Salvage Value \$4,300,000 Design Cost \$130,000,000 \$130,000,000 \$130,000,000 ROS System 8 MGD \$32,000 \$3,800,000 ROS System 8 MGD \$32,000 \$3,800,000 Chemicals 8 MGD \$130,000 \$130,000 Chemicals 8 2,080 hrs/MGD \$135 \$2,200,000 Electricity 7,240,000 KMI/Vr \$0.15 \$1,100,000 \$100 Conveyance 2% of construction cost \$250,000 \$100 \$100 Pump Station Costs \$00 hrs<	Environmental Documentation & Permits					5%	\$4,900.000		\$430.000
Instruction Instruction Singlobolic Singlobolic <thsinglobolic< th=""> Singlobolic</thsinglobolic<>	Design Cost					15%	\$15.000.000		\$1,300,000
Estimated Total Capital Cost including Implementation and Contingency \$13,000,000 \$13,00,000	Project Administration and Construction Managemen	t Cost				10%	\$9,700,000		\$850,000
ORA Operation Second Water Treatment Second Water Tr	Estimated Total Capital Cost including Implementation	on and Continge	ency				\$130,000,000		\$11,000,000
OBM Costs (Annual) M4/xnaced Varie Treatment. M4/ynaced Varie Treatment. M4/ynaced Varie Treatment. M5 ystem 8 MGD \$ 325,000 \$25,000,000 Advanced Oxidation and Disinfection 8 MGD \$ 49,000 \$38,00,000 Advanced Oxidation and Disinfection 8 MGD \$ 120,000 \$25,000 Chemicals 8 MGD \$ 120,000 \$960,000 Labor 8 2,080 hrs/MGD \$ 135 \$52,200,000 Labor 8 2,080 hrs/MGD \$ 135 \$52,000,000 Conveyance 2% of construction cost \$25,000 \$20,000 Pump Station Consumables 500 hrs \$ 135 \$58,000 Electricity 803,000 kMh/yr \$ 0.15 \$12,00,000 Pump Station Consumables 5% of pump station construction cost \$80,000 Annualized Salvage Value Annualized value of present worth \$12,000,000 Annualized Value Osts (\$/rear) 700 payments per year, spreed over Project Ufe \$28,000 Annualized Value Osts (\$/rear) \$12,000,000 \$21,000,000							Present Worth of Sa	alvage Value	\$4,500,000
Advanced Water Treatment. MGD \$ 325,000 \$ 52,600,000 RO System 8 MGD \$ 480,000 \$ 380,000 RO System 8 MGD \$ 480,000 \$ 380,000 Free Chlorine 8 MGD \$ 32,000 \$ 330,000 Chemicals 8 MGD \$ 32,000 \$ 5960,000 Labor 8 2,080 hrs/MKD \$ 120,000 \$ 5960,000 Electricity 7,240,000 kWh/yr \$ 0.15 \$ 1,100,000 Conveyance Z% of construction cost \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	O&M Costs (Annual)								
MM (DP system 8 MGD 5 420,000 S2,400,000 RO system 8 MGD 5 480,000 S3,800,000 Advanced Oxidation and Disinfection 8 MGD 5 480,000 S3,800,000 Chemicals 8 MGD 5 420,000 S960,000 Labor 8 2,080 hrs/MGD 5 135 S2,200,000 Electricity 7,240,000 kWh/yr 5 0.15 S1,100,000 Conveyance 2% of construction cost \$250,000 S660,000 Pump Station S03,000 kWh/yr 5 0.15 S120,000 Pump Station Consumables 500 hrs 5 135 S66,000 Electricity 803,000 kWh/yr 5 0.15 S120,000 Pump Station Consumables 506 of pump station construction cost \$890,000 S80,000 Annual 0&M Costs (\$ / Year) Two payments per year, spread over Project Life \$12,000,000 S12,000,000 Catial An	Advanced Water Treatment								
Advanced Oxidation and Disinfection 8 McD 5 490,000 \$3380,000 Free Chlorine 8 McD \$ 32,000 \$2560,000 Chemicals 8 McD \$ 120,000 \$9560,000 Labor 8 2,080 hrs/MCD \$ 135 \$2,200,000 Electricity 7,240,000 kWh/yr \$ 0.15 \$1,100,000 Conveyance	RO System	8		MGD	Ş	325,000	\$2,600,000		
Prec Choine 8 MGD 5 32,000 5260,000 Chemicalis 8 MGD \$ 32,000 \$566,000 Labor 8 2,080 hrs/MGD \$ 32,000 \$566,000 Electricity 7,240,000 kWh/yr \$ 0.15 \$11,00,000 Conveyance Annual 0&M 2% of construction cost \$250,000 Pump Stations Labor Costs 500 hrs \$ 135 \$68,000 Electricity 803,000 kWh/yr \$ 0.15 \$120,000 Pump Station Consumables 5% of pump station construction cost \$80,000 \$80,000 Pump Station Consumables 5% of pump station construction cost \$80,000 \$80,000 Annualized Costs (\$ / Year) Two payments per year, spread over Project LIJe \$80,000 \$12,000,000 Annualized Salvage Value Annualized value of present worth \$23,000 \$12,000,000 \$12,000,000 Total Annualized Salvage Value Annualized Value Or payments per year, spread over Project LIJe \$4,000 \$132,000,00	Advanced Oxidation and Disinfection	8		MGD	ڊ ک	480,000	\$3,800,000 \$3,000,000		
Chemicals 8 MGD \$ 120,000 \$960,000 Labor 8 2,080 hrs/MGD \$ 135 \$2,200,000 Electricity 7,240,000 kWh/yr \$ 0.15 \$1,100,000 Conveyance Annual 0&M 2% of construction cost \$250,000 Pump Stations 2% of construction cost \$250,000 Pump Station Consumables 500 hrs \$ 135 Electricity 803,000 kWh/yr \$ 0.15 \$12,000,000 Pump Station Consumables 5% of pump station construction cost \$80,000 \$80,000 \$80,000 Pump Station Consumables 5% of pump station construction cost \$80,000 \$80,000 \$80,000 Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$5,60,000 \$5,200,000 Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$12,000,000 \$12,000,000 Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$4,000 \$200 Annualized Costs (\$ / Year)	Free Chlorine	8		MGD	ŝ	32.000	\$260.000		
Labor 8 2,080 hrs/MGD \$ 135 \$2,200,000 Electricity 7,240,000 kWh/yr \$ 0.15 \$1,00,000 Conveyance 3 2% of construction cost \$250,000 Pump Stations 135 \$68,000 Electricity 803,000 kWh/yr \$ 0.15 \$120,000 Pump Station Consumables 5% of pump station construction cost \$80,000 \$80,000 Pump Station Consumables 5% of pump station construction cost \$80,000 \$80,000 Annualized Costs (\$ /Year) Two payments per year, spread over Project UJe \$80,000 \$80,000 Annualized Salvage Value Annualized ore present worth \$22,00,000 \$230,000 Annual Costs (\$/Year) Two payments per year, spread over Project UJe \$12,000,000 \$200 Setimated Unit Cost (\$/AF) Treatment \$3,800 Distribution \$200 Annual Costs (\$/Year) Dry Year Adjustment (Supply used 3/10 years) \$200 \$50,000,000 \$50,000,000 \$50,000,000 \$50,000,000 \$50,000,000 \$50,000,000	Chemicals	8		MGD	\$	120,000	\$960,000		
Electricity 7,240,000 kWh/yr \$ 0.15 \$1,100,000 Conveyance 2% of construction cost \$250,000 Pump Stations 2% of construction cost \$250,000 Pump Stations 3 135 \$668,000 Electricity 803,000 kWh/yr \$ 0.15 \$120,000 Pump Station Consumables 5% of pump station construction cost \$80,000 \$80,000 Pump Station Consumables 5% of pump station construction cost \$80,000 Annualized Costs (S / Year) Two payments per year, spread over Project Life \$66,00,000 Annualized Salvage Value Annualized on the optime per year, spread over Project Life \$66,00,000 Annualized Costs (S / Year) Two payments per year, spread over Project Life \$12,000,000 Cost (S / Year) Two payments per year, spread over Project Life \$12,000,000 Cost (S / Year) Treatment \$3,800 Distribution \$200 Cost (S / Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Cost (S / Year) Stopo,000 Annualized Costi (S / Year) Dry Year Adjustment (Supply	Labor	8	2,080	hrs/MGD	\$	135	\$2,200,000		
Conveyance 2% of construction cost \$250,000 Pump Station 135 \$68,000 Labor Costs 500 hrs \$135 \$68,000 Pump Station Consumables 5% of pump station construction cost \$80,000 \$80,000 Pump Station Consumables 5% of pump station construction cost \$80,000 Pump Station Consumables 5% of pump station construction cost \$80,000 Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$6,600,000 Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$6,600,000 Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$6,600,000 Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$12,000,000 Cotal Annualized Cost \$12,000,000 \$130,0000 Deliveries of Recycled Water 4,480 AFY \$12,000,000 Estimated Unit Cost (\$/ Year) Dry Year Adjustment (Supply used 3/10 years) \$4,000 Annualized Costs (\$ / Year) Treatment \$3,800 Distribution Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$6,600,000 Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$4,000 Annualized Costs (\$ / Year) Two payment	Electricity		7,240,000	kWh/yr	\$	0.15	\$1,100,000		
Annual Q&M 2% of construction cost \$250,000 Pump Stations I Set 1.35 Set8,000 Electricity 803,000 kWh/yr \$0.15 \$120,000 Pump Station Consumables 5% of pump station construction cost \$80,000 Pump Station Consumables 5% of pump station construction cost \$80,000 Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$6,600,000 Annualized Capital Costs (\$/Year) Two payments per year, spread over Project Life \$56,600,000 Annualized Capital Costs (\$/Year) Two payments per year, spread over Project Life \$250,000 Annualized Capital Costs (\$/Year) Two payments per year, spread over Project Life \$260,000 Annualized Capital Costs (\$/Year) Two payments per year, spread over Project Life \$12,000,000 Costal Annualized Capital Costs (\$/Year) A,480 APY \$12,000,000 Deliveries of Recycled Water 4,480 APY \$200 Annualized Capital Costs (\$/Year) Dry Year Adjustment (Supply used 3/10 years) \$200 Annualized Capital Costs (\$/Year) Two payments per year, spread over Project Life \$26,600,000 Same as constant use. Annualized Capital Costs (\$/Year) </td <td>Conveyance</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>	Conveyance								
Pump Stations Labor Costs 500 hrs \$ 135 \$68,000 Electricity 803,000 kWh/yr \$ 0.15 \$120,000 Pump Station Consumables 5% of pump station cost \$80,000 \$80,000 Pump Station Consumables 5% of pump station cost \$80,000 Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$6,600,000 Annualized Salvage Value Annualized oule of present worth -\$5230,000 Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$12,000,000 Total Annualized Cost \$12,000,000 \$12,000,000 Total Annualized Cost \$12,000,000 \$12,000,000 Estimated Unit Cost (\$/AF) Treatment \$3,800 Distribution \$200 Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$6,600,000 Same as constant use. Annualized Costs (\$ / Year) Treatment \$3,800 Distribution \$200 Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$6,600,000 Same as constant use. Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$10,120,000	Annual O&M			2% of construction	on cost		\$250,000		
Annualized Costs 500 hrs \$ 135 \$68,000 Electricity 803,000 kWh/yr \$ 0.15 \$120,000 Pump Station Consumables 5% of pump station construction cost \$80,000 \$120,000 Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$6,600,000 Annualized Costs (\$ /Year) Two payments per year, spread over Project Life \$6,600,000 Annualized Costs (\$ /Year) Two payments per year, spread over Project Life \$12,000,000 Annualized Costs \$12,000,000 \$12,000,000 Deliveries of Recycled Water 4,480 AFY \$18,000,000 Breakdown of Estimated Unit Cost (\$/AF) Treatment \$3,800 Distribution \$200 Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) \$6,00,000 Same as constant use. \$4,000 Breakdown of Estimated Unit Cost (\$/AF) Two payments per year, spread over Project Life \$6,600,000 Same as constant use. Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) \$4,000 Annualized Salvage Value Annualized value of present worth \$230,000 Same as constant use.	Pump Stations						,		
Biolon Costs Biolon Cost Biolon Cost Status Biectricity 803,000 kWh/yr \$ 0.15 \$12,000 Pump Station Consumables 5% of pump station construction cost \$80,000 Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$6,600,000 Annualized Capital Costs (\$/Year) Two payments per year, spread over Project Life \$6,600,000 Annualized Capital Costs Annualized value of present worth \$230,000 Annualized Costs \$12,000,000 \$12,000,000 Deliveries of Recycled Water 4,480 AFY \$12,000,000 Breakdown of Estimated Unit Cost (\$/AF) Treatment \$3,800 Distribution Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) \$10,000 Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$230,000 same as constant use. Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$200 Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$230,000 same as constant use. Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$10,100,000 same as constant use. Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$10,00,000 same as constant use. Annualized Costs (\$	Labor Costs		500	hrs	Ś	135	\$68,000		
Pump Station Consumables 5% of pump station construction cost \$80,000 Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$6,600,000 Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$6,600,000 Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$6,600,000 Annualized Salvage Value Annualized value of present worth -\$230,000 Annualized Cost \$12,000,000 \$12,000,000 Total Annualized Cost \$12,000,000 \$12,000,000 Deliveries of Recycled Water 4,480 AFY \$12,000,000 Estimated Unit Cost (\$/AF) Treatment \$3,800 Distribution \$200 Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) \$100 \$100 Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) \$10,100,000 \$200 Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) \$10,100,000 \$200 Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) \$10,100,000 \$200,000 Annualized Costs (\$ / Year) Dry payments per year, spread over Project Life \$6,600,000 Same as constant use	Electricity		803.000	kWb/vr	ڊ خ	0.15	\$08,000		
Total DSM puttor Edition Costs (\$/ Year) Total D&M Costs (\$/year) Annualized Capital Costs (\$/ Year) Two payments per year, spread over Project Life \$6,600,000 Annualized Salvage Value Annualized value of present worth -\$230,000 Annualized Costs \$12,000,000 Total Annualized Cost \$12,000,000 Total Annualized Cost \$12,000,000 Deliveries of Recycled Water 4,480 AFY Estimated Unit Cost (\$/AF) Treatment \$3,800 Distribution \$200 Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Costs (\$ / Year) Two payments per year, spread over Project Life Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Costs (\$ / Year) Two payments per year, spread over Project Life Annualized Salvage Value Annualized value of present worth Annualized Costs (\$ / Year) Two payments per year, spread over Project Life Annualized Costs (\$ / Year) Two payments per year, spread over Project Life Annualized Salvage Value Annualized value of present worth Annualized Costs (\$ / Year) Two payments per year, spread over Project Life Annualized Costs (\$ / Year) Two payments per year, spread over Project Life Annualized Costs (\$ / Year) Two payments per year, spread over P	Pump Station Consumables		5% of nump sta	ation construction	rost	0.15	\$80,000		
Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$6,600,000 Annualized Salvage Value Annualized value of present worth -\$230,000 Annualized Salvage Value Annualized value of present worth -\$230,000 Annualized Costs \$12,000,000 \$12,000,000 Total Annualized Cost \$12,000,000 \$12,000,000 Deliveries of Recycled Water 4,480 AFY \$18,000,000 Breakdown of Estimated Unit Cost (\$/AF) Treatment \$3,800 Distribution \$200 Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Costs (\$ / Year) Annualized value of present worth -\$230,000 Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Costs (\$ / Year) Annualized value of present worth -\$230,000 Same as constant use. Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$6,600,000 Same as constant use. -\$230,000 Same as constant use. Annualized Costs \$10,120,000 \$10,120,000 \$10,120,000 \$10,120,000 Total Annualized Cost \$10,120,000 \$10,000,000 \$16,000,000 \$10,120,000 Annuala Average Deliveries of Recycled Water 1,344 AFY </td <td>· · · · · · · · · · · · · · · · · · ·</td> <td></td> <td></td> <td>Total O&M Cost</td> <td>s (\$/yr)</td> <td></td> <td>\$12,000,000</td> <td></td> <td></td>	· · · · · · · · · · · · · · · · · · ·			Total O&M Cost	s (\$/yr)		\$12,000,000		
Annualized Capital Costs (\$/Year) Two payments per year, spread over Project Life \$6,600,000 Annual O&M Costs -\$230,000 Annual O&M Costs \$11,000,000 Total Annualized Cost \$12,000,000 Breakdown of Estimated Unit Cost (\$/AF) \$4,000 Breakdown of Estimated Unit Cost (\$/AF) Treatment \$3,800 Distribution Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Capital Costs (\$ / Year) Annualized Salvage Value Annualized value of present worth -\$230,000 Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Capital Costs (\$ / Year) Two payments per year, spread over Project Life \$6,600,000 Same as constant use. Annualized Capital Costs (\$ / Year) Two payments per year, spread over Project Life \$10,120,000 Annualized Capital Costs (\$ / Year) Two payments per year, spread over Project Life \$6,600,000 Same as constant use. Annualized Costs \$10,120,000 Same as constant use. \$10,120,000 Annualized Cost \$10,120,000 \$16,000,000 \$16,000,000 Annualized Cost \$10,120,000 \$16,000,000 \$16,000,000 Breakdown of Estimated Unit Cost (\$/A	Annualized Costs (\$ / Year)								
Annualized Salvage Value Annualized value of present worth -\$230,000 Annual O&M Costs \$12,000,000 Total Annualized Cost \$18,000,000 Deliveries of Recycled Water 4,480 AFY Estimated Unit Cost (\$/AF) \$4,000 Breakdown of Estimated Unit Cost (\$/AF) Treatment \$3,800 Distribution Annualized Costs (\$ /Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Costs (\$ /Year) Annualized Salvage Value Annualized or present worth -\$5230,000 Same as constant use. Annualized Salvage Value Annualized or present worth -\$230,000 Same as constant use. Annualized Costs (\$ /Year) Two payments per year, spread over Project Life \$6,600,000 Same as constant use. Annualized Costs (\$ /Year) Two payments per year, spread over Project Life \$10,120,000 Annualized Costs \$10,120,000 \$10,120,000 Annualized Cost \$10,120,000 \$16,000,000 Annualized Cost (\$/AF) \$13,44 AFY \$16,000,000 Breakdown of Estimated Unit Cost (\$/AF) \$12,000 \$12,000 Breakdown of Estimated Unit Cost (\$/AF) \$12,000 \$12,000	Annualized Capital Costs (\$/Year)	Two payments	per year, spread o	over Project Life			\$6,600,000		
Annual Q&M Costs \$12,000,000 Total Annualized Cost \$18,000,000 Deliveries of Recycled Water 4,480 AFY Estimated Unit Cost (\$/AF) Treatment \$3,800 Distribution Breakdown of Estimated Unit Cost (\$/AF) Treatment \$3,800 Distribution Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Costs (\$/Year) Annualized Salvage Value Annualized value of present worth -\$230,000 Same as constant use. Annualized Costs \$10,120,000 Stin_2000 Total Annualized Cost \$10,120,000 Stin_2000 Annualized Cost \$10,120,000 Stin_2000 Stimated Unit Cost (\$/AF) Stif,000,000 Stif,000,000 Breakdown of Estimated Unit Cost (\$/AF) Stif,000,000 Stif,000,000 Breakdown of Estimated Unit Cost (\$/AF) Treatment \$11,000 Distribution	Annualized Salvage Value	Annualized val	ue of present wor	th			-\$230,000		
Total Annualized Cost \$18,000,000 Deliveries of Recycled Water 4,480 AFY Estimated Unit Cost (\$/AF) \$4,000 Breakdown of Estimated Unit Cost (\$/AF) Treatment \$3,800 Distribution Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Salvage Value Annualized value of present worth Annualized Costs \$10,120,000 Samualized Cost \$10,120,000 Total Annualized Cost \$16,000,000 Samualized Cost \$10,120,000 Samualized Cost \$10,120,000 Samualized Cost \$10,120,000 Samualized Cost \$10,120,000 Samualized Cost \$10,000 Samualized Cost \$10,000 Samualized Cost \$11,000 Samualized Cost \$12,000 Samualized Cost \$12,000	Annual O&M Costs						\$12,000,000		
Deliveries of Recycled Water 4,480 AFY Estimated Unit Cost (\$/AF) \$4,000 Breakdown of Estimated Unit Cost (\$/AF) Treatment \$3,800 Distribution Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Salvage Value Annualized value of present worth \$6,600,000 Same as constant use. Annualized Salvage Value Annualized value of present worth \$10,120,000 Total Annualized Cost \$10,120,000 Annual Average Deliveries of Recycled Water 1,344 AFY Estimated Unit Cost (\$/AF) Treatment \$11,000 Distribution Breakdown of Estimated Unit Cost (\$/AF) Treatment \$11,000 Distribution	Total Annualized Cost					-	\$18,000,000		
Estimated Unit Cost (\$/AF) Treatment \$4,000 Breakdown of Estimated Unit Cost (\$/AF) Treatment (\$3,800 Distribution \$200 Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Costs (\$ / Year) Two payments per year, spread over Project Life \$6,600,000 Same as constant use. Annualized Salvage Value Annualized value of present worth -\$230,000 Same as constant use. Annual O&M Costs \$10,120,000 Total Annualized Cost \$16,000,000 Annual Average Deliveries of Recycled Water 1,344 AFY Estimated Unit Cost (\$/AF) \$12,000 Breakdown of Estimated Unit Cost (\$/AF) Treatment \$11,000	Deliveries of Recycled Water	4,48	BO AFY						
Breakdown of Estimated Unit Cost (S/AF) Treatment \$3,800 Distribution \$200 Annualized Costs (S / Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Capital Costs (S/Year) Two payments per year, spread over Project Life \$6,600,000 Same as constant use. Annualized Salvage Value Annualized value of present worth -\$230,000 Same as constant use. Annual O&M Costs \$10,120,000 Total Annualized Cost \$16,000,000 Annual Average Deliveries of Recycled Water 1,344 AFY Estimated Unit Cost (S/AF) \$12,000 Breakdown of Estimated Unit Cost (S/AF) \$12,000	Estimated Unit Cost (\$/AF)						\$4,000		
Annualized Costs (\$ / Year) Dry Year Adjustment (Supply used 3/10 years) Annualized Capital Costs (\$ / Year) Two payments per year, spread over Project Life \$6,600,000 Same as constant use. Annualized Salvage Value Annualized value of present worth -\$230,000 Same as constant use. Annual O&M Costs \$10,120,000 \$10,120,000 Total Annualized Cost \$16,000,000 Annual Verage Deliveries of Recycled Water 1,344 AFY Estimated Unit Cost (\$/AF) \$12,000 Breakdown of Estimated Unit Cost (\$/AF) Treatment \$11,000	Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$3,800		Distribution	\$200		
Annualized Capital Costs (\$/Year) Two payments per year, spread over Project Life \$6,600,000 Same as constant use. Annualized Salvage Value Annualized value of present worth -\$230,000 Same as constant use. Annual O&M Costs \$10,120,000 Total Annualized Cost \$16,000,000 Annual Verage Deliveries of Recycled Water 1,344 AFY Estimated Unit Cost (\$/AF) \$12,000 Breakdown of Estimated Unit Cost (\$/AF) Treatment \$11,000	Annualized Costs (\$ / Year)	Dry Year Adjus	tment (Supply us	ed 3/10 years)					
Annualized Salvage Value Annualized value of present worth -\$230,000 Same as constant use. Annual 0&M Costs \$10,120,000 Total Annualized Cost \$16,000,000 Annual Average Deliveries of Recycled Water 1,344 AFY Estimated Unit Cost (\$/AF) \$12,000 Breakdown of Estimated Unit Cost (\$/AF) \$1,000	Annualized Capital Costs (\$/Year)	Two payments	per year, spread o	over Project Life			\$6,600,000	Same as constant i	ise.
Annual O&M Costs \$10,120,000 Total Annualized Cost \$10,120,000 Annual Average Deliveries of Recycled Water 1,344 AFY Estimated Unit Cost (\$/AF) Freakdown of Estimated Unit Cost (\$/AF) Treatment \$11,000 Distribution \$1,000	Annualized Salvage Value	Annualized val	ue of present wor	th			-\$230,000	Same as constant i	ise.
I otal Annualized Cost \$16,000,000 Annual Average Deliveries of Recycled Water 1,344 AFY Estimated Unit Cost (\$/AF) \$12,000 Breakdown of Estimated Unit Cost (\$/AF) Treatment \$11,000	Annual O&M Costs						\$10,120,000		
Estimated Unit Cost (\$/AF) \$12,000 Strakdown of Estimated Unit Cost (\$/AF) \$12,000	Iotal Annualized Cost	1 3.				-	\$16,000,000		
Breakdown of Estimated Unit Cost (\$/AF) Treatment \$11,000 Distribution \$1.000	Estimated Unit Cost (\$/AF)	1,34	HH AFT				\$12,000		
	Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$11.000		Distribution	\$1,000		

Alternative Sat-ResSP-4							EBMUD Recycle	d Water Master Plan Lindate
Lact Indated:	20 May 18					Discount Pato	Dinob necycle	viort Life
Lindated by:	J laffe					3%	30	Vears
CCI (20 City, Dec 2017): 10870.06	1. June					370	50	lears
Item	Size	Otv	Unit		Unit Cost	Total Cost	Useful Life	Salvage Value
Canital Costs	0.20						obertai Eire	ourrage value
Treatment	Cost Estimates	for Treatment are	based on Unit Co	sts for	Each Treatmer	nt Process		
ME/LIE system	A	joi meatment are	MGD	: sts joi ج	1 225 000	\$4 900 000	30 yrs	\$0
BO System	4		MGD	Ś	1 475 000	\$5,900,000	30 yrs	\$0
Advanced Oxidation and Disinfection	4		MGD	ś	437 500	\$1,800,000	30 yrs	\$0 \$0
Chemicals (Storage and Lise)	4		WIGD	ç	125 000	\$1,800,000	30 yrs	0¢
Sitework/Pining/Structures	4		MGD	ś	3 187 500	\$13,000,000	30 yrs	\$0 \$0
Site worky riping/structures	-		Raw Const	ruction	Cost Subtotal	\$15,000,000	50 915	<u>\$0</u>
Contractor Overhead & Profit			nuw const	uction	15%	\$3,900,000		\$0 \$0
Salas Tay	50%	% of Subtotal Cost	Applicable		0%	\$1,200,000		0¢
	50%	76 01 Subtotal Cost	Estimated Subto	tal Con	struction Cost	\$1,200,000		50 \$0
			Estimated Subto	tur con	struction cost	<i>\$31,000,000</i>		<i>4</i> 0
Construction Cost Including Contingency								
construction cost menuing contingency			Estimated Subto	tal Con	struction Cost	\$31,000,000		\$0
Estimating Contingency					25%	\$7,800,000		\$0
Loundaring contingency		Estimated S	ubtotal Cost (1) ir	ncludin	g Contingency	\$39.000.000		\$0
Capital Costs (from Bids)		Lotinated o			g contingency	<i><i><i>qusjuujuu</i></i></i>		֥
Conveyance	Cost Estimates	for Pinelines are h	ased on FRMUD (onstru	ction hid cost +	estimatina contingency		
Non-urban Pipeline	16-in	1.900 LF	in-LF	Ś	30	\$912.000	75 vrs	\$550.000
Low-Density Urban Pipeline	16-in	5 900 LF	in-LF	Ś	40	\$3 776 000	75 yrs	\$2,300,000
High-Density Urban Pipeline	16-in	5,900 LF	in-LF	Ś	50	\$4 720 000	75 yrs	\$2,800,000
Trenchless Crossings	20	5,500 21		Ŷ	50	<i> </i>	75 115	\$2,000,000
San Pablo Tunnel Rehab		17 600	IF	Ś	3 500	\$62,000,000	75 vrs	\$37,000,000
HDD (All Inclusive)		1 700	L.	Ś	2 200	\$3,700,000	75 yrs	\$2,200,000
		2,700	-	Ŷ	2,200	\$5,700,000	, 5 , 15	<i>\$2,200,000</i>
Pump Stations	Cost Estimates	for Pump Stations	are based on EBN	ИUD Са	onstruction Bid	Costs		
Pump Station (Total installed HP, including standby)		EA (Cost Curv	e based on total					
· ····································	600 HP	insta	lled HP)	\$	2,400,000	\$2,400,000	50 yrs	\$960,000
		Estimated S	ubtotal Cost (2) ir	ncludin	g Contingency	\$78,000,000		\$46,000,000
Estimated Subtotal Construction Cost including Cont	ingency					\$120,000,000		\$46,000,000
Mobilization					5%	\$6,000,000		\$2,300,000
Estimated Subtotal Construction Cost including Cont	ingency and Mo	bilization				\$130,000,000		\$48,000,000
Implementation								
Environmental Documentation & Permits					5%	\$6,500,000		\$2,400,000
Design Cost					15%	\$20,000,000		\$7,200,000
Project Administration and Construction Managemer	nt Cost				10%	\$13,000,000		\$4,800,000
Estimated Total Capital Cost including Implementati	on and Conting	ency				\$170,000,000		\$62,000,000
						Present Worth of Sa	alvage Value	\$25,000,000
O&M Costs (Annual)								
Advanced Water Treatment								
MF/UF system	4		MGD	\$	325,000	\$1,300,000		
RO System	4		MGD	\$	480,000	\$1,900,000		
Advanced Oxidation and Disinfection	4		MGD	\$	49,000	\$200,000		
Free Chlorine	4		MGD	Ş	32,000	\$130,000		
Chemicais	4	2 000	IVIGD	Ş	120,000	\$480,000		
	4	2,080	nrs/MGD	Ş	135	\$1,100,000		
Electricity		3,620,000	kWh/yr	Ş	0.15	\$540,000		
Surface Water Treatment				<i>.</i>	25.4	¢270.000		
SWI O&M (Sobrante WIP)	1,460		MG	\$	254	\$370,000		
Conveyance								
Annual O&M			2% of construction	on cost		\$1,500,000		
Pump Stations								
Labor Costs		500	hrs	Ś	135	\$68.000		
Electricity		1 892 000	kWh/vr	Ś	0.15	\$280,000		
Pump Station Consumables		5% of nump sta	tion construction	rost	0.15	\$120,000		
			Total O&M Costs	: (\$/yr)		\$8,000,000		
Annualized Costs (\$ / Year)			Total Call Cost	· (•/]·/		\$0,000,000		
Annualized Capital Costs (\$/Year)	Two navments	ner vear spread o	over Project Life			\$8,600,000		
Annualized Capital Costs (0) (Car)	Annualized val	ue of present wort	h			-\$1 300 000		
Annual O&M Costs	,	ac of present word				\$8,000,000		
Total Annualized Cost						\$5,000,000		
Deliveries of Recycled Water	4.4	80 ΔΕΥ				\$15,000,000		
Estimated Unit Cost (\$/AF)	-,-	507411				\$3 300		
Broakdown of Ectimated Unit Cost (\$/AE)		Trootmont	¢1 000		Distribution	\$3,300		
Breakdown of Estimated Onit Cost (3/AF)		Treatment	\$1,500		Distribution	Ş1,400		
Annualized Costs (\$ / Year)	Dry Year Adjus	tment (Supply us	ed 3/10 years)					
Annualized Capital Costs (\$/Year)	Two payments	per year, spread o	over Project Life			\$8,600,000	Same as consta	nt use.
Annualized Salvage Value	Annualized val	ue of present wort	h			-\$1,300,000	Same as consta	nt use.
Annual O&M Costs						\$6,728,000		
Total Annualized Cost						\$14,000,000		
Annual Average Deliveries of Recycled Water	1,3	44 AFY						
Estimated Unit Cost (\$/AF)						\$10,000		
Breakdown of Estimated Unit Cost (\$/AF)		Treatment	\$5,700		Distribution	\$4,300		

APPENDIX C: ECONOMIC EVALUATION – WILLINGNESS-TO-PAY AND SHORTAGE ESTIMATES

This page is intentionally left blank.

Attachment 1

W-E Model Simulation of Demand Rationing for WSMP 2040 Portfolio E3

W-E Model Simulation of Demand Rationing <u>with</u> Increments of New Recycled Water Capacity for WSMP 2040 Portfolio E3

	Development Year							
Hydrologic Base								
Year	2020	2025	2030	2035	2040			
1921	8%	0%	0%	0%	0%			
1922	0%	0%	0%	0%	0%			
1923	0%	0%	0%	0%	0%			
1924	0%	0%	0%	0%	0%			
1925	0%	0%	0%	0%	0%			
1926	0%	0%	0%	0%	0%			
1927	0%	0%	0%	0%	0%			
1928	0%	0%	0%	0%	0%			
1929	0%	0%	0%	0%	0%			
1930	0%	0%	0%	0%	0%			
1931	0%	0%	0%	0%	8%			
1932	0%	0%	0%	0%	0%			
1933	0%	0%	0%	0%	0%			
1934	0%	0%	0%	0%	0%			
1935	0%	0%	0%	0%	0%			
1936	0%	0%	0%	17%	0%			
1937	0%	0%	0%	0%	0%			
1938	0%	0%	0%	0%	0%			
1939	0%	0%	0%	0%	0%			
1940	0%	0%	1%	0%	0%			
1941	0%	0%	17%	0%	0%			
1942	0%	0%	0%	0%	0%			
1943	0%	0%	0%	0%	0%			
1944	0%	0%	0%	0%	0%			
1945	0%	0%	0%	0%	0%			
1946	0%	8%	0%	0%	5%			
1947	0%	0%	0%	0%	20%			
1948	0%	0%	0%	0%	20%			
1949	0%	0%	0%	0%	18%			
1950	0%	0%	0%	0%	0%			
1951	8%	0%	0%	5%	0%			
1952	0%	0%	0%	20%	0%			
1953	0%	0%	0%	20%	0%			
1954	0%	0%	0%	18%	0%			

	Development Year							
Hydrologic Base								
Year	2020	2025	2030	2035	2040			
1955	0%	0%	0%	0%	0%			
1956	0%	0%	5%	0%	0%			
1957	0%	0%	20%	0%	2%			
1958	0%	0%	20%	0%	17%			
1959	0%	0%	18%	0%	13%			
1960	0%	0%	0%	0%	15%			
1961	0%	5%	0%	0%	17%			
1962	0%	20%	0%	2%	19%			
1963	0%	20%	0%	17%	0%			
1964	0%	18%	0%	11%	2%			
1965	0%	0%	0%	14%	0%			
1966	5%	0%	0%	17%	0%			
1967	20%	0%	2%	18%	0%			
1968	20%	0%	17%	0%	0%			
1969	18%	0%	13%	1%	0%			
1970	0%	0%	15%	0%	0%			
1971	0%	0%	17%	0%	0%			
1972	0%	2%	18%	0%	0%			
1973	0%	17%	0%	0%	0%			
1974	0%	12%	1%	0%	0%			
1975	0%	14%	0%	0%	0%			
1976	0%	17%	0%	0%	0%			
1977	2%	18%	0%	0%	4%			
1978	17%	0%	0%	0%	0%			
1979	15%	1%	0%	0%	0%			
1980	15%	0%	0%	0%	0%			
1981	17%	0%	0%	0%	0%			
1982	18%	0%	0%	3%	0%			
1983	0%	0%	0%	0%	0%			
1984	1%	0%	0%	0%	3%			
1985	0%	0%	0%	0%	0%			
1986	0%	0%	0%	0%	0%			
1987	0%	0%	4%	0%	0%			
1988	0%	0%	0%	0%	0%			
1989	0%	0%	0%	2%	0%			
1990	0%	0%	0%	0%	0%			
1991	0%	0%	0%	0%	0%			
1992	0%	3%	0%	0%	0%			
1993	0%	0%	0%	0%	0%			
1994	0%	0%	3%	0%	0%			

	Development Year								
Hydrologic Base									
Year	2020	2025	2030	2035	2040				
1995	0%	0%	0%	0%	0%				
1996	0%	0%	0%	0%	0%				
1997	3%	0%	0%	0%	0%				
1998	0%	0%	0%	0%	0%				
1999	0%	2%	0%	0%	0%				
2000	0%	0%	0%	0%	0%				
2001	0%	0%	0%	0%	0%				
2002	0%	0%	0%	0%	0%				
2003	0%	0%	0%	0%	0%				

W-E Model Simulation of Demand Rationing <u>without</u> Increments of New Recycled Water Capacity for WSMP 2040 Portfolio E3

	Development Year								
Hydrologic Base									
Year	2020	2025	2030	2035	2040				
1921	9%	0%	0%	0%	0%				
1922	0%	0%	0%	0%	0%				
1923	0%	0%	0%	0%	0%				
1924	0%	0%	0%	0%	0%				
1925	0%	0%	0%	0%	0%				
1926	0%	0%	0%	0%	0%				
1927	0%	0%	0%	0%	0%				
1928	0%	0%	0%	0%	0%				
1929	0%	0%	0%	0%	0%				
1930	0%	0%	0%	0%	0%				
1931	0%	0%	0%	0%	12%				
1932	0%	0%	0%	0%	0%				
1933	0%	0%	0%	0%	0%				
1934	0%	0%	0%	0%	0%				
1935	0%	0%	0%	4%	0%				
1936	0%	0%	0%	20%	0%				
1937	0%	0%	0%	0%	0%				
1938	0%	0%	0%	0%	0%				
1939	0%	0%	0%	0%	0%				
1940	0%	0%	4%	0%	0%				
1941	0%	0%	19%	0%	0%				
1942	0%	0%	0%	0%	0%				
1943	0%	0%	0%	0%	0%				
1944	0%	0%	0%	0%	0%				

	Development Year							
Hydrologic Base								
Year	2020	2025	2030	2035	2040			
1945	0%	0%	0%	0%	0%			
1946	0%	10%	0%	0%	10%			
1947	0%	0%	0%	0%	24%			
1948	0%	0%	0%	0%	24%			
1949	0%	0%	0%	0%	22%			
1950	0%	0%	0%	0%	0%			
1951	9%	0%	0%	9%	0%			
1952	0%	0%	0%	23%	0%			
1953	0%	0%	0%	23%	0%			
1954	0%	0%	0%	21%	0%			
1955	0%	0%	0%	0%	0%			
1956	0%	0%	8%	0%	0%			
1957	0%	0%	23%	0%	7%			
1958	0%	0%	23%	0%	21%			
1959	0%	0%	21%	0%	17%			
1960	0%	0%	0%	0%	19%			
1961	0%	7%	0%	0%	21%			
1962	0%	22%	0%	6%	22%			
1963	0%	22%	0%	20%	0%			
1964	0%	20%	0%	15%	6%			
1965	0%	0%	0%	17%	0%			
1966	6%	0%	0%	20%	0%			
1967	21%	0%	5%	21%	0%			
1968	21%	0%	20%	0%	0%			
1969	18%	0%	16%	5%	0%			
1970	0%	0%	18%	0%	0%			
1971	0%	0%	20%	0%	0%			
1972	0%	4%	21%	0%	0%			
1973	0%	19%	0%	0%	0%			
1974	0%	14%	5%	0%	0%			
1975	0%	16%	0%	0%	0%			
1976	0%	19%	0%	0%	0%			
1977	2%	20%	0%	0%	8%			
1978	17%	0%	0%	0%	0%			
1979	16%	3%	0%	0%	0%			
1980	16%	0%	0%	0%	0%			
1981	18%	0%	0%	0%	0%			
1982	19%	0%	0%	7%	0%			
1983	0%	0%	0%	0%	0%			
1984	2%	0%	0%	0%	7%			
	Development Year							
-----------------	------------------	------	------	------	------			
Hydrologic Base								
Year	2020	2025	2030	2035	2040			
1985	0%	0%	0%	0%	0%			
1986	0%	0%	0%	0%	0%			
1987	0%	0%	7%	0%	0%			
1988	0%	0%	0%	0%	0%			
1989	0%	0%	0%	6%	0%			
1990	0%	0%	0%	0%	0%			
1991	0%	0%	0%	0%	0%			
1992	0%	5%	0%	0%	0%			
1993	0%	0%	0%	0%	0%			
1994	0%	0%	6%	0%	0%			
1995	0%	0%	0%	0%	0%			
1996	0%	0%	0%	0%	0%			
1997	4%	0%	0%	0%	0%			
1998	0%	0%	0%	0%	0%			
1999	0%	4%	0%	0%	0%			
2000	0%	0%	0%	0%	0%			
2001	0%	0%	0%	0%	0%			
2002	0%	0%	0%	0%	0%			
2003	0%	0%	0%	0%	0%			

Attachment 2

Willingness-to-Pay for New Increments of Recycled Water Supply under WSMP 2040 Portfolio E3

	Development Year					
Hydrologic Base						
Year	2020	2025	2030	2035	2040	
Mean	\$2,366	\$2 <i>,</i> 365	\$2,402	\$2,398	\$2,401	
1921	\$2 <i>,</i> 865	\$2,113	\$2,113	\$2,113	\$2,113	
1922	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113	
1923	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113	
1924	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113	
1925	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113	
1926	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113	
1927	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113	
1928	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113	
1929	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113	
1930	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113	
1931	\$2,113	\$2,113	\$2,113	\$2,113	\$3,061	
1932	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113	
1933	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113	
1934	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113	
1935	\$2,113	\$2,113	\$2,113	\$2,088	\$2,113	
1936	\$2,113	\$2,113	\$2,113	\$4,506	\$2,113	
1937	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113	
1938	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113	
1939	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113	
1940	\$2,113	\$2,113	\$2,128	\$2,113	\$2,113	
1941	\$2,113	\$2,113	\$4,492	\$2,113	\$2,113	
1942	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113	
1943	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113	
1944	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113	
1945	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113	
1946	\$2,113	\$2,915	\$2,113	\$2,113	\$2,684	
1947	\$2,113	\$2,113	\$2,113	\$2,113	\$5,414	
1948	\$2,113	\$2,113	\$2,113	\$2,113	\$5,414	
1949	\$2,113	\$2,113	\$2,113	\$2,113	\$4,907	
1950	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113	
1951	\$2,850	\$2,113	\$2,113	\$2,612	\$2,113	
1952	\$2,113	\$2,113	\$2,113	\$5,337	\$2,113	
1953	\$2,113	\$2,113	\$2,113	\$5,337	\$2,113	

Willingness-to-Pay in \$/AF for New Increments of Potable Reuse Water Supply under WSMP 2040 Portfolio E3

	Development Year				
Hydrologic Base			•		
Year	2020	2025	2030	2035	2040
1954	\$2,113	\$2,113	\$2,113	\$4,726	\$2,113
1955	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113
1956	\$2,113	\$2,113	\$2,615	\$2,113	\$2,113
1957	\$2,113	\$2,113	\$5,299	\$2,113	\$2,342
1958	\$2,113	\$2,113	\$5,299	\$2,113	\$4,700
1959	\$2,113	\$2,113	\$4,801	\$2,113	\$3 <i>,</i> 835
1960	\$2,113	\$2,113	\$2,113	\$2,113	\$4,225
1961	\$2,113	\$2,508	\$2,113	\$2,113	\$4,722
1962	\$2,113	\$5,197	\$2,113	\$2,271	\$5,054
1963	\$2,113	\$5,197	\$2,113	\$4,570	\$2,113
1964	\$2,113	\$4,700	\$2,113	\$3,482	\$2,271
1965	\$2,113	\$2,113	\$2,113	\$3,900	\$2,113
1966	\$2,459	\$2,113	\$2,113	\$4,656	\$2,113
1967	\$5 <i>,</i> 095	\$2,113	\$2,272	\$4,923	\$2,113
1968	\$5 <i>,</i> 095	\$2,113	\$4,587	\$2,113	\$2,113
1969	\$4,488	\$2,113	\$3,718	\$2,203	\$2,113
1970	\$2,113	\$2,113	\$4,134	\$2,113	\$2,113
1971	\$2,113	\$2,113	\$4,586	\$2,113	\$2,113
1972	\$2,113	\$2,176	\$4,847	\$2,113	\$2,113
1973	\$2,113	\$4,445	\$2,113	\$2,113	\$2,113
1974	\$2,113	\$3,528	\$2,191	\$2,113	\$2,113
1975	\$2,113	\$3,934	\$2,113	\$2,113	\$2,113
1976	\$2,113	\$4,476	\$2,113	\$2,113	\$2,113
1977	\$2,124	\$4,753	\$2,113	\$2,113	\$2,480
1978	\$4,305	\$2,113	\$2,113	\$2,113	\$2,113
1979	\$3,974	\$2,093	\$2,113	\$2,113	\$2,113
1980	\$3,999	\$2,113	\$2,113	\$2,113	\$2,113
1981	\$4,414	\$2,113	\$2,113	\$2,113	\$2,113
1982	\$4,621	\$2,113	\$2,113	\$2,413	\$2,113
1983	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113
1984	\$2,048	\$2,113	\$2,113	\$2,113	\$2,363
1985	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113
1986	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113
1987	\$2,113	\$2,113	\$2,420	\$2,113	\$2,113
1988	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113
1989	\$2,113	\$2,113	\$2,113	\$2,299	\$2,113
1990	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113
1991	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113
1992	\$2,113	\$2,328	\$2,113	\$2,113	\$2,113
1993	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113

	Development Year				
Hydrologic Base					
Year	2020	2025	2030	2035	2040
1994	\$2,113	\$2,113	\$2,305	\$2,113	\$2,113
1995	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113
1996	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113
1997	\$2,285	\$2,113	\$2,113	\$2,113	\$2,113
1998	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113
1999	\$2,113	\$2,218	\$2,113	\$2,113	\$2,113
2000	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113
2001	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113
2002	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113
2003	\$2,113	\$2,113	\$2,113	\$2,113	\$2,113

Willingness-to-Pay in \$/AF for New Increments of Non-Potable Recycled Water under WSMP 2040 Portfolio E3

	Development Year				
Hydrologic Base					
Year	2020	2025	2030	2035	2040
Mean	\$1,979	\$1,977	\$2,020	\$2,016	\$2,013
1921	\$2,865	\$1,647	\$1,647	\$1,647	\$1,647
1922	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1923	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1924	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1925	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1926	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1927	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1928	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1929	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1930	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1931	\$1,647	\$1,647	\$1,647	\$1,647	\$3,061
1932	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1933	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1934	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1935	\$1,647	\$1,647	\$1,647	\$2,088	\$1,647
1936	\$1,647	\$1,647	\$1,647	\$4,506	\$1,647
1937	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1938	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1939	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1940	\$1,647	\$1,647	\$2,128	\$1,647	\$1,647
1941	\$1,647	\$1,647	\$4,492	\$1,647	\$1,647
1942	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647

	Development Year				
Hydrologic Base					
Year	2020	2025	2030	2035	2040
1943	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1944	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1945	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1946	\$1,647	\$2,915	\$1,647	\$1,647	\$2,684
1947	\$1,647	\$1,647	\$1,647	\$1,647	\$5,414
1948	\$1,647	\$1,647	\$1,647	\$1,647	\$5,414
1949	\$1,647	\$1,647	\$1,647	\$1,647	\$4,907
1950	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1951	\$2,850	\$1,647	\$1,647	\$2,612	\$1,647
1952	\$1,647	\$1,647	\$1,647	\$5,337	\$1,647
1953	\$1,647	\$1,647	\$1,647	\$5,337	\$1,647
1954	\$1,647	\$1,647	\$1,647	\$4,726	\$1,647
1955	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1956	\$1,647	\$1,647	\$2,615	\$1,647	\$1,647
1957	\$1,647	\$1,647	\$5,299	\$1,647	\$2,342
1958	\$1,647	\$1,647	\$5,299	\$1,647	\$4,700
1959	\$1,647	\$1,647	\$4,801	\$1,647	\$3 <i>,</i> 835
1960	\$1,647	\$1,647	\$1,647	\$1,647	\$4,225
1961	\$1,647	\$2,508	\$1,647	\$1,647	\$4,722
1962	\$1,647	\$5,197	\$1,647	\$2,271	\$5,054
1963	\$1,647	\$5,197	\$1,647	\$4,570	\$1,647
1964	\$1,647	\$4,700	\$1,647	\$3,482	\$2,271
1965	\$1,647	\$1,647	\$1,647	\$3,900	\$1,647
1966	\$2,459	\$1,647	\$1,647	\$4,656	\$1,647
1967	\$5,095	\$1,647	\$2,272	\$4,923	\$1,647
1968	\$5,095	\$1,647	\$4,587	\$1,647	\$1,647
1969	\$4,488	\$1,647	\$3,718	\$2,203	\$1,647
1970	\$1,647	\$1,647	\$4,134	\$1,647	\$1,647
1971	\$1,647	\$1,647	\$4,586	\$1,647	\$1,647
1972	\$1,647	\$2,176	\$4,847	\$1,647	\$1,647
1973	\$1,647	\$4,445	\$1,647	\$1,647	\$1,647
1974	\$1,647	\$3 <i>,</i> 528	\$2,191	\$1,647	\$1,647
1975	\$1,647	\$3,934	\$1,647	\$1,647	\$1,647
1976	\$1,647	\$4,476	\$1,647	\$1,647	\$1,647
1977	\$2,124	\$4,753	\$1,647	\$1,647	\$2,480
1978	\$4,305	\$1,647	\$1,647	\$1,647	\$1,647
1979	\$3,974	\$2,093	\$1,647	\$1,647	\$1,647
1980	\$3,999	\$1,647	\$1,647	\$1,647	\$1,647
1981	\$4,414	\$1,647	\$1,647	\$1,647	\$1,647
1982	\$4,621	\$1,647	\$1,647	\$2,413	\$1,647

	Development Year				
Hydrologic Base					
Year	2020	2025	2030	2035	2040
1983	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1984	\$2,048	\$1,647	\$1,647	\$1,647	\$2,363
1985	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1986	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1987	\$1,647	\$1,647	\$2,420	\$1,647	\$1,647
1988	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1989	\$1,647	\$1,647	\$1,647	\$2,299	\$1,647
1990	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1991	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1992	\$1,647	\$2,328	\$1,647	\$1,647	\$1,647
1993	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1994	\$1,647	\$1,647	\$2,305	\$1,647	\$1,647
1995	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1996	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1997	\$2,285	\$1,647	\$1,647	\$1,647	\$1,647
1998	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
1999	\$1,647	\$2,218	\$1,647	\$1,647	\$1,647
2000	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
2001	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
2002	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647
2003	\$1,647	\$1,647	\$1,647	\$1,647	\$1,647

APPENDIX D: CUSTOMER LIST FOR EAST BAYSHORE RECYCLED WATER PROJECT

This page is intentionally left blank.

		el Demanus (Filase IA)
Customer Name	Annual Average Demand, gpd	Maximum Month Demand, gpd
Oakland Raimondi Park	11,733	22,293
Oakland Mandela Parkway / 34th St.	980	1,862
Oakland Mandela Parkway / 32th St.	2,646	5,027
Oakland Mandela Parkway /W. Grant	7,615	14,469
Oakland Mandela Parkway / 14th St.	9,105	17,300
Oakland Lowell Park	15,099	28,688
Oakland DeFremery Park	7,355	13,975
Oakland Madison Park	916	1,740
Oakland Lafayette Square	1,469	2,791
Oakland Wade Johnson Park	3,005	5,710
EBALDG - Jack London Gateway Sr. Housing	1,835	3,487
Age Song at Bayside	466	885
Lake Merritt Boathouse & C. Stanford House	21,738	41,302
Peralta College - Admin Building Area	1,363	2,590
Peralta College - Laney Campus	4,314	8,197
Emeryville Park Ave. Streetscape	1,099	2,088
Oakland Willow Mini Park	616	1,170
Extended Stay America Hotel	1,899	3,608
Oakland Museum of California	4,117	7,822
Caltrans - Emeryville Segment 2 Bikeway	1,381	2,624
Caltrans - Bay Bridge - Segment 2 Bikeway	10,716	20,360
Preservation Park - Oakland	4,622	19,993
Sherwin Williams Redevelopment ¹	15,000	28,500
EBMUD Administration Building ²	22,000	28,600
IKEA ²	23,000	29,900
Pacific Park Plaza ²	30,000	39,000
Pixar Corporation ²	21,000	32,000
Allowance for 6 additional HVAC Customers in Emeryville ^{2,3}	51,000	66,300
Frontage Road Pipeline Alignment		
Target Store - Albany	9,100	17,290
Caltrans I-80 Landscape (Eastshore Hwy)	13,100	24,890
Golden Gate Fields(Pacific Racing Assn)	13,160	25,004
Caltrans I-80 Landscape (Frontage Rd)	7,479	14,210
University Village	123,500	234,650
Total Demands	443,000	769,000
¹ Estimated demand for a future development.		
² Industrial Demand		
³ Assumption for cooling tower demands		

Table D-2. List of Intermediate Term Industrial Customers and Associated Recycled Water Demands (Phase 1B)				
Customer Name	Annual Average Demand, gpd	Maximum Month Demand, gpd		
Oakland Raimondi Park	11,733	22,293		
Oakland Mandela Parkway / 34th St.	980	1,862		
Oakland Mandela Parkway / 32th St.	2,646	5,027		
Oakland Mandela Parkway / W. Grant	7,615	14,469		
Oakland Mandela Parkway / 14th St.	9,105	17,300		
Oakland Lowell Park	15,099	28,688		
Oakland DeFremery Park	7,355	13,975		
Oakland Madison Park	916	1,740		
Oakland Lafayette Square	1,469	2,791		
Oakland Wade Johnson Park	3,005	5,710		
EBALDG - Jack London Gateway Sr. Housing	1,835	3,487		
Age Song at Bayside	466	885		
Lake Merritt Boathouse & C. Stanford House	21,738	41,302		
Peralta College - Admin Building Area	1,363	2,590		
Peralta College - Laney Campus	4,314	8,197		
Emeryville Park Ave. Streetscape	1,099	2,088		
Oakland Willow Mini Park	616	1,170		
Extended Stay America Hotel	1,899	3,608		
Oakland Museum of California	4,117	7,822		
Caltrans - Emeryville Segment 2 Bikeway	1,381	2,624		
Caltrans - Bay Bridge - Segment 2 Bikeway	10,716	20,360		
Preservation Park - Oakland	4,622	19,993		
Oakland Housing Authority	3,190	6,062		
Additional Customers on Existing Alignment				
Middleshore Harbor Park	54,238	103,052		
Sherwin Williams Redevelopment ¹	15,000	28,500		
Oakland Army Base Redevelopment ¹	47,768	90,759		
EBMUD Administration Building ²	22,000	28,600		
Chiron Corporation ²	51,700	66,730		
Bayer ²	180,000	220,000		
Cool Port ²	35,000	42,000		
IKEA ²	23,000	29,900		
Pacific Park Plaza ²	30,000	39,000		
Pixar Corporation ²	21,000	32,000		
Allowance for additional 6 HVAC Customers in Emeryville ^{2,3}	51,000	66,300		
Additional Customers South of I-880	•			
Brooklyn Basin	82,500	156,750		
Schnitzer Steel ²	28,190	33,828		
Digital Realty ²	20,730	24,876		
Frontage Road Pipeline Alignment				
Target Store - Albany	9,100	17,290		

Table D-2. List of Intermediate Term Industrial Customers and Associated Recycled Water Demands (Phase 1B)					
Customer Name		Annual Average Demand, gpd	Maximum Month Demand, gpd		
Caltrans I-80 Landscape (Eastshore Hwy)		13,100	24,890		
Golden Gate Fields(Pacific Racing Assn)		13,160	25,004		
Caltrans I-80 Landscape (Frontage Rd)		7,479	14,210		
University Village		123,500	234,650		
Total D	emands	946,000	1,513,000		
¹ Estimated demand for a future development.					
² Industrial Demand					
³ Assumption for cooling tower demands					

Table D-3. List of Long Term Industrial Customers and Associated Recycled Water Demands (Phase 2)				
Customer Name	Annual Average Demand, gpd	Maximum Month Demand, gpd		
Oakland Raimondi Park	11,733	22,293		
Oakland Mandela Parkway / 34th St.	980	1,862		
Oakland Mandela Parkway / 32th St.	2,646	5,027		
Oakland Mandela Parkway /W. Grant	7,615	14,469		
Oakland Mandela Parkway / 14th St.	9,105	17,300		
Oakland Lowell Park	15,099	28,688		
Oakland DeFremery Park	7,355	13,975		
Oakland Madison Park	916	1,740		
Oakland Lafayette Square	1,469	2,791		
Oakland Wade Johnson Park	3,005	5,710		
EBALDG - Jack London Gateway Sr. Housing	1,835	3,487		
Age Song at Bayside	466	885		
Lake Merritt Boathouse & C. Stanford House	21,738	41,302		
Peralta College - Admin Building Area	1,363	2,590		
Peralta College - Laney Campus	4,314	8,197		
Emeryville Park Ave. Streetscape	1,099	2,088		
Oakland Willow Mini Park	616	1,170		
Extended Stay America Hotel	1,899	3,608		
Oakland Museum of California	4,117	7,822		
Caltrans - Emeryville Segment 2 Bikeway	1,381	2,624		
Caltrans - Bay Bridge - Segment 2 Bikeway	10,716	20,360		
Preservation Park - Oakland	4,622	19,993		
Oakland Housing Authority	3,190	6,062		
Additional Customers on Existing Alignment				
Middleshore Harbor Park	54,238	103,052		
Sherwin Williams Redevelopment ¹	15,000	28,500		
Oakland Army Base Redevelopment ¹	47,768	90,759		
EBMUD Administration Building ²	22,000	28,600		
Chiron Corporation ²	51,700	66,730		
Bayer ²	180,000	220,000		
Cool Port ²	35,000	42,000		
IKEA ²	23,000	29,900		
Pacific Park Plaza ²	30,000	39,000		
Pixar Corporation ²	21,000	32,000		
Allowance for additional 6 HVAC Customers in Emeryville ^{2, 3}	51,000	66,300		
Additional Customers South of I-880				
Brooklyn Basin	82,500	156,750		
Schnitzer Steel ²	28,190	33,828		
Digital Realty ²	20,730	24,876		
Powell Pipeline Alignment				
Spieker Properties	14,982	28,466		

Table D-3. List of Long Term Industrial Customers and Associated Recycled Water Demands (Phase 2)				
Customer Name	Annual Average Demand, gpd	Maximum Month Demand, gpd		
Watergate HOA	42,891	81,493		
Eastshore State Park - Powell St, Emeryville	15,727	29,881		
Marina Park	22,875	43,463		
Channing Way Pipeline Alignment	· · ·			
Berkeley High School	6,499	12,348		
Martin Luther King Jr. Park	4,457	8,468		
People's Park	5,493	10,437		
UC Berkeley - Hearst Ave, Berkeley	40,176	76,334		
UC Berkeley - Oxford St, Berkeley	9,829	18,675		
UC Berkeley - Warring St, Berkeley	39,317	74,702		
UC Berkeley - Bancroft Way, Berkeley	545,972	1,037,347		
UC Berkeley - Clark Kerr, Berkeley	143,157	271,998		
UC Berkeley ²	6,400	7,680		
Alameda East Alignment				
993 Atlantic Ave	2,681	5,094		
1090 Atlantic Ave # I	1,698	3,226		
701 Atlantic Ave	5,630	10,697		
1 Courageous Ct	3,753	7,131		
1 Invincible Ct	3,128	5,943		
1059 Independence Dr	1,787	3,395		
989 Independence Dr	1,608	3,055		
848 Marina Village Pky	2,055	3,905		
915 Marina Village Pky	9,472	17,997		
1001 Marina Village Pky	8,936	16,978		
1119 Atlantic Ave	8,221	15,620		
950 Independence Dr	6,970	13,243		
1305 Marina Village Pky # A	6,881	13,074		
1101 Marina Village Pky	5,004	9,508		
1055 Marina Village Pky	4,110	7,809		
2061 Challenger Dr	4,021	7,640		
1090 Marina Village Pky	4,021	7,640		
1010 Atlantic Ave	3,842	7,300		
2061 Challenger Dr	3,574	6,791		
850 Marina Village Pky	3,396	6,452		
1080 Marina Village Pky	3,396	6,452		
841 Marina Village PKY	3,396	6,452		
1015 Atlantic Ave	2,323	4,414		
813 Atlantic Ave	3,842	7,300		
Alameda Beltline	22,451	42,657		
Ralph Appezzato Memorial Beltline	6,059	11,512		
Alameda Center Alignment	•			
2351 Webster St	7,774	14,771		

Customer Name	Annual Average Demand, gpd	Maximum Month Demand, gpd
College of Alameda	58,469	111,091
Neptune Park	8,980	17,062
Alameda West Alignment		
Alameda Naval Complex	4,194	7,969
US Coast Guard Housing	10,846	20,607
Ploughshares Nursery	6,161	11,706
Alameda Point	281,926	535,659
Multipurpose Field and Soccer Fields	22,389	42,539
Michaan Auction House	2,787	5,295
Alameda City Public Works	2,371	4,505
American Bus Repairs	2,633	5,003
Alameda Naval Complex	25,267	48,007
West Essex Dr Green Belt	2,061	3,916
Alameda Naval Complex	1,277	2,426
Ferry Point Green Belt	3,696	7,022
USS Hornet Museum	25,267	48,007
Alameda Naval Complex	2,996	5,692
Veteran's Administration	32,050	54,165
Hornet Field	19,133	36,353
Encinal High	10,200	19,380
Bayship and Yacht Co ²	12,000	14,400
Allowance for additional HVAC Customers in Alameda ^{2, 3}	42,000	54,600
Frontage Road Pipeline Alignment		
Target Store - Albany	9,100	17,290
Caltrans I-80 Landscape (Eastshore Hwy)	13,100	24,890
Golden Gate Fields(Pacific Racing Assn)	13,160	25,004
Caltrans I-80 Landscape (Frontage Rd)	7,479	14,210
University Village	123,500	234,650
Total Demands	2,561,000	4,536,000
¹ Estimated demand for a future development.		
² Industrial Demand		
³ Assumption for cooling tower demands		

Table D-3. List of Long Term Industrial Customers and Associated Recycled Water Demands (Phase 2)